FERC GAS TARIFF

FIFTH REVISED VOLUME NO. 1

(Superseding Fourth Revised Volume No. 1)

Of

NORTHWEST PIPELINE LLC

Filed with

FEDERAL ENERGY REGULATORY COMMISSION

Communications concerning this Tariff should be sent to:

Bela Patel
Director, Rates & Regulatory

Northwest Pipeline LLC 2800 Post Oak Boulevard Houston, Texas 77056 P.O. Box 1396 Houston, Texas 77251

Telephone: (713)215-2659 Facsimile: (918)594-1003 bela.patel@williams.com

Table of Contents

		Sheet	No
Prelimina	ry Statement		2
Map			4
Statement	of Rates		5
Fuel Use 1	Requirements Factors	-	14
Rate Sche	dules - Index	-	<u>15</u>
TF-1	Firm Transportation		16
TI-1	Interruptible Transportation		30
	Storage Gas Service - Firm	1	50
SGS-2I	Storage Gas Service - Interruptible	(60
LS-2F	Liquefaction-Storage Gas Service - Firm	8	80
LS-2I	Liquefaction-Storage Gas Service - Interruptible		90
LS-3F	Liquefaction-Storage, Vaporization and LNG Del Serv	/−Firm 9	97
LD-4I	Liquefaction-LNG Delivery Service - Interruptible	(98
TF-2	Firm Redelivery Transportation		00
DEX	Deferred Exchange of Storage Gas		15
PAL	Park and Loan Service	12	25
TPAL	Term Park and Loan Service	13	31
TFL-1	Firm Lateral Transportation	13	35
TIL-1	Interruptible Lateral Transportation	15	50
General Te	erms and Conditions - Index	20	00
Forms of	Service Agreements - Index	3(00
Non-Confo	rming Service Agreements	39	95
Nogotiato	d Pata Sarvica Agracoments	4.0	n n

PRELIMINARY STATEMENT RELATED TO FERC GAS TARIFF, FIFTH REVISED VOLUME NO. 1 CONTAINING TRANSPORTATION AND STORAGE RATE SCHEDULES

Northwest Pipeline LLC ("Northwest" or "Transporter") is a "natural gas company" as defined by the Natural Gas Act and "an interstate pipeline" as defined by the Natural Gas Policy Act of 1978. As such, Northwest is subject to the jurisdiction of the Federal Energy Regulatory Commission. Northwest is engaged in the transportation (the term transportation, as defined herein, includes storage service), in interstate commerce, of natural gas for Shippers, pursuant to the regulations of the Federal Energy Regulatory Commission.

The major facilities of Northwest are depicted on the map located on Northwest's Internet website as identified on Sheet No. 4 of this Tariff. Northwest's pipeline system extends from a point near Blanco, New Mexico, through the states of Colorado, Utah, Wyoming, Idaho, Oregon and Washington to a point on the International Boundary near Sumas, Washington.

This Tariff contains transportation and storage rate schedules available for Shippers with whom Northwest contracts through executed Service Agreements for the transportation and storage of natural gas. It is the policy of Northwest to undertake the transportation and storage of gas only pursuant to written contracts with eligible customers upon the terms and conditions set forth in this Tariff, executed on a non-discriminatory basis after consideration of Northwest's existing commitments, delivery capacity, points of delivery and other relevant factors.

This FERC Gas Tariff is filed in compliance with Part 154, Subchapter E, Chapter I, Title 18 of the Code of Federal Regulations, as promulgated by Order of the Federal Power Commission in Docket No. R-107, dated October 30, 1948.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet No. 3 Superseding Substitute Original Sheet No. 3

RESERVED FOR FUTURE USE

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Third Revised Sheet No. 4 Superseding Second Revised Sheet No. 4

The Northwest Pipeline system map may be viewed on Northwest's Internet website at the following address:

http://www.northwest.williams.com/NWP_Portal/file_download?hfFileURL=Files/No rthwest/Downloads/SystemMap.pdf

STATEMENT OF RATES Effective Rates Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 (Dollars per Dth)

	Base		
Rate Schedule and	Tariff R	ate (1),(3)	
Type of Rate	Minimum	Maximum	
Rate Schedule TF-1 (4)(5) Reservation (Large Customer)			
System-Wide	00000	.37250	
25 Year Evergreen Exp.	.00000		
Volumetric (2)			
(Large Customer)	00005	00005	
System-Wide	.00935	.00935	
25 Year Evergreen Exp.	.00935	.00935	
(Small Customer) (6)	.00935	.66230	
Scheduled Overrun (2)	.00935	.38185	
Rate Schedule TF-2 (4)(5)			
Reservation		.37250	
Volumetric	.00935		
Scheduled Daily Overrun	.00935		
Annual Overrun	.00935	.38185	
Rate Schedule TI-1 (2)			
Volumetric (7)	.00935	.38185	
Rate Schedule TFL-1 (4)(5)			
Reservation	_	_	
Volumetric (2) Scheduled Overrun (2)	_	_	
Scheduled Overruin (2)	_	_	
Rate Schedule TIL-1 (2)			
Volumetric	-	-	

Effective Rates Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 (Continued)

(Dollars per Dth)

Entitlement Unauthorized Overrun and Underrun (8)	Rate
General System Unauthorized Daily Overrun	(9)
General System Unauthorized Daily Underrun	10.00000
General System Unauthorized Underrun Imbalances not eliminated after 72 hours	10.00000
Customer-Specific Entitlement Penalty	10.00000

Footnotes

- (1) Rate excludes surcharges approved by the Commission.
- (2) Annual Charge Adjustment ("ACA") surcharge may be applicable. Section 16 of the General Terms and Conditions describes the basis and applicability of the ACA surcharge.

Effective Rates Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 (Continued)

Footnotes (Continued)

(3) To the extent Transporter discounts the Maximum Base Tariff Rate, such discounts will be applied on a non-discriminatory basis, subject to the policies of Order No. 497.

Shippers receiving service under these rate schedules are required to furnish fuel reimbursement in-kind at the rates specified on Sheet No. 14.

An incremental facilities charge or other payment method provided for in Section 21 or 29 of the General Terms and Conditions, is payable in addition to all other rates and charges if such a charge is included in Exhibit C to a Shipper's Transportation Service Agreement.

In addition to the rates set forth on Sheet No. 5, Puget Sound Energy, Inc.'s Transportation Service Agreement Nos. 140053 and 143077 are subject to an annual incremental facility charge pursuant to Section 21 of the General Terms and Conditions. The effective annual incremental facility charge for the South Seattle Delivery Lateral Expansion Project is \$1,855,387, and it is billed in equal monthly one-twelfth increments with a daily incremental facility charge of \$0.07955 per Dth. The effective annual incremental facility charge for the North Seattle Lateral Upgrade Project is \$8,972,208, and it is billed in equal monthly one-twelfth increments with a daily incremental facility charge of \$0.14225 per Dth.

In addition to the reservation rates shown on Sheet No. 5, Shippers who contract for Columbia Gorge Expansion Project capacity are subject to a facility reservation surcharge pursuant to Section 3.4 of Rate Schedule TF-1. The facility charge used in deriving the Columbia Gorge Expansion Project facility reservation surcharge has a minimum rate of \$0 and a maximum rate of \$0.10646.

In addition to the reservation rates shown on Sheet No. 5, Shippers who contract service under Rate Schedules TF-1 (Large Customer), TF-1 25-Year Evergreen, and TF-2 are subject to a Modernization and Emissions Program Cost Recovery Mechanism (CRM) Surcharge pursuant to Section 30 of this Tariff. CRM Surcharge is currently \$0.02256.

Effective Rates Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 (Continued)

(Dollars per Dth)

Footnotes (Continued)

(4) All reservation rates are daily rates computed on the basis of 365 days per year.

For Rate Schedule TF-1, the 25-Year Evergreen Expansion reservation and volumetric rates apply to Shippers receiving service under Rate Schedule TF-1 Evergreen Expansion service agreements. The System-Wide reservation and volumetric rates apply to Shippers receiving service under all other Rate Schedule TF-1 service agreements.

For Rate Schedule TF-1, the 25-Year Evergreen Expansion maximum base tariff reservation rate is comprised of \$0.26587 for transmission costs and \$0.00495 for storage costs. The System-Wide maximum base tariff reservation rate for Rate Schedule TF-1 and the maximum base tariff reservation rate for Rate Schedule TF-2 are comprised of \$0.36755 for transmission costs and \$0.00495 for storage costs.

For Rate Schedule TF-1 (Large Customer), the maximum base tariff volumetric rates applicable to Shippers receiving service under Rate Schedule TF-1 Evergreen Expansion service agreements are comprised of \$0.00924 for transmission costs and \$0.00011 for storage costs. The maximum base tariff volumetric rates for all other services under Rate Schedule TF-1 (Large Customer) and for services under Rate Schedule TF-2 are comprised of \$0.00924 for transmission costs and \$0.00011 for storage costs.

Effective Rates Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 (Continued)

(Dollars per Dth)

Footnotes (Continued)

- (5) Rates for Rate Schedules TF-1, TF-2 and TFL-1 are also applicable to capacity release service except for short-term capacity release transactions for a term of one year or less that take effect on or before one year from the date on which Transporter is notified of the release, which are not subject to the stated Maximum Base Tariff Rate. (Section 22 of the General Terms and Conditions describes how bids for capacity release will be evaluated.) The reservation rate is the comparable volumetric bid reservation charge applicable to Replacement Shippers bidding for capacity released on a one-part volumetric bid basis.
- (6) For Rate Schedule TF-1 (Small Customer), the Maximum Base Tariff Rate is comprised of \$0.65229 for transmission costs and \$0.01001 for storage costs. Transporter will not schedule gas for delivery to a Small Customer subject to this Rate Schedule TF-1 under any transportation Service Agreement (excluding its Rate Schedule TF-2 Service Agreement at Plymouth held at the time of storage service unbundling in RP93-5) unless such Small Customer has scheduled its full Contract Demand for firm service under its Rate Schedule TF-1 (Small Customer) Service Agreement(s) for that day.
- (7) Rate Schedule TI-1 maximum base tariff volumetric rate is comprised of \$0.37679 for transmission costs and \$0.00506 for storage costs.
- (8) Applicable to Rate Schedules TF-1, TF-2, TI-1, TFL-1 and TIL-1 pursuant to Section 15.5 of the General Terms and Conditions.
- (9) The General System Unauthorized Daily Overrun Charge per Dth is the greater of \$10 or 150 percent of the highest midpoint price at NW Wyo. Pool, NW s. of Green River, Stanfield Ore., NW Can. Bdr. (Sumas), Kern River Opal, or El Paso Bondad as reflected in the Daily Price Survey published in "Gas Daily."

Effective Rates Applicable to Rate Schedules DEX-1, TPAL, and PAL

(Dollars per Dth)

Type of Rate	Base Tariff Rate (1),(3) Minimum Maximum
Rate Schedule DEX-1 (2), (4)	
Deferred Exchange	.00000 .38185
Rate Schedule TPAL	
Park/Loan Park/Loan Balance Unpark/Loan Payback	.00000 .38185 .00000 .38185 .00000 .38185
Rate Schedule PAL	
Park and Loan	.00000 .38185

Footnotes

- (1) Rate excludes surcharges approved by the Commission.
- (2) ACA surcharge may be applicable. Section 16 of the General Terms and Conditions describes the basis and applicability of the ACA surcharge.
- (3) To the extent Transporter discounts the maximum currently effective tariff rate, such discounts will be applied on a non-discriminatory basis, subject to the policies of Order No. 497.
- (4) Shippers receiving service under this rate schedule are required to furnish fuel reimbursement in-kind at the rates specified in the Statement of Rates, except as provided in Section 4 of Rate Schedule DEX-1.

Effective Rates Applicable to Rate Schedules SGS-2F and SGS-2I $\,$

(Dollars per Dth)

Rate Schedule and Type of Rate	Bas Tariff Minimum	Rate (1)
Rate Schedule SGS-2F (2) (3) (4) (5) Demand Charge Pre-Expansion Shipper	0.00000	0.02220
Expansion Shipper	0.00000	0.03393
Capacity Demand Charge Pre-Expansion Shipper	0.00000	0.00081
Expansion Shipper	0.00000	0.00291
Volumetric Bid Rates Withdrawal Charge Pre-Expansion Shipper	0.00000	0.02220
Expansion Shipper	0.00000	0.03393
Storage Charge Pre-Expansion Shipper	0.00000	0.00081
Expansion Shipper	0.00000	0.00291
Rate Schedule SGS-2I Volumetric	0.00000	0.00240

Footnotes

⁽¹⁾ Shippers receiving service under these rate schedules are required to furnish fuel reimbursement in-kind at the rates specified on Sheet No. 14.

Effective Rates Applicable to Rate Schedules SGS-2F and SGS-2I (Continued)

Footnotes (Continued)

(2) Rates are daily rates computed on the basis of 365 days per year.

Rates are also applicable to capacity release service except for short-term capacity release transactions for a term of one year or less that take effect on or before one year from the date on which Transporter is notified of the release, which are not subject to the stated Maximum Base Tariff Rate. (Section 22 of the General Terms and Conditions describes how bids for capacity release will be evaluated.) The Withdrawal Charge and Storage Charge are applicable to Replacement Shippers bidding for capacity released on a one-part volumetric bid basis.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Eighth Revised Sheet No. 8 Superseding Seventh Revised Sheet No. 8

RESERVED FOR FUTURE USE

Effective Rates Applicable to Rate Schedules LS-2F and LS-2I

(Dollars per Dth)

	Base	
Rate Schedule and	Tariff	Rate (1)
Type of Rate	Minimum	Maximum
Rate Schedule LS-2F (3)		
Demand Charge (2)	0.00000	0.03136
Capacity Demand Charge (2)	0.00000	0.00401
Volumetric Bid Rates Vaporization Demand-Related Charge (2) Storage Capacity Charge (2)	0.00000	
Liquefaction Vaporization	0.58646 0.07272	
Rate Schedule LS-2I		
Volumetric	0.00000	0.00802
Liquefaction Vaporization	0.58646 0.07272	

Footnotes

- (1) Shippers receiving service under these rate schedules are required to furnish fuel reimbursement in-kind at the rates specified on Sheet No. 14.
- (2) Rates are daily rates computed on the basis of 365 days per year.
- Rates are also applicable to capacity release service except for short-term capacity release transactions for a term of one year or less that take effect on or before one year from the date on which Transporter is notified of the release, which are not subject to the stated Maximum Base Tariff Rate. (Section 22 of the General Terms and Conditions describes how bids for capacity release will be evaluated.) The Vaporization Demand-Related Charge and Storage Capacity Charge are applicable to Replacement Shippers bidding for capacity released on a one-part volumetric bid basis.

Effective Rates Applicable to Rate Schedules LS-3F and LD-4I

(Dollars per Dth)

Rate Schedule and Type of Rate		se Rate (1) Maximum
Rate Schedule LS-3F (3)		
Demand Charge (2)	0.00000	0.03136
Capacity Demand Charge (2)	0.00000	0.00401
Volumetric Bid Rates Vaporization Demand-Related Charge (2) Storage Capacity Charge (2)	0.00000	
Liquefaction Charge (4) Vaporization Charge	0.58646 0.07272	0.58646 0.07272
Rate Schedule LD-4I		
Volumetric Charge Liquefaction Charge (4)	0.00000 0.58646	

Footnotes

- (1) Shippers receiving service under these rate schedules are required to furnish fuel reimbursement in-kind at the rates specified on Sheet No. 14.
- (2) Rates are daily rates computed on the basis of 365 days per year.
- (3) Rates are also applicable to capacity release service except for short-term capacity release transactions for a term of one year or less that take effect on or before one year from the date on which Transporter is notified of the release, which are not subject to the stated Maximum Base Tariff Rate. (Section 22 of the General Terms and Conditions describes how bids for capacity release will be evaluated.) The Vaporization Demand-Related Charge and Storage Capacity Charge are applicable to Replacement Shippers bidding for capacity released on a one-part volumetric bid basis.
- (4) The Liquefaction Charge will be trued-up annually and updated pursuant to Section 14.20 of the General Terms and Conditions.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet No. 10 Superseding Substitute Original Sheet Nos. 10 through 13

RESERVED FOR FUTURE USE

STATEMENT OF FUEL USE REQUIREMENTS FACTORS FOR REIMBURSEMENT OF FUEL USE

Applicable to Transportation Service Rendered Under Rate Schedules Contained in this Tariff, Fifth Revised Volume No. 1

The rates set forth on Sheet Nos. 5, 6, 7, 8 and 8-A are exclusive of fuel use requirements. Shipper shall reimburse Transporter in-kind for its fuel use requirements in accordance with Section 14 of the General Terms and Conditions contained herein.

The fuel use reimbursement furnished by Shippers shall be as follows for the applicable Rate Schedules included in this Tariff:

Rate Schedules TF-1, TF-2, TI-1, and DEX-1	1.39%
Rate Schedule TF-1 - Evergreen Expansion	
Incremental Surcharge (1)	0.50%
Rate Schedule TFL-1	_
Rate Schedule TIL-1	_
Rate Schedules SGS-2F and SGS-2I	0.39%
Rate Schedules LS-2F, LS-3F and LS-2I	
Liquefaction	1.31%
Vaporization	1.56%
Rate Schedule LD-4I	
Liquefaction	1.31%

The fuel use factors set forth above shall be calculated and adjusted as explained in Section 14 of the General Terms and Conditions. Fuel reimbursement quantities to be supplied by Shippers to Transporter shall be determined by applying the factors set forth above to the quantity of gas nominated for receipt by Transporter from Shipper for transportation, Jackson Prairie injection, Plymouth liquefaction, Plymouth vaporization, or for deferred exchange, as applicable.

Footnote

(1) In addition to the Rate Schedule TF-1 fuel use requirements factor, the Evergreen Expansion Incremental Surcharge will apply to the quantity of gas nominated for receipt at the Sumas, SIPI or Pacific Pool receipt points under Evergreen Expansion service agreements.

RATE SCHEDULES

INDEX

Rate Schedule	Description	Sheet	No.
TF-1	Firm Transportation	1	6
TI-1	Interruptible Transportation	3	0
SGS-2F	Storage Gas Service - Firm	5	0
SGS-2I	Storage Gas Service - Interruptible	6	0
LS-2F	Liquefaction-Storage Gas Service - Firm	8	0
LS-2I	Liquefaction-Storage Gas Service - Interruptible	9	0
LS-3F	Liquefaction-Storage, Vaporization and LNG Delivery Service - Firm	9	7
LD-4I	Liquefaction-LNG Delivery Service - Interruptibl	e 9	8
TF-2	Firm Redelivery Transportation	1	00
DEX-1	Deferred Exchange of Storage Gas	1	15
PAL	Park and Loan Service	1	25
TPAL	Term Park and Loan Service	1	31
TFL-1	Firm Lateral Transportation	1	35
TIL-1	Interruptible Lateral Transportation	1	50

RATE SCHEDULE TF-1 Firm Transportation

1. AVAILABILITY

This Rate Schedule is available as provided herein and incorporates the General Terms and Conditions to any party (hereinafter called "Shipper") for the transportation of natural gas by Transporter through Transporter's mainline transmission system under the following conditions:

- (a) Shipper desires firm service and Transporter has available capacity to render such firm service for Shipper. If at the time service is requested under this Rate Schedule, Transporter does not have capacity to receive firm transportation gas at the receipt point(s) requested for redelivery at the delivery point(s) requested, Transporter shall offer to receive firm transportation gas at other receipt point(s) where capacity may be available to enable redelivery at the delivery point(s) requested;
- (b) Transporter can commence the service contemplated without need for construction of any additional pipeline facilities, except facilities for which a facilities agreement has been entered into between Transporter and Shipper pursuant to Section 21 or 29 of the General Terms and Conditions of this Tariff; and
- (c) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

As used in this Rate Schedule, Transporter's mainline transmission system does not include Designated Laterals.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.1 Applicability. This Rate Schedule shall apply to gas transported by Transporter for Shipper pursuant to the executed Service Agreement for service under this Rate Schedule for TF-1 (Large Customer) and TF-1 (Small Customer) service. The Service Agreement will specify the customer category, i.e., whether the Shipper is a Large Customer or Small Customer and, if the Shipper is an incremental expansion customer, whether the Shipper is a Columbia Gorge Expansion customer or a 25-Year Evergreen Expansion customer.

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - (a) TF-1 (Large Customer): All rate provisions contained in this Rate Schedule apply to TF-1 (Large Customers).
 - (b) TF-1 (Small Customer): A TF-1 (Small Customer) is any pipeline or distribution company which elects Rate Schedule TF-1 (Small Customer) service and whose aggregate Transportation Contract Demand, as specified in its Service Agreement(s) hereunder, is for 10,000 Dth per day or less. (A Shipper qualified to elect either the TF-1 (Small Customer) or TF-1 (Large Customer) service hereunder may change its election permanently only in connection with the filing of a general Section 4 rate case by Transporter. Transporter shall provide such Shippers with an opportunity to make such an election prior to filing such rate applications.)
 - (c) Capacity Release Service is service initiated pursuant to Section 22 of the General Terms and Conditions and an executed Service Agreement for Rate Schedule TF-1 (Large Customer) service.
 - Any Rate Schedule TF-1 (Small Customer) Shipper may convert all of its service temporarily to TF-1 (Large Customer) service to participate in Transporter's Capacity Release Program without amending its service agreement, provided that such temporary conversion shall be for a minimum term of twelve calendar months. Notice of the intent to so convert temporarily to TF-1 (Large Customer) service must be given to Transporter electronically using the TF-1 (Small Customer) temporary conversion election screen available in Northwest Passage on Transporter's Designated Site at least one (1) hour prior to the Timely Nomination Cycle for the gas day on which the conversion is to become effective. Shipper's temporary conversion election constitutes agreement to the temporary conversion. Any TF-1 (Small Customer) Shipper participating in a temporary conversion to TF-1 (Large Customer) service shall pay all rates and charges applicable to TF-1 (Large Customer) service during the term of such temporary conversion,

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - 2.1(d) (Continued)

including reservation charges and surcharges applicable to such Shipper's full Transportation Contract Demand or such other rate to which Transporter and Shipper mutually agree, reduced as applicable by revenue credits applicable to Released Capacity. Any Rate Schedule TF-1 (Small Customer) Shipper which participates in such a temporary conversion shall be entitled to return to Rate Schedule TF-1 (Small Customer) service upon expiration of the term of the temporary conversion requested by such Shipper. However, such Shipper shall not be entitled to release its capacity rights for a term which extends beyond the term of the temporary conversion.

- 2.2 Transportation Components. Transportation service under this Rate Schedule, which does not include service on Designated Laterals, shall consist of:
 - (a) The receipt by Transporter for the account of Shipper of Shipper's gas at the Receipt Point(s) specified in the executed Service Agreement;
 - (b) The transportation of such gas through Transporter's pipeline system for the account of Shipper either directly or by displacement; and

2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)

- (c) The delivery of gas in thermally equivalent quantities after transportation (less any fuel use reimbursement furnished in-kind in accordance with Section 14 of the General Terms and Conditions) by Transporter to Shipper or for the account of Shipper at the Delivery Point(s) specified in the executed Service Agreement.
- 2.3 Character of Service. Transportation service rendered to Shipper under this Rate Schedule is firm up to Shipper's Transportation Contract Demand as specified in its executed Service Agreement, subject to the executed Service Agreement and the limitations of this Rate Schedule, and is not subject to curtailment or interruption except as expressly provided in the General Terms and Conditions. Transportation service rendered under this Rate Schedule in excess of Shipper's Transportation Contract Demand is not firm.

3. MONTHLY RATE(S)

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the sum of the amounts specified in this Section 3, as applicable. Only those rate provisions contained in Sections 3.2, 3.4, 3.5 and 3.7 apply to TF-1 (Small Customers).

3.1 Reservation Charge.

(a) For TF-1 (Large Customer) service, the Reservation Charge is the sum of the daily product of Shipper's Transportation Contract Demand as specified in the executed Service Agreement and the Base Tariff Reservation Charge stated on Sheet No. 5 of this Tariff that applies to the customer category identified in the Service Agreement. The CRM Surcharge, as stated on Sheet 5-B of this Tariff, will apply to the Large Customer category, including the 25-Year Evergreen Expansion, for contracts having a primary term of one year or more. Unless specifically adjusted pursuant to Section 3.5 herein, the Maximum Base Tariff Rate set forth on Sheet No. 5 will apply.

For capacity release service, the Reservation Charge is the sum of the daily product of the accepted reservation charge bid

3. MONTHLY RATE(S) (Continued)

price which was bid by a Replacement Shipper or a Prearranged Replacement Shipper under the bidding procedures for capacity releases set forth in Section 22 of the General Terms and Conditions and the Transportation Contract Demand acquired by the Replacement Shipper or the Prearranged Replacement Shipper.

- (b) Shipper will pay the Reservation Charges commencing with the primary term begin date set forth in the Service Agreement.
- 3.2 Volumetric Charge: The sum of (a) and (b) below:
 - (a) An amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Shipper after transportation during the month, after reduction for fuel use reimbursement furnished in kind in accordance with the terms of the executed Service Agreement and Section 14 of the General Terms and Conditions, by (ii) the TF-1 (Large Customer) or TF-1 (Small Customer) base tariff volumetric transportation rate as set forth on Sheet No. 5 of this Tariff. Unless specifically adjusted pursuant to Section 3.5 herein, the Maximum Base Tariff Rate set forth on Sheet No. 5 shall apply.
 - (b) An amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Shipper after transportation during the month, after reduction for fuel use reimbursement furnished in kind in accordance with the terms of the executed Service Agreement and Section 14 of the General Terms and Conditions, by (ii) the ACA component as referenced in footnote 2 on Sheet No. 5-A of this Tariff. This charge shall be subject to adjustment in accordance with Section 16 of the General Terms and Conditions.
- 3.3 Volumetric Release Charges: For Capacity Release service pursuant to Section 22 of the General Terms and Conditions which is provided under a volumetric bid rate the sum of (a), (b) and, if applicable, (c) below:
 - (a) The amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Replacement Shipper or Prearranged Replacement Shipper after transportation during the month, after

3. MONTHLY RATE(S) (Continued)

reduction for fuel use reimbursement furnished in kind in accordance with the terms of the executed Service Agreement and Section 14 of the General Terms and Conditions, by (ii) the accepted volumetric bid made by a Replacement Shipper or a Prearranged Replacement Shipper and by (iii) the Rate Schedule TF-1 Capacity Release Service Base Tariff Volumetric Charge, both as set forth on Sheet No. 5 of this Tariff.

- (b) The amount set forth in Section 3.2 (b) hereof.
- (c) If the Releasing Shipper has specified a minimum average load factor volumetric commitment, an amount equal to the accepted volumetric bid times the difference, if positive, between (i) the specified average load factor times Replacement Shipper's or Prearranged Replacement Shipper's Transportation Contract Demand times the number of days the Transportation Service Agreement is in effect during the month, and (ii) the quantity of Dth delivered by Transporter to Replacement Shipper or Prearranged Replacement Shipper under the subject Service Agreement during the month.
- 3.4 Additional Facility Reservation Surcharge: A Shipper who contracts for Columbia Gorge Expansion Project capacity will pay the facility reservation surcharge set forth in the footnotes to Sheet No. 5 of this Tariff and such surcharge will be in addition to all other applicable rates stated in this Section 3. The facility reservation surcharge was derived based on Transportation Contract Demand and the cost of service attributable to the Columbia Gorge Expansion Project incremental facilities and such derivation will remain in place until such time as a different allocation procedure is specified by Commission order. The monthly facilities reservation surcharge will be the sum of the daily product of Shipper's Transportation Contract Demand as specified in the executed Service Agreement and the applicable facility reservation surcharge for the Columbia Gorge Expansion Project facilities specified in the Footnotes to Sheet No. 5 of this Tariff.

3. MONTHLY RATE(S) (Continued)

3.5 Discounted Recourse Rates:

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates will not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific volumes of gas (such as volumes above or below a certain level or all volumes if volumes exceed a certain level), volumes of gas transported during specific time periods, and volumes of gas transported from specific receipt points and/or to specific delivery points, within specific corridors, or within other defined geographical areas. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

3. MONTHLY RATE(S) (Continued)

- 3.6 Charges for Capacity Release Service: The rates for capacity release service are set forth in Sheet No. 5. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, wherein the Replacement Shipper has not agreed to pay the Maximum Base Tariff Rate or a percentage of the Maximum Base Tariff Rate (as it may change from time-to-time), the Replacement Shipper will be obligated to pay:
 - (a) for capacity release transactions that are subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the lesser of the awarded bid rate and the new Maximum Base Tariff Rate unless the awarded bid rate is less than the new minimum rate, in which case the new minimum rate will apply for the remaining term of the release.
 - (b) for capacity release transactions that are not subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release.
- 3.7 Negotiated Rates: Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.8 Facilities Charge: If Transporter and Shipper enter into a facilities agreement pursuant to Section 21 or 29 of the General Terms and Conditions for Transporter to construct facilities and for Shipper to pay a facilities charge, the facilities charge will be set forth on Exhibit C to an executed Service Agreement.

4. MINIMUM MONTHLY BILL

Unless Transporter and Shipper mutually agree otherwise, the Minimum Monthly Bill will consist of the Reservation Charge specified in Section 3.1 of this Rate Schedule, as applicable.

5. TRANSPORTATION CONTRACT DEMAND

The Transportation Contract Demand is the maximum quantity of Gas, expressed in Dth, that Transporter is obligated to receive (exclusive of fuel reimbursement furnished in-kind pursuant to Section 14 of the General Terms and Conditions), transport and deliver for Shipper on a

5. TRANSPORTATION CONTRACT DEMAND (Continued)

firm basis on any one Gas Day, as specified in an executed Service Agreement for service under this Rate Schedule. Transporter's service obligation is limited to Shipper's Transportation Contract Demand as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

As long as the Transportation Contract Demand, as adjusted for any capacity releases, is not exceeded, Transporter shall be obligated to receive up to Shipper's Maximum Daily Quantity (MDQ) at each Primary Receipt Point and to deliver up to Transporter's Maximum Daily Delivery Obligation (MDDO) at each Primary Delivery Point at pressures at least as great as the pressures specified in Shipper's TF-1 Service Agreement, on a firm basis, as such MDQ and MDDO are adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

The aggregate MDQ at the Primary Receipt Points, as specified in a Service Agreement for service under this Rate Schedule, must equal the Transportation Contract Demand. The aggregate MDDO at the Primary Delivery Points, as specified in a Service Agreement for service under this Rate Schedule, must equal the Transportation Contract Demand, except for those Service Agreements that have aggregate MDDOs in excess of Transportation Contract Demand as a result of the grandfathering of pre-existing conjunctive nomination rights under the sales conversion program approved in Docket No. CP92-79.

6. SCHEDULED OVERRUN TRANSPORTATION

On any day Shipper nominates quantities of gas in excess of Shipper's Transportation Contract Demand specified in the executed Service Agreement, Transporter will schedule such quantities in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Terms and Conditions.

6. SCHEDULED OVERRUN TRANSPORTATION (Continued)

For Scheduled Quantities in excess of Shipper's Transportation Contract Demand, Shipper shall pay for the excess gas transportation on any such day the amounts specified below:

- 6.1 An amount obtained as the sum of (a) and (b):
 - (a) An amount obtained by multiplying (i) the quantity of Dth in excess of Transportation Contract Demand by (ii) the maximum base scheduled overrun transportation rate per Dth as set forth on Sheet No. 5 of this Tariff, unless otherwise agreed to by Shipper and Transporter.
 - (b) An amount obtained by multiplying (i) the quantity of Dth in excess of Transportation Contract Demand by (ii) the ACA component as referenced in footnote 2 on Sheet No. 5-A of this Tariff. This charge shall be subject to adjustment in accordance with Section 16 of the General Terms and Conditions.
 - (c) Additionally, if Shipper and Transporter have agreed to a rate other than the maximum base scheduled overrun transportation rate, Shipper may, when nominating, electronically select an option that will automatically increase (by a stated dollar increment per dth) such rate up to the rate that will increase the likelihood of such quantities being scheduled pursuant to Section 12 of the General Terms and Conditions.
- 6.2 The fuel reimbursement provided for in Section 7 of this Rate Schedule.

- 7. FUEL GAS REIMBURSEMENT AND BTU BALANCING
 Refer to Section 14 of the General Terms and Conditions.
- 8. SHIPPER'S ARRANGEMENTS PRIOR TO RECEIPT AND AFTER DELIVERY
 Refer to Section 14 of the General Terms and Conditions.
- 9. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except as modified in the executed Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

- 10. RESERVATION CHARGE ADJUSTMENTS
 - 10.1 Eligibility. Any Shipper receiving firm transportation service under this Rate Schedule may be eligible for reservation charge adjustments if Transporter fails to provide primary firm transportation due to scheduled or unscheduled maintenance, force majeure, or the Unavailability of Transporter's Facilities.
 - 10.2 Definitions. For purposes of this Section 10 only, the following definitions apply:

Annual Deficiency Volume: The Annual Deficiency Volume for any Constraint Point is the sum of the Deficiency Volumes at that Constraint Point for all Deficiency Periods during a calendar year.

Annual Exemption Volume: The Annual Exemption Volume for any Constraint Point is the lesser of (1) the Constraint Point Design Capacity or (2) the aggregate Constraint Point Firm Rights, times 365 (or 366 days for leap years), multiplied by three percent (3%).

10. RESERVATION CHARGE ADJUSTMENTS (Continued)

Constraint Point: Any receipt point, delivery point, or corridor through which primary nominations are not confirmed due to scheduled or unscheduled maintenance, force majeure, or the Unavailability of Transporter's Facilities at such constraint point or upstream or downstream of such point which reduce Transporter's ability to flow gas through the point.

Constraint Point Actual Capacity: Physical capacity at a Constraint Point which is available on any day.

Constraint Point Design Capacity: Physical design capacity at a Constraint Point when all facilities are fully functional under reasonably representative operating assumptions.

Constraint Point Firm Rights: The rights of a Shipper under Rate Schedule TF-1 to receive firm service through a Constraint Point, based on (1) primary receipt and delivery point rights and (2) corridor rights where primary receipt and/or delivery points establish a call on the relevant capacity.

Deficiency Period: A Deficiency Period begins on the day when aggregate nominations using Constraint Point Firm Rights exceed Constraint Point Actual Capacity and Transporter is unable to schedule otherwise acceptable nominations due to scheduled or unscheduled maintenance, force majeure, or the Unavailability of Transporter's Facilities. A Deficiency Period continues until the earlier of (1) the day when the Constraint Point Actual Capacity is restored to the level of the Constraint Point Design Capacity or (2) the day when Transporter is able to schedule all otherwise acceptable nominations with applicable Constraint Point Firm Rights. A Deficiency Period may be a Declared Entitlement Period, subject to the provisions of Section 14.6 of the General Terms and Conditions.

10. RESERVATION CHARGE ADJUSTMENTS (Continued)

Deficiency Volume: The quantity derived from subtracting the Constraint Point Actual Capacity from the lesser of (1) the Constraint Point Design Capacity or (2) the aggregate Constraint Point Firm Rights.

Excess Deficiency Volume: The Excess Deficiency Volume for any Constraint Point is the amount by which the Annual Deficiency Volume for a Constraint Point exceeds the Annual Exemption Volume for that Constraint Point.

Unavailability of Transporter's Facilities: Where Transporter's facilities that have been certificated, constructed, and placed into service cannot be fully utilized for their intended operation and use for reasons other than scheduled or unscheduled maintenance or force majeure.

- 10.3 Determination of Adjustments. Pursuant to extensive negotiations and the settlement reached between Transporter and Shippers as identified in Docket No. RP99-81-000, and approved by the Commission on November 6, 1998, 85 FERC \P 61,195, reservation adjustments will be calculated as follows:
 - (a) On the first day of any Deficiency Period, Transporter will post to its Designated Site the daily Deficiency Volume for the affected Constraint Point, the Constraint Point Design Capacity, the Constraint Point Actual Capacity, and the estimated duration of the Deficiency Period. Such information may be revised as applicable during the Deficiency Period.
 - (b) At least 24 hours prior to the end of any Deficiency Period, Transporter will notify affected Shippers via a posting on Transporter's Designated Site, and, if specified by Shipper on the Business Associate Information form, via an Internet E-mail or fax to the Shipper, of the date on which Transporter expects to restore the Constraint Point Design Capacity or to be able to schedule all nominations with Constraint Point Firm Rights. If Transporter is able to restore the Constraint Point Design Capacity or is able to schedule such nominations, the Deficiency Period will be deemed to have ended as specified by Transporter. Otherwise, the Deficiency Period will continue until Constraint Point Design Capacity is restored or aggregate nominations with Constraint Point Firm Rights can be scheduled.

10. RESERVATION CHARGE ADJUSTMENTS (Continued)

- (c) At the end of each calendar year, Transporter will post the Annual Deficiency Volume and the Annual Exemption Volume for each affected Constraint Point. Transporter will then compare the Annual Deficiency Volume for each affected Constraint Point to the Annual Exemption Volume for that Constraint Point. If there is an Excess Deficiency Volume for any Constraint Point, Transporter will allocate such Excess Deficiency Volume pro rata to the affected Shippers, based on Constraint Point Firm Rights during the Deficiency Period(s). Each Shipper's allocation will be the sum of its applicable Constraint Point Firm Rights for all Deficiency Periods divided by the sum of the applicable Constraint Point Firm Rights of all Shippers for those Deficiency Periods, multiplied by the Excess Deficiency Volume for a Constraint Point.
- (d) Transporter will calculate reservation charge adjustments under each Service Agreement by multiplying a Shipper's allocated share of the Excess Deficiency Volume for each Constraint Point by the weighted average Base Tariff Reservation Charge paid under such Service Agreement for such Constraint Point for the Deficiency Periods during which the Service Agreement was in effect.
- 10.4 Payment. All reservation charge adjustments will be issued or refunds paid within 90 days from the end of the applicable calendar year.

10. RESERVATION CHARGE ADJUSTMENTS (Continued)

10.5 Sole Remedy. Reservation charge adjustments pursuant to this Section 10 are Shipper's sole remedy for damages relating to Transporter's failure to provide primary firm transportation service under Rate Schedule TF-1, unless such damages result from the negligence or willful misconduct of Transporter.

11. RECEIPT AND DELIVERY POINT FLEXIBILITY

- 11.1 Permanent Changes to Primary Receipt and Delivery Points. Subject to the availability provisions of this Rate Schedule, any Shipper may permanently change primary receipt or delivery points by amending Exhibit A of the Service Agreement.
- 11.2 Use of Alternate Receipt and Delivery Points on a Temporary Basis.
 - (a) All TF-1 Shippers may use any physical receipt or delivery point without amending Exhibit A of the Service Agreement. Such points will be available for the receipt or delivery of gas on a firm basis, in accordance with the scheduling priorities delineated in Section 12 of the General Terms and Conditions.
 - (b) Transporter shall schedule service at alternate receipt and delivery points on a daily basis pursuant to Section 14 of the General Terms and Conditions.
 - (c) The scheduling of service at alternate receipt or delivery points under a Service Agreement will not result in the loss of firm contract rights to a Shipper's primary receipt or delivery points as specified in the Shipper's Service Agreement.
- 11.3 Procedures for Requesting Permanent Receipt and Delivery Point Changes. Any Shipper who wishes to amend the primary receipt or delivery points, or the associated Maximum Daily Quantity or Maximum Daily Delivery Obligation named in Exhibit A of its Service Agreement, will electronically request and execute on Transporter's Designated Site an amendment to the applicable Service Agreement

11. RECEIPT AND DELIVERY POINT FLEXIBILITY (Continued)

by 1:00 p.m. Central Clock Time (12:00 noon Mountain Clock Time) four (4) business days prior to the first of the month for which the change is desired, or for changes to occur during a month, after the first of the month, two (2) business days prior to the commencement of service, unless otherwise agreed to by the parties. If Transporter determines that a receipt or delivery point change request can be honored, the amendment to the Service Agreement will be executed by Transporter. The change will become effective on the later of the requested effective date or the date executed by Transporter, provided that such date shall not exceed fifteen days from the date of receipt of the amendment by Transporter. Notice of the resulting changes in available receipt or delivery point capacity will be posted to Transporter's Designated Site at least one (1) business day prior to implementation of such change.

Firm receipt and delivery point and associated mainline capacity will be posted to Transporter's Designated Site pursuant to Section 25.4 of the General Terms and Conditions and will be available for permanent receipt and/or delivery point changes only pursuant to the procedures outlined in Section 25 of the General Terms and Conditions.

11.4 Transporter's Maximum Service Obligation. The total volumes nominated for service on any day for all receipt or delivery points must not exceed Transportation Contract Demand under a Shipper's firm Service Agreement, except as otherwise provided in Section 6 of this Rate Schedule.

12. CONTRACT TERM EXTENSIONS

- 12.1 Standard Unilateral Evergreen Provision. If Transporter and Shipper agree to include a standard unilateral evergreen provision as indicated on Exhibit A of a long-term Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.

12. CONTRACT TERM EXTENSIONS (Continued)

- (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
- (c) The termination notice required under Section 12.1(b) will be deemed given when posted on Transporter's Designated Site.
- 12.2 Standard Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a standard bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be:
 - (i) one day for Service Agreements with a primary term of one day;
 - (ii) one month for Service Agreements with a primary term of at least two days but less than one year; or
 - (iii) one year for Service Agreements with a primary term of one year or more.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) one Business Day before the termination date if Section 12.2(a)(i) applies;
 - (ii) ten Business Days before the termination date if Section 12.2(a)(ii) applies; or
 - (iii) one year before the termination date if Section 12.2(a)(iii) applies.

12. CONTRACT TERM EXTENSIONS (Continued)

- (c) The termination notice required under Section 12.2(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.
- 12.3 Grandfathered Unilateral Evergreen Provision. If Shipper's Service Agreement contains a grandfathered unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year, at Shipper's sole option.
 - (b) Shipper may terminate all or any portion of service under its Service Agreement either at the expiration of the primary term, or upon any anniversary thereafter, by giving written notice to Transporter so stating at least twelve months in advance.
 - (c) Shipper also will have the sole option to enter into a new Service Agreement for all or any portion of the service under its Service Agreement at or after the end of the primary term of its Service Agreement. It is Transporter's and Shipper's intent that this provision provide Shipper with a "contractual right to continue such service" and to provide Transporter with concurrent pregranted abandonment of any volume that Shipper terminates within the meaning of 18 CFR 284.221(d)(2)(i) as promulgated by Order No. 636 on May 8, 1992.
 - (d) The termination notice required under Section 12.3(b) will be deemed given when posted on Transporter's Designated Site.

12. CONTRACT TERM EXTENSIONS (Continued)

12.4 Extension for a Portion of Contract Rights. If a Shipper with the contractual right in its Service Agreement to extend the primary term of its Service Agreement under a contractual rollover or evergreen provision for only a portion of its firm contract rights elects to exercise such right, the aggregate MDQs and aggregate MDDOs for the extended term must be retained by the same percentage as the Transportation Contract Demand is retained. Shipper may specify the MDQs it wishes to retain at each Receipt Point and the MDDOs it wishes to retain at each Delivery Point.

13. SEGMENTATION

- 13.1 To the extent operationally feasible, a Shipper may temporarily segment its capacity into separate parts for its own use or for the purpose of releasing capacity to a Replacement Shipper. Transporter will respond to a Shipper's segmentation request within two business days following the request; and if the segmentation request is approved, Transporter will issue new Service Agreements for each of the segments within one hour of notification to the Shipper that the request is approved.
- 13.2 Transporter will permit nominations of forward hauls up to Transportation Contract Demand and backhauls up to Transportation Contract Demand to the same point at the same time for segmented capacity.
- 13.3 Segmentation without Capacity Release. A Shipper that wishes to segment its capacity into separate parts for its own use is subject to the same procedures as apply to the temporary capacity release of segmented capacity under Sections 22 and 23 of the General Terms and Conditions except that the Shipper's release of capacity to itself will always be non-biddable. Such transactions will be treated as temporary capacity releases with the Shipper acting as both the Releasing Shipper and the Replacement Shipper. New Service Agreements will be executed with the Shipper for all new segments. Such Shipper also is specifically subject to the following:
 - (a) Capacity between a Shipper's primary receipt point(s) and primary delivery point(s) may be segmented into separate Service Agreements for a Shipper's own use, provided that such segmentation is operationally feasible and provided that firm capacity is available at and between the designated points;

13. SEGMENTATION (Continued)

- (b) Transportation Contract Demand under any Service Agreement created as a result of segmentation may not exceed the Transportation Contract Demand under Shipper's Service Agreement as it existed before the segmentation; and
- (c) Each Service Agreement will have a primary receipt point and a primary delivery point, but Shipper will have the right to use any available alternate receipt and delivery points on a secondary firm basis, with a scheduling priority consistent with Section 12 of the General Terms and Conditions of Transporter's tariff.
- 13.4 Segmentation with Capacity Release. A Shipper that wishes to segment its capacity for the purpose of releasing all or part of that capacity to a Replacement Shipper is subject to the capacity release provisions of Sections 22 and 23 of the General Terms and Conditions of Transporter's tariff. New Service Agreements will be executed with the Releasing Shipper for the retained segment(s) and the Replacement Shipper for the released segment.

First Revised Sheet No. 29 Superseding Substitute Original Sheet No. 29

RATE SCHEDULE TI-1 Interruptible Transportation

1. AVAILABILITY

This Rate Schedule is available on a first-come, first-served basis to any party (hereinafter called "Shipper") for the transportation of natural gas by Transporter through Transporter's mainline transmission system under the following conditions:

- (a) Transporter can render the service contemplated without need for construction of any additional pipeline facilities, except for facilities for which a facilities agreement has been entered into between Transporter and Shipper pursuant to Section 21 of the General Terms and Conditions of this Tariff; and
- (b) Shipper and Transporter shall have executed a Service Agreement for service under this Rate Schedule.

As used in this Rate Schedule, Transporter's mainline transmission system does not include Designated Laterals.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to gas transported by Transporter for Shipper pursuant to the executed Service Agreement for service under this Rate Schedule.
- 2.2 Transportation Components. Transportation service under this Rate Schedule, which does not include service on Designated Laterals, shall consist of:
 - (a) The receipt by Transporter for the account of Shipper of Shipper's gas at the Receipt Point(s) specified in the executed Service Agreement;
 - (b) The transportation of such gas through Transporter's pipeline system for the account of Shipper either directly or by displacement; and

2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)

- (c) The delivery of gas in thermally equivalent quantities after transportation (less any fuel use reimbursement furnished in-kind in accordance with Section 14 of the General Terms and Conditions) by Transporter to Shipper or for the account of Shipper at the Delivery Point(s) specified in the executed Service Agreement.
- 2.3 Character of Services. Transportation service rendered under this Rate Schedule is interruptible, and subject to curtailment or interruption at any time as expressly delineated in the General Terms and Conditions.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the amounts specified in this Section 3, as applicable.

- 3.1 Transportation Service. The sum of (a) and (b) below:
 - (a) An amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Shipper after transportation during the month, after reduction for fuel reimbursement provided in-kind in accordance with Section 14 of the General Terms and Conditions, by (ii) the base tariff interruptible transportation rate as set forth on Sheet No. 5 of this Tariff. Unless specifically adjusted pursuant to Section 3.2 herein, the Maximum Base Tariff Rate set forth on Sheet No. 5 shall apply.
 - (b) An amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Shipper after transportation during the month, after reduction for fuel use reimbursement provided in-kind in accordance with Section 14 of the General Terms and Conditions, by (ii) the ACA component as referenced in footnote 2 on Sheet No. 5-A of this Tariff. This charge shall be subject to adjustment in accordance with Section 16 of the General Terms and Conditions.

3. MONTHLY RATE (Continued)

3.2 Discounted Recourse Rates.

- Transporter reserves the right to temporarily discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific quantities of gas (such as quantities above or below a certain level or all quantities if quantities exceed a certain level), quantities of gas transported during specific time periods, and quantities of gas transported from specific receipt points and/or to specific delivery points, within specific corridors, or within other defined geographical areas. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) If Transporter elects to grant a temporary discount, it will do so without amending Shipper's Service Agreement. The temporary discount will be electronically granted using the Rate Schedule TI-1 Temporary Discount form available in Northwest Passage on Transporter's Designated Site. Shipper's electronic completion of the Temporary Discount form constitutes agreement by the Shipper to the discount and Transporter's electronic acceptance constitutes Transporter's granting of the discount.
- (c) In completing the electronic Temporary Discount form, Shipper may select an option which will automatically increase (by a stated dollar increment per dth) the granted rate up to the rate that will increase the likelihood of such quantities being scheduled pursuant to Section 12 of the General Terms and Conditions.
- (d) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

3. MONTHLY RATE (Continued)

- 3.3 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.4 Facilities Construction. If Transporter and Shipper enter into a facilities agreement pursuant to Section 21 of the General Terms and Conditions for Transporter to construct facilities, payment for such facilities will be set forth in the facilities agreement.

4. SHIPPER'S MAXIMUM DAILY QUANTITY

The executed Service Agreement shall specify a maximum daily quantity of gas to be transported thereunder for Shipper. The maximum daily quantity refers to the quantity of gas to be delivered by Transporter to Shipper after transportation, at the Delivery Point(s), after fuel use reimbursement provided in-kind in accordance with the terms of Section 14 of the General Terms and Conditions.

5. FUEL GAS REIMBURSEMENT

Refer to Section 14 of the General Terms and Conditions.

6. SHIPPER'S ARRANGEMENTS PRIOR TO RECEIPT AND AFTER DELIVERY

Refer to Section 14 of the General Terms and Conditions.

7. RECEIPT AND DELIVERY POINT FLEXIBILITY

Transporter will allow amendments to receipt or delivery points designated in Service Agreements under this Rate Schedule. Any Shipper who wishes to amend the receipt or delivery points named in its Service Agreement will electronically request and execute on Transporter's Designated Site an amendment to the applicable Service Agreement which implements a revised Exhibit A.

Interruptible shippers who are bumped by changes to receipt and delivery points by firm Shippers, as specified in Section 11 of Rate Schedule TF-1, shall have the right to switch receipts or deliveries on a flexible basis to any other available receipt or delivery point, consistent with the priority of service provisions contained in Section 12 and the operating conditions specified in Section 14 of the General Terms and Conditions.

Nominations for interruptible service will be honored at such receipt or delivery points consistent with the priority of service provisions contained in Section 12 of the General Terms and Conditions. The total volumes nominated for all receipt and delivery points must not exceed Shipper's maximum daily quantity, except as otherwise provided in Section 4 of this Rate Schedule.

8. EVERGREEN PROVISION

- 8.1 Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as contemplated by Section 4 of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one month.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon

8. EVERGREEN PROVISION (Continued)

the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least thirty days before the termination date.

(c) Termination notice under Section 8.1(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

9. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except Sections 22, 23, 24 and 25 and except as modified in the executed Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

First Revised Sheet No. 36 Superseding Substitute Original Sheet Nos. 36 Through 49

RATE SCHEDULE SGS-2F Storage Gas Service - Firm

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the purchase of natural gas storage service from Transporter when Shipper and Transporter have executed a Service Agreement for the storage of gas under this Rate Schedule and have arranged for the related transportation of gas to and from the Jackson Prairie Storage Project under one of Transporter's transportation rate schedules.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to firm storage gas service at the Jackson Prairie Storage Project. The executed Service Agreement for service under this Rate Schedule will specify the Shipper category, i.e., whether the Shipper is a Pre-Expansion Shipper or an Expansion Shipper. The Jackson Prairie Storage Project capacity available for this Rate Schedule will be Transporter's undivided interest as an owner in the Project, excluding any portion of such interest which may be authorized for use by Transporter for transportation balancing. Delivery of natural gas by Shipper to Transporter for injection and by Transporter to Shipper upon withdrawal shall be at the point of interconnection between the Jackson Prairie Storage Project and Transporter's main transmission line.
- 2.2 Storage Components. Firm storage service consists of Transporter's injection storage and withdrawal of Shipper's gas.
- 2.3 Character of Service. Storage gas service rendered to Shipper under this Rate Schedule, up to Shipper's Storage Demand and Storage Capacity and subject to the limitations of this Rate Schedule and the executed Service Agreement, shall be firm and shall not be subject to curtailment or interruption except as expressly provided in this Rate Schedule and in the General Terms and Conditions. Storage gas service rendered to Shipper under this Rate Schedule in excess of Shipper's Storage Demand and Storage Capacity is not firm.
- 2.4 Capacity Release. Shippers releasing firm storage rights shall do so in accordance with the capacity release provisions outlined in Section 22 of the General Terms and Conditions. Any such release is subject to the terms and conditions of this Rate Schedule.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the amounts specified in this Section 3, as applicable.

- 3.1 Storage Service. The sum of (a) and (b) below:
 - (a) The demand charge will be the sum of the daily product of Shipper's Storage Demand and the Demand Charge rate stated on the Statement of Rates in this Tariff that applies to the customer category identified in the Service Agreement.
 - (b) The capacity demand charge is the sum of the daily product of Shipper's Storage Capacity and the Capacity Demand Charge rate stated on the Statement of Rates in this Tariff that applies to the customer category identified in the Service Agreement.

The related transportation of gas to and from the Jackson Prairie storage facility shall be subject to separate transportation charges under applicable open-access Rate Schedules. The rates set forth in the sub-paragraphs above are exclusive of the aforementioned transportation charges.

3.2 Discounted Recourse Rates.

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific volumes of gas such as volumes injected, withdrawn or stored above or below a certain level or all volumes if volumes exceed a certain level, and volumes of gas injected, withdrawn or stored during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

3. MONTHLY RATE (Continued)

- 3.3 Charges for Capacity Release Service: The rates for capacity release service are set forth in Sheet No. 7. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, the Replacement Shipper will be obligated to pay:
 - (a) the lesser of the awarded bid rate and the new Maximum Base Tariff Rate, or the greater of the awarded bid rate and the new minimum base tariff rate, as applicable, for the remaining term of the release for capacity release transactions with a term of more than one year and where the awarded bid rate was not tied to the Maximum Base Tariff Rate as it may change from time to time;
 - (b) the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release for capacity release transactions with a term of one year or less that take effect on or before one year from the date on which Transporter is notified of the release and where the award bid rate was not tied to the Maximum Base Tariff Rate; or
 - (c) the new Maximum Base Tariff Rate or, if applicable, the percentage of the new Maximum Base Tariff Rate for capacity release transactions where the awarded bid rate was tied to the Maximum Base Tariff Rate as it may change from time to time.

For capacity release service subject to demand charges, the payments by the Replacement Shipper shall be equal to the sum of the daily product of the accepted Demand Charge bid rate and the Storage Demand, plus the sum of the daily product of the accepted Capacity Demand Charge bid rate and the Storage Capacity.

For capacity release service subject to volumetric bid rates, the payments by the Replacement Shipper shall be equal to the accepted volumetric bid rate for withdrawals multiplied by the actual volumes withdrawn each day plus the accepted volumetric bid rate for storage multiplied by the actual volumes in storage each day.

3. MONTHLY RATE (Continued)

3.4 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.

4. MINIMUM MONTHLY BILL

Unless Transporter and Shipper mutually agree otherwise, the Minimum Monthly Bill will consist of the sum of the demand and capacity demand charges specified in Section 3 of this Rate Schedule, as applicable.

5. FUEL GAS REIMBURSEMENT

Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions.

6. STORAGE DEMAND

The Storage Demand shall be the largest number of Dth Transporter is obligated to withdraw and deliver to Shipper, and Shipper is entitled to receive from Transporter, at Jackson Prairie on any one day, to the limitations set forth in Section 9 below, and shall be specified in the executed Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Demand, as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

7. STORAGE CAPACITY

Shipper's Storage Capacity shall be the maximum quantity of gas in Dth which Transporter is obligated to store for Shipper's account and shall be specified in the executed Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Capacity, as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

8. DEFINITIONS

- 8.1 A Storage Cycle is the twelve-month period beginning October 1 of any calendar year and ending September 30 of the following calendar year.
- 8.2 Shipper's Working Gas Inventory shall be the actual quantity of working gas in storage for Shipper's account at the beginning of any given day.
- 8.3 Shipper's Working Gas Quantity for a Storage Cycle shall be determined as of October 1 and shall be the lesser of:
 - (a) Shipper's Working Gas Inventory as of October 1, the beginning of the Storage Cycle; or
 - (b) The minimum quantity of Shipper's Working Gas Inventory at any time between August 31 and September 30 of the preceding Storage Cycle divided by 0.80; or
 - (c) The minimum quantity of Shipper's Working Gas Inventory at any time between June 30 and September 30 of the preceding Storage Cycle divided by 0.35.

8. DEFINITIONS (Continued)

The above method of determining Shipper's Working Gas Quantity may be modified consistent with any comparable modification under the January 15, 1998 Gas Storage Project Agreement, or superseding agreement, permitted by the Jackson Prairie Storage Project Management Committee. A Shipper's Working Gas Quantity will be adjusted for any Working Gas Quantity released as indicated on Exhibit A to a Replacement Shipper's Service Agreement.

- 8.4 Shipper's Available Working Gas on any day during the Storage Cycle shall be equal to Shippers' Working Gas Inventory less Shipper's Unavailable Working Gas.
- 8.5 Shipper's Unavailable Working Gas on any day during the Storage Cycle shall be equal to the highest level of Shipper's Working Gas Inventory during the preceding days of the current Storage Cycle less Shipper's Working Gas Quantity.

9. WITHDRAWALS OF STORAGE GAS

9.1 General Procedure. Shipper may nominate to withdraw gas on any day, specifying the quantity of gas within Shipper's Available Working Gas which it desires withdrawn under this Rate Schedule during such day. Transporter will schedule the withdrawal of the quantity of gas so nominated, subject to the limitations set forth in this Rate Schedule and subject as necessary to confirmation of the nomination changes for the related transportation service agreement.

- 9. WITHDRAWALS OF STORAGE GAS (Continued)
 - 9.2 Withdrawal Obligation. Transporter's daily withdrawal obligation shall be at 100 percent of the Shipper's Storage Demand as long as Shipper's Available Working Gas is greater than or equal to 60 percent of Shipper's Storage Capacity. On any day when Shipper's Available Working gas is less than 60 percent of Shipper's Storage Capacity, Transporter's daily withdrawal obligation shall be reduced by two percent of Shipper's Storage Demand for each one percent that Shipper's Available Working Gas is less than 60 percent of Shipper's Storage Capacity, until a minimum daily withdrawal rate equal to 10 percent of Shipper's Storage Demand is reached.
- 10. INJECTIONS OF WORKING GAS FOR SHIPPER'S ACCOUNT

Upon Transporter's request, Shipper shall provide written notice to Transporter prior to May 1 of each year, of the quantities of gas to be injected for the account of Shipper during the period of May 1 through September 30 of such year. Shipper may nominate to inject gas on any day, specifying the quantity of gas it desires injected under this Rate Schedule during such day, including the applicable fuel reimbursement requirements. Transporter will schedule the injection of the quantity of gas so nominated, subject to the limitations set forth in this Rate Schedule and subject to delivery of such quantity, and shall retain any fuel use reimbursement furnished in-kind in accordance with Section 14 of the General Terms and Conditions in addition to any fuel reimbursement required from the party under whose Service Agreement the gas is to be transported to Jackson Prairie.

12. LIMITATIONS ON INJECTIONS AND WITHDRAWALS FROM STORAGE

Shipper may Nominate gas to be injected into or withdrawn from storage for Shipper's account at any time during the year. In no event shall the balance of gas stored in Shipper's account exceed Shipper's Storage Capacity as defined under Section 6 of this Rate Schedule. Transporter will schedule available injection capacity consistent with the priority of service provisions and curtailment policy in Section 12 of the General Terms and Conditions.

After the commencement of an annual Storage Cycle, withdrawals from Shipper's Available Working Gas may be replaced both to maintain deliverability and to restore the quantity available for withdrawals. Additional working gas may also be injected during the Storage Cycle; provided, however, that Shipper's Unavailable Working Gas as defined in Section 8 above shall not be available for withdrawal during the current Storage Cycle.

13. WITHDRAWALS IN EXCESS OF FIRM ENTITLEMENT (BEST-EFFORTS WITHDRAWALS)

Shipper may nominate to withdraw quantities in excess of Shipper's Storage Demand on a best-efforts basis; provided, however, that the total quantity withdrawn may not exceed the level of Shipper's Available Working Gas. Transporter may make such excess withdrawal, consistent with the priority of service provisions contained in Section 12 of the General Terms and Conditions.

14. TRANSFER OF WORKING GAS INVENTORY

Shippers that are subject to this Rate Schedule may agree to transfer their respective Jackson Prairie Working Gas Inventories to any capacity holder in the Jackson Prairie Storage facility under Rate Schedules SGS-2F, SGS-2I, TPAL, and PAL. Participating Shippers must notify Transporter's Marketing Services personnel of their intent to transfer such inventory, in writing. Transfers of Working Gas Inventory may not result in any Shipper taking title to quantities that exceed such Shipper's contractual rights.

Pursuant to the January 15, 1998 Gas Storage Project Agreement, owners of the Jackson Prairie Storage Project may transfer portions of their respective available working gas inventories, as defined in the Project Agreement, to each other. Upon agreement of the parties, and subject to the terms of the Project Agreement, Transporter may utilize its ownership account on behalf of a Rate Schedule SGS-2F Shipper to transfer such Shipper's Working Gas Inventory to an owner's available working gas inventory account. Conversely, an owner may transfer its available working gas inventory to a Rate Schedule SGS-2F Shipper's Working Gas Inventory account.

Transfers from a SGS-2F to SGS-2I, TPAL, and PAL contracts will be scheduled pursuant to the priority of service provisions and curtailment policy in Section 12 of the General Terms and Conditions.

15. EVERGREEN PROVISION

- 15.1 Standard Unilateral Evergreen Provision. If Transporter and Shipper agree to include a standard unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
 - (c) The termination notice required under Section 15.1(b) will be deemed given when posted on Transporter's Designated Site.
- 15.2 Standard Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a standard bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:

- 15. EVERGREEN PROVISION (Continued)
 - (a) The established rollover period will be:
 - (i) one month for a Service Agreement with a primary term of less than one year; or
 - (ii) one year for a Service Agreement with a primary term of one year or more.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) ten Business Days before the termination date if Section 15.2(a)(i) applies; or
 - (ii) one year before the termination date if Section 15.2(a) (ii) applies.
 - (c) The termination notice required under Section 15.2(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.
 - 15.3 Grandfathered Unilateral Evergreen Provision. If Shipper's Service Agreement contains a grandfathered unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year, at Shipper's sole option.

15. EVERGREEN PROVISION (Continued)

- (b) Shipper may terminate all or any portion of service under its Service Agreement either at the expiration of the primary term, or upon any anniversary thereafter, by giving written notice to Transporter so stating at least twelve months in advance.
- (c) Shipper also will have the sole option to enter into a new Service Agreement for all or any portion of the service under its Service Agreement at or after the end of the primary term of its Service Agreement. It is Transporter's and Shipper's intent that this provision provide Shipper with a "contractual right to continue such service" and to provide Transporter with concurrent pregranted abandonment of any volume that Shipper terminates within the meaning of 18 CFR 284.221(d)(2)(i) as promulgated by Order No. 636 on May 8, 1992.
- (d) The termination notice required under Section 15.3(b) will be deemed given when posted on Transporter's Designated Site.

16. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, are applicable to this Rate Schedule and are hereby made a part hereof.

First Revised Sheet No. 59 Superseding Substitute Original Sheet No. 59

1. AVAILABILITY

This open-access Rate Schedule is available to any Shipper for the purchase of natural gas storage service from Transporter when Shipper and Transporter have executed a Service Agreement for the storage of gas under this Rate Schedule and have arranged for the related transportation of gas to and from the Jackson Prairie Storage Project under one of Transporter's Transportation Rate Schedules.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to interruptible storage service at the Jackson Prairie Storage Project. Delivery of natural gas by Shipper to Transporter for injection and by Transporter to Shipper upon withdrawal shall be at the point of interconnection between the Jackson Prairie Storage Project and Transporter's main transmission line.
- 2.2 Service Components. Interruptible storage service consists of Transporter's injection, storage and withdrawal of Shipper's gas.
- 2.3 Character of Service. Storage gas service rendered to Shipper under this Rate Schedule shall be interruptible up to Shipper's Interruptible Storage Capacity, and shall be subject to the executed Service Agreement, and the limitations of this Rate Schedule. The scheduling priorities and curtailment policy applicable to this service will be in accordance with Section 12 of the General Terms and Conditions.

If and to the extent Shipper's Working Gas Inventory cannot continue to be accommodated within the storage capacity available to Transporter, for any reason, or such utilized storage capacity is needed to meet Transporter's firm obligations under Rate Schedule SGS2F or Transporter's system balancing requirements, Transporter will notify Shipper that, within seven days of such notice, Shipper is required to reduce the level of its Working Gas Inventory existing on the day of such notice to a level specified by Transporter.

Prior to the end of any seven day period (or any extension of such period, as applicable,) Transporter may notify Shipper that it has revised the required level applicable to Shipper's Working Gas Inventory. If the new level is higher than the previously specified

2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)

level, the seven day period continues to apply to that portion of Shipper's Working Gas Inventory that is in excess of the revised level. If the revised level is lower than the previously specified level, requiring additional gas to be withdrawn, the seven day period applicable to the volume of Shipper's Working Gas Inventory in excess of the previously specified level is still valid but a new seven day period will apply to the volume of Shipper's Working Gas Inventory that is in excess of the newly specified level.

However, if Shipper has nominated Working Gas Inventory from the Jackson Prairie Storage Project using primary Receipt and Delivery Points under a Rate Schedule TF-1 or Rate Schedule TF-2 service agreement and Transporter is unable to provide the related transportation, then such seven day period shall be extended by one day for each day Transporter is unable to render such transportation. Transporter shall not extend the seven day period if Transporter is unable to provide the related transportation under a Rate Schedule TF-1 Service Agreement using alternate points or overrun nominations or under a Rate Schedule TI-1 Service Agreement. If Shipper fails to reduce its Working Gas Inventory to the level specified by Transporter within seven days (or any applicable extensions) of receiving notice from Transporter, then Transporter shall take title to all of Shipper's Working Gas Inventory in excess of the Working Gas Inventory level specified by Transporter, free and clear of any adverse claims.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the amount specified in this Section 3, as applicable.

3.1 Volumetric Charge. Shipper will pay Transporter for service rendered hereunder a volumetric charge in an amount equal to the sum of the daily product of Shipper's Working Gas Inventory and the Maximum Base Tariff Rate per Dth as set forth on Sheet No. 7 of this Tariff.

The related transportation of gas to and from the Jackson Prairie Storage Project shall be subject to separate transportation charges under Transporter's applicable open-access Transportation Rate Schedules. The volumetric charge set forth in this Section 3 is exclusive of the aforementioned transportation charges.

3. MONTHLY RATE (Continued)

3.2 Discounted Recourse Rates.

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific volumes of gas, such as volumes injected, withdrawn or stored above or below a certain level or all volumes if volumes exceed a certain level, and volumes of gas injected, withdrawn or stored during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.3 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.

4. FUEL GAS REIMBURSEMENT

Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions.

5. DEFINITIONS

- 5.1 Shipper's Interruptible Storage Capacity shall be the maximum quantity of gas in Dth which Transporter may store for Shipper's account and shall be specified in the executed Service Agreement between Transporter and Shipper.
- 5.2 Shipper's Working Gas Inventory shall be the actual quantity of working gas in storage for Shipper's account at the beginning of any given day.

6. WITHDRAWALS FOR SHIPPER'S ACCOUNT

- 6.1 General Procedure. Shipper may nominate to withdraw gas at any time during the year. Shipper may nominate to withdraw gas under this Rate Schedule on any day, specifying the quantity of gas within Shipper's Working Gas Inventory. Transporter will schedule the quantity of gas so nominated subject to the limitations set forth in this Rate Schedule and subject to acceptance of such quantity by the party under whose Service Agreement the gas is to be transported. Transporter will schedule withdrawals in accordance with the priority of service and curtailing policy delineated in Section 12 of the General Terms and Conditions.
- 6.2 Withdrawal Limitation. In no event shall withdrawals for a Shipper exceed the balance of gas stored in Shipper's account.

7. INJECTIONS FOR SHIPPER'S ACCOUNT

7.1 General Procedure. Shipper may nominate to inject gas at any time during the year. Shipper may nominate to inject gas under this Rate Schedule on any day, specifying the quantity of gas it desires injected, including applicable fuel reimbursement requirements. Transporter will schedule the quantity of gas so nominated, subject to the limitations set forth in this Rate Schedule and subject to delivery of such quantity, and shall retain any fuel use reimbursement furnished in-kind in accordance with Section 14 of the General Terms and Conditions.

7. INJECTIONS FOR SHIPPER'S ACCOUNT (Continued)

7.2 Injection Limitation. Injections may be scheduled only to the extent that injection and storage capacity is available to Transporter which is not being utilized for Rate Schedule SGS-2F service or for Transporter's balancing requirements.

In no event shall injections result in the balance of gas stored in Shipper's account exceeding Shipper's Interruptible Storage Capacity.

8. TRANSFER OF WORKING GAS INVENTORY

Shippers that are subject to this Rate Schedule may agree to transfer their respective Jackson Prairie Working Gas Inventories to any capacity holder in the Jackson Prairie Storage facility under Rate Schedule SGS-2F, SGS-2I, PAL, and TPAL whose contractual rate is equal to or greater than the transferring capacity holder's contractual rate. Participating Shippers must notify Transporter's Marketing Services personnel of their intent to transfer such inventory, in writing, prior to the beginning of the gas day in which such transfer will occur. Transfers of Working Gas Inventory may not result in any Shipper taking title to quantities that exceed such Shipper's contractual rights.

Pursuant to the January 15, 1998 Gas Storage Project Agreement, owners of the Jackson Prairie Storage Project may transfer portions of their respective available working gas inventories, as defined in the Project Agreement, to each other. Upon agreement of the parties, and subject to the terms of the Project Agreement, Transporter may utilize its ownership account on behalf of a Rate Schedule SGS-2I Shipper to transfer such Shipper's Working Gas Inventory to an owner's available working gas inventory account. Conversely, an owner may transfer its available working gas inventory to a Rate Schedule SGS-2I Shipper's Working Gas Inventory account.

9. EVERGREEN PROVISION

- 9.1 Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one month.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least thirty days before the termination date.
 - (c) Termination notice under Section 9.1(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

10. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the General Terms and Conditions contained in this Tariff.

First Revised Sheet Nos. 66 Through 69 Superseding Substitute Original Sheet Nos. 66 Through 69

Fifth Revised Sheet No. 70 Superseding Fourth Revised Sheet No. 70

Fifth Revised Sheet No. 71 Superseding Fourth Revised Sheet No. 71

Fifth Revised Sheet No. 72 Superseding Fourth Revised Sheet No. 72

Fourth Revised Sheet No. 73 Superseding Third Revised Sheet No. 73

Sixth Revised Sheet No. 74 Superseding Fifth Revised Sheet No. 74

Fourth Revised Sheet No. 75 Superseding Third Revised Sheet No. 75

First Revised Sheet Nos. 76 Through 79 Superseding Substitute Original Sheet Nos. 76 Through 79

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the purchase of a liquefaction-storage gas service from Transporter, when Shipper and Transporter have executed a Service Agreement for the purchase of such service under this Rate Schedule and have arranged for the related transportation of gas to and from the Plymouth LNG Facility under one of Transporter's transportation rate schedules.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to the firm liquefaction-storage gas service rendered by Transporter to Shipper under the Service Agreement for such service.
- 2.2 Service Components. Storage service under this Rate Schedule shall consist of the liquefaction and storage by Transporter for Shipper's account of gas transported to the Plymouth LNG Facility under a separate Service Agreement, the vaporization of such stored gas, and delivery to Shipper for transportation under a separate Service Agreement. Delivery of natural gas by Shipper to Transporter for liquefaction and by Transporter to Shipper upon vaporization shall be at the point of interconnection between Transporter's Plymouth LNG Facility and Transporter's main transmission line.
- 2.3 Character of Service. Service rendered to Shipper under this Rate Schedule, within the limitations described in the Service Agreement and this Rate Schedule, shall be firm and shall not be subject to curtailment or interruption except as expressly provided in the General Terms and Conditions.
- 2.4 Capacity Release. Shippers releasing firm storage rights shall do so in accordance with the capacity release provisions outlined in Section 22 of the General Terms and Conditions. Any such release is subject to the terms and conditions of this rate schedule and may require a transfer of the Storage Capacity Balance as specified in Section 10 of this Rate Schedule.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the sum of the amounts specified in this Section 3, as applicable.

3. MONTHLY RATE (Continued)

- 3.1 Storage Service. The sum of (a) through (d) below:
 - (a) The demand charge will be the sum of the daily product of Shipper's Storage Demand and the Demand Charge rate.
 - (b) The capacity demand charge will be the sum of the daily product of Shipper's Storage Capacity and the Capacity Demand Charge rate.
 - (c) The liquefaction charge will be the sum of the daily product of Shipper's gas per Dth scheduled for liquefaction into Shipper's storage account (except as provided in Section 9 of this Rate Schedule) and the Liquefaction rate.
 - (d) The vaporization charge will be the sum of the daily product of Shipper's gas per Dth scheduled for vaporization and the Vaporization rate.

The unit rates that are applicable to this Rate Schedule shall be those as set forth from time to time in the Statement of Rates of this Tariff.

- 3.2 Charges for Capacity Release Service: The rates for capacity release service that are applicable to this Rate Schedule are set forth in the Statement of Rates of this Tariff. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, wherein the Replacement Shipper has not agreed to pay the Maximum Base Tariff Rate or a percentage of the Maximum Base Tariff Rate (as it may change from time-to-time), the Replacement Shipper will be obligated to pay:
 - (a) for capacity release transactions that are subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the lesser of the awarded bid rate and the new Maximum Base Tariff Rate unless the awarded bid rate is less than the new minimum rate, in which case the new minimum rate will apply for the remaining term of the release.
 - (b) for capacity release transactions that are not subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release.

3. MONTHLY RATE (Continued)

For capacity release service subject to demand charges, the payments by the Replacement Shipper, in addition to the Liquefaction and Vaporization Charges, shall be equal to the sum of the daily product of the awarded Demand Charge bid rate and the Storage Demand, plus the sum of the daily product of the awarded Capacity Demand Charge bid rate and the Storage Capacity.

For capacity release service subject to volumetric bid rates, the payments by the Replacement Shipper, in addition to the liquefaction and vaporization charges, shall be equal to the awarded volumetric bid Vaporization Demand-Related Charge multiplied by the volumes delivered to Shipper each day plus the awarded volumetric bid Storage Capacity Charge multiplied by Shipper's Storage Capacity Balance each day.

3. MONTHLY RATE (Continued)

3.3 Discounted Recourse Rates.

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth in the Statement of Rates of this Tariff that are applicable to this Rate Schedule, or any superseding tariff. Such discounted Recourse Rates may apply to: specific volumes of gas, such as volumes of gas liquefied, vaporized, stored above or below a certain level; or all volumes if volumes exceed a certain level; and/or volumes of gas liquefied, vaporized or stored during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.4 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.5 Transportation Charges. The related transportation of gas to and from the Plymouth LNG Facility shall be subject to separate transportation charges under applicable Rate Schedules. The rates set forth in sub-paragraphs 3.1 through 3.4 are exclusive of the aforementioned transportation charges.

4. MINIMUM MONTHLY BILL

Unless Transporter and Shipper mutually agree otherwise, the Minimum Monthly Bill will consist of the sum of the demand charge and the capacity demand charge specified in Section 3 of this Rate Schedule.

5. FUEL GAS REIMBURSEMENT

5.1 Fuel Gas Reimbursement. Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions upon liquefaction and vaporization of Shipper's gas.

5.2 Vaporization Fuel. Shipper's fuel reimbursement quantities for vaporization will not be supplied from Shipper's gas nominated for vaporization, but instead will be supplied from Shipper's Boil-off balance. Any amount of Shipper's fuel reimbursement quantities for vaporization that would reduce Shipper's Boil-off balance below zero will be reimbursed to Transporter in-kind within 30 days of Shipper's vaporization nomination, unless other arrangements are made between Transporter and Shipper.

6. DEFINITIONS

- 6.1 Storage Demand. The Storage Demand shall be the largest number of Dth Transporter is obligated to vaporize for, and Shipper is entitled to receive from, the Plymouth LNG Facility under this Rate Schedule on any one day, subject to the limitations described in Section 8 of this Rate Schedule, and shall be specified in the Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Demand as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.
- 6.2 Storage Capacity. The Storage Capacity shall be the maximum quantity of gas in Dth which Transporter is obligated to liquefy and store in liquid form for Shipper's account and shall be specified in the Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Capacity as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.
- 6.3 Liquefaction Period. The Liquefaction Period shall be the seven consecutive months beginning on April 1 of any year and extending through the next succeeding October 31.
- 6.4 Vaporization Period. The Vaporization Period shall be the five consecutive months beginning on November 1 of any year and extending through the next succeeding March 31.
- 6.5 Storage Capacity Balance. Shipper's Storage Capacity Balance at any particular time shall be the quantity of gas in storage in liquid form for Shipper at such time.

6. DEFINITIONS (Continued)

Annual Liquefaction Quantity. Shipper's Annual Liquefaction Quantity shall be the quantity of gas in Dth, up to Shipper's Storage Capacity, which Shipper desires to have liquefied and stored in liquid form by Transporter for Shipper's account during the Liquefaction Period, and shall be provided to Transporter in writing on or before April 1 of each year. In the event that Shipper does not submit an Annual Liquefaction Quantity by April 1, Shipper's Annual Liquefaction Ouantity for the Liquefaction Period shall be the quantity difference between Shipper's Storage Capacity and Shipper's Capacity balance on April 1. Shipper, upon 10 days written notice to Transporter, may elect to change its Annual Liquefaction Quantity during the Liquefaction Period. Such change shall not reduce the Annual Liquefaction Quantity below Shipper's pro rata share of gas that has been liquefied at the time of the election change. A Shipper's pro rata share will be derived by multiplying (a) the quantity of gas liquefied to date to meet the Annual Liquefaction Quantities of all LS-2F and LS-2I Shippers by (b) the ratio of the Shipper's Annual Liquefaction Quantity to the total Annual Liquefaction Quantities for LS-2F and LS-2I Shippers prior to the election change.

7. LIQUEFACTION INTO STORAGE FOR SHIPPER'S ACCOUNT

During a Liquefaction Period, Shipper shall nominate sufficient quantities of gas to fill Shipper's Annual Liquefaction Quantity. Such nominations shall commence on April 1 and shall consist of uniform daily quantities equal to 1/200th of Shipper's Annual Liquefaction Quantity (except for the last day of liquefaction) unless a different nomination pattern is operationally feasible and mutually agreed upon by Transporter and Shipper. In addition, Transporter may schedule the Liquefaction Period and rate of liquefaction to fit system operating conditions.

Transporter shall not be obligated to liquefy and store gas for Shipper in excess of Shipper's Storage Capacity.

Nominations shall be made by Shipper scheduling such tendered volumes on any day as transportation volumes delivered under a Service Agreement for liquefaction and storage.

Shipper may nominate gas for liquefaction and storage during a Vaporization Period in replacement of gas vaporized during such Vaporization Period; provided, however, the liquefaction of such gas shall be at such times as may be agreed upon between Transporter and Shipper.

8. VAPORIZATION FROM STORAGE AND DELIVERY TO SHIPPER

- 8.1 General Procedure. Shipper may nominate vaporization of gas on any day during the Vaporization Period, specifying the quantity of gas it desires vaporized under this Rate Schedule during such day. Transporter shall vaporize and deliver for transportation the quantity of gas so nominated out of Shipper's Storage Capacity Balance, subject to the limitations set forth in this Rate Schedule.
- 8.2 Notice Required. The notice given by Shipper to Transporter for vaporization on any day shall be prior to the commencement of such day; provided, however, that commencement of actual delivery for transportation shall be determined by system operating conditions to the limitations set forth in this Rate Schedule.
- 8.3 Daily Vaporization in Excess of Shipper's Storage Demand. Shipper may Nominate a quantity of gas in excess of Shipper's Storage Demand and Transporter will schedule such quantities in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Term and Conditions.
- 8.4 Vaporization During a Liquefaction Period. Shipper may nominate gas out of Shipper's Storage Capacity Balance for vaporization and delivery for transportation to Shipper on any day during the Liquefaction Period. However, such vaporization and delivery will be in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Terms and Conditions.

9. LNG BOIL-OFF

- 9.1 Calculation of LNG Boil-off. Shipper will be allocated a pro rata share of monthly LNG Boil-off quantities of gas in Dths by multiplying the monthly Boil-off quantity by the quotient of the Shipper's average daily Storage Capacity Balance for the month and the sum of the average daily Storage Capacity Balances for the month of all Shippers' Service Agreements under Rate Schedules LS-2F, LS-3F and LS-2I.
- 9.2 LNG Boil-off Balance. Shipper's allocated share of monthly LNG Boil-off will be subtracted from Shipper's Storage Capacity Balance no later than the 15th day of the following month and will be added to Shipper's Boil-off balance due from Transporter. This Boil-off balance will be deemed to be at the Plymouth LNG Facility. Shipper may either choose to nominate the Boil-off balance for re-liquefaction or transport the Boil-off balance from the Plymouth LNG Facility under one of Transporter's transportation rate schedules. Nominations for the reliquefaction of Boil-off will be subject to fuel use reimbursement.

9. LNG Boil Off (Continued)

9.3 LNG Boil-off Balance Tolerance. The sum of Shipper's Storage Capacity Balance and Shipper's Boil-off balance cannot exceed Shipper's Storage Capacity. Additional tolerances regarding the Boil-off balance are detailed in Section 15 of the General Terms and Conditions.

10. TRANSFER OF STORAGE CAPACITY BALANCE

Shippers subject to this Rate Schedule and to Rate Schedule LS-2I may agree to transfer all or part of their respective Storage Capacity Balance among themselves. Participating Shippers must notify Transporter's nominations personnel of their intent to transfer such inventory, in writing, prior to the beginning of the gas day in which such transfer will occur. Transfers of Storage Capacity Balances may not result in any given Shipper taking title to inventory that exceed such Shipper's Storage Capacity, unless the participating Shippers before such transfer occurs enter into a Capacity Release arrangement providing for additional Storage Capacity in accordance with Section 22 of the General Terms and Conditions.

11. EVERGREEN PROVISION

- 11.1 Five Year Notice Unilateral Evergreen Provision. If Transporter and Shipper agree to include a five year notice unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
 - (c) The termination notice required under Section 11.1(b) will be deemed given when posted on Transporter's Designated Site.
- 11.2 One Year Notice Unilateral Evergreen Provision. For a Service Agreement that has a primary term of at least 15 years, Transporter and Shipper may agree to include a one year notice unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, in which case the following conditions will apply:
 - (a) The established rollover period will be one year.

11. EVERGREEN PROVISION (Continued)

- (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least one year before the termination date.
- (c) The termination notice required under Section 11.2(b) will be deemed given when posted on Transporter's Designated Site.
- 11.3 Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be:
 - (i) one month for a Service Agreement with a primary term of less than one year; or
 - (ii) one year for a Service Agreement with a primary term of one year or more.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) ten Business Days before the termination date if Section 11.3(a)(i) applies; or
 - (ii) one year before the termination date if Section 11.3(a)(ii) applies.
 - (c) The termination notice required under Section 11.3(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

12. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except as modified in the Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

First Revised Sheet Nos. 88 Through 89 Superseding Substitute Original Sheet Nos. 88 Through 89

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the purchase of a liquefaction-storage gas service from Transporter, when Shipper and Transporter have executed a Service Agreement for the purchase of such service under this Rate Schedule and have arranged for the related transportation of gas to and from the Plymouth LNG Facility under one of Transporter's transportation rate schedules.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to the interruptible liquefaction-storage gas service rendered by Transporter to Shipper under the Service Agreement for such service.
- 2.2 Service Components. Service under this Rate Schedule shall consist of the liquefaction and storage by Transporter for Shipper's account of gas transported to the Plymouth LNG Facility under a separate Service Agreement and the vaporization of such stored gas, and delivery to Shipper for transportation under a separate Service Agreement. Delivery of natural gas by Shipper to Transporter for liquefaction and by Transporter to Shipper upon vaporization shall be at the point of interconnection between Transporter's Plymouth LNG Facility and Transporter's main transmission line.
- 2.3 Character of Service. Service rendered to Shipper under this Rate Schedule, up to Shipper's Storage Capacity shall be interruptible and shall be subject to curtailment or interruption as provided herein and in the General Terms and Conditions.

2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)

2.4 Mandatory Vaporization. If storage capacity being utilized hereunder by Shipper is needed to meet Transporter's firm obligations under Rate Schedules LS-2F or LS-3F Transporter shall require Shipper to, within seven days: (1) vaporize and withdraw all LS-2I volumes held in storage by Transporter for or on behalf of Shipper and (2) eliminate Shipper's Boil-off balance; provided, however, if Shipper has arranged for transportation of its gas from the Plymouth LNG Facility under a Rate Schedule TF-1 or TF-2 agreement and Transporter is unable to provide such related transportation, then such seven day period shall be extended by one day for each day Transporter is unable to render such transportation. Transporter shall not extend the seven day period if Transporter is unable to provide the related transportation under a Rate Schedule TI-1 agreement. If Shipper fails to vaporize and withdraw all of its LS-2I Storage Capacity Balance held in storage by Transporter for or on behalf of Shipper and eliminate its Boil-off balance within seven days upon receiving notice from Transporter, then Transporter shall take title to any such remaining LS-2I Storage Capacity Balance and Boil-off balance free and clear of any adverse claims.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the amounts specified in this Section 3, as applicable.

- 3.1 Storage Service. The sum of (a) through (c) below:
- (a) The volumetric charge will be the sum of the daily product of Shipper's Storage Capacity Balance and the Volumetric rate.
- (b) The liquefaction charge will be the sum of the daily product of Shipper's gas per Dth scheduled for liquefaction into Shipper's storage account (except as provided in Section 8 of this Rate Schedule) and the Liquefaction rate.
- (c) The vaporization charge will be the sum of the daily product of Shipper's gas per Dth scheduled for vaporization and the Vaporization rate.

The unit rates that are applicable to this Rate Schedule shall be those as set forth from time to time in the Statement of Rates of this Tariff.

3. MONTHLY RATE (Continued)

- 3.2 Discounted Recourse Rates.
 - Transporter reserves the right to discount at any time the (a) Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth in the Statement of Rates of this Tariff that are applicable to this Rate Schedule, or any superseding tariff. Such discounted Recourse Rates may apply to: specific volumes of gas, such as volumes of gas liquefied, vaporized, stored above or below a certain level; or all volumes if volumes exceed a certain level; and/or volumes of gas liquefied, vaporized or stored during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
 - (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.3 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.4 Transportation Charges. The related transportation of gas to and from the Plymouth LNG Facility shall be subject to separate transportation charges under applicable Rate Schedules. The rates set forth in sub-paragraphs 3.1 through 3.3 are exclusive of the aforementioned transportation charges.

4. FUEL GAS REIMBURSEMENT

4.1 Fuel Gas Reimbursement. Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions upon liquefaction and vaporization of Shipper's gas.

4.2 Vaporization Fuel. Shipper's fuel reimbursement quantities for vaporization will not be supplied from Shipper's gas nominated for vaporization, but instead will be supplied from Shipper's Boil-off balance. Any amount of Shipper's fuel reimbursement quantities for vaporization that would reduce Shipper's Boil-off balance below zero will be reimbursed to Transporter in-kind within seven days of Shipper's vaporization nomination, unless other arrangements are made between Transporter and Shipper.

5. DEFINITIONS

- 5.1 Storage Capacity. The Storage Capacity shall be the maximum quantity of gas in Dth which Transporter may liquefy and store in liquid form for Shipper's account and shall be specified in the Service Agreement between Transporter and Shipper.
- 5.2 Liquefaction Period. The Liquefaction Period shall be the seven consecutive months beginning on April 1 of any year and extending through the next succeeding October 31.
- 5.3 Vaporization Period. The Vaporization Period shall be the five consecutive months beginning on November 1 of any year and extending through the next succeeding March 31.
- 5.4 Storage Capacity Balance. Shipper's Storage Capacity Balance at any particular time shall be the quantity of gas in storage in liquid form for Shipper at such time.

6. LIQUEFACTION INTO STORAGE FOR SHIPPER'S ACCOUNT

Shipper may nominate gas for liquefaction and storage during the Liquefaction Period, specifying the volume of gas to be liquefied. Transporter will schedule the volume of gas so nominated subject to the limitations set forth in this Rate Schedule and subject to Transporter's obligations to provide firm storage services under Rate Schedules LS-2F and LS-3F. In addition, Transporter may schedule the rate of liquefaction to fit system operating conditions.

6. LIQUEFACTION INTO STORAGE FOR SHIPPER'S ACCOUNT (Continued)

Transporter shall not be obligated to liquefy and store gas for Shipper in excess of Shipper's Storage Capacity.

The tender by Shipper to Transporter shall be made by Shipper scheduling such tendered volumes on any day as transportation volumes delivered under a Service Agreement for liquefaction and storage.

Shipper may nominate gas for liquefaction and storage during a Vaporization Period in replacement of gas vaporized during such Vaporization Period; provided, however, the liquefaction of such gas shall be at such times as may be agreed upon between Transporter and Shipper, subject to Transporter's obligations to provide firm storage services under Rate Schedules LS-2F and LS-3F.

7. VAPORIZATION FROM STORAGE AND DELIVERY TO SHIPPER

7.1 General Procedure. Shipper may nominate vaporization of gas on any day during the Vaporization Period, specifying the volume of gas. Transporter will schedule the volume of gas so nominated out of Shipper's Storage Capacity Balance, subject to the limitations set forth in this Rate Schedule and subject to Transporter's obligations to provide firm storage service under Rate Schedules LS-2F and LS-3F.

- 7. VAPORIZATION FROM STORAGE AND DELIVERY TO SHIPPER (Continued)
 - 7.2 Notice Required. The notice given by Shipper to Transporter for vaporization on any day shall be prior to the commencement of such day; provided, however, that commencement of actual delivery for transportation shall be determined by system operating conditions.
 - 7.3 Vaporization During a Liquefaction Period. Shipper may nominate gas out of Shipper's Storage Capacity Balance for vaporization and delivery for transportation to Shipper on any day during the Liquefaction Period. However, such vaporization and delivery will be subject to the priority of service and curtailment policy in Section 12 of the General Terms and Conditions.

8. LNG BOIL-OFF

- 8.1 Calculation of LNG Boil-off. Shipper will be allocated a pro rata share of monthly LNG Boil-off quantities of gas in Dths by multiplying the monthly Boil-off quantity by the quotient of the Shipper's average daily Storage Capacity Balance for the month and the sum of the average daily Storage Capacity Balances for the month of all Shippers' Service Agreements under Rate Schedules LS-2F, LS-3F and LS-2I.
- 8.2 LNG Boil-off Balance. Shipper's allocated share of monthly LNG Boil-off will be subtracted from Shipper's Storage Capacity Balance no later than the 15th day of the following month and will be added to Shipper's Boil-off balance due from Transporter. This Boil-off balance will be deemed to be at the Plymouth LNG Facility. Shipper may either choose to nominate the Boil-off balance for re-liquefaction or transport the Boil-off balance from the Plymouth LNG Facility under one of Transporter's transportation rate schedules. Nominations for the reliquefaction of Boil-off will be subject to fuel use reimbursement.
- 8.3 LNG Boil-off Balance Tolerance. The sum of Shipper's Storage Capacity Balance and Shipper's Boil-off balance cannot exceed Shipper's Storage Capacity. Additional tolerances regarding the Boil-off balance are detailed in Section 15 of the General Terms and Conditions.

9. TRANSFER OF STORAGE CAPACITY BALANCE

Shippers subject to this Rate Schedule and to Rate Schedule LS-2F may agree to transfer all or part of their respective Storage Capacity Balance among themselves. Participating Shippers must notify Transporter's nominations personnel of their intent to transfer such inventory, in writing, prior to the beginning of the gas day in which such transfer will occur. Transfers of Storage Capacity Balances may not result in any given Shipper taking title to inventory that exceed such Shipper's Storage Capacity, unless the participating Shippers before such transfer occurs enter into a Capacity Release arrangement providing for additional Storage Capacity in accordance with Section 22 of the General Terms and Conditions.

10. EVERGREEN PROVISION

- 10.1 Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one month.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least thirty days before the termination date.
 - (c) Termination notice under Section 10.1(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

11. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except as modified in the Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

Second Revised Sheet No. 96 Superseding First Revised Sheet Nos. 96 Through 99

RATE SCHEDULE LS-3F

Liquefaction-Storage, Vaporization and LNG Delivery Service - Firm

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the purchase of liquefaction-storage, vaporization and LNG delivery service from Transporter, when Shipper and Transporter have executed a Service Agreement for the purchase of such service under this Rate Schedule and Shipper has arranged for the related transportation of gas to and from the Plymouth LNG Facility under one of Transporter's transportation rate schedules.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to the firm liquefaction-storage, vaporization and LNG delivery service rendered by Transporter to Shipper under the Service Agreement for such service.
- 2.2 Storage Service Component. Storage service under this Rate Schedule shall consist of liquefaction into storage by Transporter for Shipper's account of gas transported to the Plymouth LNG Facility under a separate Service Agreement. Delivery of natural gas by Shipper to Transporter for liquefaction shall be at the point of interconnection between Transporter's Plymouth LNG Facility and Transporter's main transmission line.
- 2.3 Vaporization Service Component. Vaporization service under this Rate Schedule shall consist of the vaporization of stored gas at the Plymouth LNG Facility and delivery to Shipper for transportation. Delivery of natural gas by Transporter to Shipper upon vaporization shall be at the point of interconnection between Transporter's Plymouth LNG Facility and Transporter's main transmission line.
- 2.4 LNG Delivery Service Component. LNG delivery service under this Rate Schedule shall consist of the delivery of stored LNG at the Plymouth LNG Facility by Transporter to Shipper. Delivery of LNG by Transporter to Shipper shall be at LNG off-system delivery points directly connected to the Plymouth LNG Facility storage tanks.
- 2.5 Character of Service. Service rendered to Shipper under this Rate Schedule, within the limitations described in the Service Agreement and this Rate Schedule, shall be firm and shall not be subject to curtailment or interruption except as expressly provided in the General Terms and Conditions.
- 2.6 Capacity Release. Shippers releasing firm storage rights shall do so in accordance with the capacity release provisions outlined in Section 22 of the General Terms and Conditions. Any such release is subject to the terms and conditions of this Rate Schedule and may require a transfer of Storage Capacity Balance inventory as specified in Section 11 of this Rate Schedule.

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the sum of the amounts specified in this Section 3, as applicable.

- 3.1 Service Charges. The sum of (a) through (d) below:
 - (a) The demand charge will be the sum of the daily product of Shipper's Storage Demand and the Demand Charge rate.
 - (b) The capacity demand charge will be the sum of the daily product of Shipper's Storage Capacity and the Capacity Demand Charge rate.
 - (c) The liquefaction charge will be the sum of the daily product of Shipper's gas per Dth scheduled for liquefaction into Shipper's storage account and the Liquefaction Charge rate.

The amount paid by Shipper under Section 3.1(c) will be trued-up on a calendar year basis and the true-up amount will be billed/refunded, as described in Section 14 of the General Terms and Conditions.

(d) The vaporization charge will be the sum of the daily product of Shipper's gas per Dth scheduled for vaporization and the Vaporization Charge rate.

The unit rates that are applicable to this Rate Schedule shall be those as set forth from time to time in the Statement of Rates of this Tariff.

3.2 Charges for Capacity Release Service: The rates for capacity release service that are applicable to this Rate Schedule are set forth in the Statement of Rates of this Tariff. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, wherein the Replacement Shipper has not agreed to pay the Maximum Base Tariff Rate or a percentage of the Maximum Base Tariff Rate (as it may change from time-to-time), the Replacement Shipper will be obligated to pay:

3. MONTHLY RATE (Continued)

- (a) for capacity release transactions that are subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the lesser of the awarded bid rate and the new Maximum Base Tariff Rate unless the awarded bid rate is less than the new minimum rate, in which case the new minimum rate will apply for the remaining term of the release.
- (b) for capacity release transactions that are not subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release.

For capacity release service subject to demand charges, the payments by the Replacement Shipper, in addition to the liquefaction charge and vaporization charge, shall be equal to the sum of the daily product of the awarded Demand Charge bid rate and the Storage Demand, plus the sum of the daily product of the awarded Capacity Demand Charge bid rate and the Storage Capacity.

For capacity release service subject to volumetric bid rates, the payments by the Replacement Shipper, in addition to the liquefaction charge and vaporization charge, shall be equal to the awarded volumetric bid Vaporization Demand-Related Charge multiplied by the volumes delivered to Shipper each day plus the awarded volumetric bid Storage Capacity Charge multiplied by Shipper's Storage Capacity Balance each day.

3.3 Discounted Recourse Rates.

(a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth in the Statement of Rates of this Tariff that are applicable to this Rate Schedule, or any superseding tariff. Such discounted Recourse Rates may apply to: specific volumes of gas, such as volumes of gas liquefied, vaporized, stored above or below a certain level; or all volumes if volumes exceed a certain level; and/or volumes of gas liquefied, vaporized or stored during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.

(b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

3. MONTHLY RATE (Continued)

- 3.4 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.5 Transportation Charges. The related transportation of gas to and from the Plymouth LNG Facility shall be subject to separate transportation charges under applicable Rate Schedules. The rates set forth in sub-paragraphs 3.1 through 3.4 are exclusive of the aforementioned transportation charges.

4. MINIMUM MONTHLY BILL

Unless Transporter and Shipper mutually agree otherwise, the Minimum Monthly Bill will consist of the sum of the demand charge and the capacity demand charge specified in Section 3 of this Rate Schedule.

5. FUEL GAS REIMBURSEMENT

- 5.1 Fuel Gas Reimbursement. Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions upon liquefaction and vaporization of Shipper's gas.
- 5.2 Vaporization Fuel. Shipper's fuel reimbursement quantities for vaporization will not be supplied from Shipper's gas nominated for vaporization, but instead will be supplied from Shipper's Boil-off balance. Any amount of Shipper's fuel reimbursement quantities for vaporization that would reduce Shipper's Boil-off balance below zero will be reimbursed to Transporter in-kind within 30 days of Shipper's vaporization nomination, unless other arrangements are made between Transporter and Shipper.

RATE SCHEDULE LS-3F

Liquefaction-Storage, Vaporization and LNG Delivery Service - Firm (Continued)

6. DEFINITIONS

- 6.1 Storage Demand. The Storage Demand shall be the largest number of Dth Transporter is obligated to vaporize for and Shipper is entitled to receive from the Plymouth LNG Facility under this Rate Schedule on any one day, subject to the limitations described in Section 8 of this Rate Schedule, and shall be specified in the Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Demand as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.
- 6.2 Storage Capacity. The Storage Capacity shall be the maximum quantity of gas in Dth which Transporter is obligated to liquefy and store in liquid form for Shipper's account and shall be specified in the Service Agreement between Transporter and Shipper. Transporter's service obligation is limited to Shipper's Storage Capacity as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.
- 6.3 Liquefaction Period. The Liquefaction Period shall be January 1 through December 31.
- 6.4 Vaporization Period. The Vaporization Period shall be the five consecutive months beginning on November 1 of any year and extending through the next succeeding March 31.
- 6.5 Storage Capacity Balance. Shipper's Storage Capacity Balance at any particular time shall be the quantity of gas in storage in liquid form for Shipper at such time.

7. LIQUEFACTION INTO STORAGE FOR SHIPPER'S ACCOUNT

- 7.1 During the Liquefaction Period, Shipper may nominate sufficient quantities of gas to fill Shipper's Storage Capacity. Transporter may schedule the rate of liquefaction to fit system operating conditions. Transporter may batch, in its sole discretion, confirmed liquefaction nominations from all Plymouth LNG Facility rate schedule Service Agreements over multiple days in order to operate the liquefiers in an efficient and responsible manner. Once liquefaction begins, Transporter will deliver the batched nominations to each individual Shipper in an operationally efficient manner, subject to system operating conditions.
- 7.2 Transporter shall not be obligated to liquefy and store gas for Shipper in excess of Shipper's Storage Capacity.
- 7.3 Nominations shall be made by Shipper scheduling such tendered volumes on any day as transportation volumes delivered under a Service Agreement for liquefaction and storage.

8. VAPORIZATION FROM STORAGE AND DELIVERY TO SHIPPER

- 8.1 General Procedure. Shipper may nominate vaporization of gas during the Vaporization Period, specifying the quantity of gas it desires vaporized under this Rate Schedule. Transporter shall vaporize and deliver for transportation the quantity of gas so nominated out of Shipper's Storage Capacity Balance, subject to the limitations set forth in this Rate Schedule.
- 8.2 Notice Required. The notice given by Shipper to Transporter for vaporization shall be in accordance with the NAESB WGQ Nominations Related Standards; provided, however, that commencement of actual delivery for transportation shall be determined by system operating conditions.
- 8.3 Daily Vaporization in Excess of Shipper's Storage Demand. Shipper may Nominate a quantity of gas in excess of Shipper's Storage Demand and Transporter will schedule such quantities in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Term and Conditions.
- 8.4 Out-Of-Period Vaporization. Shipper may nominate gas out of Shipper's Storage Capacity Balance for vaporization and delivery for transportation to Shipper outside of the vaporization period stated in 6.4 of this rate schedule; provided, however, such vaporization and delivery will be in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Terms and Conditions.

9. LNG DELIVERIES FROM STORAGE

9.1 General Procedure. Shipper may nominate deliveries of LNG by specifying the quantity of LNG it desires delivered under this Rate Schedule. Transporter shall deliver the quantity of LNG so nominated out of Shipper's Storage Capacity Balance in an operationally efficient manner, subject to the limitations set forth in this Rate Schedule. Such LNG deliveries will be in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Terms and Conditions.

Consistent with Transporter's option to batch liquefaction nominations, Shipper may only schedule delivery of LNG quantities that are physically available in Shipper's Storage Capacity Balance, and may not rely on scheduled liquefaction nominations held for batching purposes.

9. LNG DELIVERIES FROM STORAGE (Continued)

9.2 Notice Required. The notice given by Shipper to Transporter for LNG deliveries shall be in accordance with the NAESB WGQ Nominations Related Standards; provided, however, that commencement of actual LNG delivery shall be determined by system operating conditions.

10. TREATMENT OF LNG BOIL-OFF

- 10.1 Calculation of LNG Boil-off. Shipper will be allocated a pro rata share of monthly LNG Boil-off quantities of gas in Dths by multiplying the monthly Boil-off quantity by the quotient of the Shipper's average daily Storage Capacity Balance for the month and the sum of the average daily Storage Capacity Balances for the month of all Shippers' Service Agreements under Rate Schedules LS-2F, LS-3F and LS-2I.
- 10.2 LNG Boil-off Balance. Shipper's allocated share of monthly LNG Boil-off will be subtracted from Shipper's Storage Capacity Balance no later than the 15th day of the following month and will be added to Shipper's Boil-off balance due from Transporter. This Boil-off balance will be deemed to be at the Plymouth LNG Facility. Shipper may either choose to nominate the Boil-off balance for re-liquefaction or transport the Boil-off balance from the Plymouth LNG Facility under one of Transporter's transportation rate schedules. Nominations for the reliquefaction of Boil-off will be subject to both the liquefaction charge and fuel use reimbursement.
- 10.3 LNG Boil-off Balance Tolerance. The sum of Shipper's Storage Capacity Balance and Shipper's Boil-off balance cannot exceed Shipper's Storage Capacity. Additional tolerances regarding the Boil-off balance are detailed in Section 15 of the General Terms and Conditions.

11. TRANSFER OF STORAGE CAPACITY BALANCE

Shippers subject to this Rate Schedule may agree to transfer all or part of their respective Storage Capacity Balance among themselves. Participating Shippers must notify Transporter's nominations personnel of their intent to transfer such inventory, in writing, prior to the beginning of the gas day in which such transfer will occur. Transfers of Storage Capacity Balances may not result in any given Shipper taking title to inventory volumes that exceed such Shipper's Storage Capacity, unless the participating Shippers before such transfer occurs enter into a Capacity Release arrangement providing for additional Storage Capacity in accordance with Section 22 of the General Terms and Conditions.

12. EVERGREEN PROVISION

When converting from Rate Schedule LS-2F to Rate Schedule LS-3F, Shipper may retain its existing Rate Schedule LS-2F evergreen provision. Otherwise, the following evergreen provisions are available under this Rate Schedule.

- 12.1 Five Year Notice Unilateral Evergreen Provision. If Transporter and Shipper agree to include a five year notice unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
 - (c) The termination notice required under Section 12.1(b) will be deemed given when posted on Transporter's Designated Site.
- 12.2 One Year Notice Unilateral Evergreen Provision. For a Service Agreement that has a primary term of at least 15 years, Transporter and Shipper may agree to include a one year notice unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, in which case the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least one year before the termination date.
 - (c) The termination notice required under Section 12.2(b) will be deemed given when posted on Transporter's Designated Site.
- 12.3 Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:

12. EVERGREEN PROVISION (Continued)

- (a) The established rollover period will be:
 - (i) one month for a Service Agreement with a primary term of less than one year; or
 - (ii) one year for a Service Agreement with a primary term of one year or more.
- (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) ten Business Days before the termination date if Section 12.3(a)(i) applies; or
 - (ii) one year before the termination date if Section 12.3(a)(ii) applies.
- (c) The termination notice required under Section 12.3(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

- 13. RATE SCHEDULE LS-2F CONVERSION TO RATE SCHEDULE LS-3F.
 - 13.1A Rate Schedule LS-2F Shipper may permanently convert all or part of its LS-2F service to a Rate Schedule LS-3F Service Agreement. Shipper's permanent conversion to Rate Schedule LS-3F service must be performed electronically using Transporter's Designated Site.
 - 13.2 The Storage Demand and Storage Capacity on the LS-3F Service Agreement must be maintained in the same proportion that exists on the converting Rate Schedule LS-2F Service Agreement.
 - 13.3 At the time of conversion, a converting Rate Schedule LS-2F Shipper may transfer its related Storage Capacity Balance to the corresponding Rate Schedule LS-3F Service Agreement; provided, however, the quantity transferred cannot be more than the Storage Capacity being converted.
 - 13.4 Conversions from Rate Schedule LS-2F service to Rate Schedule LS-3F service must become effective on the first day of a future month designated by Shipper.

14. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except as modified in the Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

RATE SCHEDULE LD-41 Liquefaction-LNG Delivery Service - Interruptible

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the purchase of a liquefaction-LNG delivery service from Transporter, when Shipper and Transporter have executed a Service Agreement for the purchase of such service under this Rate Schedule and Shipper has arranged for the related transportation of gas to the Plymouth LNG Facility under one of Transporter's transportation rate schedules and LNG deliveries to a connected delivery facility.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to the interruptible liquefaction-LNG delivery service rendered by Transporter to Shipper under the Service Agreement for such service.
- 2.2 Liquefaction Components. Liquefaction service under this Rate Schedule shall consist of the liquefaction by Transporter of gas transported to the Plymouth LNG Facility by Shipper under a separate Service Agreement. Delivery of natural gas by Shipper to Transporter for liquefaction shall be at the point of interconnection between Transporter's main transmission line and Transporter's Plymouth LNG Facility.
- 2.3 LNG Delivery Service Components. LNG delivery service under this Rate Schedule shall consist of the delivery of LNG upon liquefaction of Shipper's gas, as stated in Section 2.2, at the Plymouth LNG Facility by Transporter to Shipper. Delivery of LNG by Transporter to Shipper shall be at the point of interconnection between Transporter's Plymouth LNG Facility and LNG off-system delivery points directly connected to the Plymouth LNG liquefiers.
- 2.4 Character of Service. Service rendered to Shipper under this Rate Schedule, within the limitations set forth in this Rate Schedule and the Service Agreement, shall be interruptible and shall be subject to curtailment or interruption as provided herein and in the General Terms and Conditions.

RATE SCHEDULE LD-4I Liquefaction-LNG Delivery Service - Interruptible (Continued)

3. MONTHLY RATE

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the amounts specified in this Section 3, as applicable.

- 3.1 Liquefaction Service. The sum of (a) and (b) below:
 - (a) The volumetric charge will be the sum of the daily product of Shipper's gas per Dth of gas scheduled for liquefaction and the Volumetric Charge rate.
 - (b) The liquefaction charge will be the sum of the daily product of Shipper's gas per Dth scheduled for liquefaction and the Liquefaction Charge rate.

The amount paid by Shipper under Section 3.1 (b) will be trued-up on a calendar year basis and the true-up amount will be billed/refunded, as described in Section 14 of the General Terms and Conditions.

The unit rates that are applicable to this Rate Schedule shall be those as set forth from time to time in the currently effective Statement of Rates of this Tariff.

3.2 Discounted Recourse Rates.

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth in the Statement of Rates of this Tariff that are applicable to this Rate Schedule, or any superseding tariff. Such discounted Recourse Rates may apply to: specific volumes of gas, such as volumes of gas liquefied above or below a certain level; or all volumes if volumes exceed a certain level; and/or volumes of gas liquefied during specific time periods. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

RATE SCHEDULE LD-4I Liquefaction-LNG Delivery Service - Interruptible (Continued)

- 3.3 Negotiated Rates. Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in this Section 3.
- 3.4 Transportation Charges. The related transportation of gas to the Plymouth LNG Facility shall be subject to separate transportation charges under applicable Rate Schedules. The rates set forth in subparagraphs 3.1 through 3.3 are exclusive of the aforementioned transportation charges.

4. FUEL GAS REIMBURSEMENT

Shipper shall reimburse Transporter for fuel use in-kind, as detailed in Section 14 of the General Terms and Conditions upon liquefaction of Shipper's gas.

5. LIQUEFACTION FOR LNG DELIVERIES

- 5.1 Shipper may nominate liquefaction for LNG deliveries by specifying the quantity of LNG it desires delivered under this Rate Schedule. Transporter shall deliver the quantity of LNG so nominated, subject to the limitations set forth in this Rate Schedule. Such liquefaction for LNG deliveries will be in accordance with the priority of service and curtailment policy delineated in Section 12 of the General Terms and Conditions.
- 5.2 The notice given by Shipper to Transporter for liquefaction shall be in accordance with the NAESB WGQ Nominations Related Standards; provided, however, that commencement of actual LNG delivery shall be determined by system operating conditions. Transporter may batch, in its sole discretion, confirmed liquefaction nominations from all Plymouth LNG rate schedule Service Agreements over multiple days in order to operate the liquefiers in an efficient and responsible manner. Once liquefaction begins, Transporter will deliver the batched nominations to each individual Shipper in an operationally efficient manner, subject to system operating conditions.
- 5.3 Nominations shall be made by Shipper scheduling such tendered volumes on any day as transportation volumes delivered under a Service Agreement for liquefaction.

RATE SCHEDULE LD-4I Liquefaction-LNG Delivery Service - Interruptible (Continued)

6. TREATMENT OF LNG BOIL-OFF

LNG Boil-off Balance. Shipper's LNG Boil-off quantities received for redelivery at a Plymouth LNG Facility receipt point from a non-jurisdictional LNG facility will be reflected as a balance due from Transporter. This Boil-off balance will be deemed to be at the Plymouth LNG Facility. Shipper must nominate the Boil-off balance for re-liquefaction in order to reduce or eliminate such balance. Nominations for the re-liquefaction of Boil-off will be subject to the volumetric charge, the liquefaction charge and fuel use reimbursement. Tolerances regarding Shipper's Boil-off balance are detailed in Section 15 of the General Terms and Conditions.

7. EVERGREEN PROVISION

Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:

- (a) The established rollover period will be one month.
- (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least thirty days before the termination date.
- (c) Termination notice under Section 7(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

8. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except as modified in the Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

RATE SCHEDULE TF-2 Firm Redelivery Transportation

1. AVAILABILITY

This Rate Schedule is available to any party (hereinafter called "Shipper") for firm Redelivery Transportation of natural gas by Transporter through Transporter's mainline transmission system under the following conditions:

- (a) Shipper desires firm service and Transporter has available capacity to render such firm service for Shipper;
- (b) Transporter can commence the service contemplated without need for construction of any additional pipeline facilities except for facilities for which a facilities agreement has been entered into between Transporter an Shipper pursuant to Section 21 of the General Terms and Conditions of this Tariff;
- (c) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule; and
- (d) Shipper has a service agreement for storage service at a storage facility described in Section 10.1, below, or is an owner with storage rights in such storage facility, which will be the source of the quantities redelivered hereunder.

As used in this Rate Schedule, Transporter's mainline transmission system does not include Designated Laterals.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule shall apply to gas transported by Transporter for Shipper pursuant to the executed Service Agreement for service under this Rate Schedule. Shipper shall execute a separate Service Agreement for Redelivery Transportation from each Eligible Storage Facility Receipt Point to the delivery point(s) requested.
- 2.2 Character of Service. Redelivery Transportation rendered to Shipper under this Rate Schedule, up to Shipper's Transportation Contract Demand and subject to the limitations of this Rate Schedule and the executed Transportation Service Agreement, will be firm and will not be subject to curtailment except as expressly provided in the General Terms and Conditions. Redelivery Transportation rendered to Shipper under this Rate Schedule in excess of Shipper's Transportation Contract Demand is not firm.

2. APPLICABILITY AND CHARACTER OF SERVICE

2.3 Capacity Release Service. Capacity Release Service is service initiated pursuant to Section 22 of the General Terms and Conditions and an executed Service Agreement for Rate Schedule TF-2 service.

3. MONTHLY RATE(S)

Each month, Shipper will pay Transporter for service rendered under this Rate Schedule the sum of the amounts specified in this Section 3, as applicable.

3.1 Reservation Charge:

(a) The sum of the daily product of Shipper's Monthly Billing Quantity as specified in the executed Service Agreement and the reservation charge unit rate which is stated on Sheet No. 5 of this Tariff and the CRM Surcharge as stated on Sheet 5-B of this Tariff. The Monthly Billing Quantity shall be determined for each Shipper and Service Agreement as specified in Section 6 herein. Unless specifically adjusted pursuant to Section 3.3 herein, the Maximum Base Tariff Rate set forth on Sheet No. 5 will apply.

For capacity release service, the Reservation Charge is the sum of the daily product of the accepted reservation charge bid price which was bid by a Replacement Shipper or a Prearranged Replacement Shipper under the bidding procedures for capacity releases set forth in Section 22 of the General Terms and Conditions and the Monthly Billing Quantity applicable to the Replacement Shipper or the Prearranged Replacement Shipper.

- (b) Shipper shall commence payment of the Reservation Charges on the primary term begin date set forth in the Service Agreement, unless otherwise agreed to by the parties.
- 3.2 Volumetric Charge: An amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Shipper after transportation during the month, after reduction for fuel use reimbursement furnished in kind in accordance with the terms of the executed Service Agreement and Section 14 of the General Terms and Conditions, by (ii) the base volumetric transportation rate, as applicable, as set forth on Sheet No. 5 of this Tariff. Unless specifically adjusted pursuant to Section 3.3 herein, the Maximum Base Tariff Rate set forth on Sheet No. 5 shall apply.

3. MONTHLY RATE(S) (Continued)

3.3 Discounted Recourse Rates:

- (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific volumes of gas (such as volumes above or below a certain level or all volumes if volumes exceed a certain level), volumes of gas transported during specific time periods, and volumes of gas transported to specific delivery points, within specific corridors, or within other defined geographical areas. If Transporter discounts any Recourse Rates to any Shipper, Transporter will file with the Commission any required reports reflecting such discounts.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.4 Volumetric Release Charges: For Capacity Release service pursuant to Section 22 of the General Terms and Conditions which is provided under a volumetric bid rate the sum of (a), (b) and, if applicable, (c) below:
 - (a) The amount obtained by multiplying (i) the quantity of Dth scheduled for delivery by Transporter to Replacement Shipper or Prearranged Replacement Shipper after transportation during the month, after reduction for fuel use reimbursement furnished in kind in

3. MONTHLY RATE(S) (Continued)

accordance with the terms of the executed Service Agreement and Section 14 of the General Terms and Conditions, by (ii) the accepted volumetric bid made by a Replacement Shipper or a Prearranged Replacement Shipper not to exceed the maximum Rate Schedule TF-2 Volumetric Bid Reservation Charge as set forth on Sheet No. 5 of this Tariff.

- (b) The amount set forth in Section 3.2 hereof.
- (c) If the Releasing Shipper has specified a minimum average load factor volumetric commitment, an amount equal to the accepted volumetric bid times the difference, if positive, between (i) the specified average load factor times Replacement Shipper's or Prearranged Replacement Shipper's Transportation Contract Demand times the number of days the Transportation Service Agreement is in effect during the month, and (ii) the quantity of Dth delivered by Transporter to Replacement Shipper or Prearranged Replacement Shipper under the subject Service Agreement during the month.
- 3.5 Annual Overrun Charge: In addition to any other charges applicable to Redelivery Transportation services pursuant to this Section 3, each Shipper receiving Redelivery Transportation service shall pay Annual Overrun Charges to the extent that the quantity of Dth transported for Shipper, after reduction for fuel use reimbursement furnished in kind, during any twelve-month period ending September 30 is greater than Shipper's Annual Contract Quantity as defined in Section 6 herein. Each month Transporter shall determine the periodto-date quantity of Dth transported for Shipper and shall bill an annual overrun charge to Shipper if the quantity transported on a cumulative basis has exceeded the Annual Contract Quantity, but only to the extent that such overrun charges have not already been billed to Shipper during a prior month. The overrun charge billed during any month shall be the product of such previously unbilled overrun quantity in Dth and the per unit Annual Overrun Charge rate set forth on Sheet No. 5 of this Tariff.
- 3.6 Daily Overrun Transportation Charge: For any Scheduled Quantities in excess of Shipper's Transportation Contract Demand, as described in Sections 5 and 7, Shipper shall pay

3. MONTHLY RATE(S) (Continued)

for the excess gas transported on any such day an amount obtained by multiplying (i) the quantity of Dth in excess of Transportation Contract Demand by (ii) the maximum base scheduled daily overrun transportation rate per Dth as set forth on Sheet No. 5 of this Tariff, unless otherwise agreed to by Shipper and Transporter.

Any Shipper receiving daily overrun transportation service shall also be responsible for the fuel reimbursement provided for in Section 13 of this Rate Schedule.

- 3.7 Negotiated Rates: Notwithstanding the general provisions of this Section 3, if Transporter and Shipper mutually agree to Negotiated Rates for service hereunder, such Negotiated Rates will apply in lieu of the otherwise applicable rates identified in Sections 3.1, 3.2 and 3.3 of this Rate Schedule.
- 3.8 Charges for Capacity Release Service: The rates for capacity release service are set forth in Sheet No. 5. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, wherein the Replacement Shipper has not agreed to pay the Maximum Base Tariff Rate or a percentage of the Maximum Base Tariff Rate (as it may change from time-to-time), the Replacement Shipper will be obligated to pay:
 - (a) for capacity release transactions that are subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the lesser of the awarded bid rate and the new Maximum Base Tariff Rate unless the awarded bid rate is less than the new minimum rate, in which case the new minimum rate will apply for the remaining term of the release.
 - (b) for capacity release transactions that are not subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release.

- 3. MONTHLY RATE(S) (Continued)
 - 3.9 Facilities Charge: If Transporter and Shipper enter into a facilities agreement pursuant to Section 21 of the General Terms and Conditions for Transporter to construct facilities and for Shipper to pay a facilities charge, the facilities charge will be set forth on Exhibit C to an executed Service Agreement.
- 4. MINIMUM MONTHLY BILL

Unless Transporter and Shipper mutually agree otherwise, the Minimum Monthly Bill will consist of the Reservation Charge specified in Section 3.1 of this Rate Schedule, as applicable.

5. TRANSPORTATION CONTRACT DEMAND

The Transportation Contract Demand is the maximum quantity of Gas, expressed in Dth, that Transporter is obligated to receive (exclusive of fuel reimbursement furnished in-kind pursuant to Section 14 of the General Terms and Conditions), transport and deliver for Shipper on a firm basis on any one Gas Day, as specified in an executed Service Agreement for service under this Rate Schedule. Transporter's service obligation is limited to Shipper's Transportation Contract Demand as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

Transportation Contract Demand initially may not exceed the quantity of gas Shipper is entitled to withdraw on a daily basis under the related firm storage agreement, or ownership rights to firm withdrawal capacity, and, in the aggregate for any twelve-month period ending September 30, may not exceed Annual Contract Quantity. However, subsequent reductions in such Shipper's firm withdrawal rights will not require corresponding changes to the related Transportation Contract Demand.

As long as the Transportation Contract Demand and Annual Contract Quantity, as adjusted for any capacity releases, are not exceeded, Transporter shall be obligated to receive up to Shipper's Maximum Daily Quantity (MDQ) at the Primary Receipt Point and to deliver up to Transporter's Maximum Daily Delivery Obligation (MDDO) at each Primary Delivery Point at pressures at least as great as the pressures specified in Shipper's TF-2 Service Agreement, on a firm basis, as such MDQ and MDDO are adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions. Capacity that is posted pursuant to Section 25.4 of the General Terms and Conditions will be available for increases in Transportation Contract Demand only pursuant to Section 25 of the General Terms and Conditions.

The MDQ at the Primary Receipt Point specified in a Service Agreement for service under this Rate Schedule, must equal the specified Transportation Contract Demand. The aggregate MDDO at the Primary Delivery Points, as specified in a Service Agreement for service under this Rate Schedule, must equal the specified Transportation Contract Demand, except for Service Agreements subject to Section 11 and Service Agreements that have aggregate MDDOs in excess of Transportation Contract Demand as a result of the grandfathering of pre-existing conjunctive nomination rights under the bundled storage restructuring approved in Docket No. RP93-5-011.

- 6. ANNUAL CONTRACT QUANTITY AND MONTHLY BILLING QUANTITY
 - 6.1 Annual Contract Quantity: The Annual Contract Quantity is a quantity in Dth set forth in each executed Service Agreement and, as adjusted for any capacity releases pursuant to Section 22 of the General Terms and Conditions, represents the quantity of transportation service which each Shipper may receive during each twelve-month period ending September 30 without being subject to the Annual Overrun Charges described in Section 3.5 herein. A Shipper's Annual Contract Quantity initially may not be less than such Shipper's Storage Capacity specified in the related firm storage Service Agreement initiated pursuant to any storage service rate schedule contained in this Tariff (or in the tariff of any third party providing storage services to Shipper) or such Shipper's related entitlement to firm storage capacity as an owner in the storage facility. Subsequent increases in such Shipper's Storage Capacity or ownership rights to firm storage capacity will not require corresponding changes to the related Annual Contract Quantity.

The Annual Contract Quantity may be increased at any time, subject to availability of capacity, but may only be decreased due to permanent releases of capacity pursuant to Section 22.5 of the General Terms and Conditions or in connection with the filing of a general Section 4 rate case by Transporter. Transporter shall provide Shippers with an opportunity to make such an election prior to filing such rate applications.

6.2 Monthly Billing Quantity: The Monthly Billing Quantity is a quantity in Dth set forth in each executed Service Agreement and represents the quantity used by Transporter to bill the Reservation Charges specified in Section 3.1 to each Shipper. The Monthly Billing Quantity for each Service Agreement shall be derived by dividing Shipper's Annual Contract Quantity by 365 days.

First Revised Sheet No. 105 Superseding Substitute Original Sheet No. 105

RATE SCHEDULE TF-2 Firm Redelivery Transportation (Continued)

7. SCHEDULED OVERRUN TRANSPORTATION

On any day Shipper nominates quantities of gas in excess of Shipper's Transportation Contract Demand or Annual Contract Quantity specified in the executed Service Agreement, Transporter will schedule such excess quantities in accordance with the priority of service and curtailing policy delineated in Section 12 of the General Terms and Conditions of Transporter's Tariff.

8. CREDITING OF ANNUAL OVERRUN CHARGES

- 8.1 General. To the extent that Transporter receives annual overrun charges as specified in Section 3.5 associated with services provided under this Rate Schedule, Transporter shall credit one hundred (100) percent of such revenues to its eligible firm transportation Shippers as further described below.
- 8.2 Applicability. Revenue credits from Annual Overrun Charges received by Transporter shall be provided to Transporter's Rate Schedules TF-1 (Large Customers) and TF-2 Shippers (applicable firm Shippers), excluding Shippers receiving service at a discounted firm transportation rate or under a Capacity Release Service Agreement.
- 8.3 Timing of Credits. Within 30 days after the accounting closing for September of each year, Transporter shall determine the total amount of the applicable Rate Schedule TF-2 Annual Overrun Charge revenues received during the twelve-month period ending September 30 of such year and the distribution of the revenue credits due to eligible Shippers as described below. Such revenue credits shall be reflected as a credit billing adjustment to the next invoices rendered to the eligible Shippers. In the event that such credit billing adjustment would result in a credit total invoice to any Shipper, Transporter will refund the excess credit billing adjustment to the Shipper in cash within 15 days after determination of the amount of the credit due to the Shipper.
- 8.4 Exclusion. Revenue credits shall not be awarded for that portion of revenues associated with Annual Overrun volumes that relate to the recovery, by Transporter, of volumetric transportation charges, nor shall they be awarded to any Shipper who received an annual overrun charge penalty any time during the twelve-month period ending September 30 of the applicable year.
- 8.5 Distribution Method. The revenue credits shall be credited to each eligible Shipper on a pro-rata basis in proportion to the Maximum Base Tariff rate reservation charges (exclusive of surcharges) billed to each applicable firm Shipper during the twelve-month period ending September 30 of the applicable year divided by the total reservation charges billed to all applicable firm Shippers for the same period.

- 8. CREDITING OF ANNUAL OVERRUN CHARGES (Continued)
 - 8.6 Interest. Transporter shall pay interest to Shippers if the revenue credits specified herein are not distributed in a timely manner as specified in Section 8.3, above. Such interest shall be calculated based upon the rate of interest specified in Section 154.501(d) of the Commission's regulations.
- 9. RECEIPT AND DELIVERY POINT FLEXIBILITY
 - 9.1 Permanent Changes to Primary Delivery Points. Subject to the availability provisions of this Rate Schedule and Section 9.4 below, any Shipper may permanently change primary delivery points by amending Exhibit A of the Service Agreement.
 - 9.2 Receipt Point Changes. Any Shipper electing to receive Rate Schedule TF-2 service shall not be entitled to change primary receipt points. Rate Schedule TF-2 Shippers shall not be entitled to utilize the alternate receipt point provisions described in Section 11 of Rate Schedule TF-1, nor shall any Replacement Shipper or Prearranged Replacement Shipper be entitled to change the primary receipt point.
 - 9.3 Use of Alternate Delivery Points on a Temporary Basis.
 - (a) All TF-2 Shippers may use any physical delivery point without amending Exhibit A of the Service Agreement. Such points will be available for the delivery of gas on a firm basis, with a scheduling priority superior to interruptible, balancing and overrun transportation service requested through the affected points, but inferior to firm service requested by Shippers using such points as primary delivery points.
 - (b) Transporter shall schedule service at alternate delivery points on a daily basis pursuant to Section 14 of the General Terms and Conditions.
 - (c) The scheduling of service at alternate delivery points under a Service Agreement will not result in the loss of firm contract rights to a Shipper's primary delivery points as specified in the Shipper's Service Agreement.

- 9. RECEIPT AND DELIVERY POINT FLEXIBILITY (Continued)
 - 9.4 Procedures for Requesting Permanent Delivery Point Changes. Any Shipper who wishes to amend only the primary delivery points or the associated Maximum Daily Delivery Obligation named in Exhibit A of its Service Agreement will electronically request and execute on Transporter's Designated Site an amendment to the applicable Service Agreement by 1:00 pm Central Clock Time (noon Mountain Clock Time) four (4) business days prior to the first of the month for which the change is desired, or for changes to occur during a month, after the first of the month, two (2) business days prior to the commencement of service, unless otherwise agreed to by the parties. If Transporter determines that a delivery point change request can be honored, the amendment to the Service Agreement will be executed by Transporter. The change will become effective on the later of the requested service date or the date executed by Transporter, provided that such date shall not exceed fifteen days from receipt of the amendment by Transporter. Notice of the resulting changes in available delivery point capacity will be posted to Transporter's Designated Site at least one (1) business day prior to implementation of such change.

Firm delivery point and mainline capacity will be posted pursuant to Section 25.4 of the General Terms and Conditions and will be available for permanent delivery point changes only pursuant to the procedures outlined in Section 25 of the General Terms and Conditions.

- 9.5 Transporter's Maximum Service Obligation. The total volumes nominated for service on any day for all receipt or delivery points must not exceed Transportation Contract Demand under a Shipper's firm Service Agreement, except as otherwise provided in Section 7 of this Rate Schedule.
- 10. ELIGIBLE STORAGE FACILITY RECEIPT POINTS AND SERVICE REQUESTS
 - 10.1 Eligible Storage Facility Receipt Points. Transportation service provided under this Rate Schedule shall be only for transportation from an Eligible Storage Facility Receipt Point to the delivery point(s) specified in the Service Agreement. An Eligible Storage Facility Receipt Point is a mainline receipt point at which any storage facility is connected to Transporter's system by a pipeline or other facilities installed solely to make that connection or at which any storage facility located in the service area of any of Transporter's on-system local distribution customers accesses Transporter's system.

- 10. ELIGIBLE STORAGE FACILITY RECEIPT POINTS AND SERVICE REQUESTS (Continued)
 - 10.2 Service Requests. Requests for service shall be made pursuant to the procedures contained in Section 28 of the General Terms and Conditions. Transporter shall be entitled to deny transportation service under this Rate Schedule if, in Transporter's sole discretion, Transporter does not have the capacity to provide the firm Redelivery Transportation services requested from any storage facility without adding mainline transmission facilities.
- 11. MAXIMUM DAILY DELIVERY OBLIGATIONS

Transporter shall not be obligated on any day to transport from the Plymouth LNG Facility a volume of gas for Shipper which, in conjunction with other services by Transporter for Shipper under any other firm rate schedule included in this Tariff, shall be in excess of the capacity of Transporter's facilities. Transporter shall aggregate the maximum daily delivery obligations, not to exceed Transportation Contract Demand, under all firm rate schedules included in this Tariff in determining the capacity of Transporter's facilities at delivery points to accommodate both transportation redeliveries of gas originating from the Plymouth LNG Facility and other firm services under all other firm rate schedules.

Transporter will preserve Shippers' historical conjunctive billing flexibility and will not degrade the redelivery transportation service related to Rate Schedule TF-2 services which originated from pre-existing conjunctive nomination rights under the bundled storage restructuring approved in Docket No. RP93-5-011.

12. FUEL GAS REIMBURSEMENT AND BTU BALANCING

Refer to Section 14 of the General Terms and Conditions.

13. SHIPPER'S ARRANGEMENTS PRIOR TO RECEIPT AND AFTER DELIVERY

Refer to Section 14 of the General Terms and Conditions.

14. EVERGREEN PROVISION

- 14.1 Standard Unilateral Evergreen Provision. If Transporter and Shipper agree to include a standard unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
 - (c) The termination notice required under Section 14.1(b) will be deemed given when posted on Transporter's Designated Site.
- 14.2 Standard Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a standard bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be:
 - (i) one month for Service Agreements with a primary term of less than one year; or
 - (ii) one year for Service Agreements with a primary term of one year or more.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) ten Business Days before the termination date if Section 14.2(a)(i) applies; or
 - (ii) one year before the termination date if Section 14.2(a)(ii) applies.
 - (c) The required termination notice under Section 14.2(b) will be deemed given when posted on Transporter's Designated Site. If

14. EVERGREEN PROVISION (Continued)

Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

- 14.3 Grandfathered Unilateral Evergreen Provision. If Shipper's Service Agreement contains a grandfathered unilateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year, at Shipper's sole option.
 - (b) Shipper may terminate all or any portion of service under its Service Agreement either at the expiration of the primary term, or upon any anniversary thereafter, by giving written notice to Transporter so stating at least twelve months in advance.
 - (c) Shipper also will have the sole option to enter into a new Service Agreement for all or any portion of the service under its Service Agreement at or after the end of the primary term of its Service Agreement. It is Transporter's and Shipper's intent that this provision provide Shipper with a "contractual right to continue such service" and to provide Transporter with concurrent pregranted abandonment of any volume that Shipper terminates within the meaning of 18 CFR 284.221(d)(2)(i) as promulgated by Order No. 636 on May 8, 1992.
 - (d) The termination notice required under Section 14.3(b) will be deemed given when posted on Transporter's Designated Site.

15. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions contained in this Tariff, except Sections 13 and 16 and as modified in the executed Service Agreement, are applicable to this Rate Schedule and are hereby made a part hereof.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 112 Through 114 Superseding Substitute Original Sheet Nos. 112 Through 114

RESERVED FOR FUTURE USE

RATE SCHEDULE DEX-1 Deferred Exchange of Storage Gas

1. AVAILABILITY

This Rate Schedule is available as provided herein to any party (Shipper) for the deferred exchange of natural gas in storage under the following conditions:

- (a) Shipper has Working Gas Inventory in the Target Storage Facility;
- (b) Transporter needs to increase the quantity of gas available to it for system balancing in the Target Storage Facility; and
- (c) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule will apply to the deferred exchange of storage gas, consisting of a transfer by Shipper to Transporter of some or all of Shipper's Working Gas Inventory in the Target Storage Facility and the delivery by Transporter to Shipper during a future time period of the Deferred Exchange Quantity at the Deferred Exchange Storage Facility. At Shipper's option, Transporter will deliver the Deferred Exchange Quantity either by performing a transfer to Shipper's account in the Deferred Exchange Storage Facility or by making the Deferred Exchange Quantity available to Shipper at the Deferred Exchange Storage Facility receipt point on Transporter's system.
- 2.2 Character of Service. Service rendered to Shipper under this Rate Schedule will be firm up to Shipper's Deferred Exchange Quantity and shall not be subject to curtailment except as expressly provided in the General Terms and Conditions. Rate Schedule DEX-1 service will be offered when Transporter, in its sole discretion, determines that it needs to increase the level of storage gas available to it for system balancing at the Target Storage Facility in order to enhance or maintain the operational integrity of its system. Service rendered under this Rate Schedule in excess of Shipper's Deferred Exchange Quantity is not firm.

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - 2.2 Character of Service. (Continued)

Transporter will notify Shippers of its desire to provide Rate Schedule DEX-1 service by posting the following information on its Internet web site for a minimum period of 24 hours:

- (a) the Target Storage Facility;
- (b) the quantity of gas in Dth that Transporter wishes to acquire;

2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)

- (c) the date by which Shipper must transfer its storage gas to Transporter;
- (d) the Deferred Exchange Storage Facility;
- (e) the Deferred Exchange Period; and
- (f) the Daily Delivery Quantity.

Interested Shippers may submit bids for any quantity, up to the lesser of Shipper's Working Gas Inventory in the Target Storage Facility or the quantity posted. Unless Transporter has specified in its posting that it is willing to accept Negotiated Rate bids, the rate bid cannot exceed the maximum base deferred exchange rate for service under Rate Schedule DEX-1 as set forth on Sheet No. 6 of this Tariff. If the aggregate quantity bid by all Shippers exceeds the quantity posted, service will be awarded pursuant to the priority of service provisions in Section 12 of the General Terms and Conditions.

If applicable, Negotiated Rate bids will be evaluated in accordance with Section 19.3 of the General Terms and Conditions.

3. RATE

Shipper will pay Transporter a total amount equal to the product of the Deferred Exchange Quantity and the rate per Dth that Shipper has bid, plus applicable surcharges. Transporter will submit an invoice to Shipper for the total amount owed on or before the tenth $10^{\rm th}$ day of the month immediately following the month during which Shipper delivers the Deferred Exchange Quantity to Transporter.

Gas that is transported on Transporter's system to the Target Storage Facility and/or from the Deferred Exchange Storage Facility will be subject to transportation and fuel charges under the applicable transportation rate schedule. Likewise, gas that is stored on Transporter's system prior to Shipper's delivery to Transporter and/or after Transporter's delivery to Shipper will be subject to appropriate charges under the applicable storage rate schedule.

4. FUEL REIMBURSEMENT

Service rendered under this Rate Schedule is subject to fuel reimbursement in accordance with Section 14 of the General Terms and Conditions. However, if Transporter offers to provide Rate Schedule DEX-1 service in accordance with Section 2.2 of this Rate Schedule and no bids are submitted, or if the total quantity bid is less than the total Deferred Exchange Quantity needed, Transporter may post an offer to provide Rate Schedule DEX-1 service for the Deferred Exchange Quantity for which no bids were received, and the fuel reimbursement requirement for that quantity will be waived.

5. DELIVERY OF DEFERRED EXCHANGE QUANTITY BY TRANSPORTER

- 5.1 General Procedure. During the Deferred Exchange Period, Transporter will deliver the Deferred Exchange Quantity at the Daily Delivery Quantity reflected in Exhibit A to Shipper's Service Agreement, subject to any withdrawal limitations at the Deferred Exchange Storage Facility.
- 5.2 Shipper may request Transporter to deliver the Deferred Exchange Quantity during a time period other than the Deferred Exchange Period specified on Exhibit A of the Service Agreement or to increase or decrease the Daily Delivery Quantity specified on Exhibit A. Such requests will be accommodated if Transporter, in its sole discretion, determines that accommodating such requests will not jeopardize the operational integrity of its system and will not interfere with firm service to other Shippers.

If Shipper is unable, for any reason, to accept all or any portion of the Deferred Exchange Quantity under the terms specified in Exhibit A of Shipper's Service Agreement, then Shipper will reimburse Transporter for any penalties and/or additional charges related to the Deferred Exchange Quantity that Transporter may incur. Further, upon expiration of the Deferred Exchange Period or any mutually agreeable extension thereof, Transporter will take title to any remaining Deferred Exchange Quantity, free and clear of any claims by Shipper.

6. DEFINITIONS

The following terms, when used in this Rate Schedule, will have the following meanings:

- 6.1 Daily Delivery Quantity: The quantity of gas in Dth that Transporter agrees to deliver to Shipper each day during the Deferred Exchange Period.
- 6.2 Deferred Exchange Period: The period of time during which Transporter agrees to deliver the Deferred Exchange Quantity to Shipper.
- 6.3 Deferred Exchange Quantity: The quantity of gas in Dth that Shipper agrees to transfer to Transporter at the Target Storage Facility and that Transporter agrees to subsequently deliver to Shipper at the Deferred Exchange Storage Facility.
- 6.4 Deferred Exchange Storage Facility: The gas storage facility where Transporter will deliver gas to Shipper.
- 6.5 Target Storage Facility: The storage facility at which Transporter wishes to increase the gas available to it for system balancing.
- 6.6 Working Gas Inventory: The quantity of working gas in Dth stored in Shipper's account.

7. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the following Sections of the General Terms and Conditions contained in this Tariff, except as modified in the executed Service Agreement: Sections 1 through 12, 14, 17, 18, 19, 26, 27, 28 and 32.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 119 Through 124 Superseding Substitute Original Sheet Nos. 119 Through 124

RESERVED FOR FUTURE USE

RATE SCHEDULE PAL Park and Loan Service

1. AVAILABILITY

- 1.1 This Rate Schedule is available to any party (hereinafter called "Shipper") for the parking and loaning of natural gas under the following conditions:
 - (a) Shipper has made a valid request for service under this Rate Schedule by establishing a credit limit satisfactory to Transporter pursuant to Section 5.10 of the General Terms and Conditions and by providing Transporter with the information necessary to fill in the blanks in the Form of Service Agreement for service under this Rate Schedule; and
 - (b) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 This Rate Schedule applies to all park and loan service rendered by Transporter for Shipper pursuant to Transporter's blanket certificate under Part 284 of the regulations of the Commission and the executed Service Agreement for service under this Rate Schedule.
- 2.2 Service under this Rate Schedule is interruptible and will be scheduled in accordance with the priority of service provisions delineated in Section 12 of the General Terms and Conditions and only to the extent that Transporter, in its reasonable discretion, determines it has sufficient operational flexibility to temporarily utilize a portion of its line pack, storage service rights at the Jackson Prairie Storage Project and/or contracted Clay Basin storage to provide such service without adversely affecting its ability to accommodate its system balancing requirements.

2.3 Park Service under this Rate Schedule consists of:

(a) Transporter's receipt of gas (Parked Quantity) for Shipper's account at the logical point (Park Point) designated in the executed Service Agreement, either at the Clay Basin Park Point adjacent to Transporter's physical Clay Basin Receipt/Delivery Point, at the Jackson Prairie Park Point adjacent to Transporter's physical Jackson Prairie Receipt/Delivery Point, or any other mutually agreeable non-storage Park Point, up to the Maximum Park Quantity specified in the executed Service Agreement; and

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - (b) Transporter's subsequent delivery of the Parked Quantity of gas to Shipper at the Park Point designated in the executed Service Agreement.
 - 2.4 Loan service under this Rate Schedule consists of:
 - (a) Transporter's delivery of gas (Loaned Quantity) for Shipper's account at the logical point (Loan Point) designated in the executed Service Agreement, either at the Clay Basin Loan Point adjacent to Transporter's physical Clay Basin Receipt/Delivery Point, at the Jackson Prairie Loan Point adjacent to Transporter's physical Jackson Prairie Receipt/Delivery Point, or any other mutually agreeable non-storage Loan point, up to the Maximum Loan Quantity specified in the executed Service Agreement; and
 - (b) Transporter's subsequent receipt of the Loaned Quantity from Shipper at the Loan Point designated in the executed Service Agreement.
 - 2.5 Transportation of gas for or on behalf of Shipper to or from the Park Point or Loan Point will not be performed under this Rate Schedule. Shipper is obligated to make arrangements with Transporter or other parties for the related transportation services to and from the Park Point or Loan Point. Transporter's receipts or deliveries of gas for Shipper's account at the Park Point or Loan Point under this Rate Schedule will be scheduled only for quantities that are matched by Scheduled Quantities for the related transportation services to or from the Park Point or Loan Point or for quantities that are matched by a confirmed transfer (a) between Transporter's storage account and Shipper's storage account in the storage field adjacent to the Park Point or Loan Point or (b) between Shippers at the same non-storage Park or Loan Point.

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - 2.6 Transfers: Shippers must notify Transporter's Marketing Services personnel, via email, of their intent to transfer such inventory. Transfers of quantities may not result in any Shipper taking title to quantities that exceed such Shipper's contractual rights under this or other Rates Schedules.
 - (a) Non-Storage Points: Shippers that are subject to this Rate Schedule may agree to transfer their respective Park Quantity or Loan Quantity to another PAL or TPAL contract at the same non-storage point. Transfers to a PAL or TPAL contract can only occur if the receiving contract rate is equal to or greater than the transferring Shipper's contractual rate.
 - (b) Clay Basin Storage Point: Shippers that are subject to this Rate Schedule may agree to transfer their respective Park Quantity or Loan Quantity at the Clay Basin storage facility to another Shipper with a Clay Basin Park Quantity or Loan Quantity under Rates Schedules PAL or TPAL, or to any capacity holder in the Clay Basin storage facility. Transfers to PAL or TPAL contract can only occur if the receiving contract rate is equal to or greater than the transferring Shipper's contractual rate. Transfers between PAL and another Clay Basin capacity holder requires an agreement among the PAL Shipper, Transporter and the other Clay Basin capacity holder. Upon agreement of the parties, Transporter may utilize its Clay Basin contracted capacity on behalf of a Rate Schedule PAL Shipper to transfer such Shipper's PAL quantity to or from the other Clay Basin capacity holder's inventory account.
 - (c) Jackson Prairie Storage Point: Shippers that are subject to this Rate Schedule may agree to transfer their respective Jackson Prairie Park or Loan Quantities to any capacity holder in the Jackson Prairie Storage facility under Rate Schedules SGS-2F, SGS-2I, TPAL, and PAL whose contractual rate is equal to or greater than the transferring Shipper's contractual rate.

Pursuant to the January 15, 1998 Gas Storage Project Agreement, owners of the Jackson Prairie Storage Project may transfer portions of their respective available working gas inventories, as defined in the Project Agreement, to each other. Upon agreement of the parties, and subject to the terms of the Project Agreement, Transporter may utilize its ownership account on behalf of a Rate Schedule PAL Shipper to transfer such Shipper's PAL quantity to an owner's available working gas

inventory account. Conversely, an owner may transfer its available working gas inventory to a Rate Schedule PAL Shipper's PAL contract.

3. RATE

- 3.1 Transporter's Maximum Base Tariff Rate, as set forth on the Statement of Rates, will apply to all service rendered under this Rate Schedule unless Transporter agrees to discount its Recourse Rate as described below or unless Shipper and Transporter mutually agree to a Negotiated Rate pursuant to Section 19 of the General Terms and Conditions.
- 3.2 Discounted Recourse Rates.
 - (a) Transporter reserves the right to discount, on a non-discriminatory basis, the Recourse Rate for any individual Shipper under any Service Agreement without discounting any other Recourse Rate for that or another Shipper, provided, however, that such discounted Recourse Rate will not be less than the minimum base rate set forth on the Statement of Rates in this Tariff. Such discounted Recourse Rates may apply to specific quantities of gas (such as quantities above or below a certain level or all quantities if quantities exceed a certain level) or service during specific time periods.
 - (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.3 Transporter will establish on its Designated Site quantities and rates that Transporter will accept for new PAL service. Posted minimum rates apply to new service only. For quantities previously parked/loaned, Transporter will schedule in accordance with the priority of service provisions and curtailment policy in Section 12 of the General Terms and Conditions based on maximum/minimum quantities established on its Designated Site.
- 3.4 For each billing period for service under a Service Agreement, Shipper will pay Transporter a total amount equal to the sum of the daily products of (1) the Daily Rate set forth in the executed Service Agreement and (2) the maximum Parked or Loaned Quantity, as applicable, on such day.
- 3.5 ACA surcharges and fuel reimbursement will not apply to service under this Rate Schedule.

4. ALLOCATION OF SERVICE

In allocating or interrupting service among Shippers under this Rate Schedule, Transporter will adhere to the priority of service provisions in Section 12 of the General Terms and Conditions.

5. SERVICE RESTRICTIONS AND PENALTIES

- 5.1 Shipper may be required to remove some or all of the Parked Quantity or return some or all of the Loaned Quantity prior to termination of its Service Agreement if Transporter, in its reasonable discretion, determines such action is necessary to accommodate higher priority services, maintain system integrity or to accommodate system balancing requirements for firm services. Notification of any such requirements will be made to Shipper by posting on Transporter's Designated Site and, if specified by Shipper on the Business Associate Information form, via an E-mail to the Shipper. Transporter's notification will indicate the time within which specified Parked Quantities shall be removed by Shipper and/or specified Loaned Quantities shall be returned to Transporter, consistent with Transporter's operating conditions. In no event will the specified time frame be less than three (3) gas days from the date of Transporter's notification.
- 5.2 Any Parked Quantity not removed within the time frame specified by Transporter's notice and any Parked Quantity not removed by the termination date of the Service Agreement will become the property of Transporter at no cost to Transporter, free and clear of any adverse claims. Any Loaned Quantity not returned within the time frame specified by Transporter's notice and any Loaned Quantity not returned by the termination date of the Service Agreement will be deemed sold to Shipper under Section 32 of the General Terms and Conditions at 150% of the highest weekly weighted average price at NW Wyoming Pool, NW south of Green River, NW Stanfield, NW Sumas, or El Paso Bondad as reflected in the Daily Price Survey as published in "Gas Daily" for the week in which the Shipper failed to comply.

Notwithstanding the above, Transporter may allow additional time to comply following notification to remove the Parked Quantity or return the Loaned Quantity when operational conditions permit and such additional time does not impair Transporter's ability to provide reliable firm service. Furthermore, if Shipper makes a timely nomination to comply with a notification to remove the Parked Quantity or return the Loaned Quantity and such nomination is not scheduled and confirmed due to Transporter's lack of capacity to schedule and confirm a matching nomination by Shipper of firm transportation service within the Shipper's primary path, then the specified compliance time frame will be extended by one day for each day Transporter is unable to render such transportation. Transporter will not extend the specified compliance time frame if Transporter is unable to provide the related

5. SERVICE RESTRICTIONS AND PENALTIES (Continued)

transportation under a firm Service Agreement outside of Shipper's primary path or on an overrun basis or under a Rate Schedule TI-1 Service Agreement.

- 5.3 Revenue received by Transporter resulting from the deemed sale of Loaned Quantities will be used by Transporter to purchase gas to replace the gas deemed sold to the Shipper, after first taking into account any offsetting confiscated Parked Quantities that have not been reflected in Transporter's fuel reimbursement factor. Any excess revenues will be refunded to Shippers after March 1 of the applicable calendar year using the refund methodology specified in Section 32.2 of the General Terms and Conditions. Any confiscated Parked Quantities in excess of those used to offset Loaned Quantities that are deemed sold will be refunded to customers through Transporter's fuel reimbursement factor pursuant to Section 14.12 of the General Terms and Conditions that goes into effect April 1 of each year.
- 6. RESERVED FOR FUTURE USE
- 7. EVERGREEN PROVISION
 - 7.1 Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one day.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least three days before the termination date.
 - (c) Termination notice under Section 7.1(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

First Revised Sheet No. 130 Superseding Substitute Original Sheet No. 130

RATE SCHEDULE PAL Park and Loan Service (Continued)

8. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the General Terms and Conditions contained in this Tariff.

RATE SCHEDULE TPAL Term Park and Loan Service

1. AVAILABILITY

- 1.1 This Rate Schedule is available to any party (hereinafter called "Shipper") for the parking and loaning of natural gas under the following conditions:
 - (a) Shipper has made a valid request for service under this Rate Schedule by establishing a credit limit satisfactory to Transporter pursuant to Section 5.10 of the General Terms and Conditions and by providing Transporter with the information necessary to fill in the blanks in the Form of Service Agreement for service under this Rate Schedule; and
 - (b) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 This Rate Schedule applies to all park and loan service rendered by Transporter for Shipper pursuant to Transporter's blanket certificate under Part 284 of the regulations of the Commission and the executed Service Agreement for service under this Rate Schedule.
- 2.2 Service under this Rate Schedule is interruptible and will be scheduled in accordance with the priority of service provisions delineated in Section 12 of the General Terms and Conditions and only to the extent that Transporter, in its reasonable discretion, determines it has sufficient operational flexibility to temporarily utilize a portion of its line pack, storage service rights at the Jackson Prairie Storage Project and/or contracted Clay Basin storage to provide such service without adversely affecting its ability to accommodate its system balancing requirements.

2.3 Park Service under this Rate Schedule consists of:

(a) Transporter's receipt of gas (Parked Quantity) for Shipper's account at the logical point (Park Point) designated in the executed Service Agreement, either at the Clay Basin Park Point adjacent to Transporter's physical Clay Basin Receipt/Delivery Point, at the Jackson Prairie Park Point adjacent to Transporter's physical Jackson Prairie Receipt/Delivery Point, or any other mutually agreeable non-storage Park Point, up to the Maximum Parked Quantity specified in the executed Service Agreement; and

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - (b) Transporter's subsequent delivery of the Parked Quantity of gas to Shipper at the Park Point designated in the executed Service Agreement.
 - 2.4 Loan service under this Rate Schedule consists of:
 - (a) Transporter's delivery of gas (Loaned Quantity) for Shipper's account at the logical point (Loan Point) designated in the executed Service Agreement, either at the Clay Basin Loan Point adjacent to Transporter's physical Clay Basin Receipt/Delivery Point, at the Jackson Prairie Loan Point adjacent to Transporter's physical Jackson Prairie Receipt/Delivery Point, or any other mutually agreeable non-storage Loan Point, up to the Maximum Loaned Quantity specified in the executed Service Agreement; and
 - (b) Transporter's subsequent receipt of the Loaned Quantity from Shipper at the Loan Point designated in the executed Service Agreement.
 - 2.5 Service under this Rate Schedule will have defined quantities and a fixed term of when and where parking or loaning activity can occur. If, due to operating conditions, Transporter is unable to schedule Shipper's nominations, within the park/loan schedule limits identified in the Service Agreement Transporter and Shipper may mutually agree to amend the Service Agreement with a revised park/loan activity schedule when operating conditions permit, if Transporter, in its reasonable discretion, determines such action will not prevent Transporter from accommodating higher priority services, maintaining its system integrity or accommodating system balancing requirements for firm services.
 - 2.6 If Transporter determines quantities are available to be parked or loaned under this Rate Schedule, such quantities will be subject to the competitive bid procedures described in Section 25.5 of the General Terms and Conditions, except that the competitive bid timelines described in Section 25.5(b) of the General Terms and Conditions shall not apply. All competitive bid postings for service under Rate Schedule TPAL will be for a bid period of no less than one hour, provided the bid period will begin during Transporter's normal business hours. Transporter will notify Shippers of the TPAL competitive bid posting via its Designated Site.

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - 2.7 Transportation of gas for or on behalf of Shipper to or from the Park Point or Loan Point will not be performed under this Rate Schedule. Shipper is obligated to make arrangements with Transporter or other parties for the related transportation services to and from the Park Point or Loan Point. Transporter's receipts or deliveries of gas for Shipper's account at the Park Point or Loan Point under this Rate Schedule will be scheduled only for quantities that are matched by Scheduled Quantities for the related transportation services to or from the Park Point or Loan Point or for quantities that are matched by a confirmed transfer (a) between Transporter's storage account and Shipper's storage account in the storage field adjacent to the Park Point or Loan Point or (b) between Shippers at the same non-storage Park or Loan Point.

- 2. APPLICABILITY AND CHARACTER OF SERVICE (Continued)
 - 2.8 Transfers: Transporter will post on Transporter's Designated Site whether TPAL transfers will be allowed. If allowed at a given Park Point or Loan Point, participating Shippers must notify Transporter's Marketing Services personnel, via email, of their intent to transfer such inventory. Transfers of quantities may not result in any Shipper taking title to quantities that exceed such Shipper's contractual rights under this or other Rates Schedules.
 - (a) Non-Storage Points: Shippers that are subject to this Rate Schedule may agree to transfer their respective Parked Quantity or Loaned Quantity to another PAL or TPAL contract at the same non-storage point.
 - (b) Clay Basin Storage Point: Shippers that are subject to this Rate Schedule may agree to transfer their respective Parked Quantity or Loaned Quantity at the Clay Basin storage facility to another Shipper with a Clay Basin Parked Quantity or Loaned Quantity under Rates Schedules PAL or TPAL, or to any capacity holder in the Clay Basin storage facility. Transfers between TPAL and another Clay Basin capacity holder requires an agreement among the TPAL Shipper, Transporter and the other Clay Basin capacity holder. Upon agreement of the parties, Transporter may utilize its Clay Basin contracted capacity on behalf of a Rate Schedule TPAL Shipper to transfer such Shipper's TPAL quantity to or from the other Clay Basin capacity holder's inventory account.
 - (c) Jackson Prairie Storage Point: Shippers that are subject to this Rate Schedule may agree to transfer their respective Jackson Prairie Park or Loan Quantities to any capacity holder in the Jackson Prairie Storage facility under Rate Schedules SGS-2F, SGS-2I, TPAL, and PAL.

Pursuant to the January 15, 1998 Gas Storage Project Agreement, owners of the Jackson Prairie Storage Project may transfer portions of their respective available working gas inventories, as defined in the Project Agreement, to each other. Upon agreement of the parties, and subject to the terms of the Project Agreement, Transporter may utilize its ownership account on behalf of a Rate Schedule TPAL Shipper to transfer such Shipper's TPAL quantity to an owner's available working gas inventory account. Conversely, an owner may transfer its available working gas inventory to a Rate Schedule TPAL Shipper's TPAL contract.

3. RATE

3.1 Transporter's Maximum Base Tariff Rate for Rate Schedule TPAL, as set forth on Statement of Rates of this Tariff, will apply to all service rendered under this Rate Schedule unless Transporter agrees to discount its Recourse Rate as described below or unless Shipper and Transporter mutually agree to a Negotiated Rate pursuant to Section 19 of the General Terms and Conditions.

3.2 Discounted Recourse Rates.

- (a) Transporter reserves the right to discount, on a non-discriminatory basis, the Recourse Rate for any individual Shipper under any Service Agreement without discounting any other Recourse Rate for that or another Shipper, provided, however, that such discounted Recourse Rate will not be less than the minimum base rate set forth on the Statement of Rates in this Tariff. Such discounted Recourse Rates may apply to specific quantities of gas (such as quantities above or below a certain level or all quantities if quantities exceed a certain level) or service during specific time periods.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.
- 3.3 For each billing period for service under a TPAL Service Agreement, Shipper will pay Transporter a total amount equal to the sum of the following charges described below:
 - (a) The sum of the daily product of (1) the daily Park/Loan Rate set forth in the executed Service Agreement and (2) the daily scheduled quantity (including transfers) of gas parked or loaned;
 - (b) The sum of the daily products of (1) the daily Park/Loan Balance Rate set forth in the executed Service Agreement and (2) the daily quantity of gas parked or loaned at the end of the Gas Day (such charge shall be calculated based on the end of the Gas Day balances for each Gas Day less any new gas parked or loaned on that Gas Day); and
 - (c) The sum of the daily products of (1) the daily UnPark/Loan Payback Rate set forth in the executed Service Agreement and (2) the daily scheduled quantity (including transfers) of park gas delivered to Shipper or loaned gas received from Shipper.

3. RATE (Continued)

Except for Negotiated Rates, on any Gas Day, the sum of the daily Park/Loan Rate, the Park/Loan Balance Rate and the UnPark/Loan Payback Rate assessed for any park or loan may not exceed the Maximum Base Tariff Park/Loan Rate on a per unit basis.

3.4 $\ \$ ACA surcharges and fuel reimbursement will not apply to service under this Rate Schedule.

4. ALLOCATION OF SERVICE

In allocating or interrupting service among Shippers under this Rate Schedule, Transporter will adhere to the priority of service provisions in Section 12 of the General Terms and Conditions.

RATE SCHEDULE TPAL Term Park and Loan Service (Continued)

5. SERVICE RESTRICTIONS AND PENALTIES

- 5.1 Shipper may be required to remove some or all of the Parked Quantity or return some or all of the Loaned Quantity prior to termination of its Service Agreement if Transporter, in its reasonable discretion, determines such action is necessary to accommodate higher priority services, maintain system integrity or to accommodate system balancing requirements for firm services. Notification of any such requirements will be made to Shipper by posting on Transporter's Designated Site and, if specified by Shipper on the Business Associate Information form, via an e-mail to the Shipper. Transporter's notification will indicate the time within which specified Parked Quantities shall be removed by Shipper and/or specified Loaned Quantities shall be returned to Transporter, consistent with Transporter's operating conditions. In no event will the specified time frame be less than three (3) gas days from the date of Transporter's notification.
- 5.2 Any Parked Quantity not removed within the time frame specified by Transporter's notice and any Parked Quantity not removed by the termination date of the Service Agreement will become the property of Transporter at no cost to Transporter, free and clear of any adverse claims. Any Loaned Quantity not returned within the time frame specified by Transporter's notice and any Loaned Quantity not returned by the termination date of the Service Agreement will be deemed sold to Shipper under Section 32 of the General Terms and Conditions at 150% of the highest weekly average price of the following trading points: the NW Wyoming Pool, NW south of Green River, Stanfield, Oregon, NW Canadian Border (Sumas), or El Paso Bondad as reflected in the Daily Price Survey as published in "Gas Daily" for the week in which the Shipper failed to comply.

Notwithstanding the above, Transporter may allow additional time to comply following notification to remove the Parked Quantity or return the Loaned Quantity when operational conditions permit and such additional time does not impair Transporter's ability to provide reliable firm service. Furthermore, if Shipper makes a timely nomination to comply with a notification to remove the Parked Quantity or return the Loaned Quantity and such nomination is not scheduled and confirmed due to Transporter's lack of capacity to schedule and confirm a matching nomination by Shipper of firm transportation service within the Shipper's primary path, then the specified compliance time frame will be extended by one day for each day Transporter is unable to render such transportation. Transporter will not extend the specified compliance time frame if Transporter is unable to provide the related transportation under a firm Service Agreement outside of Shipper's primary path or on an overrun basis or under a Rate Schedule TI-1 Service Agreement.

RATE SCHEDULE TPAL Term Park and Loan Service (Continued)

5. SERVICE RESTRICTIONS AND PENALTIES (Continued)

5.3 Revenue received by Transporter resulting from the deemed sale of Loaned Quantities will be used by Transporter to purchase gas to replace the gas deemed sold to the Shipper, after first taking into account any offsetting confiscated Parked Quantities that have not been reflected in Transporter's fuel reimbursement factor. Any excess revenues will be refunded to Shippers after March 1 of the applicable calendar year using the refund methodology specified in Section 32.2 of the General Terms and Conditions. Any confiscated Parked Quantities in excess of those used to offset Loaned Quantities that are deemed sold will be refunded to customers through Transporter's fuel reimbursement factor pursuant to Section 14.12 of the General Terms and Conditions that goes into effect April 1 of each year.

6. QUANTITY

Each Service Agreement shall specify the maximum quantities to be parked or loaned and a schedule of park or loan quantities under a specific transaction. The Service Agreement may provide for flexibility in total volumes and in the daily volumes parked or loaned, in the timing of the park or the loan (or any portion thereof), in the duration of the park or loan (or portion thereof), and/or in the timing of the completion of the park or loan (or portion thereof) by the return of gas to the Shipper or to Transporter, and shall specify the limits of the flexibility allowed.

7. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the General Terms and Conditions contained in this Tariff.

RATE SCHEDULE TFL-1 Firm Lateral Transportation

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the Transportation of Natural Gas by Transporter on a Designated Lateral when:

- (a) Shipper desires incremental firm service on a Designated Lateral and Transporter has or will have available capacity to render such firm service for Shipper;
- (b) Transporter can render the desired service without modifying the Designated Lateral facilities, except for facilities for which a facilities agreement has been entered into between Transporter an Shipper pursuant to Section 21 of the General Terms and Conditions; and
- (c) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule will apply to Transportation by Transporter for Shipper on a Designated Lateral pursuant to an executed Service Agreement under this Rate Schedule.
- 2.2 Character of Service. Transportation service under this Rate Schedule is firm up to Shipper's Transportation Contract Demand. Transportation service under this Rate Schedule is limited to receipt and delivery points on the Designated Lateral covered by the Service Agreement and is not subject to curtailment or interruption except as expressly provided in the General Terms and Conditions.

3. RATES AND MONTHLY CHARGES

- 3.1 Monthly Charges.
 - (a) Unless Shipper and Transporter have agreed to a Negotiated Rate under Section 3.7, Shipper will pay Transporter for service rendered during each Month under this Rate Schedule the sum of the reservation charge specified in Section 3.2, the volumetric charge specified in Section 3.3 and, if applicable, the scheduled overrun charge specified in Section 3.4.
 - (b) As applicable, all Shippers, including Negotiated Rate Shippers, also will pay Transporter each Month for any facilities charge as specified in Section 3.5 and, pursuant to the General Terms and Conditions, any imbalance penalties not waived and any unauthorized overrun and/or underrun charges.

RATES AND MONTHLY CHARGES (Continued)

Further, all Shippers, including Negotiated Rate Shippers, will reimburse Transporter in-kind for fuel use requirements in accordance with Section 14 of the General Terms and Conditions based on the applicable Factor set forth on Sheet No. 14.

3.2 Reservation Charge.

- (a) Except for released capacity acquired on a volumetric bid basis, the reservation charge is the sum of the daily product of Shipper's Transportation Contract Demand (CD) and, as applicable,:
 - (1) the Rate Schedule TFL-1 applicable maximum base reservation rate for the Designated Lateral stated on Sheet No. 5;
 - (2) a discounted reservation rate that may be granted pursuant to Section 3.6; or
 - (3) the reservation rate determined pursuant to the capacity release provisions in Section 22 of the General Terms and Conditions.
- (b) For released capacity acquired on a volumetric bid basis (including prearranged volumetric releases under Section 22 of the General Terms and Conditions), the Replacement Shipper will pay a reservation charge that is:
 - (1) the sum of the daily product of the applicable volumetric reservation rate determined pursuant to the capacity release provisions in Section 22 of the General Terms and Conditions and the Replacement Shipper's daily Scheduled Quantity in Dth; plus,
 - (2) if the Releasing Shipper has specified a minimum average load factor volumetric commitment, an amount equal to the accepted volumetric bid times the difference, if positive, between (i) the specified average load factor times Replacement Shipper's Transportation Contract Demand times the number of days the Transportation Service Agreement is in effect during the month, and (ii) Replacement Shipper's Scheduled Quantity under the subject Service Agreement during the month.

- 3. RATES AND MONTHLY CHARGES (Continued)
 - (c) Charges for Capacity Release Service: The rates for capacity release service are set forth in Sheet No. 5. See Section 22 of the General Terms and Conditions for information about rates for capacity release service, including information about acceptable bids. In the event of a base tariff maximum and/or minimum rate change, wherein the Replacement Shipper has not agreed to pay the Maximum Base Tariff Rate or a percentage of the Maximum Base Tariff Rate (as it may change from time-to-time), the Replacement Shipper will be obligated to pay:
 - (i) for capacity release transactions that are subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the lesser of the awarded bid rate and the new Maximum Base Tariff Rate unless the awarded bid rate is less than the new minimum rate, in which case the new minimum rate will apply for the remaining term of the release.
 - (ii) for capacity release transactions that are not subject to the Maximum Base Tariff Rate pursuant to FERC regulations: the greater of the minimum base tariff rate and the awarded bid rate for the remaining term of the release.
 - (d) Shipper will pay the Reservation Charges commencing with the primary term begin date set forth in the Service Agreement.
 - 3.3 Volumetric Charge. The volumetric charge is the sum of (i) the daily product of Shipper's Scheduled Quantity in Dth and the Rate Schedule TFL-1 applicable maximum base volumetric rate for the Designated Lateral stated in the Statement of Rates of this Tariff, and (ii) the daily product of Shipper's Scheduled Quantity in Dth and the ACA component as referenced in the Statement of Rates of this Tariff, unless the condition in General Terms and Conditions Section 16.2 applies.
 - 3.4 Overrun Charge. For any volume transported in excess of Shipper's Transportation Contract Demand on any day, Shipper shall pay an amount obtained by multiplying (i) the quantity of Dth in excess of Transportation Contract Demand by (ii) the maximum base scheduled overrun rate per Dth for the Designated Lateral stated on Sheet No. 5, unless otherwise agreed to by Shipper and Transporter.

3. RATES AND MONTHLY CHARGES (Continued)

- 3.5 Facilities Charge: If Transporter and Shipper enter into a facilities agreement pursuant to Section 21 of the General Terms and Conditions for Transporter to construct facilities and for Shipper to pay a facilities charge, the facilities charge will be set forth on Exhibit C to an executed Service Agreement.
- 3.6 Discounted Recourse Rates.
 - (a) Transporter reserves the right to discount at any time the Recourse Rates for any individual Shipper under any Service Agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates will not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates will be set forth in the Shipper's Service Agreement and may apply to specific volumes of gas (such as volumes above or below a certain level or all volumes
 - (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

3. RATES AND MONTHLY CHARGES (Continued)

if volumes exceed a certain level), volumes of gas transported during specific time periods, and volumes of gas transported from specific receipt points and/or to specific delivery points, or within specific corridors. A load factor based discounted reservation rate may be computed by dividing the reservation revenues payable pursuant to a volumetric formula by Shipper's Transportation Contract Demand; i.e., the discounted daily reservation rate would be equal to a stated amount per Dth times the actual daily load factor for Shipper's use of its Transportation Contract Demand.

- (b) Under the Section 3.6(a) load factor based discounted reservation rate option, a minimum average load factor provision also may apply. In that event, the monthly reservation charge under Section 3.2(a) will be equal to the greater of (i) the sum of the daily reservation rate charges calculated using the stated amount per Dth and Shipper's actual daily load factors or (ii) the sum of the daily reservation charges calculated using the stated amount per Dth and the specified minimum average load factor.
- 3.7 Negotiated Rate. If Transporter and Shipper mutually agree to a Negotiated Rate for service hereunder as provided in Section 19 of the General Terms and Conditions and set forth such rate in the Service Agreement, the charges established by such Negotiated Rate will apply. In addition to any such Negotiated Rate, Shipper will pay any applicable charges identified in Section 3.1(b), plus the ACA surcharge component of the volumetric charge otherwise applicable to Scheduled Ouantities under Rate Schedule TFL-1.

4. TRANSPORTATION CONTRACT DEMAND

4.1 The Transportation Contract Demand is the maximum quantity of gas, expressed in Dth, that Transporter is obligated to receive (exclusive of fuel reimbursement furnished in-kind pursuant to Section 14 of the General Terms and Conditions), transport and deliver for Shipper on a firm basis on any one Gas Day, as specified in an executed Service Agreement for service under this Rate Schedule. Transporter's service obligation is limited to Shipper's Transportation Contract Demand as adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.

- 4.2 As long as the Transportation Contract Demand, as adjusted for any capacity releases, is not exceeded, Transporter shall be obligated to receive up to Shipper's Maximum Daily Quantity (MDQ) at each Primary Receipt Point and to deliver up to Transporter's Maximum Daily Delivery Obligation (MDDO) at each Primary Delivery Point at pressures at least as great as the pressures specified in Shipper's TFL-1 Service Agreement, on a firm basis, as such MDQ and MDDO are adjusted for any released capacity pursuant to Section 22 of the General Terms and Conditions.
- 4.3 The aggregate MDQ at the Primary Receipt Points, as specified in a Service Agreement for service under this Rate Schedule, must equal the Transportation Contract Demand. The aggregate MDDO at the Primary Delivery

Points, as specified in a Service Agreement for service under this Rate Schedule, must equal the Transportation Contract Demand.

5. SCHEDULED OVERRUN TRANSPORTATION

5.1 On any day Shipper nominates quantities of gas in excess of Shipper's Transportation Contract Demand specified in the executed Service Agreement, Transporter will schedule such quantities in accordance with the priority of service and curtailing policy delineated in Section 12 of the General Terms and Conditions.

6. RECEIPT AND DELIVERY POINT FLEXIBILITY

6.1 Permanent Changes to Primary Receipt and Delivery Points. Subject to the availability provisions of this Rate Schedule, any Shipper may permanently change primary receipt or delivery points on the Designated Lateral by amending Exhibit A of the Service Agreement.

- 6. RECEIPT AND DELIVERY POINT FLEXIBILITY (Continued)
 - 6.2 Procedures for Requesting Permanent Receipt and Delivery Point Changes. Any Shipper who wishes to amend the primary receipt or delivery points, or the associated MDQ or MDDO named in Exhibit A of its Service Agreement, will electronically request and execute on Transporter's Designated Site an amendment to the applicable Service Agreement at least one business day, but no more than sixty days, prior to the requested effective date. If Transporter determines that a receipt or delivery point change request can be honored, the amendment to the Service Agreement will be executed by Transporter. The change will become effective on the later of the requested effective date or the date executed by Transporter, provided that such date shall not exceed fifteen days from the date of receipt of the amendment by Transporter. Notice of the resulting changes in available receipt or delivery point capacity will be posted to Transporter's Designated Site at least one business day prior to implementation of such change.

On a not unduly discriminatory basis, Transporter may allow Shipper to submit an amendment request more than sixty days prior to the requested effective date if such request (i) is associated with an open season for available capacity posted under Section 25 of the General Terms and Conditions, or (ii) is associated with a concurrent request by Shipper for the construction of facilities under Section 21 of the General Terms and Conditions to create the requested receipt and/or delivery point capacity.

6. RECEIPT AND DELIVERY POINT FLEXIBILITY (Continued)

Capacity posted on Transporter's Designated Site pursuant to Section 25.4 of the General Terms and Conditions will be available for permanent receipt and/or delivery point changes only pursuant to the procedures outlined in Section 25 of the General Terms and Conditions.

6.3 Transporter's Maximum Service Obligation. The total volumes nominated for service on any day for all receipt or delivery points must not exceed Transportation Contract Demand under a Shipper's firm Service Agreement, except as otherwise provided in Section 5 of this Rate Schedule.

7. CONTRACT TERM EXTENSIONS

- 7.1 Standard Unilateral Evergreen Provision. If Transporter and Shipper agree to include a standard unilateral evergreen provision as indicated on Exhibit A of a long-term Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one year.
 - (b) Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving written notice to Transporter so stating at least five years before the termination date.
 - (c) The termination notice required under Section 7.1(b) will be deemed given when posted on Transporter's Designated Site.
- 7.2 Standard Bi-Lateral Evergreen Provision. If Transporter and Shipper agree to include a standard bi-lateral evergreen provision as indicated on Exhibit A of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be:
 - (i) one day for Service Agreements with a primary term of one day;
 - (ii) one month for Service Agreements with a primary term of at least two days but less than one year; or

7. CONTRACT TERM EXTENSIONS (continued)

- (\mbox{iii}) one year for Service Agreements with a primary term of one year or more.
- (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least:
 - (i) one Business Day before the termination date if Section 7.2(a)(i) applies;
 - (ii) ten Business Days before the termination date if Section 7.2(a)(ii) applies; or
 - (iii) one year before the termination date if Section 7.2(a) (iii) applies.
- (c) The termination notice required under Section 7.2(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

8. SEGMENTATION

8.1 To the extent operationally feasible, a Shipper may temporarily segment its capacity on a Designated Lateral into separate parts for its own use or for the purpose of releasing capacity to a Replacement Shipper. Shipper may submit a segmentation request no more than sixty days prior to the requested effective date. Transporter will respond to a Shipper's segmentation request within two business days following the request; and if the segmentation request is approved, Transporter will issue new Service Agreements for each of the segments within one hour of notification to the Shipper that the request is approved. Capacity posted on Transporter's Designated Site pursuant to Section 25.4 of the General Terms and Conditions will be available for capacity release segmentation requests only pursuant to the procedures outlined in Section 25 of the General Terms and Conditions.

On a not unduly discriminatory basis, Transporter may allow Shipper to submit a segmentation request more than sixty days prior to the requested effective date if such request (i) is associated with an

8. SEGMENTATION (Continued)

open season for available capacity posted under Section 25 of the General Terms and Conditions, or (ii) is associated with a concurrent request by Shipper for the construction of facilities under Section 21 of the General Terms and Conditions to create the requested receipt and/or delivery point capacity.

- 8.2 Transporter will permit nominations of forward hauls up to Transportation Contract Demand and backhauls up to Transportation Contract Demand on the Designated Lateral to the same point at the same time for segmented capacity.
- 8.3 Segmentation without Capacity Release. A Shipper that wishes to segment its capacity into separate parts for its own use is subject to the same procedures as apply to the temporary capacity release of segmented capacity under Sections 22 and 23 of the General Terms and Conditions except that the Shipper's release of capacity to itself will always be non-biddable. Such transactions will be treated as temporary capacity releases with the Shipper acting as both the Releasing Shipper and the Replacement Shipper. New Service Agreements will be executed with the Shipper for all new segments. Such Shipper also is specifically subject to the following:
 - (a) Capacity between a Shipper's primary receipt point(s) and primary delivery point(s) on the Designated Lateral may be segmented into separate Service Agreements for a Shipper's own use, provided that such segmentation is operationally feasible and provided that firm capacity is available at and between the designated points.
 - (b) Transportation Contract Demand under any Service Agreement created as a result of segmentation may not exceed the Transportation Contract Demand under Shipper's Service Agreement as it existed before the segmentation.
 - (c) Each Service Agreement will have a primary receipt point and a primary delivery point on the Designated Lateral, but Shipper will have the right to use any available alternate receipt and delivery points on the Designated Lateral on a secondary firm basis, with a scheduling priority consistent with Section 12 of the General Terms and Conditions of Transporter's tariff.

8. SEGMENTATION (Continued)

8.4 Segmentation with Capacity Release. A Shipper that wishes to segment its capacity for the purpose of releasing all or part of that capacity to a Replacement Shipper is subject to the capacity release provisions of Sections 22 and 23 of the General Terms and Conditions of Transporter's tariff. New Service Agreements will be executed with the Releasing Shipper for the retained segment(s) and the Replacement Shipper for the released segment.

9. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the General Terms and Conditions, except as such General Terms and Conditions may be modified in a Service Agreement.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 145 Through 149 Superseding Substitute Original Sheet Nos. 145 Through 149

RESERVED FOR FUTURE USE

RATE SCHEDULE TIL-1 Interruptible Lateral Transportation

1. AVAILABILITY

This Rate Schedule is available to any Shipper for the Transportation of Natural Gas by Transporter on a Designated Lateral when:

- (a) Shipper desires interruptible service on a Designated Lateral;
- (b) Transporter can render the desired service without modifying the Designated Lateral facilities, except for facilities for which a facilities agreement has been entered into between Transporter an Shipper pursuant to Section 21 of the General Terms and Conditions; and
- (c) Shipper and Transporter have executed a Service Agreement for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 Applicability. This Rate Schedule will apply to Transportation by Transporter for Shipper on a Designated Lateral pursuant to an executed Service Agreement under this Rate Schedule.
- 2.2 Character of Service. Transportation service under this Rate Schedule is interruptible up to Shipper's Maximum Daily Quantity. Transportation service under this Rate Schedule is limited to receipt and delivery points on the Designated Lateral covered by the Service Agreement and is subject to curtailment or interruption at anytime as expressly delineated in the General Terms and Conditions.

RATE SCHEDULE TIL-1 Interruptible Lateral Transportation (continued)

3. RATES AND MONTHLY CHARGES

- 3.1 Monthly Charges.
 - (a) Unless Shipper and Transporter have agreed to a Negotiated Rate, Shipper will pay Transporter for service rendered during each Month under this Rate Schedule the sum of the volumetric charge specified in Section 3.2.
 - (b) As applicable, all Shippers, including Negotiated Rate Shippers, will pay Transporter imbalance penalties not waived.

RATE SCHEDULE TIL-1 Interruptible Lateral Transportation (Continued)

3. RATES AND MONTHLY CHARGES (Continued)

- (c) Further, all Shippers, including Negotiated Rate Shippers, will reimburse Transporter in-kind for fuel use requirements in accordance with Section 14 of the General Terms and Conditions based on the applicable Factor set forth on Sheet No. 14.
- 3.2 Volumetric Charge. The volumetric charge is the sum of (i) the daily product of Shipper's Scheduled Quantity in Dth and either the Rate Schedule TIL-1 applicable maximum base volumetric rate for the Designated Lateral stated on the Statement of Rates or the discounted Recourse Rate established pursuant to Section 3.4, and (ii) the daily product of Shipper's Scheduled Quantity in Dth and the ACA component as referenced in the Statement of Rates of this Tariff, unless the condition in General Terms and Conditions Section 16.2 applies.
- 3.3 Facilities Construction: If Transporter and Shipper enter into a facilities agreement pursuant to Section 21 of the General Terms and Conditions for Transporter to construct facilities, payment for such facilities will be set forth in the facilities agreement.

3.4 Discounted Recourse Rates.

- (a) Transporter reserves the right to temporarily discount at any time the Recourse Rates for any individual Shipper under any service agreement without discounting any other Recourse Rates for that or another Shipper; provided, however, that such discounted Recourse Rates shall not be less than the minimum base rates set forth on the Statement of Rates in this Tariff, or any superseding tariff. Such discounted Recourse Rates may apply to specific quantities of gas (such as quantities above or below a certain level or all quantities if quantities exceed a certain level), quantities of gas transported during specific time periods, and quantities of gas transported from specific receipt points and/or to specific delivery points, or within specific corridors.
- (b) Discounted Recourse Rates also may be calculated using a formula based on index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points. Index-based, discounted rates will be no lower than the minimum and no higher than the Maximum Base Tariff Rate.

RATE SCHEDULE TIL-1 Interruptible Lateral Transportation (Continued)

- 3. RATES AND MONTHLY CHARGES (Continued)
 - (b) If Transporter elects to grant a temporary discount, it will do so without amending Shipper's Service Agreement. The temporary discount will be electronically granted using the Rate Schedule TIL-1 Temporary Discount form available in Northwest Passage on Transporter's Designated Site. Shipper's electronic completion of the Temporary Discount form constitutes agreement by the Shipper to the discount and Transporter's electronic acceptance constitutes Transporter's granting of the discount.
 - 3.5 Negotiated Rate. If Transporter and Shipper mutually agree to a Negotiated Rate for service hereunder as provided in Section 19 of the General Terms and Conditions and set forth such rate in the Service Agreement, the charges established by such Negotiated Rate will apply. In addition to any such Negotiated Rate, Shipper will pay any applicable charges identified in Section 3.1(b), plus the ACA surcharge component of the volumetric charge otherwise applicable to Scheduled Quantities under Rate Schedule TIL-1.
- 4. SHIPPER'S MAXIMUM DAILY QUANTITY

The executed Service Agreement shall specify a Maximum Daily Quantity of gas to be transported thereunder for Shipper. The Maximum Daily Quantity refers to the quantity of gas to be delivered by Transporter to Shipper after transportation, at the Delivery Point(s), after fuel use reimbursement provided in-kind in accordance with the terms of Section 14 of the General Terms and Conditions.

5. RESERVED FOR FUTURE USE

RATE SCHEDULE TIL-1 Interruptible Lateral Transportation (Continued)

6. EVERGREEN PROVISION

- 6.1 Evergreen Provision. If Transporter and Shipper agree to include an evergreen provision as contemplated by Section 4 of the Service Agreement, the following conditions will apply:
 - (a) The established rollover period will be one month.
 - (b) Either Transporter or Shipper may terminate the Service Agreement in its entirety upon the primary term end date or upon the conclusion of any evergreen rollover period thereafter by giving the other party termination notice at least thirty days before the termination date.
 - (c) Termination notice under Section 6.1(b) will be deemed given when posted on Transporter's Designated Site. If Transporter gives termination notice, such termination notice also will be given via Internet E-mail or fax if specified by Shipper on the Business Associate Information form.

7. GENERAL TERMS AND CONDITIONS

This Rate Schedule incorporates the General Terms and Conditions except as such General Terms and Conditions may be modified in a Service Agreement.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 154 Through 199 Superseding Substitute Original Sheet Nos. 154 Through 199

RESERVED FOR FUTURE USE

GENERAL TERMS AND CONDITIONS

INDEX

SECTION	TITLE	SHEET NUMBER
1.	DEFINITIONS	201
2.	RECEIPT AND DELIVERY POINTS AND PRESSURES	203
3.	QUALITY	204
4.	MEASUREMENT OF GAS	206
5.	BILLING AND PAYMENT	210
6.	STATUTORY REGULATION	213
7.	LIABILITY AND RISK OF LOSS	214
8.	TITLE TO GAS	215
9.	FORCE MAJEURE	216
10.	INTERRUPTIONS OF SERVICE	
11.	SERVICE AGREEMENT AND SERVICE CONDITIONS	218
12.	SCHEDULING PRIORITIES AND CURTAILMENT	
13.	GAS TECHNOLOGY INSTITUTE VOLUNTARY CONTRIBUTIONS	224
14.	OPERATING CONDITIONS	
15.	DETERMINATION OF DELIVERIES AND IMBALANCES	
16.	FERC ANNUAL CHARGE ADJUSTMENT	
17.	STANDARDS OF CONDUCT	
18.	COMPLAINT PROCEDURES	
19.	NEGOTIATED RATES	
20.	RESERVED FOR FUTURE USE	
21.	INTERCONNECTS	
22.	CAPACITY RELEASE	
23.	REVENUE CREDITING FOR CAPACITY RELEASES (FIRM)	
24.	CLAY BASIN STORAGE SERVICE REVENUE CREDITING	
25.	AVAILABLE CAPACITY	
26.	NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS	
27.	RESERVED FOR FUTURE USE	
28.	SERVICE REQUEST PROCEDURES	
29.	MAINLINE PATH ALTERATION	
30.	MODERNIZATION AND EMISSIONS REDUCTION PROGRAM CRM	
31.	RESERVED FOR FUTURE USE	
32.	SALES OF EXCESS GAS	297

GENERAL TERMS AND CONDITIONS

1. DEFINITIONS

Except where the context expressly establishes another meaning, the following terms when used in this Tariff and in a Service Agreement or other agreement with Transporter incorporating said Tariff will have the following meanings:

<u>Agent</u>: Any party retained by Service Requester for purposes of administering Service Requester's Service Agreement with Transporter. An Agent has only those rights designated in writing to Transporter by such Service Requester.

Associated Receipt Point: A Receipt Point identified by Transporter from which gas may be nominated to a specific pool without incurring transportation and fuel charges. Associated Receipt Point(s) will be posted on Transporter's Designated Site.

Balancing-Approved Day: A Day in which Transporter has indicated on its website that it is accepting Balancing Nominations.

<u>Balancing Nomination</u>: A separately identified Nomination to resolve or reduce imbalance(s) with Transporter.

Begin Date: The Gas Day specified by Service Requester when a transaction is
to begin.

<u>Biogas</u>: Raw gas substantially composed of methane that is produced by the breakdown of organic matter in the absence of oxygen.

<u>Boil-off</u>: Gas vapors generated during the storage and use of liquefied natural gas.

British Thermal Unit ("Btu"): As defined in the American Gas Association Report No. 3, as revised from time to time. "Standardize the reporting basis for Btu as 14.73 psia and 60 degrees F (101.325 kPa and 15 degrees C, and dry). Standardize the reporting basis for gigacalorie as 1.035646 Kg/cm² and 15.6 degrees C and dry. Standardize the reporting basis for gas volumes as cubic foot at standard conditions of 14.73 psia, 60 degrees, F and dry. For gas volumes reported in cubic meters, the standard conditions are 101.325 kPa, 15 degrees C, and dry." NAESB WGQ Standard 2.3.9.

<u>Business Associate Information</u>: The company identifier information, designated representative information, notification preferences and related data provided by a party to Transporter as specified in the Business Associate Information form contained in the informational posting section on Transporter's Designated Site.

<u>Business Day</u>: "Monday through Friday, excluding Federal Banking Holidays for transactions in the U.S., and similar holidays for transactions occurring in Canada and Mexico." NAESB WGQ Standard 3.2.1

 $\underline{\text{Commission}}$: The Federal Energy Regulatory Commission or any successor entity thereto.

1. DEFINITIONS (Continued)

<u>Contract Demand</u>: For the purposes of the General Terms and Conditions of this Tariff, the term "Contract Demand" is used generically to represent all or any combination of the following terms, as applicable: "CD," "Transportation Contract Demand," "Storage Demand" and "Storage Capacity."

 $\underline{\text{Cubic Foot}}$: The volume of gas which occupies one cubic foot of space when held at a base temperature of sixty degrees Fahrenheit (60 F) and a base pressure of 14.73 psia.

Dekatherm ("Dth"): "The standard quantity for nominations, confirmation and scheduling is dekatherms per gas day in the United States, gigajoules per gas day in Canada and gigajoules per gas day in Mexico. (For reference 1 dekatherm = 1,000,000 Btu'sand 1 gigajoule = 1,000,000,000 joules.) For commercial purposes, the standard conversion factor between dekatherms and gigajoules is 1.055056 gigajoules per dekatherm. The standard Btu is the International Btu, which is also called the Btu(IT); the standard joule is the joule specified in the SI system of units." NAESB WGQ Standard 1.3.14. For purposes of this Tariff and associated Service Agreements, MMBtu and Dth are synonymous.

<u>Designated Lateral</u>: A receipt or delivery lateral designated by Transporter and authorized by the Commission to be utilized solely for transportation of Natural Gas under Rate Schedules TFL-1 and TIL-1. Each Designated Lateral will be identified on the Statement of Rates on Sheet No. 5.

<u>Designated Site</u>: An Internet address designated by Transporter or Service Requester for informational postings and the receipt or delivery of electronic data transactions. Transporter's Designated Site may be its Internet web site containing informational postings and access to Northwest Passage and/or its Internet site used for conducting EDI transactions.

Elapsed-Prorated-Scheduled Quantity: "Elapsed-prorated-scheduled quantity means that portion of the scheduled quantity that would have theoretically flowed up to the effective time of the intraday nomination being confirmed, based upon a cumulative uniform hourly quantity for each nomination period affected." NAESB WGQ Standard 1.2.12.

Electronic Data Interchange (EDI): The transfer of information between Transporter's and Service Requester's Designated Sites using NAESB approved standard data sets developed and maintained by the American National Standards Institute's (ANSI) Accredited Standards Committee X12 (ASC X12). EDI data exchange is governed by an executed Trading Partner Agreement between Transporter and Service Requester.

Electronic Delivery Mechanism (EDM): The computer technology model which allows all parties sending and receiving data to accept a Transmission Control Protocol/Internet Protocol (TCP/IP) using Point to Point Protocol (PPP) connection. At a minimum, sending and receiving parties should

1. DEFINITIONS (Continued)

designate an Internet address as a Designated Site for the receipt and delivery of NAESB standardized data sets.

End Date: The Gas Day specified by Service Requester when a transaction is
to end.

<u>Firm Service</u>: Transportation or Storage service which is defined as firm in the Rate Schedule in Transporter's Tariff applicable to such Shipper's Executed Firm Service Agreement. Firm Service for Transportation can yield Primary Firm Service Rights or Secondary Firm Service Rights.

Gas Day: A period of twenty-four (24) consecutive hours beginning and ending as near as practicable at nine o'clock (9 a.m.) Central Clock Time (8 a.m. Mountain Clock Time). Per NAESB WGQ Standard 1.3.1, "Standard time for the Gas Day should be 9 a.m. to 9 a.m. (Central Clock Time)." Reference to a specific day shall mean the twenty-four hours starting at 9 a.m. Central Clock Time (8 a.m. Mountain Clock Time) of that calendar day.

<u>Gas or Natural Gas</u>: Any mixture of hydrocarbons or of hydrocarbons and non-combustible gases, in a gaseous state, consisting essentially of methane. Such gas may be either in its natural state produced from wells, including casinghead and residue gas, or after manufacture or other methods of producing synthetic gas.

Gross Heating Value: The total Btu content for a cubic foot of gas on a dry basis as determined by calculation from compositional analysis using physical properties of gases at 14.73 psia and 60 F as prescribed by the American Gas Association.

<u>Interruptible Service</u>: Transportation or Storage service which is defined as interruptible in the Rate Schedule in Transporter's Tariff applicable to such Shipper's Executed Service Agreement.

<u>IntraDay Nomination</u>: "An intraday nomination is a nomination submitted after the nomination deadline whose effective time is no earlier than the beginning of the Gas Day and runs through the end of that Gas Day." NAESB WGQ Standard 1.2.4.

LNG: Liquefied natural gas.

Mcf: One thousand cubic feet of gas.

 $\underline{\text{MMBtu}}$: One million Btus. For purposes of this Tariff and associated Service Agreements, MMBtu and Dth are synonymous.

<u>Mainline Transportation System</u>: Transporter's mainline(s) extending generally from the U.S./Canadian border near Sumas, Washington to a point near Blanco, New Mexico, and all receipt and delivery laterals and ancillary facilities owned in whole or in part by Transporter which are utilized to provide jurisdictional transportation of natural gas.

1. DEFINITIONS (Continued)

<u>Maximum Base Tariff Rate</u>: The rate, excluding all applicable surcharges, designated on Sheet Nos. 5, 6, 7 or 8.1 of this Tariff.

<u>Maximum Daily Delivery Obligation ("MDDO")</u>: The maximum quantity of gas Transporter is obligated to deliver to a Shipper at a specific delivery point designated on an executed firm Service Agreement.

Maximum Daily Quantity ("MDQ"): For purposes of Rate Schedules TI-1 and TIL-1, the maximum daily quantity of gas, expressed in dekatherms, to be delivered by Transporter to Shipper after transportation, at the Delivery Point(s), after fuel use reimbursement provided in-kind in accordance with the terms of Section 14 of the General Terms and Conditions. For purposes of Rate Schedules TF-1, TF-2, TFL-1, the maximum daily quantity of gas, expressed in dekatherms, to be received by Transporter at the primary receipt point(s) specified on a Shipper's Service Agreement which will be grossed up by fuel use reimbursement provided in-kind in accordance with the terms of Section 14 of the General Terms and Conditions.

<u>Month</u>: A calendar month; provided, however, that Month shall mean Transporter's normal business month for purposes of metering at the wellhead.

Mountain Clock Time: Mountain Standard Time or Daylight Savings Time in Salt Lake City, Utah, whichever is currently effective.

 ${\tt NAESB\ Standards}$: The business practices and electronic communication practices promulgated by the Wholesale Gas Quadrant ("WGQ") of the North American Energy Standards Board ("NAESB"), as codified in Section 284.12(a) of the Commission's regulations.

<u>Negotiated Rate</u>: A rate or formula for computing a rate for Part 284 service which may be greater than, equal to or less than the Recourse Rate, but which may not be less than the minimum base rate as set forth on Sheet No. 5, 6, 7 or 8.1. A Negotiated Rate must be mutually agreed upon by Transporter and Shipper and may be based on a rate design other than straight fixed-variable.

<u>Nomination</u>: A line item pertaining to the transportation or storage of gas under an executed Service Agreement which contains all NAESB issued and FERC approved standard data elements.

<u>Nominating Party</u>: The person or party designated by Shipper to communicate Shipper's nominations to Transporter.

Northwest Passage: Transporter's interactive proprietary software used to conduct electronic data transactions. Northwest Passage is accessible on Transporter's Designated Site pursuant to a completed Business Associate Information form by Service Requester.

1. DEFINITIONS (Continued)

Operational Balancing Agreement ("OBA"): "An OBA is a contract between two parties which specified the procedures to manage operating variances at an interconnect." NAESB WGQ Standard 2.2.1.

Operational Flow Order ("OFO"): "An operational flow order is an order issued to alleviate conditions, inter alia, which threaten or could threaten the safe operations or system integrity, of the transportation service provider's system or to maintain operations required to provide efficient and reliable firm service. Whenever a Transportation Service Provider experiences these conditions, any pertinent order should be referred to as an Operational Flow Order." NAESB WGQ Standard 1.2.6.

Operational Impact Area: "Operational Impact Area is the term used to describe a Transportation Service Provider's designation of the largest possible area(s) on its system in which imbalances have a similar operational effect." NAESB WGQ Standard 2.2.2. Transporter designates its entire system, except the laterals under Rate Schedules TFL-1 and TIL-1, as one Operational Impact Area for purposes of netting and trading Shipper Imbalances and Receiving Party Imbalances. Each lateral under Rate Schedules TFL-1 and TIL-1 is designated as an Operational Impact Area for purposes of netting and trading Shipper Imbalances and Receiving Party Imbalances.

<u>Point of Unbundling</u>: With respect to sales provided pursuant to Section 32, Point of Unbundling means the Clay Basin and/or Jackson Prairie storage facility receipt point(s) into Transporter's system.

 $\underline{\text{Pooling Party}}$: Any party utilizing the pooling services outlined in Section 14.19 of the General Terms and Conditions of this tariff.

Primary Firm Service Rights: the quantity of Primary Firm Service Rights pursuant to a Firm Transportation Service Agreement at Receipt and Delivery Points is the greater of the quantity designated respectively as "MDQs" and "MDDOs" in such Firm Service Agreement or the Scheduled Quantity at any receipt or delivery point for the Day on such Agreement (excluding Scheduled Overrun). The quantity of Primary Firm Service Rights at any given point on the mainline or on a lateral is the greater of:

1. DEFINITIONS (Continued)

a) The lesser of the sum of the MDQs upstream of any applicable point and the sum of the MDDOs downstream of any point in such Firm Transportation Service Agreement; or b) the Scheduled Quantity (excluding Scheduled Overrun) at any given point for the Day on such Agreement. MDQs, MDDOs and Contract Demand are adjusted for any released capacity pursuant to Section 22.

Psia: Pressure expressed in pounds per square inch absolute.

Receiving Party: The party who controls the facilities into which the gas is delivered for Shipper.

Receiving Party Imbalance: A Receiving Party Imbalance will result when there is a difference between scheduled deliveries and actual measured deliveries for a given delivery point operator.

Recourse Rate: The rate designated as the Maximum Base Tariff Rate on Sheet Nos. 5, 6, 7 or 8.1 of this Tariff. For purposes of this Tariff, Recourse Rate and Maximum Base Tariff Rate are synonymous.

Related Income Taxes:

- a) Related Income Taxes, if associated with a contribution in aid of construction, is defined as the current federal and state income taxes resulting from the reimbursement of actual facility costs less the present value of the income tax benefit received by Transporter from the tax depreciation of the new facilities. The cash flow discount factor for the present value calculation will be the weighted average cost of capital as stated in the facilities agreement or, if not stated in a facilities agreement, the last stated litigated or settled general rate case weighted average cost of capital.
- b) Related Income Taxes, if associated with a facilities agreement buy-out, is defined as the current federal and state income taxes on the net book value and the present value amount of the unpaid removal obligation charge reimbursed by the Shipper to Transporter less the income tax benefit received by Transporter resulting from the facilities defined in the facilities agreement. The income tax benefit received by Transporter is the sum of: 1) the present value of any future tax benefits received by Transporter from the tax depreciation of the facilities after the effective date of the buyout, discounted at the weighted average cost of capital currently in effect for the facilities agreement, and 2) the balance of accumulated deferred Income tax as of the effective date of the buy-out.

1. DEFINITIONS (Continued)

c) For both a) and b) above, Related Income Taxes include a gross-up calculation to cover the income tax that Transporter must pay on the Related Income Taxes that are collected from the Shipper.

Renewable Natural Gas ("RNG"): Renewable Natural Gas, also known as biomethane, refers to the portion of biogas that has been purified. Sources may include landfill gas, dairies or feedlots, publicly owned treatment works, sewage treatment plants, and wastewater plants. RNG must conform to the gas quality specifications and testing/monitoring requirements in Section 3 of the GT&C and be free from bacteria, pathogens, and any other substances injurious to pipeline facilities or that would cause the gas to be unmarketable.

Scheduled Quantity: The quantity of gas Transporter has scheduled to transport from specific receipt point(s) to specific delivery point(s), Transportation Service Agreements or zones for a defined period pursuant to an executed Service Agreement; the quantity of gas Transporter has scheduled for aggregation/dis-aggregation from a Pool.

Secondary Firm Service Rights: For purposes of Section 12.1, the quantity of Secondary Firm Service Rights pursuant to a Firm Transportation Service Agreement at any Receipt and/or Delivery Point, on the mainline or on a lateral is equal to the Contract Demand on a Firm Service Agreement (as adjusted for any released capacity pursuant to Section 22) less the quantity at such point that is designated as Primary Firm Service Rights, as defined in this Section 1.

- a) Secondary Firm Service Rights include rights designated as subordinate in any Firm Service Agreement with non-conforming provisions that have been approved by the Commission; and beginning April 1, 2024, rights acquired in the LaPlata to Green River corridor that are designated as subordinate on Exhibit A of any TF-1 Firm Service Agreement pursuant to the settlement approved in Docket No. RP22-1155.
- b) Secondary Firm Service Rights include rights delineated in Rate Schedule TF-2, Section 11.

<u>Service Requester</u>: A Shipper or any other party that executes with Transporter a Business Associate Information form and/or a Trading Partner Agreement.

<u>Shipper</u>: A party that executes a Service Agreement with Transporter under one of the Rate Schedules contained in this tariff.

Shipper Imbalance: A Shipper Imbalance will result when there is a difference between a Shipper's scheduled net receipts and deliveries due to an upstream and/or downstream confirmation.

1. DEFINITIONS (Continued)

Thermally Equivalent: An equal number of Dths.

 $\overline{\text{Trading Partner Agreement}}$: An agreement entered into by Transporter and requesting party which governs the EDI between Transporter and Service Requester.

Transportation: The receipt of Shipper's gas at input points on Transporter's mainline transportation system (referred to as "Receipt Points") and the delivery of thermally equivalent quantities by Transporter to Shipper or for Shipper's account (after making allowance for any requisite fuel reimbursement furnished in-kind) at other points on Transporter's mainline transportation system (referred to as "Delivery Points") regardless of the direction of flow of gas in Transporter's pipelines between the Receipt and Delivery Points, or of changes in such directions of flow. Transportation also means the receipt of Shipper's gas for injection or liquefaction into Transporter's storage facilities and the storage and subsequent withdrawal or vaporization of thermally equivalent quantities, less fuel use, for Shipper's account. All storage references in this tariff, shall be considered to be a part of the term Transportation.

<u>Transporter</u>: Northwest Pipeline LLC, Northwest or Transportation Service Provider.

2. RECEIPT AND DELIVERY POINT AND PRESSURES

2.1 Receipt Points. The Receipt Point(s) at which Transporter will receive gas for transportation are to be identified by the executed Service Agreement. Such Agreements shall be amended as necessary to reflect the addition or deletion of Receipt Points mutually agreed to by the parties in writing, subject to the Receipt Point Flexibility and Priority of Service provisions of this Tariff.

These points include all physical receipt points, as well as any physical delivery point or any other point on Transporter's Mainline Transportation System which Shipper and Transporter have contractually agreed to utilize as a receipt point.

- 2.2 Shipper's Delivery Pressure. Gas shall be delivered to Transporter at the specified Receipt Points at pressures sufficient to allow the gas to enter the facilities of Transporter but shall not be delivered at pressures in excess of the maximum allowable operating pressure of Transporter's facilities at such Receipt Points. Transporter is under no obligation to modify its line pressures to permit the entry of Shipper's gas into its system.
- 2.3 Delivery Points. Transporter shall deliver gas for Shipper's account at the delivery points identified by the executed Service Agreement. The parties may mutually agree in writing to amend said Service Agreement to add or delete delivery points, subject to the Delivery Point Flexibility and Priority of Service provisions of this tariff.

These points include all physical delivery points, as well as any physical receipt point or any other point on Transporter's Mainline Transportation System which Shipper and Transporter have contractually agreed to utilize as a delivery point.

2.4 Transporter's Delivery Pressure. Transporter shall deliver gas for Shipper's account at the pressures existing from time to time in Transporter's facilities at the points of delivery but no less than the pressures mutually agreed upon by Transporter and Shipper and specified by the executed firm Service Agreement.

First Revised Sheet No. 203-A Superseding Substitute Original Sheet No. 203-A

GENERAL TERMS AND CONDITIONS (Continued)

2. RECEIPT AND DELIVERY POINT AND PRESSURES (Continued)

Any delivery pressure commitments that are subject to specified conditions in order to ensure Transporter's quality of service will also be specified by the executed firm Service Agreement.

3. QUALITY

- 3.1 Gas Quality at Receipt Points. All Gas delivered by Shipper to Transporter shall conform to the applicable specifications in either Section 3.1(a) or Section 3.1(b). As used in this section, the La Plata Facilities are defined as those facilities commencing at a measurement facility downstream of the discharge side of Northwest's La Plata B compressor station southward to the Blanco Hub, including the La Plata A compressor station and certain plant interconnects, all located in southern Colorado and northern New Mexico.
 - (a) All Gas delivered by Shipper to Transporter at Receipt Points not connected to the La Plata Facilities shall conform to the following specifications:
 - (1) Hydrocarbon Liquids and Liquefiables: The hydrocarbon dew point of the gas delivered shall not exceed fifteen degrees Fahrenheit at any pressure between 100 psia and 1,000 psia as calculated from the gas composition and shall be free from hydrocarbons in the liquid state. At all times, any and all liquid or liquefiable hydrocarbons, or any other constituent or by-product, recovered from the gas by Transporter, after delivery of gas to Transporter shall be and remain the exclusive property of Transporter.
 - (2) Hydrogen Sulfide and Total Sulfur: The gas shall contain not more than one quarter grain of hydrogen sulfide per one hundred cubic feet and not more than five grains total sulfur per one hundred cubic feet.
 - (3) Carbon Dioxide: The gas shall contain not more than two percent by volume of carbon dioxide, except as otherwise provided in Section 3.5.
 - (4) Inerts: The gas shall contain not more than a combined total of three percent by volume of inerts including, but not limited to, carbon dioxide, nitrogen and oxygen, except as otherwise provided in Section 3.1(c) and Section 3.5.

3. QUALITY (Continued)

- (5) Dust, Gums, etc.: The gas shall be commercially free from objectionable odors (excluding odorant added to natural gas for safety reasons or to comply with federal and/or state regulations), solid matter, dust, gums, and gum forming constituents, or any other substance which interferes with the intended purpose of merchantability of the gas, or causes interference with the proper and safe operation of the lines, meters, regulators, or other appliances through which it may flow.
- (6) Heating Value: The total gross heating value of the gas deliverable hereunder shall not be less than 985 Btu, except as otherwise provided in Section 3.1(c).
- (7) Oxygen: The gas shall not contain in excess of two-tenths of one percent by volume of oxygen, and the parties agree to exercise every reasonable effort to keep the gas completely free of oxygen.
- (8) Temperature: The temperature of the gas at the point of delivery shall not exceed one hundred twenty degrees Fahrenheit.
- (9) Water: The gas delivered shall be free from liquid water and shall not contain more than seven pounds of water in vapor phase per million cubic feet.
- (10) Mercury: The gas shall be free from any detectable mercury.
- (11) Toxic or Hazardous Substance: The gas shall not contain any toxic or hazardous substance in concentrations which, in the normal use of the gas, may be hazardous to health, injurious to pipeline facilities, or be a limit to merchantability or be contrary to applicable government standards.

3. QUALITY (Continued)

- (12) Bacteria: The gas, including any associated liquids, shall not contain any microbiological organism, active bacteria or bacterial agent capable of causing or contributing to: (i) injury to Transporter's pipelines, meters, regulators, or other facilities and appliances through which such gas flows or (ii) interference with the proper operation of the Transporter's facilities. Microbiological organisms, include, but are not limited to, sulfate reducing bacteria (SRB) and acid producing bacteria (ACB). When bacteria or microbiological organisms are considered a possibility, Shipper(s) desiring to Nominate such gas, upon Transporter's request, shall cause such gas to be tested for bacteria or bacterial agents utilizing the American Petroleum Institute test method API-RP38 or other acceptable test method as determined by both parties.
- (b) All Gas delivered by Shipper to Transporter at Receipt Points connected to the La Plata Facilities shall conform to the following specifications:
 - (1) Hydrocarbon Liquids and Liquefiables: The hydrocarbon dew point of the gas delivered shall not exceed fifteen degrees Fahrenheit at any pressure between 100 psia and 1,000 psia as calculated from the gas composition and shall be free from hydrocarbons in the liquid state. At all times, any and all liquid or liquefiable hydrocarbons, or any other constituent or by-product, recovered from the gas by Transporter, after delivery of gas to Transporter shall be and remain the exclusive property of Transporter.
 - (2) Hydrogen Sulfide and Total Sulfur: The gas shall contain not more than one quarter grain of hydrogen sulfide per one hundred cubic feet of gas. The gas shall contain not more than 0.3 grains of mercaptan sulfur per one hundred cubic feet of gas. The gas shall contain not more than 0.75 grains of total sulfur per one hundred cubic feet of gas.

3. QUALITY (Continued)

- (3) Carbon Dioxide: The gas shall contain not more than two percent by volume of carbon dioxide, except as otherwise provided in Section 3.5.
- (4) Inerts: The gas shall contain not more than a combined total of three percent by volume of inerts including, but not limited to, carbon dioxide, nitrogen and oxygen, except as otherwise provided in Section 3.5.
- (5) Dust, Gums, etc.: The gas shall be commercially free from objectionable odors (excluding odorant added to natural gas for safety reasons or to comply with federal and/or state regulations), solid matter, dust, gums, and gum forming constituents, or any other substance which interferes with the intended purpose of merchantability of the gas, or causes interference with the proper and safe operation of the lines, meters, regulators, or other appliances through which it may flow.
- (6) Heating Value: The total gross heating value of the gas deliverable hereunder shall not be less than $985~\mathrm{Btu}$ and not more than a maximum of $1,110~\mathrm{Btu}$.
- (7) Oxygen: The gas shall not contain in excess of one-tenth of one percent by volume of oxygen, and the parties agree to exercise every reasonable effort to keep the gas completely free of oxygen.
- (8) Temperature: The temperature of the gas at the point of delivery shall not be less than forty (40) degrees Fahrenheit, and not exceed one hundred twenty degrees (120) Fahrenheit.
- (9) Water: The gas delivered shall be free from liquid water and shall not contain more than seven pounds of water in vapor phase per million cubic feet.
- (10) Mercury: The gas shall be free from any detectable mercury.

3. QUALITY (Continued)

- (11) Toxic or Hazardous Substance: The gas shall not contain any toxic or hazardous substance in concentrations which, in the normal use of the gas, may be hazardous to health, injurious to pipeline facilities, or be a limit to merchantability or be contrary to applicable government standards.
- (12) Bacteria: The gas, including any associated liquids, shall not contain any microbiological organism, active bacteria or bacterial agent capable of causing or contributing to: (i) injury to Transporter's pipelines, meters, regulators, or other facilities and appliances through which such gas flows or (ii) interference with the proper operation of the Transporter's facilities. Microbiological organisms, include, but are not limited to, sulfate reducing bacteria (SRB) and acid producing bacteria (ACB). When bacteria or microbiological organisms are considered a possibility, Shipper(s) desiring to Nominate such gas, upon Transporter's request, shall cause such gas to be tested for bacteria or bacterial agents utilizing the American Petroleum Institute test method API-RP38 or other acceptable test method as determined by both parties.
- (c) RNG delivered by Shipper to Transporter at Receipt Point mentioned in Section 3.1(a) shall conform to the following specifications and testing /monitoring requirements:
 - (1) RNG gas delivered into Transporter's system shall conform to the gas quality specifications set forth in Section 3.1(a) above except when Transporter, in its reasonable judgement, determines it can accept RNG: (a) containing not more than a combined total of four percent by volume of inerts including, but not limited to, carbon dioxide, nitrogen and oxygen and/or (b) with a total gross heating value of not less than 970 Btu.
 - (2) Transporter prohibits the delivery of RNG to Transporter's Receipt Points from any landfill permitted under the Resource Conservation and Recovery Act Subtitle C (42 U.S.C. § 6921 6932), whether by the United States Environmental Protection Agency or by a state under a program authorized by the United States Environmental Protection Agency ("Hazardous Waste Landfills"). A Hazardous Waste Landfill includes all continuous land and structures, and other appurtenances and improvements, on the land used for the treatment, transfer, storage, resource recovery, disposal or recycling hazardous waste management units, or combinations of these units.

3. QUALITY (Continued)

The landfill operator shall certify in writing to Transporter that the RNG is not being produced from landfill gas collected from a Hazardous Waste landfill before delivering the RNG into Transporter's pipeline system.

- (3) RNG to be delivered to and transported on Transporter's pipeline system shall be subject to periodic testing and monitoring based on the biogas source. Transporter's RNG operating policy and testing requirements for the acceptance of RNG into Transporter's system will be posted on Transporter's Designated Site.
- 3.2 Gas Quality at Delivery Point(s). The gas delivered by Transporter to Shipper at the Delivery Point shall be natural gas containing a gross heating value of at least 985 Btus. Such gas shall be commercially free of dust, gums, dirt, impurities and other solid matter and shall not contain more than one-quarter grain hydrogen sulfide per one hundred cubic feet as determined by using commercially available on-line analyses and/or such analytical methods that are generally accepted in industry practice; provided that Transporter may install and utilize a recording hydrogen sulfide analyzer to monitor the gas at points at which it deems such continuous monitoring to be desirable. The gas to be delivered shall not contain more than five grains of total sulfur per on hundred cubic feet, unless Transporter determines that there is a regulatory requirement for gas to be odorized upstream of the Delivery Point, in which case, the gas to be delivered shall not contain more than six grains of total sulfur per one hundred cubic feet.

The gas shall be free of water and hydrocarbons in liquid form at the temperature and pressure at which the gas is delivered and in no event shall have a water content in excess of seven pounds in vapor phase per million cubic feet.

3. QUALITY (Continued)

- 3.3 Determination of Gross Heating Value and Component Analysis. party operating the measurement equipment shall determine the gross heating value of the gas delivered and its component analysis at reasonable intervals. Such determination shall be made using either an on-line chromatograph or by chromatographic analysis of a representative sample of gas taken with a continuous sampler. Transporter may at its option allow the use of spot samples. If at any time and for any reason Shipper or Transporter should question the results of any spot sampling, a redetermination shall be made and the redetermination mutually acceptable to the parties shall be used; provided, however, if neither party questions such results within a period of sixty (60) days following the determination thereof, then such results shall be deemed conclusive and binding upon the parties. Btu measuring equipment shall be installed at a location or locations where the gross heating value of the gas received or delivered hereunder may be reasonably determined.
- 3.4 Failure to Meet Specifications. Transporter or Shipper shall have the right, exercisable by the giving of written or oral notice to the other party, to require the remedy of any failure to deliver or redeliver gas in accordance with the quality specifications set forth in Sections 3.1 and 3.2. In the event gas delivered by either party fails to conform to such specifications, as evidenced by the latest chromatograph analysis derived from an on-line chromatograph or from a sample taken manually and analyzed by a chromatograph, or from any other verifiable evidence, the receiving party may refuse to accept all or any portion of such gas.
- 3.5 Accepting Gas Which Fails to Meet Specifications. Transporter will accept gas that does not meet quality specifications on a non-discriminatory basis to all similarly situated Shippers, provided:
 1) such acceptance does not jeopardize Transporter's ability to meet its obligations to deliver gas to downstream interconnecting pipelines or markets; and, 2) such acceptance occurs on a short-term basis as a result of an unplanned upstream event such as a plant start-up, plant upset or line freeze-off.
- 3.6 Gas Analysis Equipment. If Transporter, in its reasonable judgment, determines that any additional or modified Gas analysis or control equipment is needed to accurately monitor the quality of Gas received at an existing Receipt Point and control the receipt of Gas failing to conform to the applicable quality specifications, then the Shipper(s) desiring to nominate at such Receipt Point will cause the interconnecting party at such Receipt Point to install such necessary additional or modified equipment.

3. QUALITY (Continued)

Unless otherwise mutually agreed, if Transporter installs such additional or modified Gas analysis or control equipment, the interconnecting party will provide a contribution in aid of construction to Transporter for all actual costs incurred by Transporter, and reimburse Transporter for any Related Income Taxes.

3.7 NAESB Standards Related to Gas Quality. NAESB WGQ Quadrant Electronic Delivery Mechanism Related Standards referenced below, will apply and are incorporated by reference:

NAESB WGQ Standards:

4.3.89 through 4.3.93

4. MEASUREMENT OF GAS

4.1 Unit of Volume. NAESB WGQ Flowing Gas Related Standard 2.3.10 will apply.

The unit of volume for the purpose of measurement shall be one (1) cubic foot of gas at a base temperature of 60 F. and at a base pressure of 14.73 psia. Such measured volumes, converted to Mcf, shall be multiplied by their gross heating value per cubic foot and divided by 1,000 to determine Dth received and delivered hereunder. The unit of volume for the purpose of balancing total receipts and deliveries of gas hereunder shall be one Dth.

- 4.2 Computation of Volume. The volumes of gas measured hereunder shall be computed in accordance with specifications prescribed by the American Gas Association, applied in a practical and appropriate manner. The absolute atmospheric pressure used for volume calculations shall be the average atmospheric pressure determined by calculations based on the actual elevation above sea level of the meter at the place of measurement.
 - Meters or Measurement Equipment. For purposes of this Section, Meters or Measurement Equipment shall mean the primary measurement element (i.e., orifice plate, turbine meter, etc.), and recording instruments. Gas delivered to Transporter at the Receipt Points shall be measured by measurement equipment installed, operated and maintained at or near such Receipt Points. Gas delivered by Transporter at the Delivery Points shall be measured by measurement equipment installed, operated and maintained at or near such Delivery Points. Where orifice meters are used they shall be installed and operated in accordance with the specifications prescribed in ANSI/API 2530 as supplemented and modified from time to time, applied in a practical and appropriate manner. Where positive displacement meters, turbine meters, or other measuring devices are used, they shall be installed and operated in accordance with recommendations of the American Gas Association, where available. The respective meter, meter readings and meter charts shall be accessible at all reasonable times to inspection and examination by the non-operating party. Calibration and adjustment of meters and changing of charts shall be done only by the party owning and operating the measurement facilities, unless otherwise agreed to by the parties.

Substitute Original Sheet No. 206A Superseding Original Sheet No. 206-A

- 4. MEASUREMENT OF GAS (Continued)
 - 4.4 Temperature and Specific Gravity. The following measurement factors shall be given due consideration.
 - (a) The temperature of the natural gas flowing through each meter shall be an assumed temperature arrived at by agreement or the actual measured temperature.
 - (b) The specific gravity of the gas shall be determined by the party operating the meter by calculation, adjusted for any difference between the specific gravity in the ideal state and in the real state in accordance with published procedures accepted by the American Gas Association.
 - 4.5 Notice of Equipment Tests. Unless otherwise agreed to by the parties, the party operating the measurement facilities shall give notice to the other party of the time and location of all tests of gas delivered hereunder or of any equipment used in measuring or determining the nature or quality of such gas, in order that such other party may conveniently have its representative present. Should the non-operating party reasonably not be satisfied with any such test, it shall so notify the operating party and the operating party shall perform such retests as may be necessary to assure an accurate test.
 - 4.6 Meter Testing. The accuracy of measurement equipment shall be verified at reasonable intervals by the party operating such equipment, and if requested, in the presence of representatives of the other party, unless otherwise agreed to by the parties. If either party notifies the other that it desires a special test of any measurement equipment, the operating party shall cooperate to secure an immediate verification of the accuracy of such measurement equipment and joint observation of any adjustments. The payment for all such tests shall be borne by the party requesting the test, unless otherwise agreed to by the parties.
 - 4.7 Check Meters and Non-Interference. Either party may install and operate check meters at its option and expense to check the other party's primary measurement equipment, but measurement of gas shall be by the measurement equipment of the parties identified in Section 4.3. Such check meters shall be installed so as not to interfere with the operation of the primary measurement equipment of such parties. The parties shall

4. MEASUREMENT OF GAS (Continued)

exercise care in the installation, maintenance and operation of check measuring, pressure regulating equipment or gas compressors so as to prevent any inaccuracy in the determination of the quantity or quality of gas being measured. If unacceptable square root error or gauge line error shifts occur, the installation of pulsation filters may be required by Northwest. The party responsible for the source of any pulsation shall also be responsible for the installation cost of remedial devices or filtering equipment to reduce or eliminate such pulsation. If disagreements over pulsation arise, a third-party consultant acceptable to both Transporter and Shipper will be consulted to resolve the problem.

4.8 Adjustment of Inaccuracies. NAESB WGQ Flowing Gas Related Standards 2.3.13 and 2.3.14 will apply.

If the percentage of inaccuracy upon any test shall be an amount of two (2) percent or higher, the registration of such meter shall be corrected at the rate of such inaccuracy for any period which is definitely known or agreed upon. In the event the period is not definitely known or agreed upon, such correction shall be for a period extending back one-half (1/2) of the time elapsed since the date of the last calibration. In no event shall inaccuracies of less than 100 Dth over the period be corrected.

Following any test, measurement equipment found inaccurate shall be immediately restored by the operating party as closely as possible to a condition of accuracy. If the measurement equipment is out of service or out of repair for any reason so that the amount of gas delivered cannot be estimated or computed from the reading thereof, the amount of gas delivered through the period such meter is out of service or out of repair shall be estimated and agreed upon by the Transporter and Receiving Party upon the basis of the best data available using the first of the following methods which is feasible:

- (a) by using the registration of the other party's check meter if installed and accurately registering; or
- (b) by correcting the error if the percentage of error is ascertainable by calibration, test or mathematical calculation; or

- 4. MEASUREMENT OF GAS (Continued)
 - (c) by estimating the quantity of deliveries by comparison with deliveries during preceding periods under similar conditions when the meter was registering accurately.
 - 4.9 Correction of Volume Calculation Error. If, upon inspection and verification, any volume calculation error shall result in a volume inaccuracy of two (2) percent or higher, then the inaccuracy shall be corrected to zero error following verification by Transporter and Shipper, however, in no event shall inaccuracies less than 100 Dth over the period be corrected.
 - 4.10 Prior Period Adjustments. NAESB WGQ Flowing Gas Related Standards 2.3.11 and 2.3.12 will apply.
 - 4.11 Records Preservation. Each party shall preserve all test and measurement data, charts or similar records for a period of at least one (1) year or such other periods as shall be required under Part 225 of the Commission's regulations or the lawful requirements of any other governmental body, Federal or State, prescribing longer retention periods for any category of record.

5. BILLING AND PAYMENT

5.1 Statement by Shipper. NAESB WGQ Flowing Gas Related Standard 2.3.7 will apply.

Except where Transporter is the party designated as the party operating and maintaining the delivery point, Operator shall report or cause to be reported to Transporter the daily volumes and gross heating value of gas delivered pursuant to the Service Agreement by telephone, fax, Internet E-mail or other electronic means agreed to between the parties to Transporter's Gas Measurement Department by the 5th business day of the calendar month following the month of delivery of such volumes. Shipper shall furnish Transporter's Gas Measurement Department with a written statement that confirms such reported volumes by the 10th business day of the calendar month following the month of delivery.

5.2 Statement by Northwest. NAESB WGQ Invoicing Related Standard 3.3.14; NAESB WGQ Flowing Gas Related Standard 2.3.28, Invoicing Related Standards 3.3.3 through 3.3.13, 3.3.22 through 3.3.26; and Capacity Release Related Standards 5.3.22 and 5.3.23 will apply.

Transporter's invoice and imbalance statements will include information consistent with the standards referenced in this subsection.

5.3 Payment by Shipper. NAESB WGQ Invoicing Related Standards 3.3.17 and 3.3.18 will apply.

On or before the 25th day of each calendar month, Shipper shall make payment to Transporter for all amounts due Transporter as set forth on the statement referred to in Section 5.2. Unless otherwise agreed to by Transporter, such payment shall be made by wire transfer on or before the due date to a depository designated by Transporter on its invoice. If Transporter agrees to accept payment by check, and unless otherwise agreed to by Transporter, such payment must be received on or before the 25th day of each calendar month at the address specified by Transporter on its Designated Site. If the 25th falls on a day that the designated depository is not open in normal course of business to receive Shipper's payment, or the 25th falls on a day that payment by check cannot be received by Transporter in normal course of business, then Shipper's payment by wire transfer shall be made, or payment by check shall be received, on or before the business day prior to the 25th that such depository or delivery is available.

First Revised Sheet No. 210-A Superseding Substitute Original Sheet No. 210-A

Second Revised Sheet No. 210-B Superseding First Revised Sheet No. 210-B

Third Revised Sheet No. 210-C Superseding Second Revised Sheet No. 210-C

5. BILLING AND PAYMENT (Continued)

- 5.4 Failure to Pay Bills. Should Shipper fail to pay all of the amount of any bill, when the same becomes due, Transporter shall have the right to charge Shipper a late charge on the unpaid balance which shall accrue on each calendar day from the due date at the rate specified in Section 154.501(d) of the Commission's regulations. If such failure to pay continues for thirty (30) days after payment is due, Transporter, in addition to any other remedy it may have under the Service Agreement, may; (1) discontinue transportation service for self-implementing transactions without Commission approval, or (2) after application to and authorization by the Commission suspend further service until such amount is paid. However, if Shipper in good faith shall dispute the amount of any such bill or part thereof and shall pay to Transporter such amounts as it concedes to be correct, then Transporter shall not be entitled to suspend further service.
- 5.5 Right to Set Off Unpaid Amounts. In the event that Shipper does not pay the full amount due in accordance with Section 5.3 hereof, without prejudice to any other rights or remedies it may have, Transporter shall have the right to withhold and set off payment of any amounts of monies due by Transporter to Shipper, whether in connection with the Service Agreement in dispute or otherwise, against any and all amounts or monies due or owing by Shipper to Transporter for services performed under the Service Agreement in dispute.
- 5.6 Disputed Statements and Prior Period Adjustments. NAESB WGQ Flowing Gas Related Standard 2.3.31 and Invoicing Related Standards 3.3.15, 3.3.16, 3.3.19 and 3.3.21 will apply.
- 5.7 Rate Refunds. In the event that the Commission orders refunds of any rates charged by Transporter, Transporter will make such refunds to all affected, eligible Shippers.

Rate refunds that involve capacity release transactions shall be made as follows:

- (a) For temporary capacity releases that are more than one year in duration or take effect more than one year from the date that Transporter receives notification of the release, Transporter will make such rate refunds to the Replacement Shipper.
- (b) For temporary releases of one year or less in duration and that take effect within one year of Transporter receiving notification of the release, Transporter will make such refunds to the Releasing Shipper.

5. BILLING AND PAYMENT (Continued)

- (c) For permanent releases, Transporter will make refunds pro rata, between the Releasing Shipper and the Replacement Shipper, based on the effective date of the release, provided that refunds related to temporary capacity releases occurring before or after a permanent release will be made consistent with Sections 5.7(a) and (b).
- (d) Additionally, for all capacity release transactions, the Releasing Shipper and the Replacement Shipper may agree in the Special Terms and Conditions of such release, that subsequent to Transporter's disbursement of any rate refunds pursuant to Sections 5.7(a), (b) or (c), the party receiving the refund may disburse it to the other party.
- 5.8 Record Retention. Transporter and Shipper shall retain all records pertaining to billing and payment for at least three (3) years.
- 5.9 Access to Records. Each party hereto, or its designated representative, shall have access to the books and records of the other party during normal business hours to the extent such records are applicable to measurement and quantities of gas delivered hereunder.
- 5.10 Establishment of Credit. NAESB WGQ Creditworthiness Standards 0.3.3 through 0.3.10 will apply.

Transporter shall have the right to require Shipper to provide proof of ability to pay, satisfactory to Transporter, including advance deposits if deemed required by Transporter. Transporter shall pay interest on Shipper's advance deposit in accordance with Section 154.501(d) of the Commission's regulations.

5.11 Discount Policy. Any discounts will be applied to the base tariff charge, excluding any surcharges.

6. STATUTORY REGULATION

- Regulation. Transporter's rates, charges, classifications and services as set forth in this tariff are subject to regulation by the Commission under the Natural Gas Act, the Natural Gas Policy Act of 1978 and any successor or supplemental act. Transporter shall have the right to file from time to time with the Commission under Section 4 of the Natural Gas Act such new rate schedules and changes in its existing effective tariff as Transporter may find necessary from time to time to: (a) assure Transporter just and reasonable rates and charges, as well as a rate of return sufficient, inter alia, to service the debt, attract capital, permit facilities expansion and (b) efficiently and reliably manage its Transportation System to provide adequate natural gas service to all of Transporter's customers. Without in any way limiting the generality of the foregoing, Transporter shall have the right to file new rate schedules and or new tariff sheets fairly and appropriately reflecting changes in its costs or changes in the conditions of service attributable to such transportation or any other service provided. Shipper shall have the right to protest any such new rate schedules and changes before the Commission. Notwithstanding any other provision hereof, these provisions are subject to all valid legislation and all valid present or future orders, rules and regulations of duly constituted authorities now or hereafter having jurisdiction.
- 6.2 Filing Fees. Shipper shall reimburse Transporter for any and all filing fees incurred by Transporter in seeking necessary governmental authorizations for service or facilities requested by the Shipper under any Rate Schedule included in this tariff.
- 6.3 Imported Natural Gas. For imported Natural Gas tendered by Shipper for Transportation by Transporter under Rate Schedules TF-1, TF-2 or TI-1, Shipper agrees that it or the entity selling the imported Natural Gas to Shipper is the "importer of record" and Shipper will comply with all requirements for reporting and submitting payment of duties, fees, and taxes to the United States or agencies thereof to be made on imported Natural Gas and for making the declaration of entry pursuant to 19 CFR Section 141.19. Shipper agrees to indemnify and hold Transporter harmless from any and all claims of damage or violation of any applicable laws, ordinances and statutes which pertain to the importation of the gas transported hereunder and which require reporting and/or filing of fees in connection with said import.

6. STATUTORY REGULATION (Continued)

6.4 Warranty Of Eligibility For Transportation. Shipper warrants that all Gas delivered to Transporter for Transportation under all Rate Schedules included in this Tariff will be eligible for Transportation in interstate commerce under the rules, regulations or orders of the Commission in existence at the time the Gas is shipped. Shipper will indemnify Transporter and save it harmless from all suits, actions, damages, costs, losses, expenses (including reasonable attorney fees) and regulatory proceedings, arising from breach of this warranty.

7. LIABILITY AND RISK OF LOSS

- 7.1 Liability for Facilities. Each party assumes full responsibility and liability arising from the operation of the pipeline and facilities owned by it and agrees to hold the other party harmless from any liability of any nature arising from the owning party's installation, ownership, and operation thereof. Transporter shall have no obligation to odorize gas delivered to Shipper under this tariff nor to maintain any odorant level.
- 7.2 Risk of Loss. Except as otherwise provided in the Service Agreement, risk of loss to the volumes of gas transported shall pass to Transporter upon delivery to Transporter for Shipper's account and to Shipper upon delivery by Transporter for Shipper's account.
- 7.3 Limitation of Liability. Except as otherwise provided elsewhere in this tariff, in no event shall Transporter or Shipper be liable to the other for special, indirect, consequential (including loss of profits), incidental or punitive damages whether or not such damages arise out of breach of contract, negligence, tort, strict liability; provided, however, unless otherwise agreed to by Transporter and Shipper, the foregoing shall not limit Transporter's liability, if any, to Shipper, nor Shipper's liability, if any, to Transporter, arising out of gross negligence, willful misconduct, or bad faith actions. Nothing herein shall limit Transporter's liability, if any, to Shipper, nor Shipper's liability, if any, to Transporter, for direct damages.

8. TITLE TO GAS

8.1 Warranty of Title. Transporter accepts Shipper's gas at the Receipt Point(s) subject to the understanding that Shipper warrants that it will at the time of delivery of gas to Transporter for transportation, have good title to all gas so delivered to Transporter, free and clear of all liens, encumbrances and claims whatsoever, and that it will at such time of delivery have the right to deliver such gas. Transporter may also request Shipper's proof of right to purchase or transport such gas.

Subject to the foregoing, Transporter warrants that it will at the time of delivery of the transported gas to Shipper at the Delivery Point have the right to deliver said gas free and clear of all liens, encumbrances and claims whatsoever.

- 8.2 Buy/Sell. "Buy/sell" arrangements are those in which Shipper takes title to natural gas at or upstream of Transporter's receipt point in a transaction with a third party with the understanding and commitment that after the gas has been transported on Transporter's transmission system and delivered to Shipper at the delivery point, such gas will be returned to the third party and title to the gas reconveyed to the third party. All buy/sell arrangements existing prior to November 1, 1993, the date Transporter's capacity release mechanism went into effect, are grandfathered. Shipper warrants that the gas which it tenders for firm transportation will not be acquired by Shipper under buy/sell arrangements executed on or after such date.
- 8.3 Indemnity. Each party shall indemnify, save and hold the other party, its subsidiaries and/or affiliates and their directors, officers, employees and agents, free and harmless from all suits, actions, debts, accounts, damages, costs, losses and expenses arising from or out of adverse claims of all persons to the gas delivered by it to such other party.

8. TITLE TO GAS (Continued)

8.4 Waiver of Shipper Must Have Title Requirement. If Transporter acquires transportation and/or storage capacity on an off-system pipeline, Transporter will use such capacity for operational purposes and/or will render service to Shippers using such capacity pursuant to the terms and conditions of its tariff and at Commission-approved rates. For transactions subject to this Section 8.4, the "Shipper must have title" requirement is waived.

9. FORCE MAJEURE

Neither Transporter nor Shipper shall be liable in damages to the other for any act, omission or circumstance occasioned by or in consequence of any blockades, insurrections, riots, epidemics, flood, washouts, landslides, mudslides, earthquakes, extreme cold or freezing weather, lightning, rulers and peoples, civil disturbances, explosions, breakage or imminent breakage or freezing of or accident to machinery or line of pipe, the order of any court or governmental authority having jurisdiction, and any other cause, whether of the kind herein enumerated or otherwise, not reasonably within the control of the party claiming suspension and which by the exercise of due diligence such party is unable to prevent or overcome. Failure to prevent or settle any strike or strikes shall not be considered a matter within the control of the party claiming suspension.

Such causes or contingencies affecting the performance under the executed Service Agreement by either Transporter or Shipper, however, shall not relieve it of liability in the event of its concurring negligence or in the event of its failure to use due diligence to remedy the situation and to remove the cause in an adequate manner and with all reasonable dispatch, nor shall such causes or contingencies affecting such performance relieve either party from its obligations to make payments of amounts then due, including reservation charges.

10. INTERRUPTIONS OF SERVICE

- 10.1 Tests and Repairs. Transporter shall have the right, without liability to Shipper, to interrupt the transportation of gas for Shipper when necessary to test, alter, modify, enlarge or repair any facility or property comprising a part of, or appurtenant to, Transporter's pipeline system, or otherwise related to the operation thereof. Transporter shall endeavor to cause a minimum of inconvenience to Shipper because of such interruptions.
- 10.2 Operational Control. Transporter shall retain full operational control of its pipeline system and shall at all times be entitled to schedule deliveries and to operate its pipeline system in a manner which, in Transporter's sole opinion, is consistent with the obligations and operating conditions as they may exist from time to time on Transporter's system, or which will allow Transporter to optimize the receipt and/or utilization of gas supplies on its transmission system now and in the future. Nothing in any Service Agreement shall be deemed to impinge on such operating flexibility.

11. SERVICE AGREEMENT AND SERVICE CONDITIONS

- 11.1 Form of Agreement. Transporter and Shipper shall electronically execute a Service Agreement covering the service contemplated using Transporter's standard Form of Service Agreement. The executed Service Agreement, together with the General Terms and Conditions and applicable Rate Schedule shall constitute the entire agreement between the parties for service provided therein and shall be modified only by mutual agreement, electronically in Northwest Passage. Except for the granting of temporary discounts pursuant to Section 3.3(b) of Rate Schedule TI-1 and Section 3.5(b) of Rate Schedule TIL-1, all modifications to Service Agreements must be made in accordance with Section 11.4 of the General Terms and Conditions of Transporter's Tariff.
- 11.2 Electronic Execution. The Service Agreement is executed by Shipper when Shipper submits the required information electronically in Northwest Passage on Transporter's Designated Site. The Service Agreement is executed by Transporter if and when Transporter awards the Service Agreement in Northwest Passage. Electronic execution of a Service Agreement will be binding and enforceable in the same manner and to the same extent as execution of a Service Agreement in writing, and evidence of electronic execution, if introduced on paper, will be admissible to the same extent as other business records originated in written form. Shipper's use of the electronic execution feature of Northwest Passage to execute a Service Agreement, together with Transporter's award thereof, will constitute an executed Service Agreement between the parties in satisfaction of any applicable "statute of frauds."
- 11.3 Notices. Any notices called for in any Service Agreement or communications concerning such Service Agreement will be in written or electronic form and will be considered as having been duly given when delivered personally, or when mailed, faxed, or sent by other electronic measures to either Shipper or Transporter at the place designated. The place designated for Shipper will be the mailing address and contact information for Shipper as set forth on the Business Associate Information form submitted by Shipper, as revised from time to time. The place designated for Transporter will be set forth on Transporter's Designated Site.
- 11.4 Modifications to Service Agreements. For Service Agreements that currently are not in the format of the Form of Service Agreement contained in this Tariff, Shipper may voluntarily restate its agreement using the current Form of Service Agreement with preservation of all pre-existing contractual rights.

11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)

If Transporter and Shipper desire to modify an existing Service Agreement, the following provisions will apply:

- (a) For modifications that are to become effective immediately, the Shipper must electronically submit a request in Northwest Passage to restate its existing Service Agreement in the format of the Form of Service Agreement in the Tariff. Such request must incorporate any effective Addendum to Service Agreement and any effective Amendment(s) made pursuant to Sections 11.5 or 22.12 of the General Terms and Conditions of this Tariff. However, any Addendum to Service Agreement and/or Amendment(s) that were executed before such restatement request which are not yet effective will continue.
- (b) For modifications that are to become effective at a future date, Shipper must: i) electronically submit a request in Northwest Passage to amend its existing Service Agreement using the Form of Service Agreement Amendment in the Tariff; and ii) have an existing Service Agreement in the format of the Form of Service Agreement in the Tariff. Such request must incorporate any effective Addendum to Service Agreement and any effective Amendment(s) made pursuant to Sections 11.5 or 22.12 of the General Terms and Conditions of this Tariff. However, any Addendum to Service Agreement and/or Amendment(s) that were executed before such amendment request which are not yet effective will continue.
- (c) If a Service Agreement states that a date (such as an effective date, a primary term begin date or a primary term end date) is dependent upon the occurrence of a specified event, then once the described event has occurred, Transporter and Shipper will restate the Service Agreement in the format of the current Form of Service Agreement in the Tariff in accordance with Section 11.4(a) above, and will replace the description of the date with the actual date.

- 11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)
 - 11.5 Successors and Assigns. An executed Service Agreement will be binding upon and inure to the benefit of the parties and to their respective successors and assigns.
 - (a) If a Shipper has a corporate name change, including a name change due to a merger or acquisition of substantially all its assets by a third party, the following provisions will apply:
 - i. Shipper must immediately provide written notice and the necessary supporting documentation of the name change to Transporter;
 - ii. Shipper must electronically submit a request to
 Transporter to restate or amend all of its Service
 Agreement(s) or submit an Addendum to Service Agreement to
 reflect the name change;
 - iii. Shipper must satisfy Transporter's credit requirements;
 - iv. Shipper's electronic submission of the Addendum to Service Agreement will constitute Shipper's execution of an amendment to the Service Agreement. Transporter's acceptance of Shipper's name change and supporting documents and approval of Shipper's creditworthiness shall constitute Transporter's execution of an amendment to the Service Agreement;
 - v. Name changes must become effective on the first day of a month unless otherwise agreed to by Transporter and Shipper; and
 - vi. When a Releasing Shipper's Service Agreement has been revised to reflect a name change, the Service Agreement(s) of the related Replacement Shipper(s) will automatically be deemed amended to reflect the new name of the Releasing Shipper and an Addendum to Service Agreement documenting the name change will be added to such Replacement Shipper's related Service Agreement(s).

11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)

- (b) A Shipper may temporarily or permanently release its firm contract rights to a Replacement Shipper pursuant to Section 22 of the General Terms and Conditions.
- (c) A Shipper may assign a Service Agreement, instead of releasing its contract rights, only if the assignee is an affiliate of Shipper.
- If a Shipper desires to assign its capacity to an affiliate of the Shipper, instead of releasing its capacity, the following provisions apply:
 - i. Shipper (Assignor) must immediately provide Transporter with written notice and the necessary supporting documentation of the assignment;
 - ii. Shipper (Assignor) must electronically submit a request to restate or amend all of its Service Agreement(s) or electronically submit an Addendum to Service Agreement to reflect the assignment;
 - iii. Shipper (Assignee) must electronically agree to such
 assignment request;
 - iv. Shipper (Assignee) must satisfy Transporter's credit
 requirements;
 - v. Transporter's consent to such assignment is required and will be demonstrated by its electronic execution of such assignment request. Transporter's consent will not unreasonably be withheld;

11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)

- vi. Shipper's electronic submission of the Addendum to Service Agreement will be deemed Shipper's execution of an amendment to the Service Agreement. Transporter's acceptance of the assignment documentation and approval of Shipper's creditworthiness shall constitute Transporter's execution of an amendment to the Service Agreement;
- vii. Assignments must become effective on the first day of the month unless otherwise agreed to by Transporter and Shippers (Assignor and Assignee);
- viii. When a Releasing Shipper's Service Agreement has been revised to reflect an assignment, the Service Agreement(s) of the related Replacement Shipper(s) will be deemed amended to replace the Assignor's name with the Assignee's name and an Addendum to Service Agreement will automatically be added to such Replacement Shipper's related Service Agreement(s) to reflect such assignment.

11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)

- 11.6 Post-Termination Obligations. Upon termination, a Service Agreement will cease to have any force or effect, save as to any unsatisfied obligations or liabilities of either party arising under the Service Agreement prior to the date of such termination, or arising thereafter as a result of such termination. Provided, however that this provision will not supersede any abandonment authorization which may be required.
- 11.7 Extension of Firm Service Agreement(s) Prior to Expiration. Prior to the expiration of the term of a firm Service Agreement(s) and prior to Transporter posting the availability of such capacity for competitive bid subject to a Shipper's right of first refusal, if any, Transporter and Shipper may mutually agree to negotiate the terms of Shipper's Service Agreement(s) in exchange for Shipper's extension of all or a part of its existing service(s) under its Service Agreement(s). Extensions of such Service Agreement(s) will be negotiated on a case-by-case basis in a not unduly discriminatory manner. To the extent that Transporter and Shipper have mutually agreed to extend such service, the posting and bidding requirements of Section 25 of the General Terms and Conditions of the Tariff will not apply.
- 11.8 Shipper Buy-Out of Firm Service Agreement.
 - (a) General. When Transporter is negotiating with a Shipper that is willing to subscribe to posted unsubscribed capacity or capacity offered in an open season, Transporter may agree to allow such Shipper the right to buy out of the resulting Service Agreement in its entirety.
 - (b) Eligibility. Transporter, on a not unduly discriminatory basis, will only agree to a buy-out if, in Transporter's reasonable judgment, it is financially beneficial to Transporter.
 - To determine if the buy-out is financially beneficial, Transporter may take into account the term of the agreement, the rates, and the value, demand for, and alternative uses of, the capacity under the firm Service Agreement and any exit fee agreed to by Shipper as part of the negotiations to buy out of a firm Service Agreement.
 - (c) Negotiable Terms of Buy-out. Transporter will use its reasonable judgment to establish: 1) the time period in which Shipper must provide written notification to Transporter to implement its right to buy out of a firm Service Agreement; and 2) the payment schedule for Shipper to remit the negotiated exit fees to Transporter. All components of such exit fee will be negotiated between

- 11. SERVICE AGREEMENT AND SERVICE CONDITIONS (Continued)
 - 11.8 Shipper Buy-Out of Firm Service Agreement. (Continued)

Transporter and Shipper, including any present value discount rate (should that be one of the components of the exit fee).

- (d) Requirements to Exercise. To exercise its right to buy out of its firm Service Agreement, Shipper must, as of the effective date of the buy-out, have:
 - i. no unpaid invoice amounts;
 - ii. a zero imbalance;
 - iii. no outstanding releases of capacity pursuant to Section 22 of this Tariff; and
 - iv. no outstanding facilities payment obligation pursuant
 to Exhibit C on the Service Agreement.
- (e) Agreement. The buy-out right negotiated between Transporter and Shipper will be stated on Exhibit E of Shipper's firm Service Agreement.
- 11.9 Conditional Service Agreement Extensions. In negotiating the extension of a firm Service Agreement pursuant to Section 11.7 of the General Terms and Conditions, Transporter and Shipper may mutually agree to give Shipper the option to subsequently reduce the length of such extension under certain conditions provided: (a) the Service Agreement contains a Standard Unilateral Evergreen Provision; (b) the primary term end date of the Service Agreement(s), is extended beyond the primary term end date or evergreen rollover period; (c) the extended Service Agreement(s) containing the option is contemplated to serve anticipated new infrastructure that will utilize Transporter's system; (d) the option to reduce the extension period must be based upon events specified in the Service Agreement related to the anticipated new infrastructure. Shipper's right to reduce the length of any extension pursuant to this provision is forfeited if the Shipper does not notify Transporter on or before the date specified in the Conditional Service Agreement Extension.
- 11.10 Conversion of Rate Schedule LS-2F Capacity. When a Shipper's Rate Schedule LS-2F Service Agreement has been revised to reflect a conversion of capacity, an Addendum to the Service Agreement will automatically be added to the Shipper's base LS-2F Service Agreement to reflect such conversion.

- 12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY
 - 12.1 Scheduling Priorities for Transportation

If Transporter determines that its operationally available Transportation capacity through any point on its system is insufficient to accommodate all otherwise acceptable new Nominations for services under Rate Schedules TF-1, TF-2, TFL-1, TI-1 and TIL-1 in any Nomination cycle through such point, Transporter will schedule its operationally available capacity for Nominations that have not already been scheduled for a given Gas Day in accordance with the following priorities of service:

- (a) First, Nominations for Primary Firm Service will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated pro rata based on the Primary Firm Service Rights associated with such Nominations through the constraint point.
- (b) Second, Nominations for Secondary Firm Service will be scheduled as follows:
 - i) Nominations made pursuant to Rate Schedule TF-2 from the Plymouth LNG Facility where the Nomination is at or in between the receipt and delivery point(s) listed on Shipper's associated TF-1 Firm Service Agreement;
 - ii) Nominations under any Commission-approved Firm Service Agreement with non-conforming provisions which provide Shipper with subordinate rights; and beginning April 1, 2024, nominations on any conforming TF-1 Firm Service Agreement with subordinate rights as indicated on Exhibit A; and

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

iii) All other Nominations for Secondary Firm Service.

If all such Nominations cannot be accommodated, available capacity will be allocated pro rata based on nominating Shippers' Contract Demand.

- (c) Third, Nominations for interruptible services and overrun services will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated based on the rates that apply to such services, from the highest to the lowest. For services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations. Interruptible and overrun services are subject to bumping, as described in Section 12.7. For services with Negotiated Rates, the rates used for scheduling priorities are delineated in Section 12.6.
 - (i) Shippers that are not nominating at the Maximum Base Tariff Rates and desire to elevate their scheduling priority by agreeing to pay a higher rate, not to exceed the Maximum Base Tariff Rate, in the event such higher rate is necessary to allow an otherwise acceptable and confirmable Nomination to be scheduled, must indicate their willingness to do so at the time of Nomination.
- (d) Fourth, Nominations for Makeup and Payback will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated pro rata based on such Nominations.

12.2 Scheduling Priorities for Jackson Prairie Storage

If Transporter determines that its operationally available injection, withdrawal and storage capacities at the Jackson Prairie storage facility are insufficient to accommodate Transporter's balancing needs as well as all otherwise acceptable new Nominations for services under Rate Schedules SGS-2F, SGS-2I, DEX-1, TPAL, and PAL in any Nomination cycle, Transporter will schedule

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

its operationally available capacity for Nominations that have not already been scheduled for a given Gas Day in accordance with the following priorities of service:

- (a) First, Nominations for firm service under Rate Schedule SGS-2F and Transporter's firm balancing rights will be scheduled. If all such Nominations cannot be accommodated, available capacity for injections will be allocated pro rata based on Transporter's and Shipper's Storage Capacities and available capacity for withdrawals will be allocated pro rata based on Transporter's and Shipper's Storage Demand, which is subject to the limitations in Rate Schedule SGS-2F, Section 9.2.
- (b) Second, Nominations for deliveries of Deferred Exchange Quantities at Jackson Prairie under Rate Schedule DEX-1, up to the Daily Delivery Quantity, will be scheduled. If all such Nominations cannot be accommodated, capacity will be allocated based on rate, highest to lowest. For services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations.
- (c) Third, Nominations for Transporter's best-efforts system balancing requirements will be scheduled.
- (d) Fourth, Nominations (including Shipper's requests for inground storage transfers) for interruptible storage capacity under Rate Schedules SGS-2F (as best-efforts), SGS-2I, PAL and/or TPAL will be scheduled. If all such storage capacity Nominations cannot be accommodated, available capacity will be allocated based on rate, highest to lowest. For Rate Schedules SGS-2F (as best-efforts), SGS-2I, PAL and/or TPAL services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations. To the extent there is more storage capacity than there is injection and/or withdrawal capacity, Transporter will separately and subsequently allocate capacity for in-ground transfer requests using the methodologies herein.

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

After scheduling interruptible storage capacity, Nominations for injections/withdrawals for interruptible service under Rate Schedules SGS-2I, SGS-2F (best-efforts), DEX-1 (Deferred Exchange Quantities that exceed the Daily Delivery Quantity), and PAL will be scheduled. If all such injection and/or withdrawal Nominations cannot be accommodated, available capacity will be allocated pro rata based on otherwise acceptable and confirmable Nominations.

For services with Negotiated Rates, the rates used for scheduling priorities are delineated in Section 12.6.

12.3 Scheduling Priorities for Clay Basin Storage

If Transporter determines that its operationally available injection, withdrawal, and storage capacities at the Clay Basin storage facility are insufficient to accommodate all otherwise acceptable new Nominations related to system balancing and/or, for services under Rate Schedules DEX-1, TPAL, and PAL in any Nomination cycle, Transporter will schedule the operationally available capacity for Nominations that have not already been scheduled for a given Gas Day in accordance with the following priorities of service:

- (a) First, Nominations for Transporter's system-balancing requirements will be scheduled.
- (b) Second, Nominations for deliveries of Deferred Exchange Quantity under Rate Schedule DEX-1, up to the Daily Delivery Quantity, will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated based on rate, highest to lowest. For services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations.
- (c) Third, Nominations (including Shipper's requests for in-ground storage transfers) for interruptible storage capacity under Rate Schedules PAL and/or TPAL will be scheduled. If all such storage

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

capacity Nominations cannot be accommodated, available capacity will be allocated based on rate, highest to lowest. For Rate Schedules PAL and/or TPAL services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations. To the extent there is more storage capacity than there is injection and/or withdrawal capacity, Transporter will separately and subsequently allocate capacity for in-ground transfer requests using the methodologies herein.

After scheduling interruptible storage capacity, Nominations for injections/withdrawals for interruptible service under Rate Schedules DEX-1 (Deferred Exchange Quantities that exceed the Daily Delivery Quantity) and PAL will be scheduled. If all such injection and/or withdrawal Nominations cannot be accommodated, available capacity will be allocated pro rata based on otherwise acceptable and confirmable Nominations.

For services with Negotiated Rates, the rates used for scheduling priorities are delineated in Section 12.6.

12.4 Scheduling Priorities for Plymouth LNG Facilities

If Transporter determines that its operationally available liquefaction, vaporization and/or LNG delivery capacities at the Plymouth LNG Facility are insufficient to accommodate all otherwise acceptable new Nominations for services under Rate Schedules LS-2F, LS-3F, LS-2I and LD-4I in any Nomination cycle, Transporter will schedule its operationally available capacity for Nominations that have not already been scheduled for a given Gas Day in accordance with the following priorities of service:

(a) First, Nominations for firm service under Rate Schedules LS-2F and LS-3F will be scheduled. If all such Nominations cannot be accommodated, available liquefaction capacity and available LNG delivery capacity will be allocated pro rata based on Shipper's Storage Capacities and available vaporization capacity will be allocated based on Shipper's Storage Demand.

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

- (b) Second, Nominations for interruptible service under Rate Schedules LS-2I and LD-4I, and Nominations for overrun service under Rate Schedules LS-2F and LS-3F will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated according to the rates applicable to the service, from highest to lowest. For services at the same rates, any necessary allocations will be pro rata based on Nominations. For services with Negotiated Rates, the rates used for scheduling priorities are delineated in Section 12.6.
- 12.5 Scheduling Priorities for Park and Loan at Non-Storage Points

Nominations (including Shipper's requests for transfers) for interruptible service under Rate Schedules PAL and/or TPAL will be scheduled. If all such Nominations cannot be accommodated, available capacity will be allocated based on rate, highest to lowest. For Rate Schedules PAL and/or TPAL services at the same rates, any necessary allocations of available capacity will be pro rata based on otherwise acceptable and confirmable Nominations.

For services with Negotiated Rates, the rates used for scheduling priorities are delineated in Section 12.6.

12.6 Negotiated Rate Used to Establish Priority

For non-firm services subject to Negotiated Rates, the rate used for scheduling priorities will be:

- (1) the lower of the Negotiated Rate or the corresponding Recourse Rate, if the Negotiated Rate is a stated rate;
- (2) the lower of the minimum Negotiated Rate or the corresponding Recourse Rate, if the Negotiated Rate is a formula rate with a specified minimum; or
- (3) the corresponding minimum base rate set forth on the Statement of Rates, if the Negotiated Rate is a formula rate without a specified minimum.

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

12.7 Bumping

Previously Scheduled Quantities for interruptible, overrun or bestefforts, and park and loan services may be bumped in accordance with the following procedures:

- (a) Transporter will give scheduling priority to Firm Nominations submitted during the evening Nomination, Intraday 1 and Intraday 2 Nomination Cycles over previously Scheduled Quantities for interruptible, overrun or best-efforts, and park and loan services. The elapsed-prorated-scheduled quantity process will apply to previously Scheduled Quantities bumped in the Intraday 1 and Intraday 2 Nomination Cycles.
- (b) If previously Scheduled Quantities will be reduced as a result of bumping, Transporter will report such reductions on its Designated Site .
- 12.8 Previously Scheduled Quantities at the Agreement Level

To the extent that a Shipper, by nomination, revises Scheduled Quantities on a service agreement through a receipt or delivery point, on the mainline or on a lateral in subsequent Nomination cycles, the quantity that remains scheduled on such Agreement through such point after the revision will not be considered a new Nomination through the point.

12.9 Curtailment Policy

If Transporter determines that operationally available capacity through any constraint point on its system has been reduced for any reason whatsoever to a level insufficient to accommodate all previously Scheduled Quantities, in the next scheduling cycle, Transporter will fill the reduced operationally available capacity as follows:

- i) Elapsed-Prorated Scheduled Quantities will be scheduled first;
- ii) Previously Scheduled Quantities will be scheduled second and will be scheduled using the priority levels delineated in Sections 12.1, 12.2, 12.3, 12.4 or 12.5 above, except that Transporter will use the Contract Demand of the Firm Service Agreement as the basis for filling the reduced operationally available capacity (once scheduled in a NAESB cycle, Secondary Firm Nominations are deemed Primary Firm Nominations).

12. SCHEDULING PRIORITIES AND CURTAILMENT POLICY (Continued)

12.10 Shipper Notice Responsibility

Upon receipt of Transporter's notification of Scheduled Quantities under Section 12, Shipper will have the responsibility to inform its end-users, suppliers and all others involved in the affected transaction, except parties that interconnect with Transporter.

12.11 Operating Limitations

Notwithstanding the other provisions of this Section 12, Transporter will not be required to interrupt service to any Shipper served from those locations on Transporter's system where, due to operational or capacity limitations, Transporter's ability to maintain the priorities provided herein would not be improved. Transporter will not be obligated to modify its system operations or its operating procedures related to firm service in order to ameliorate specific operating and capacity limitations which result in interruptible service curtailment.

12.12 Curtailment / Allocation Liability

If Transporter is required to curtail Scheduled Quantities or allocate available capacity during the term of the Service Agreement and if the curtailment or an allocation is implemented for any reason, Transporter will not be liable for any damages caused by implementation of curtailment and/or allocation procedures as set forth herein except to the extent that such curtailment and/or allocation is shown to be the result of negligence or misfeasance by Transporter.

12.13 Nominations Subsequent to the Intraday 3 Nomination Cycle

To the extent Transporter's existing transportation and storage obligations are not compromised, Shipper may request changes in Scheduled Quantities following the Intraday 3 Nomination Cycle. All Nominations must be confirmed by the upstream/downstream operators and Transporter.

First Revised Sheet No. 223 Superseding Substitute Original Sheet No. 223

13. GAS TECHNOLOGY INSTITUTE VOLUNTARY CONTRIBUTIONS

- 13.1 Purpose. The Gas Technology Institute ("GTI"), an Illinois not-for-profit corporation, has been organized for the purpose of sponsoring research, development and demonstration programs in the field of natural and manufactured gas for the purpose of assisting all segments of the gas industry in providing adequate, reliable, safe, economic and environmentally acceptable gas service for the benefit of gas consumers and the general public.
- 13.2 Voluntary Contributions. Transporter will provide a "check the box" option on invoices for Shippers to voluntarily contribute funds to \mathtt{GTI} .
- 13.3 Remittance to GTI. Transporter shall collect from Shippers and remit to GTI, not later than fifteen (15) days after the receipt thereof, all monies received by virtue of voluntary GTI contributions.

14. OPERATING CONDITIONS

14.1 Nomination Procedure for Gas Receipts and Deliveries.

(a) Nominations. NAESB WGQ Nominations Related Standards 1.2.1, 1.3.3 through 1.3.9, 1.3.11, 1.3.13, 1.3.20, 1.3.21, 1.3.24, 1.3.25, 1.3.27, 1.3.32, 1.3.33, 1.3.35 through 1.3.46, 1.3.48, 1.3.53, 1.3.55, 1.3.56, 1.3.58, 1.3.62, 1.3.79, and 1.3.80 will apply.

The following timelines in Central Clock Time (with Mountain Clock Times in parentheses) reflect NAESB WGQ Standard 1.3.2:

Timely Nomination Cycle (time on day prior to flow)

Nomination leaves control of Nominating Party	1:00 P.M. (12:00 Noon)				
Receipt of nomination by Transporter	1:15 (12:15) P.M.				
Quick response by Transporter for validity of data elements	1:30 (12:30) P.M.				
Receipt of completed confirmations by Transporter from upstream and downstream connected parties	4:30 (3:30) P.M.				
Scheduled quantities available to Shippers and point operators	5:00 (4:00) P.M.				
Scheduled quantities commence flow	9:00 (8:00) A.M.(Gas Day)				
ing Nomination Cycle (time on day prior to flow)					

Evening Nomination Cycle (time on day prior to flow)

Nomination leaves control of Nominating Party	6:00	(5:00)	P.M.
Receipt of nomination by Transporter	6:15	(5:15)	P.M.
Quick response by Transporter for validity of data elements	6:30	(5:30)	P.M.
Receipt of completed confirmations by Transporter from upstream and downstream connected parties	8:30	(7:30)	P.M.

Scheduled quantities available to Shippers, point operators and bumped parties	9:00 (8:00) P.M.				
Scheduled quantities commence flow Day)	9:00 (8:00) A.M.(Gas				
Intraday 1 Nomination Cycle (time on Gas Day)					
Nomination leaves control of Nominating Party	10:00 (9:00) A.M.				
Receipt of nomination by Transporter	10:15 (9:15) A.M.				
Quick response by Transporter for validity of data elements	10:30 (9:30) A.M.				
Receipt of completed confirmations by Transporter from upstream and downstream connected parties	12:30 P.M. (11:30 A.M.)				
Scheduled quantities available to Shippers, point operators and bumped parties	1:00 P.M. (12:00 Noon)				
Scheduled quantities commence flow	2:00 (1:00) P.M.				
Intraday 2 Nomination Cycle (time on Gas Day)					
Nomination leaves control of Nominating Party	2:30 (1:30) P.M.				
Receipt of nomination by Transporter	2:45 (1:45) P.M.				
Quick response by Transporter for validity of data elements	3:00 (2:00) P.M.				
Receipt of completed confirmations by Transporter from upstream and downstream connected parties	5:00 (4:00) P.M.				
Scheduled quantities available to Shippers and point operators	5:30 (4:30) P.M.				
Scheduled quantities commence flow	6:00 (5:00) P.M.				

14. OPERATING CONDITIONS (Continued)

Intraday 3 Nomination Cycle (time on Gas Day)

Nomination leaves control of Nominating Party	7:00	(6:00)	P.M.
Receipt of nomination by Transporter	7:15	(6:15)	P.M.
Quick response by Transporter for validity of data elements	7:30	(6:30)	P.M.
Receipt of completed confirmations by Transporter from upstream and downstream connected parties	9:30	(8:30)	P.M.
Scheduled quantities available to Shippers and point operators	10:00	(9:00)	P.M.
Scheduled quantities commence flow	10:00	(9:00)	P.M.

Bumping is not allowed during the Intraday 3 Nomination Cycle.

For the purposes of this section, the word "provides" shall mean, for transmittals pursuant to NAESB WGQ Standards 1.4.x, receipt at the designated site, and for purposes of other forms of transmittal, it shall mean send or post.

RESERVED FOR FUTURE USE

First Revised Sheet No. 225-I Superseding Substitute Original Sheet No. 225-I

GENERAL TERMS AND CONDITIONS (Continued)

14. OPERATING CONDITIONS (Continued)

(b) Balancing Nominations. Nominating Party shall similarly nominate to Transporter's Nomination personnel the quantity of natural gas to be used to balance Receiving Party Imbalances and Shipper Imbalances, as defined in Section 15 hereof, and shall schedule such balancing gas in accordance with Transporter's nomination procedures, subject to available capacity. A nomination request may include both daily nominations and balancing nominations but such nominations must be separately identified so that priorities of service can be maintained.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet No. 226 Superseding Substitute Original Sheet No. 226

RESERVED FOR FUTURE USE

- 14. OPERATING CONDITIONS (Continued)
 - (i) NAESB WGQ Nominations Related Standard 1.3.19 will apply.
 - (ii) Nominating Party/Shipper is responsible to ensure that the gas supply is available in sufficient quantity at the designated Receipt Points before the nominations are communicated to Transporter. Preceding the confirmation of nominations, Transporter has the right to verify the consistency of each nomination with Transporter's service agreements with Shipper(s), availability of capacity up and downstream of Transporter's system, and the availability of mainline capacity necessary to transport the gas. Transporter may reject nominations if such information cannot be verified. Transporter will notify Receiving Party and/or Nominating Party and/or Shipper of the quantity that Transporter can receive and deliver pursuant to the procedures in Section 14.4 below.
 - (iii) Nominating Party/Shipper is responsible for the accuracy of all nomination requests. Transporter will only receive nominations through Transporter's Designated Site. All nomination information, except volume data, is considered correct and confirmed by Nominating Party when it is communicated to Transporter.
 - (iv) Transporter shall not schedule gas for delivery to a Small Customer subject to Rate Schedule TF-1 under any transportation Service Agreement (excluding its Rate Schedule TF-2 Service Agreement at Plymouth held at the time of storage service unbundling in RP93-5) unless such Small Customer has scheduled its full Contract Demand for firm service under its Rate Schedule TF-1 (Small Customer) Service Agreement(s) for that day.

- 14.2 Consolidated Nominations. Transporter may approve a Shipper's request to consolidate the nominations under its Service Agreements from the same Rate Schedule having the same receipt and delivery point, provided that the consolidated nominations would not negatively impact other Shippers and Transporter can reasonably administer the request under this Tariff, utilizing Transporter's current nomination system.
- 14.3 Nomination Changes After the Start of the Gas Day. Nomination changes may be made after the start of the Gas Day pursuant to the Intraday 1, Intraday 2, and Intraday 3 Nomination Cycles. Such transportation nominations will be subject as necessary to confirmation of the nomination changes with any involved third party upstream of the transportation receipt point and downstream of the transportation delivery point.

- 14. OPERATING CONDITIONS (Continued)
 - 14.4 Confirmation and Scheduling of Gas Receipts and Deliveries. NAESB WGQ Nominations Related Standards 1.3.22 and 1.3.23 will apply.

All nominations are considered scheduled and confirmed for receipt and/or delivery when confirmations are made available to Nominating Party and/or Shipper and/or Receiving Party on Transporter's Designated Site.

14.5 Limitation Upon Transporter's Daily Transportation Obligation. Transporter shall not be obligated to deliver to Receiving Party or an interconnecting pipeline on any day a greater quantity of gas than it has received and scheduled for delivery. Transporter is not obligated to manipulate its pipeline operation in order for the Receiving Party to avoid unauthorized overruns as provided in Section 15 of the General Terms and Conditions, or the Receiving Party to avoid operating expenses of compression or other related facilities on its system.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet No. 228-B Superseding Substitute Original Sheet No. 228-B

RESERVED FOR FUTURE USE

- 14. OPERATING CONDITIONS (Continued)
 - 14.6 Declared Entitlement Period.
 - (a) General. A Declared Entitlement Period is a time period, declared by Transporter, during which unauthorized overrun or underrun provisions apply as a result of an interruption or curtailment due to capacity constraints (including constraints that cause Deficiency Periods as defined in Section 10.2 of Rate Schedule TF-1), supply interruptions, or the existence of any undertake or overtake situation which jeopardizes system integrity.
 - (b) Declared Entitlement Notice, Contents and Procedures.
 - (i) Declaration of Entitlement Period via Portal Posting

As early as reasonably possible but no later than two hours prior to commencement of the gas day, Transporter will notify Receiving Party(s) when a Declared Entitlement Period is in effect.

Transporter may not announce a Declared Entitlement for the coming gas day after close of the confirmation process (Evening Nomination Cycle) on the day prior to such gas day unless (i)

Transporter has given a warning based on available data by posting to Transporter's Designated Site the possibility of a Declared Entitlement Period prior to the close of such confirmation period, or (ii) changes to operational circumstances or data occur after the close of such confirmation process which Transporter determines require a Declared Entitlement.

Operating conditions that may trigger post-confirmation, same-day entitlements are extensive supply failure, storage facility failure, pipeline or compression failure, excess linepack drafting, unexpected and extremely cold weather, or firm curtailment. Transporter will provide Shippers with actual notice of such operating conditions in the form of posting an All Shippers Notice to Transporter's Designated Site. Each Shipper shall designate an operational contact, and/or a third party designee, for Transporter to notify on a 24-hour a day, 365 days a year basis. If Transporter is unable to notify any Shipper because that Shipper's contact is unavailable, such Shipper shall be solely responsible for any consequences arising from such failure of communication.

(ii) Declared Entitlement Period Notice

To issue a general system or customer-specific Declared Entitlement Period, Transporter will post an entitlement notification that contains the following:

- (a) Time and Date of issuance;
- (b) Date that the entitlement is considered effective;

14. OPERATING CONDITIONS (Continued)

- Identify that portion of the system contributing to the operational problem and, in the case of a customer-specific entitlement, provide notification to each Receiving Party that (1) has contributed to the adverse operating conditions for at least two consecutive days; and (2) whose actual measured deliveries have deviated from scheduled deliveries for the previous Gas Day by the greater of 10,000 Dth or 5 percent. Before issuing a customer-specific entitlement, Transporter will attempt to remedy the adverse operating condition through requests for voluntary action; provided, however, exigent circumstances may exist which require immediate issuance of a customer-specific entitlement. Transporter will not issue a customer-specific Receiving Party entitlement for the same time period and portion of the system in which a general system Declared Entitlement Period has been issued. Due to operational constraints on Transporter's system, there may be instances when a general system Declared Entitlement Period will apply to Transporter's entire system;
- (d) Specify the entitlement as either an overrun or an underrun entitlement and will specify the percentage, as detailed in Section 15.5(c), by which a Receiving Party's unauthorized overruns or underruns may differ from daily entitlements, as outlined in Section 15.5(b), before Section 15.5 penalties are imposed; and
- (e) Any other terms Transporter may reasonably require to ensure the effectiveness of the Declared Entitlement Period.

After an entitlement has been lifted, Transporter will post to its Designated Site information on factors that caused the entitlement to be issued and then lifted.

If deliveries to Receiving Party exceed scheduled and authorized volumes during any day that such a general system or customer-specific Declared Entitlement Period is in effect or deliveries are less than scheduled and authorized volumes in an undertake situation, the unauthorized overrun or underrun charge provisions as described in Section 15 of the General Terms and Conditions will apply.

The provisions of Section 14.3, Nomination Changes After the Start of the Gas Day, are applicable to allow Shippers to adjust receipts and/or deliveries as necessary during such Declared Entitlement Period.

14. OPERATING CONDITIONS (Continued)

- 14.7 Records of Scheduled Quantities. Transporter shall keep accurate records of the quantities of gas nominated and scheduled for transportation, and any imbalances, which records shall be made available to Receiving Party or Shipper upon request. If Receiving Party's gas is commingled with other gas at the point of receipt or at the point of delivery, the scheduling arrangements and Transporter's records shall include procedures for the allocation of the total quantity at such points.
- 14.8 Arrangements Prior to Receipt and After Delivery. It shall be Shipper's obligation to make arrangements with other parties for delivery of gas into Transporter's system at the Receipt Point(s) and for the receipt of gas after transportation at the Delivery Point(s). Shipper shall be obligated in such arrangements to require dispatching and operating coordination with Transporter and access to appropriate charts and records. Such arrangements shall not be accepted by Transporter unless found to be satisfactory by Transporter. Shipper shall pay for any and all third party services including, but not limited to compression, processing, gathering, transportation or assigned nominating and balancing responsibilities or services.
- 14.9 Operating Information and Estimates. Upon request of Transporter, Shipper shall from time to time submit its best estimates of the daily, monthly and annual volumes of gas to be transported, including peak day requirements, together with such other operating data as Transporter may require in order to schedule its operations.
- 14.10 System Operations and Storage. Transporter shall not be required to compress transportation gas into its pipeline, lower its system operating pressure, alter the direction of gas flow, or alter other operation or utilization of its facilities or otherwise change its normal pipeline operations in order to receive, transport or deliver gas hereunder.

Transporter shall utilize its system storage capacity to accommodate transportation imbalances to the extent that existing transportation and storage obligations are not compromised.

14.11 Receipt and Delivery Points. Refer to Section 2 of the General Terms and Conditions for operating procedures and pressures at receipt and delivery points. Once confirmed as outlined in Section 14.4, service to an alternate receipt and/or delivery point shall have the same priority of service rights as a primary receipt and/or delivery point.

14. OPERATING CONDITIONS (Continued)

14.12 Fuel Gas Reimbursement. NAESB WGQ Nominations Related Standards 1.3.15, 1.3.16 and 1.3.28 through 1.3.31 will apply.

In addition to the payments for transportation and storage, Shipper will reimburse Transporter for Shipper's pro rata share of gas used for fuel, including lost or gained and unaccounted-for gas, based on a fuel use requirements factor (Factor) as set forth on Sheet No. 14. The Factor for Rate Schedules TF-1, TF-2, TI-1 and DEX-1 (Transportation Rate Schedules) and the Factor(s) for Rate Schedules TFL-1 and TIL-1 (Lateral Rate Schedules) will be determined semiannually to become effective on April 1 and on October 1 of each year. A separate Factor will be determined for each Designated Lateral under the Lateral Rate Schedules. The Factor for Rate Schedules SGS-2F and SGS-2I, (Underground Storage Rate Schedules) and Rate Schedules LS-2F, LS-3F, LS-2I and LD-4I (LNG Rate Schedules) will be determined annually to become effective on April 1 of each year. Fuel usage and lost or gained and unaccounted-for volumes incurred at the Jackson Prairie Storage Project attributable to system balancing requirements will be included in the Factor for the Transportation Rate Schedules. The Factor for the Transportation Rate Schedules and the Factor(s) for the Lateral Rate Schedules may also include a component for volumetric increases or decreases to Transporter's system occurring during the prior calendar year as described below. The Factors will be calculated as follows:

- (a) Projected fuel usage volumes will be determined for services under the Transportation Rate Schedules and for services on each separate Designated Lateral under the Lateral Rate Schedules for the upcoming six months beginning April 1 or October 1. Projected fuel usage will be determined for services under the Underground Storage Rate Schedules and for services under the LNG Rate Schedules for the upcoming twelve months beginning April 1.
- (b) Projected lost or gained and unaccounted-for volumes will be determined for services under the Transportation Rate Schedules and for services on each separate Designated Lateral under the Lateral Rate Schedules for the upcoming six months beginning April 1 or October 1. Projected lost or gained and unaccounted-for volumes will be determined for services under the Underground Storage Rate Schedules and for services under the LNG Rate Schedules for the upcoming twelve months beginning April 1.

- (c) For the six months ending December 31 (to be included in the applicable April 1 calculations) or for the six months ending June 30 (to be included in the applicable October 1 calculations), Transporter will determine the net increase or decrease in volumes to Transporter's system pursuant to Sections 15.10(b), 15.11 and 22.2(b) of the General Terms and Conditions, Section 2.2 of Rate Schedule SGS-2I, Section 2.4 of Rate Schedule LS-2I, Section 5.2 of Rate Schedule DEX-1, Section 5.3 of Rate Schedule TPAL, and Section 5.3 of Rate Schedule PAL. Those volumes under Sections 15.10(b) and 15.11 associated with service on a Designated Lateral will be incorporated in the Factor calculation for such Designated Lateral.
- (d) Projected volumes to be nominated for receipt by Transporter will be determined for services under the Transportation Rate Schedules and for services on each separate Designated Lateral under the Lateral Rate Schedules for the upcoming six months beginning April 1 or October 1. Projected volumes to be nominated for injection by Transporter for services under the Underground Storage Rate Schedules and for liquefaction and vaporization by Transporter for services under the LNG Rate Schedules will be determined for the upcoming twelve months beginning April 1.
- The sum of the projected volumes determined pursuant to Sections 14.12(a) and (b) that apply to services under the Transportation Rate Schedules and the volumes determined pursuant to Section 14.12(c), except such volumes associated with service on Designated Laterals, will be divided by the projected receipt volumes determined pursuant to Section 14.12(d) that apply to services under the Transportation Rate Schedules to derive the Factor for the Transportation Rate Schedules. The sum of the projected volumes determined pursuant to Sections 14.12(a) and (b) that apply to Transporter's services on each Designated Lateral and the volumes determined pursuant to Section 14.12(c) associated with service on each Designated Lateral will be divided by the projected receipt volumes determined pursuant to Section 14.12(d) that apply to services on each Designated Lateral to derive the Factor for each Designated Lateral under the Lateral Rate Schedules.

- (f) The sum of the projected volumes determined pursuant to Sections 14.12(a) and (b) that apply to services under the Underground Storage Rate Schedules will be divided by the projected injection volumes determined pursuant to Section 14.12(d) to derive the Factor for the Underground Storage Rate Schedules. The sum of the projected volumes determined pursuant to Sections 14.12(a) and (b) that apply to services under the LNG Rate Schedules will be assigned to either the liquefaction or vaporization processes and will be divided by the projected liquefaction and vaporization volumes determined pursuant to Section 14.12(d) to derive the Factors for the LNG Rate Schedules.
- (g) All projected data will include true-up adjustments for variances between fuel reimbursements and fuel used, including lost or gained and unaccounted-for volumes. For the Factors applicable to the Transportation Rate Schedules and for each Designated Lateral under the Lateral Rate Schedules, the true-up adjustments will be determined for the prior six months ending December 31 (to be included in the April 1 calculations) or for the prior six months ending June 30 (to be included in the October 1 calculations). For the Factors applicable to the Underground Storage Rate Schedules and the LNG Rate Schedules, the true-up adjustments will be determined for the prior twelve months ending December 31.
- (h) In addition to the Rate Schedule TF-1 Factor, a fixed 0.50% Evergreen Expansion Incremental Surcharge will apply to the quantity of gas nominated for receipt at the Sumas, SIPI or Pacific Pool receipt points under Evergreen Expansion service agreements. The projected fuel usage for services under the Transportation Rate Schedules determined pursuant to Section 14.12(a) will be adjusted to exclude fuel usage equivalent to projected fuel usage reimbursements from the Evergreen Expansion Incremental Surcharge.

First Revised Sheet No. 231-C Superseding Substitute Original Sheet No. 231-C

GENERAL TERMS AND CONDITIONS (Continued)

- 14.13 Balancing. Balancing of thermally equivalent volumes of gas received and delivered shall be achieved as nearly as feasible on a daily basis, with any cumulative imbalance accounted for as provided in Section 15. Imbalance makeup shall be nominated and scheduled in accordance with the nomination procedures of this Section or netted or traded as provided in Section 15.
- 14.14 System Integrity. Nothing herein shall limit Transporter's right to take actions or to require Receiving Parties or Shippers hereunder to take actions of whatever nature may be required to correct any situations which may threaten the operation or integrity of its system, including maintenance of service to other customers. Among the actions Transporter shall be entitled to take is the issuance of Operational Flow Orders in accordance with Section 14.15 below.

- 14. OPERATING CONDITIONS (Continued)
 - 14.15 Operational Flow Order ("OFO")
 - (a) General. NAESB WGQ Nominations Related Standards 1.3.26 and 1.3.34 will apply.

This Section 14.15 will apply to any Shipper, including any Replacement Shipper, receiving firm transportation service under any Rate Schedule contained in Transporter's FERC Gas Tariff. Transporter will issue OFOs when in its reasonable judgment Transporter determines OFOs are necessitated by the condition described in Section 14.15(b) hereof, and when all other available operating measures, including storage options and excluding any firm service curtailment, have been exhausted.

This Section 14.15 also will apply to the implementation and administration of contract-specific OFOs issued by Transporter within the contract-specific OFO parameters set forth on Exhibit A or a FERC approved non-conforming provision on Exhibit B of Rate Schedule TF-1 and TF-2 Service Agreements.

- (b) OFO Operational Conditions. An OFO may be issued when primary firm nominations through a given constraint point exceed the greater of the physical design capacity or the posted operationally available capacity for a given day. Transporter will not issue an OFO when Transporter's inability to meet firm transportation commitments is due to scheduled or unscheduled maintenance, force majeure, or the Unavailability of Transporter's Facilities.
- Reliance on Scheduled Quantities for OFO Calculations. When calculating OFO obligations to alleviate the adverse operating condition, Transporter relies on Scheduled Quantities from the Timely Nomination Cycle that are (i) on the downstream side of the constraint point and scheduled on a Service Agreement subject to a Realignment OFO, or (ii) flowing through the OFO constraint point in the nonconstrained direction. Due to this reliance, Shippers may not reduce such Scheduled Quantities in subsequent Nomination cycles during the Gas Day for which Transporter has issued an OFO. If such Scheduled Quantities are reduced in a subsequent Nomination cycle during the Gas Day on which an OFO has been issued and not restored in the next available Nomination cycle, they will be subject to the same OFO penalties as those outlined in Section 14.15(f) herein, unless a Shipper reduces a corresponding quantity flowing through the constraint point, in the constrained direction, or takes other actions acceptable to Transporter.
- (d) Order of Operational Remedies. Transporter will take the following actions in order to alleviate adverse operating conditions utilizing the mechanisms below in the following order:

14. OPERATING CONDITIONS (Continued)

- (i) Entitlement and Scheduling Remedies to the extent Transporter determines they will alleviate the adverse operating condition, invoke the scheduling and entitlement mechanisms provided for by Transporter's Tariff;
- (ii) Contract-Specific OFO to the extent Transporter determines they will alleviate the adverse operating condition, invoke, pro rata based on applicable contract quantities, specific contractual provisions which require, as a condition of a Capacity Release or a commitment of capacity, certain Shippers to flow gas upon Transporter's request;

If after a good faith effort a Shipper is unable to find a market for OFO compliance gas, Transporter will deem a Shipper's nomination as confirmed, even if a downstream interconnecting pipeline or delivery point where the gas is intended to flow fails to confirm a market or accept the gas. Transporter will treat such nomination for gas that did not actually move through the delivery point as a Shipper Imbalance and handle any imbalance resulting from OFO compliance as provided in Section 15.10.

If a Shipper subject to a Contract-Specific OFO is unable to comply with the terms of an OFO due to failure of gas supply, penalties shall not be assessed to Shipper. Failure of gas supply occurs when a Shipper is unable to make a nomination at a receipt point from the downstream side of an OFO constraint point because such Shipper cannot, after exerting good faith efforts, physically obtain gas supply at any such receipt points by the timeline pursuant to Section 14.15(e). Shipper will provide to Transporter a signed affidavit, the form of which is posted on Transporter's Designated Site, prior to the Intraday 1 Nomination Cycle on the Gas Day for which the OFO has been issued, of its good faith efforts to obtain gas supply before Transporter will honor Shipper's request to be exempt from penalties.

(iii) Realignment OFO - Service Agreements that are subject to a Realignment OFO, requiring Shippers with unutilized MDQs on the downstream side of the constraint point to realign to those points, are those Service Agreements that meet the following criteria: have capacity that contains receipt points on both sides of the constrained point with primary firm corridor rights through such constrained point.

A Shipper's Realignment OFO obligation on any given day for a Service Agreement that is subject to a Realignment OFO is based upon the percentage of primary rights that are scheduled through a given constraint point and its unutilized MDQ on the downstream

14. OPERATING CONDITIONS (Continued)

side of such constraint point. In no event will a Shipper be required to bring on supplies on the downstream side of a constraint point that exceed the scheduled quantity that was scheduled through the constraint point or its unutilized MDQ on the downstream side of the constraint point.

Nominating Shippers may elect to voluntarily decrease their nominations through a constrained area by the volume which is subject to the OFO, in lieu of realigning to supplies on the opposite side of the constraint point. If Shipper is unable to comply with the terms of an OFO due to failure of gas supply, penalties shall not be assessed to Shipper, and reservation charge adjustments will not be paid to such Shipper by Transporter; however, such Shipper's nominations will be decreased by a volume equivalent to that quantity of gas that does not comply with such OFO. In addition, Transporter will decrease Shipper's nomination by a volume equivalent to that quantity of gas that does not comply with such OFO if Northwest has invoked a Supply Shortage OFO pursuant to Section 14.15(v).

Shippers subject to a Realignment OFO will be required to, at a minimum, maintain their scheduled quantities from the Timely Nomination Cycle from receipt points on the downstream side of an OFO constraint point through the end of the Gas Day on which the OFO was declared. Realignment OFO Shippers that reduce such receipts on the downstream side of an OFO constraint point, and who do not restore such reduced receipts in the next available Nomination cycle, or who offset their storage withdrawal with a storage injection nomination after the Timely Nomination Cycle must also reduce their Scheduled Quantities through the OFO constraint point; and

(iv) Must-flow OFO - Service Agreements that are subject to a Must-flow OFO, requiring Shippers to nominate their pro rata share (based on primary firm corridor rights) of the displacement requirement necessary to provide displacement nominations through a given constraint point in order for Transporter to obtain a net Scheduled Quantity that does not exceed the greater of the physical design capacity or the posted operationally available capacity for a given Gas Day are those Service Agreements that have capacity capable of providing displacement gas through a given constraint point utilizing their primary corridor rights.

Transporter will issue a Must-flow OFO to such Shippers for the Gas Day(s) the Must-flow OFO is to be in effect, requiring such Shippers to nominate their pro rata share (based on corridor rights) of the displacement requirement necessary to alleviate the operating condition (but not to exceed Contract Demand), regardless of amounts such Shippers may have nominated.

14. OPERATING CONDITIONS (Continued)

A Shipper subject to the Must-flow OFO will only be required to flow up to the same total percentage of their corridor rights as all Shippers, regardless of the percentage flowing before any OFO is issued, not to exceed its corridor rights through the constrained point.

If after a good faith effort a Shipper is unable to find a market for OFO compliance gas, Transporter will deem a Shipper's nomination as confirmed, even if a downstream interconnecting pipeline or delivery point where the gas is intended to flow fails to confirm a market or accept the gas. Transporter will treat such nomination for gas that did not actually move through the delivery point as a Shipper Imbalance and handle any imbalance resulting from OFO compliance as provided in Section 15.10.

If a Shipper subject to a Must-flow OFO is unable to comply with the terms of an OFO due to failure of gas supply, penalties shall not be assessed to Shipper. Failure of gas supply occurs when a Shipper is unable to make a nomination at a receipt point from the downstream side of an OFO constraint point because such Shipper cannot, after exerting good faith efforts, physically obtain gas supply at any such receipt points by the timeline pursuant to Section 14.15(e). Shipper will provide to Transporter a signed affidavit, the form of which is posted on Transporter's Designated Site, prior to the Intraday 1 Nomination Cycle on the Gas Day for which the OFO has been issued, of its good faith efforts to obtain gas supply before Transporter will honor Shipper's request to be exempt from penalties.

(v) Supply Shortage OFO - Service Agreements that are subject to a Supply Shortage OFO, requiring pro-rata reductions to Shipper's nominations (based on primary firm corridor rights) to provide the necessary relief through the OFO constraint point in the constrained direction to obtain a net Scheduled Quantity that does not exceed the greater of the physical design capacity or the posted operationally available capacity for a given Gas Day, are those Service Agreements that have primary firm corridor rights scheduled through the constraint point in the

14. OPERATING CONDITIONS (Continued)

constrained direction and the remedies outlined in (i) through (iv) above have not alleviated the adverse operating condition.

When a Supply Shortage OFO is required due to a supply interruption that would result in operational problems, Transporter will fill the OFO constraint point in the Intraday 1 Nomination Cycle to the greater of the physical design capacity or the posted operationally available capacity by allocating Shipper's primary firm nomination requests pro rata based on the Shipper's primary firm corridor rights through the OFO constraint point. If Transporter reduces Shipper nominations pursuant to a Supply Shortage OFO, reservation charge adjustments will not be paid to such Shipper by Transporter.

In no event will Transporter issue a Realignment or Must-flow OFO associated with any contract whose OFO obligation is $100\ \mathrm{Dth}$ or less.

Rate Schedule TF-2 Shippers will not be subject to a Must-Flow OFO obligation once they have fully utilized their Annual Contract Quantity (ACQ) stated on the service agreement.

Transporter will remove OFOs in an order reversed from the order set forth above.

(e) OFO Notice, Contents and Procedures.

Portal Posting Requirements

Transporter will make reasonable efforts to notify Shippers that an operational problem is developing which could potentially necessitate the issuance of an OFO. Transporter will declare an OFO for one or more specific constraint points via an All Shipper Notice, identifying the effective date and duration of the OFO (if none is specified, the OFO will be effective until further notice).

Transporter will post to its designated site, summary level information each Gas Day the OFO is in effect.

Issuance of an OFO

Subsequent to the Portal Posting on the day prior to commencement of the Gas Day, no later than $5:00~\rm p.m.$ Central Clock Time ($4:00~\rm p.m.$ Mountain Clock Time), during the scheduling and confirmation procedure, Transporter will provide notification of the Contract-specific, Realignment and Must-flow OFO obligations via an Internet E-mail to the affected Shipper(s).

14. OPERATING CONDITIONS (Continued)

Each Shipper shall designate an operational contact, and/or a third-party designee, for Transporter to notify on a 24-hour a day, 365 days a year basis. If Transporter is unable to notify any Shipper because that Shipper's contact is unavailable, such Shipper shall be solely responsible for any consequences arising from such failure of communication. Each OFO notification will contain the following provisions:

- (i) time and date of issuance;
- (ii) time that the OFO is considered to be effective;
- (iii) OFO type of operational remedy as defined in Section
 14.15 (d) (ii), (iii), (iv) or (v);
- (iv) the Service Agreement subject to the OFO;
- (v) constrained point location;
- (vi) specific action necessary to remedy the condition;
- (vii) any other terms Transporter may reasonably require to ensure the effectiveness of the OFO; and
- (viii) to the extent that a Shipper has been requested to take action pursuant to Section 14.15(d)(ii), (iii) or (iv) above and such Shipper has temporarily released all or a portion of its capacity subject to the OFO recall provisions of Section 22.2(a)(2), then Transporter will provide, at a contract level, the associated Replacement Shipper's scheduled quantities used to calculate its OFO obligation.

Unless Transporter has invoked a Supply Shortage OFO, or unless otherwise specified in the OFO, compliance must commence at the beginning of the applicable gas day (Evening Nomination Cycle). If Transporter has invoked a Supply Shortage OFO, compliance by Shippers subject to a Contract-Specific or Must-flow OFO must commence no later than the Intraday 1 Nomination Cycle.

Notwithstanding the foregoing, Transporter and Shipper may agree to take any other mutually agreeable action that would assist in alleviating the operating condition which necessitated the issuance of the OFO.

14. OPERATING CONDITIONS (Continued)

Failure to Comply with OFO. If Shipper (or Shipper's (f) designee or successor) fails to comply with the terms of an OFO for reasons other than force majeure or Transporter's negligence/misconduct, such Shipper, its designee or successor shall be liable for a penalty equal to the greater of ten dollars (\$10.00) or four (4) times the highest absolute price at NW Wyoming Pool, NW south of Green River, NW Stanfield, NW Sumas, or El Paso Bondad as reflected in the Daily Price Survey published in "Gas Daily" for each of the days during which a Shipper fails to comply with an OFO for each Dth of gas that does not comply. Transporter may in its reasonable judgment waive penalties due to noncompliance with an issued OFO. No penalties shall be assessed on Shipper for noncompliance with an issued OFO when such OFO is due to Transporter's negligence or misconduct or due to force majeure invoked by Transporter.

Upon request from a Releasing Shipper that has been assessed an OFO penalty, Transporter will provide scheduled quantity data for its associated Replacement Shippers that was used to determine such OFO penalty.

Shipper acknowledges that Transporter does not have access to third-party information that may be necessary to detect or prevent certain actions that could be taken by other Shippers for the purpose of avoiding their OFO obligations. Shipper's acknowledgement shall not be interpreted to limit Shipper's obligations under 18 CFR § 1c.1.

Penalty Revenue Crediting. One hundred percent of all penalty revenues received by Transporter pursuant to Section 14.15(f) during any month will be credited to firm transportation Shippers. Such credits will be allocated to all firm transportation Shippers under Rate Schedule TF-1 (Large Customer), excluding discounted firm transportation service and Shippers receiving service under a capacity release Service Agreement, Rate Schedule TF-1 (Small Customer) for the months they paid a reservation charge due to conversion to Rate Schedule TF-1(Large Customer) service, and Rate Schedule TF-2 (collectively, the specified Shippers), except each penalty revenue credit shall be allocated only to those specified Shippers who were not assessed such penalty on the day(s) for which such penalty revenues are to be credited. The credits will be allocated to the specified Shippers pro rata in proportion to reservation revenue (total revenue, net of credits from capacity

14. OPERATING CONDITIONS (Continued)

releases as described in Section 23 of the General Terms and Conditions, less volumetric charges and all surcharges) received by Transporter from each and all of the specified Shippers for each applicable month. Such revenue credits shall be reflected as a credit billing adjustment to the next month's billings rendered after the applicable month. In the event that such credit billing adjustments would result in a credit total invoice to any Shipper, Transporter will refund the credit billing adjustment to the Shipper in cash within 30 days after determination of the amount of the credit due to the Shipper.

- (h) Shipper Not Liable. Shipper will not be liable for any non-OFO penalties which might otherwise be imposed by Transporter with respect to any action taken by such Shipper in conformance with an OFO issued by Transporter.
- (i) Transporter Not Liable. Transporter will not be liable for any damages, costs or expenses incurred as a result of an OFO unless such damages result from the negligence or willful misconduct of Transporter.
- 14.16 Disaster Recovery Procedures. In the event of a disaster such that Transporter's or Receiving Party's communication capability is not available for confirmation and scheduling of gas receipts and deliveries, nominations will remain in effect as confirmed for the most current day, unless alternate emergency communication methods are established. Transporter may declare an entitlement period on the system until normal communications are restored.
- 14.17 Interconnect Balancing Agreements. NAESB WGQ Flowing Gas Related Standard 2.3.29 will apply.

Transporter may enter into and operate pursuant to interconnect balancing agreements with other pipelines or gas suppliers relating to operating procedures governing nominating, scheduling and balancing for the receipt or delivery of gas at points of interconnection. As part of the provisions of these agreements, Transporter will allow the trading of imbalances between receipt points or between delivery points located in the same geographical area. For receipt points, the same geographical area means receipt points located in the same market hub, or producing basins located within contiguous gathering areas. For delivery points, the same geographical area means the zones used by Northwest for nominations.

14.18 Pre-Determined Allocations ("PDAs"). In the event an OBA is not in effect at a particular interconnecting point on Transporter's system, NAESB WGQ Flowing Gas Related Standards 2.3.1 through 2.3.6, 2.3.8, 2.3.15 through 2.3.23, and 2.3.25 through 2.3.27 will apply.

14. OPERATING CONDITIONS (Continued)

14.19 Pooling.

- (a) NAESB WGQ Standards. NAESB WGQ Nominations Related Standards 1.3.17, 1.3.18 and 1.3.64 through 1.3.77 will apply.
- (b) General.
 - (i) Pools will be available as an optional service for supply aggregation from any requested receipt point to a pool and for delivery from a pool to any requested delivery point on Transporter's system.
 - (ii) Associated Receipt Points for each pool are posted on Transporter's Designated Site and will be updated as needed to reflect the connection of new receipt points or the abandonment of existing receipt points. Proposed changes to pool boundaries will be posted on Transporter's Designated Site and will be submitted to the Commission at least 30 days prior to such changes being implemented. Any changes to pool boundaries will be proposed to become effective on the first day of a calendar month.
 - (iii) Sections 1, 2, 3, 4, 6, 9, 10, 14, 18 and 26 of the General Terms and Conditions of this tariff may apply to a Pooling Party that is not a Shipper. In those instances, the term "Shipper" will be synonymous with "Pooling Party," and the Pooling Party will be subject to the relevant provisions.

- (c) Nominations. All Pooling Parties are subject to the applicable operating conditions outlined in this Section 14.
 - (i) Associated Receipt Point. To aggregate gas from an Associated Receipt Point to a pool, a Pooling Party will nominate from that receipt point to the designated pool. The Pooling Party is not required to be a Shipper nor to have a transportation service agreement. A Pooling Party may have multiple packages of gas from any receipt point within a particular pool, provided the nominations to and from the pool contain unique package identification numbers.
 - (ii) Non-Associated Receipt Point. To aggregate gas from a non-associated receipt point to a pool, a Pooling Party will nominate from a receipt point to the designated pool, using its own transportation service agreement. A Pooling Party may have multiple packages of gas from any receipt point within a particular pool, provided the nominations to and from the pool contain unique package identification numbers.
 - (iii) Delivery Point. To arrange for delivery of gas from a pool, a Shipper will nominate a total volume to be delivered under its transportation service agreement from the pool to the delivery point. In order for Transporter to schedule such delivery in accordance with the Shipper's capacity rights and priorities of service, the Pooling Party or Third Party Account Administrator must confirm the total volume to be delivered from the pool by receipt point (including package identification numbers, where applicable).
- (d) Storage Facility Nominations. Pooling nominations at receipt or delivery points for storage facilities are subject to the same nomination and confirmation deadlines as nominations for all other receipt and delivery points.

- 14. OPERATING CONDITIONS (Continued)
 - (e) Title Transfer Tracking.
 - (i) An Account Holder using Transporter as the Title Transfer Tracking Service Provider may transfer title by:
 - (1) nominating deliveries into a pool pursuant to Sections 14.19(c)(i) or 14.19(c)(ii) for the account of a different Pooling Party in that pool; and/or
 - (2) disaggregating from its account in a pool to one or more Shippers nominating deliveries from that pool pursuant to Section 14.19(c)(iii).
 - (ii) If a Third Party Account Administrator provides Title Transfer Tracking service, nominations made by the Third Party Account Administrator will be subject to a contract with Transporter and will be subject to the applicable operating conditions outlined in this Section 14.

To use a Third Party Account Administrator as the Title Transfer Tracking Service Provider, an Account Holder must nominate deliveries into a pool pursuant to Sections 14.19(c)(i) or 14.19(c)(ii) for the account of the Third Party Account Administrator. After providing Title Transfer Tracking service for an Account Holder, the Third Party Account Administrator then must disaggregate the pooled gas to one or more Shippers nominating deliveries from the pool pursuant to Section 14.19(c)(iii).

(f) Balancing. No imbalances in pools are allowed. If volumes nominated by receipt point for aggregation in a designated pool each day do not match volumes nominated and confirmed by receipt point for delivery that day from the designated pool, Transporter will schedule the lower volume.

14. OPERATING CONDITIONS (Continued)

- (g) Capacity Rights/Priority of Service. Pools have no receipt or delivery point rights for purposes of scheduling.
 - (i) Associated Receipt Point. Capacity rights for receipt by Transporter into a pool from an Associated Receipt Point will be determined by the receipt point MDQs, Contract Demand, and applicable corridor rights pertaining to the transportation service agreement used to deliver gas from the pool to the ultimate delivery point.
 - (ii) Non-Associated Receipt Point. Capacity rights for receipt by Transporter into a pool from a non-associated receipt point will be determined by the receipt point MDQs, Contract Demand, and applicable corridor rights pertaining to the Pooling Party's transportation service agreement used to nominate gas into the pool.
 - (iii) Delivery Point. Capacity rights from a pool to the delivery point will be determined by the delivery point MDDOs, Contract Demand and applicable corridor rights pertaining to the transportation service agreement used to deliver gas from the pool to the ultimate delivery point.
 - (iv) Priorities of service for gas that is pooled are the same as priorities of service for gas that is not pooled.

(h) Charges.

- (i) Associated Receipt Point. Nominations to a pool from an Associated Receipt Point will not incur transportation or fuel charges.
- (ii) Non-Associated Receipt Point. Nominations to a pool from a non-associated receipt point will incur transportation and fuel charges.
- (iii) Delivery Point. Nominations from a pool to any delivery point will incur transportation and fuel charges.

14. OPERATING CONDITIONS (Continued)

- (i) Warranty and Liability.
 - (i) Pooling Party warrants that it has title to or the right to transfer title to gas under the pooling provisions of this tariff. Pooling Party will indemnify Transporter and save it harmless from all suits, actions, damages, costs, losses, and expenses (including reasonable attorneys' fees) arising from breach of this warranty.
 - (ii) Transporter will have no liability to Pooling Party for any and all damages and costs arising out of an intended or completed pooling transaction except to the extent such liability is shown to be the result of the negligence or willful misconduct of Transporter.
 - (iii) Pooling Party agrees to defend, indemnify, and hold harmless Transporter, its officers, agents, employees and contractors ("Indemnitee") against any liability, suit, action, claim, damage, or expenses whatsoever, including costs and attorneys' fees ("Liabilities and Expenses") occurring in connection with or relating in any way to Pooling Party's use or reliance on Transporter's pooling services, whether such Liabilities and Expenses are suffered by Indemnitee as a direct or indirect result of Pooling

Party's use of or reliance on Transporter's pooling services or other omissions or tortious acts by Pooling Party, except to the extent such Liabilities and Expenses are shown to be the result of the negligence or willful misconduct of Transporter.

(iv) Neither Transporter nor Pooling Party shall be liable to the other or to any other party for any consequential, punitive, special, or other damages arising in any way from any error, omission, or inaccuracy of Transporter's pooling services.

14. OPERATING CONDITIONS (Continued)

- 14.20 Liquefaction Charge True-Up Adjustment. A Plymouth LNG Facility Liquefaction Charge true-up adjustment for Rate Schedule LS-3F and LD-4I Shippers will be determined for the prior twelve months ending December 31 and will be billed/refunded on the subsequent March invoice.
 - (a) The true-up adjustment applicable to each Shipper's Service Agreement will be the difference between the actual Liquefaction Charge rate and the estimated Liquefaction Charge rate that was billed to the Shipper, multiplied by the Dths assessed the estimated Liquefaction Charge rate on Shipper's invoices during the prior calendar year.
 - (b) The actual Liquefaction Charge rate will be the liquefaction costs for the prior calendar year divided by the total Dths assessed the estimated Liquefaction Charge rate for LS-3F and LD-4I services during the prior calendar year. The liquefaction costs will be the total Plymouth LNG Facility power costs plus other variable liquefaction material costs incurred (e.g., chemicals) less those same costs associated with: (i) the vaporization service, and (ii) liquefaction services (i.e., Boil-off, liquefaction and general purpose) for Plymouth LNG Facility Rate Schedules LS-2F and LS-2I.

For purposes of this calculation, the power costs associated with liquefaction nominations during the Liquefaction Period for Rate Schedules LS-2F and LS-2I will be deemed to have been incurred in the months and order of April, October, May, September, June, July and August until the prior calendar year total liquefaction quantity during the Liquefaction Period for these Rate Schedules is met. Power costs associated with liquefaction nominations during the Vaporization Period for Rate Schedules LS-2F and LS-2I will be deemed to have been incurred in the month(s) the liquefaction nominations are made.

14. OPERATING CONDITIONS (Continued)

(c) The estimated Liquefaction Charge rate will be based on the prior calendar year's actual Liquefaction Charge rate, adjusted for any material changes in forecast costs, and will be updated and filed concurrently with Northwest's annual filing of the fuel reimbursement factor for the LNG Rate Schedules pursuant to Section 14.12 of the Tariff, to become effective on April 1 of each year. Notwithstanding the foregoing, Northwest may file a revised estimated Liquefaction Charge before the annual filing if significant cost changes arise in response to unforeseen changes in Rate Schedule LS-3F and/or LD-4I activity. In the latter case, the revised estimated Liquefaction Charge for Rate Schedules LS-3F and LD-4I will become effective the first of the month after Commission approval.

Third Revised Sheet No. 233 Superseding Second Revised Sheet No. 233

GENERAL TERMS AND CONDITIONS (Continued)

- 15. DETERMINATION OF DELIVERIES AND IMBALANCES
 - 15.1 RESERVED FOR FUTURE USE.
 - 15.2 Imbalances and Penalties at Delivery Points. Penalties and overrun and underrun charges as described in Sections 15.3 and 15.5 will be levied by Transporter against the Receiving Party who controls the facilities into which gas is delivered unless the Receiving Party has executed an OBA with Transporter and that OBA also covers receipts by Transporter from the Receiving Party. If the Receiving Party has not executed an OBA and incurs either an overage or underage in takes from Transporter, due to a failure to take quantities of gas equivalent to the total of all Nominating Party nominations for deliveries into its facilities, such penalties and overrun or underrun charges will be applied as described in this Section. Such Receiving Party will be responsible for all Receiving Party Imbalances and associated penalties and charges. Unless otherwise agreed upon by the parties, Receiving Party imbalances will reside on one of the Receiving Party's executed Service Agreements.

15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)

Transporter will waive any penalties due to Receiving Party/Shipper Imbalances which result from errors made by Transporter or which result from force majeure invoked by Transporter. Transporter also will waive Receiving Party/Shipper Imbalance penalties for other good cause, including Transporter's reasonable judgment that Receiving Party's/Shipper's actions have not impaired Transporter's ability to provide reliable firm service. Such waivers will be applied uniformly in a not unduly discriminatory manner.

15.3 Receiving Party Imbalances and Penalties. If Receiving Party's cumulative monthly Receiving Party Imbalance exceeds the greater of 5,000 Dth or (a) 5 percent during the period of March through July or (b) 3 percent during the period of August through February above or below total confirmed nominations for that month ("Tolerance Limit"), Transporter shall notify Receiving Party that Receiving Party's Imbalance exceeds the allowed tolerance. Such notice shall be provided by the fifteenth day of the month following the month service is rendered. Transporter will notify Receiving Party when their Receiving Party Imbalance exceeds the Tolerance Limit, and specify whether a penalty situation exists. Receiving Party will have 45 nonentitlement, Balancing-Approved Days from the date of notification by Transporter to bring its Receiving Party Imbalance within tolerance. The 45 non-entitlement, Balancing-Approved Day period will be extended proportionately if, and to the extent that Shipper's failure to bring its imbalance within tolerances is caused by Transporter's inability to accommodate Shipper's make-up or payback nomination(s).

If the Receiving Party fails to get within tolerance at any point during the 45 non-entitlement, Balancing-Approved Day period, Receiving Party shall pay a penalty to Transporter equal to \$10.00/Dth on the imbalance over the Tolerance Limit, as described above. Receiving Party Imbalances shall be cumulative and Receiving Party must specifically adjust nominations as necessary to eliminate such imbalances.

Under a customer-specific Declared Entitlement Period, a Receiving Party will no longer be subject to the customer-specific entitlement once the system's integrity, which was jeopardized by the Receiving Party's actions, has been restored or Transporter has issued a general system Declared Entitlement Period that involves such Receiving Party.

15.4 Receiving Party Imbalance Netting. Upon Shipper's request, Transporter will net Receiving Party Imbalances with Receiving Party Imbalances, and Receiving Party Imbalances with Shipper Imbalances (as defined in Section 15.7 hereof), between a Shipper's Service Agreements within an Operational Impact Area. Netting between a Shipper's Service Agreements, when offsetting a Receiving Party Imbalance with a Shipper Imbalance, will be limited to a maximum quantity equal to the smallest imbalance under the agreements being netted.

15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)

If a Receiving Party fails to eliminate Receiving Party Imbalances within the allowed period referenced above, in addition to any penalties applied hereunder, Transporter shall have the right to adjust nominations, receipts or deliveries in order to achieve a balance. Receiving Party imbalances may also be eliminated by trading imbalances between Service Agreements with third parties as set forth in Section 15.9.

- 15.5 Daily Entitlement Overrun and Underrun Provisions.
 - (a) Unauthorized Overrun and Underrun Volumes. If on any day during a Declared Entitlement Period, the total physical quantity of gas taken by Receiving Party from Transporter exceeds or is less than the total quantity of gas which Receiving Party is entitled to take on such day as defined below, then all gas taken in excess of such entitlement or not taken within said entitlement shall constitute unauthorized overrun or underrun volume. Each general system or customer-specific Declared Entitlement Period shall be specified as either an overrun or an underrun entitlement for Receiving Party such that only one penalty condition may exist at one time.
 - (b) Receiving Party's Daily Entitlement. The total quantity of gas which Receiving Party is entitled to take from Transporter on any day consists of the sum of the following quantities:
 - (i) The volume of gas in Dth authorized by a confirmed nomination for delivery to Receiving Party on such day under firm transportation Service Agreements.
 - (ii) The volume of gas in Dth authorized by a confirmed nomination for delivery to Receiving Party on such day under interruptible transportation Service Agreements.
 - (c) Daily Unauthorized Overrun and Underrun Charges. The Daily Unauthorized Overrun or Underrun Charges shall be applied, as explained below, during a Declared Entitlement Period, either as a result of capacity constraints, supply interruption, or an overtake or an undertake situation applicable to any Receiving Party, when Transporter has so notified such Receiving Party prior to the start of the Gas Day. These charges are not intended to apply to occasional or minor supply interruptions which would not have a detrimental impact on Transporter's operations or on other Receiving Parties. Under such circumstances, Transporter would not notify Receiving Party that a Declared Entitlement Period is in effect.

15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)

In the event an undertake situation exists, Transporter will evaluate operational conditions and determine if immediate relief is needed to return the system to a reasonable operational level. If immediate relief is needed, Transporter will notify each Receiving Party of the quantity required to be taken off the system within the next 72 hours as imbalance make-up to eliminate the underrun imbalances so as to return the system to a reasonable operational level and avoid the imbalance penalty as specified in Section 15.5(c)(ii). Such quantities shall be based upon cumulative underrun imbalances which contributed to the underrun imbalance problem as determined by Transporter's operational data up to the date of the Declared Entitlement Period and calculated for each Receiving Party in proportion to such party's undertakes. Daily unauthorized underrun charges as specified in Section 15.5(c)(i) will also apply if Receiving Party causes additional unauthorized underrun imbalances during a Declared Entitlement Period.

Receiving Party shall pay Transporter for all unauthorized overrun or underrun quantities, as defined in Section 15.5(a) above, that exceed the percentage specified by Transporter in its Declared Entitlement. For a general system or customer-specific Declared Entitlement Period, such percentage will be: (i) in Transporter's sole discretion 3 percent, or, in the case of a Declared Entitlement Period announced on the day it is to be in effect, 5 percent for that day (Stage I), 8 percent (Stage II) or 13 percent (Stage III) of Receiving Party's entitlement as set forth in Section 15.5(b) with respect to unauthorized overrun quantities, and (ii) 5 percent (Stage I), 8 percent(Stage II) or 13 percent (Stage III) of Receiving Party's entitlement as set forth in Section 15.5(b) with respect to unauthorized underrun quantities (individually, the "Threshold Percentage").

The following amounts will be payable in addition to the amounts otherwise payable by Receiving Party to Transporter under Transporter's applicable Rate Schedules, subject only to the provisions set forth in (d) below:

(i) During a general system or customer-specific Declared Entitlement Period, the daily unauthorized overrun or underrun charge will be the product of the daily unauthorized overrun/underrun quantity (i.e., the quantity which exceeds the Threshold Percentage of Receiving Party's entitlement for such day), multiplied by a rate per Dth as set forth in the Statement of Rates of this Tariff.

- 15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)
 - (ii) In addition to (i) above, for that part of the unauthorized underrun imbalances where immediate relief is requested but not provided within a 72 hour period as set forth in Section 15.5(c), the additional 72 hour penalty will be the product of the underrun imbalance quantity not eliminated multiplied by a rate per Dth as set forth in the Statement of Rates of this tariff.
 - (iii) If an unauthorized overrun or underrun quantity causes Transporter to pay demand or reservation charge credits to any Shipper under the provisions of this Tariff for any day, the Shipper(s) which caused such demand or reservation charge credits to be incurred shall on a pro rata basis reimburse Transporter for such demand or reservation charge credit payments through a special charge included on Transporter's invoices to each affected Shipper.
 - (iv) If a Receiving Party is initially below the Threshold Percentage based on estimated measured volumes but exceeds the Threshold Percentage once actual measured volumes are available, the Receiving Party will be provided with the opportunity to avoid a penalty situation for that Gas Day if the Receiving Party submits a valid request, within ten (10) days of receiving notification from Transporter that the Threshold Percentage has been exceeded, to increase/remove the Scheduled Quantities that exceed the Threshold Percentage and such request is confirmed by the upstream/downstream operators and Transporter. Transporter will only confirm the change in Scheduled Quantities if Transporter could have authorized the change, based on operationally available capacity, for the Gas Day in which the Threshold Percentage was exceeded.
 - (d) Exception. The Daily Unauthorized Overrun and Underrun Charges provided in Section 15.5(c) shall not apply on any day when either (i), (ii), or (iii) below is applicable:
 - (i) When the unauthorized overrun volume on the particular day during either a general system or customer-specific Declared Entitlement Period does not exceed the Threshold Percentage as set forth in Section 15.5(c) of Receiving Party's entitlement for such day as set forth in Section 15.5(b), or 2,000 Dth, whichever is larger.
 - (ii) When the unauthorized underrun volume on the particular day during either a general system or customer-specific Declared Entitlement Period does not exceed the Threshold Percentage as set forth in Section 15.5(c) of Receiving Party's entitlement for such day as set forth in Section 15.5(b), or 2,000 Dth, whichever is larger.

- 15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)
 - (iii) If Transporter has not made notification to Receiving Party of a Declared Entitlement Period in accordance with Section 14.6.
 - (e) Waiver. Transporter will waive all or a portion of the Daily Unauthorized Overrun and Underrun Charges provided in Section 15.5(c) when Transporter fails to make Scheduled Quantities available to point operators and Shippers, including bumped parties, at the times specified in Section 14.1(a) for any of the Nomination Cycles, provided that:
 - (i) Transporter was solely at fault for the failure to make the Scheduled Quantities available; and
 - (ii) the affected party demonstrates that the failure to provide Scheduled Quantities on a timely basis materially impacted such party's ability to stay within Declared Entitlement levels.

The waiver will not apply (1) to gas that flowed more than one hour after the start of the Gas Day if Scheduled Quantities are made available prior to the start of the Gas Day or (2) to gas that flowed more than one hour after Scheduled Quantities were made available for nominations submitted during the Intraday 1, Intraday 2 or Intraday 3 Nomination Cycles.

15.6 Penalty Crediting. Except for the penalty payments pursuant to Section 15.5(c)(iii), one hundred percent (100%) of all penalties received by Transporter during any applicable month will be credited to firm and interruptible transportation Shippers. Such credits will be allocated to all Shippers under Rate Schedule TF-1 (excluding discounted firm transportation service and Shippers receiving service under a capacity release Service Agreement), Rate Schedule TF-2 and Rate Schedule TI-1 paying the Maximum Base Tariff Rate, (collectively, the Specified Shippers), that were not assessed an unauthorized overrun or underrun penalty on the day for which such penalties are to be credited.

15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)

The penalty credits will be allocated to the Specified Shippers pro rata in proportion to total revenue excluding surcharges, net of credits from capacity releases as described in Section 23 of the General Terms and Conditions, received by Transporter from each and all of the Specified Shippers for each applicable month. Such penalty credits shall be reflected as a credit billing adjustment to each March billing for credits received during the prior calendar year. In the event that such credit billing adjustments would result in a credit total invoice to any Shipper, Transporter will refund the credit billing adjustment to the Shipper by check within 15 days after determination of the amount of the credit due to the Shipper.

- If a Shipper's cumulative monthly Shipper Imbalance is more than 5,000 Dth or 5 percent, above or below total confirmed nominations for the month, whichever is greater, the Shipper will have exceeded allowed tolerances. If Shipper exceeds allowed tolerances, Transporter will provide an imbalance statement to Shipper on or before Transporter's next invoice date and Shipper will then have 15 non-entitlement Balancing-Approved Days from the date of Transporter's invoice to eliminate such excess imbalance; otherwise, Shipper will be subject to a Shipper Imbalance penalty as described in Section 15.10. The fifteen-day period will be extended proportionately if, and to the extent that, Shipper's failure to eliminate its excess imbalance is caused by Transporter's inability, due to operating or force majeure conditions to accommodate Shipper's make-up nominations, provided such nominations are from Shippers primary receipt point(s). A Shipper may eliminate Shipper Imbalances by scheduling imbalance gas in accordance with the procedures outlined in Section 14 hereof.
- 15.8 Shipper Imbalance Netting. Upon Shipper's request, Transporter will net Shipper Imbalances with Shipper Imbalances, and Shipper Imbalances with Receiving Party Imbalances, between its Service Agreements. Netting between a Shipper's Service Agreements, when offsetting a Receiving Party Imbalance with a Shipper Imbalance, will be limited to a maximum quantity equal to the smallest imbalance under the agreements being netted. Shipper Imbalances may also be eliminated by trading imbalances with third parties as set forth in Section 15.9.

- 15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)
 - 15.9 Shipper/Receiving Party Imbalance Trading. A Shipper or Receiving Party may eliminate imbalances by trading imbalances between Service Agreements with third parties within the same Operational Impact Area. Trading between Service Agreements of third parties, when a Receiving Party Imbalance is traded with a Shipper Imbalance or another Receiving Party Imbalance, will be limited to a maximum quantity equal to the smallest imbalance under the agreements being traded.
 - (a) All imbalance trading shall be conducted through Transporter's Designated Site. Shippers or Receiving Parties may post an open or prearranged trade. Trading can be done at any time and will occur once the confirming trader accepts the posting by the initiating trader. If Shipper or Receiving Party desires to trade imbalances, Shipper or Receiving Party must consent to Transporter's release of necessary information regarding its imbalance.
 - (b) Shippers or Receiving Parties are responsible for making whatever arrangements among themselves that they deem necessary prior to finalizing the imbalance trade on Transporter's Designated Site.

- 15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)
 - (c) Transporter will not be liable for any losses incurred by a Shipper if a Shipper is unable to complete an imbalance trade once Shipper has notified Transporter of his desire to consummate an imbalance trade.
 - (d) Any imbalance gas being traded will still be deemed to be imbalance gas and will be subject to: 1) scheduling provisions in Section 14; 2) the priority of service provisions in Section 12; and 3) the penalty provisions in Section 15.
 - 15.10 Assessment of Shipper Imbalance Penalties. Shipper shall have 15 non-entitlement, Balancing-Approved Days from the date of Transporter's invoice to adjust excess Shipper Imbalances as described in Section 15.7 to allowed tolerance levels or be subject to the following:
 - (a) For excess Shipper Imbalances resulting from confirmed nominations for deliveries by Transporter exceeding confirmed Shipper nominations for receipts from an interconnecting pipeline, Shipper will pay a penalty of \$5.00 per Dth, such penalty to be payable within 10 days after receipt of any assessment from Transporter.
 - (b) For excess Shipper Imbalances resulting from confirmed nominations for receipt from Shipper exceeding confirmed nominations for deliveries by Transporter to an interconnecting pipeline, any such excess volumes shall become the property of Transporter at no cost to Transporter and shall be free and clear of any adverse claims related thereto.

Transporter shall exempt Shippers from Shipper Imbalance penalties resulting from OFO compliance and provide an additional 30 days to correct OFO-related Shipper Imbalances.

15.11 Assessment of Charges on Imbalances at Termination of Service Agreement. Transporter and Shipper shall eliminate any cumulative imbalances between receipts and deliveries of gas existing at the termination of a Service Agreement within 15 non-entitlement, Balancing-Approved Days after the date of Transporter's invoice. The fifteen-day period will be extended proportionately if, and to the extent that, Shipper's failure to eliminate its remaining cumulative imbalance is caused by Transporter's inability, due to operating or force majeure conditions to accommodate Shipper's make-up nominations provided such nominations are from Shipper's primary receipt point(s).

15. DETERMINATION OF DELIVERIES AND IMBALANCES (Continued)

At the end of the 15 day period, any gas which was delivered by Transporter for which equivalent volumes were not received by Transporter from Shipper shall be paid for by Shipper at \$5.00 per Dth. Shipper shall pay such amount within 10 days after receipt of such bill. Any gas received by Transporter in excess of that delivered for Shipper which remains in Transporter's system at the end of the 15 day period shall become the property of Transporter at no cost to Transporter and free and clear of any adverse claims related thereto.

- 15.12 Imbalances with Other Parties. Transporter shall not be responsible for eliminating any imbalances in volumes transported that accrue between Receiving Party or Shipper and any other party, including without limitation any volume imbalances that accrue between local distribution companies and specific end-users or between producers or pipelines and local distribution companies or end-users. Furthermore, Transporter shall not be obligated to adjust or deviate from its standard operating and accounting procedures in order to alleviate such imbalances.
- 15.13 Netting and Trading within Operational Impact Area. NAESB WGQ Flowing Gas Related Standard 2.3.30 will apply.
- 15.14 LNG Boil-off Balance in Excess of Allowed Tolerances. If Shipper's Boil-off balance is more than 10,000 Dths or 10 percent of Shipper's Storage Capacity, whichever is greater, Transporter will notify Shipper that the Boil-off balance exceeds allowed tolerances. Shipper will then have 45 non-entitlement, Balancing-Approved Days to eliminate such excess balance. The 45 non-entitlement, Balancing-Approved Day period will be extended proportionately if, and to the extent that Shipper's failure to bring its balance within tolerances is caused by Transporter's inability, due to operating or force majeure conditions to accommodate Shipper's re-liquefaction/transportation request. Any excess quantities not brought within tolerance within the required timeframe shall become the property of Transporter at no cost to Transporter and will be free and clear of any adverse claims related thereto.

In the event a Service Agreement containing a Boil-off balance terminates, Shipper will have 45 non-entitlement, Balancing-Approved Days from the date of termination to eliminate the entire Boil-off balance. For any quantities owed to Shipper after this period, Transporter will take title to such quantities free and clear of any adverse claims. For any quantities owed to Transporter after this period, Transporter shall invoice Shipper for each Dth owed the greater of five dollars or two times the highest absolute price at NW Wyoming Pool, NW south of Green River, NW Stanfield, NW Sumas, or El Paso Bondad as reflected in the Daily Price Survey published in "Gas Daily" for the first Gas Day after Shipper's Service Agreement termination date. Shipper shall pay such amount within 10 days of receipt of such invoice.

16. FERC ANNUAL CHARGE ADJUSTMENT

- 16.1 Purpose. For the purpose of funding Transporter's share of the Commission's gas program costs, this section establishes an ACA provision to be applicable to service under Transporter's Rate Schedules as referenced on the Statement of Rates contained in this tariff. The Commission-authorized ACA rate, which is revised annually, is specified in an annual notice issued by the Commission entitled "FY [Year] Gas Annual Charges Correction for Annual Charges Unit Charge." This document can be found on the Annual Charges page of the Natural Gas section of the Commission's website, located at http://www.ferc.gov.
- 16.2 Basis of the Annual Charge Adjustment. The ACA charge shall be adjusted to Transporter's pressure base and heating value, if required, which is listed on the Commission's Annual Charges Billing. The ACA charge will be assessed on all transportation volumes including those moving into, but not out of, the Jackson Prairie Storage Project or Plymouth LNG Facility. The ACA charge will not be assessed on volumes transported on a Designated Lateral by a Rate Schedule TFL-1 or TIL-1 Service Agreement if those same transportation volumes were delivered on to or off of the Designated Lateral by a Rate Schedule TF-1, TF-2 or TI-1 Service Agreement.
- 16.3 Remittance to the Commission. Transporter shall remit to the Commission, on/or before the date which is specified on the Annual Charges Billing, the Total Annual Charge stated on such billing.
- 16.4 Accounting Procedures. Transporter will record annual charges in Commission Account No. 928. Pursuant to Section 154.402(b) of the Commission's regulations, Transporter will not attempt to recover any annual charges previously recorded in Account No. 928 in a Natural Gas Act Section 4 rate case.

First Revised Sheet No. 239 Superseding Substitute Original Sheet No. 239

GENERAL TERMS AND CONDITIONS (Continued)

17. STANDARDS OF CONDUCT

Transporter will comply with 18 CFR Part 358 of the Commission's regulations including posting in the informational posting section on Transporter's Designated Site the required information set forth in 18 CFR Part 358.

First Revised Sheet Nos. 240 Through 242 Superseding Substitute Original Sheet Nos. 240 Through 242

18. COMPLAINT PROCEDURES

- 18.1 Shippers are first encouraged to work with Transporter to resolve problems on an informal basis, prior to filing a formal complaint.
- 18.2 In the event of an unresolved problem, Shipper should submit a complaint in writing to Transporter at the following address:

Attention: Director, Commercial Services Northwest Pipeline LLC 2800 Post Oak Boulevard Houston, TX 77056 P.O. Box 1396 Houston, TX 77251

- 18.3 Transporter will respond initially within 48 hours and in writing within 30 days to such complaints.
- 18.4 If Shipper has exhausted all options in Sections 18.1 through 18.3, further discussion may be pursued with the Vice President, Customer Services and Rates.
- 18.5 A Shipper may also file a complaint with the Commission pursuant to 18 CFR Section 385.206 of the Commission's regulations.

19. NEGOTIATED RATES

19.1 Availability. Shipper and Transporter may agree to Negotiated Rates for a specific term of service under any open-access rate schedule contained in this tariff, provided that Shipper has not acquired its capacity through a temporary capacity release transaction pursuant to Section 22 of the General Terms and Conditions of this tariff. The rates shown on Transporter's Statement of Rates are available as Recourse Rates for any Shipper that does not negotiate a rate with Transporter.

19.2 Applicability.

(a) Existing Service. Notwithstanding anything to the contrary contained in this tariff, Transporter and Shipper may mutually agree to Negotiated Rates and contract term for all or any portion of the capacity under any existing open-access Service Agreement, provided that Shipper has not acquired its capacity through a temporary capacity release. If only a portion of the capacity under any existing open-access Service Agreement will be priced at Negotiated Rates, the original Service Agreement must first be bifurcated, and Recourse Rates will continue to apply to the Service Agreement not subject to the Negotiated Rates.

(b) New Service.

(1) Requests for Service. Transporter and Shipper may mutually agree to Negotiated Rates and contract term for any available capacity that is requested by Shipper, provided that such capacity has been posted for informational purposes, but is not posted for bid at the time of Shipper's request. If Shipper and Transporter mutually agree on a Negotiated Rate for the capacity, Transporter will post the terms of any such pre-arranged agreement in accordance with Section 25.5 of the General Terms and Conditions of Transporter's tariff, and other parties will have an opportunity to bid on the capacity. The pre-arranged Shipper will have a one-time right to match any higher bid in order to retain the capacity.

19. NEGOTIATED RATES (Continued)

- (2) Bids on Posted Capacity. If available capacity, including capacity under expiring or terminating Service Agreements, is posted for bid pursuant to Section 25.5 of the General Terms and Conditions and Transporter has determined that it is willing to consider bids at Negotiated Rates, Transporter must specify in its posting that it will consider Negotiated Rate bids in addition to Recourse Rate and discounted Recourse Rate bids.
- (c) Option Fee. Where negotiated rates include an option fee, the applicable rate will be calculated as follows for the purpose of scheduling pursuant to Section 12: (i) any stated daily or calculated daily formulaic rate, plus (ii) the quotient of the total value of the option fee divided by the volume applicable to the option fee. For the purposes of determining the best bid pursuant to Section 25, the option fee will be included in the same manner in the Net Present Value calculation.
- 19.3 Evaluation of Negotiated Rate Bids. If Transporter determines that it is willing to consider Negotiated Rate bids for available capacity that is posted pursuant to Section 25 of the General Terms and Conditions or deferred exchange service that is posted pursuant to Section 2.2 of Rate Schedule DEX-1, Transporter will state in its posting the specific basis on which Negotiated Rate bids will be considered and one of the following evaluation methods for determining the winning bid:
 - (a) Net Present Value (reservation rate x quantity x term, discounted using the discount rate(s) shown in the posting);
 - (b) Revenue (reservation rate x quantity); or
 - (c) Reservation Rate/Deferred Exchange Rate.

If Transporter specifies in its posting that it will accept Negotiated Rate bids tied to an index or indices, Transporter will post an assumed value for such index or indices, where such assumed value is based on both historical and projected index prices. This assumed value will be used to compare Negotiated Rate bids with Recourse Rate and discounted Recourse Rate bids; provided, however, that if a Shipper with a Negotiated Rate bid tied to an index or indices has submitted the bid with the highest value and is awarded the capacity, such Shipper will pay according to the actual value of the index or indices at the time service is rendered.

Negotiated Rate bids in excess of Transporter's Maximum Base Tariff Rate for the applicable rate schedule will be treated as Maximum Base Tariff Rate offers for bid evaluation purposes.

19. NEGOTIATED RATES (Continued)

- 19.4 Filing Requirement. Transporter will submit to the Commission a tariff sheet stating the name of the Shipper, the Negotiated Rate, the rate schedule, the receipt and delivery points, and the contract quantities applicable to any Negotiated Rate Service Agreement. Unless Transporter executes and files a non-conforming service agreement, such tariff sheet will contain a statement that the Negotiated Rate service agreement does not deviate in any material aspect from the form of service agreement in the tariff for the applicable rate schedule.
- 19.5 Rate Treatment. Transporter will not seek in future general rate proceedings discount-type adjustments to demand charge billing determinants for capacity converted from Recourse Rate service agreements to Negotiated Rate service agreements, unless the Recourse Rate had been discounted. In those situations where Transporter had granted a market-justified discount to the Recourse Rate and subsequently converted the service agreement to a Negotiated Rate service agreement, Transporter may seek a discount-type adjustment. Such adjustment would be based on the greater of: (i) the Negotiated Rate revenues received or (ii) the discounted Recourse Rate revenues which otherwise would have been received.
- 19.6 Limitations. This section 19 does not authorize Transporter to negotiate terms and conditions of service.

First Revised Sheet Nos. 247 Through 249 Superseding Substitute Original Sheet Nos. 247 Through 249

Third Revised Sheet No. 250 Superseding Second Revised Sheet No. 250

Second Revised Sheet No. 251 Superseding First Revised Sheet No. 251

21. INTERCONNECTS

- 21.1 General. This Section 21 applies to all facilities for receiving Gas into or delivering Gas out of Transporter's Mainline Transportation System ("Interconnect"). Interconnect facilities generally include a Tap, Meter Station and Other Interconnect Facilities, as such terms are defined in Section 21.4 below. This Section 21 does not apply to facilities required to accommodate the Transportation of new Contract Demand on Transporter's Mainline Transportation System.
- 21.2 Request for Interconnect. A party desiring to establish an Interconnect ("Interconnecting Party"), will deliver to Transporter a written request that provides sufficient and reliable information for Transporter to evaluate the Interconnect and assess the associated operational impacts that the Interconnect will have on Transporter's Mainline Transportation System. Information required with such request must include the proposed location of the Interconnect, the anticipated minimum and maximum capacity needs, the anticipated pressure requirements at desired delivery point(s), the anticipated Gas composition and quality for receipt points and such other information as Transporter may deem relevant. Transporter will respond within 60 days after the foregoing information has been received with either an answer to the request or an estimate of the additional time Transporter will need to provide an answer to the request.
- 21.3 Conditions for Access. Transporter will allow access for the Interconnect once all of the following conditions have been satisfied:
 - (a) The Interconnect will not adversely affect Mainline Transportation System operations;
 - (b) The Interconnect and resulting Transportation service will not result in diminished services to Transporter's existing customers;
 - (c) The Interconnect will not cause Transporter to be in violation of any environmental or safety laws or regulations;
 - (d) The Interconnect will not cause Transporter to be in violation of its right-of-way agreements or any other contractual obligations;
 - (e) Transporter and/or Interconnecting Party will have acquired any necessary regulatory approvals, permits and real property rights to construct, own, operate and maintain the Interconnect facilities;
 - (f) If applicable, the party responsible for confirming Nominations at the receipt point or interconnecting pipeline will have agreed to execute an OBA for such Interconnect prior to Transporter placing the Interconnect in-service; and

21. INTERCONNECTS (Continued)

- (g) The Transporter and the Interconnecting Party will have executed a facilities agreement and, if applicable, an Exhibit C to the Interconnecting Party's valid Service Agreement with respect to the Interconnect facilities that Transporter will design and construct.
- 21.4 Design, Construction, Ownership, Operation and Maintenance of Interconnect. Upon satisfaction of the conditions in Section 21.3:
 - (a) Transporter will design, construct, own, operate and maintain the Tap for the Interconnect. The "Tap" will mean the tap(s), valve(s) and appurtenant facilities that physically connect with Transporter's Mainline Transportation System.
 - (b) Transporter will design, construct, own, operate and maintain the Meter Station for the Interconnect, subject to the following exceptions:
 - (1) If Transporter determines in its sole discretion that an Interconnecting Party's ownership of all or part of the Meter Station will not be detrimental to its Mainline Transportation System, then Transporter may allow the Interconnecting Party to acquire an ownership interest in the Meter Station in proportion to construction costs for which the Interconnecting Party is responsible. However, Transporter will not transfer any ownership interest to the Interconnecting Party prior to payment in full of such construction costs; or
 - (2) If the Interconnecting Party is another pipeline performing interstate Gas transportation subject to the Commission's jurisdiction, then which pipeline will design, construct, own, operate and/or maintain the Meter Station will be mutually agreed upon by Transporter and the Interconnecting Party.

The "Meter Station" will mean Measurement Equipment, valves and appurtenant facilities (including the incidental piping connecting the Tap to the Meter Station) that will perform custody transfer measurement and ensure that Gas receipts or deliveries will not adversely affect the reliability or the operational integrity of Transporter's Mainline Transportation System.

21. INTERCONNECTS (Continued)

- (c) Transporter will not be required to design, construct, own, operate or maintain any other facilities necessary to effectuate the Interconnect besides the Tap and Meter Station ("Other Interconnect Facilities"), but may agree to do so in a facilities agreement.
- (d) For any Interconnect facilities that Transporter will design and construct:
 - (1) The terms governing the design, construction, ownership, operation and maintenance of and payment for such facilities will be set forth in a facilities agreement between Transporter and the Interconnecting Party; and
 - (2) All real property rights required for the construction, ownership, operation and maintenance of the Interconnect facilities, including rights of ingress and egress, will be obtained in Transporter's name, unless Transporter otherwise agrees.
- (e) For any Interconnect facilities that the Interconnecting Party will design and construct:
 - (1) Such facilities must comply with Sections 21.3 and 21.4(f);
 - (2) The Interconnecting Party assumes full responsibility and liability for the design and construction of such facilities and will hold Transporter harmless from any and all liability arising from the design and construction of such facilities by the Interconnecting Party or third party on the Interconnecting Party's behalf; and
 - (3) If such facilities will be constructed in close proximity to Transporter's facilities such that Transporter reasonably determines its facilities and/or personnel may be at risk, then: the Interconnecting Party will coordinate the construction of such proximate facilities with Transporter so that such proximate facilities comply with Transporter's reasonable specifications, Transporter may pre-approve related design drawings and bills of materials, and Transporter may have inspectors present during construction. Approval to proceed with construction from Transporter and Transporter's inspectors will not be unreasonably delayed or withheld. The Interconnecting Party will reimburse Transporter for all costs associated with Transporter's review and inspection of such proximate

21. INTERCONNECTS (Continued)

facilities within 30 days of Transporter submitting an invoice detailing such costs. Interest calculated in accordance with 18 CFR Section 154.501(d) will accrue on any balance remaining after the due date specified in the invoice.

(f) Regardless of which party designs and constructs the Interconnect facilities, Transporter will have the right to require the installation of any equipment necessary to: accurately monitor the quality of Gas received into its Mainline Transportation System to ensure that such Gas meets the specifications of its Tariff, maintain the reliability and operational integrity of its Mainline Transportation System, and enable accurate custody transfer measurement.

21.5 Payment for Interconnect.

- (a) For any Tap and/or Meter Station that Transporter will design and construct, the facilities agreement will provide that the Interconnecting Party will make prepayments to Transporter, as specified in the facilities agreement, for Transporter's design and construction costs, Related Income Taxes and any other costs directly related to the construction of the Tap and/or Meter Station, with a final true-up payment/refund after completion.
- (b) For any Other Interconnect Facilities that Transporter agrees to design and construct, the facilities agreement and, if applicable, Exhibit C or Exhibit D to the Interconnecting Party's valid Service Agreement will provide that the Interconnecting Party will pay to Transporter over an agreed period of time an incremental facilities charge. Such charge will include, as applicable and appropriate, all Transporter's costs associated with the Other Interconnect Facilities, including without limitation operating and maintenance expenses, administrative and general expenses, return on debt, return on equity, income taxes, other taxes, depreciation and net negative salvage ("Cost Requirement").
- (c) Notwithstanding Sections 21.5(a) and (b), Transporter and the Interconnecting Party may agree to a different payment method, as applicable and appropriate, for Transporter to design, construct, own, operate and maintain any given Interconnect facility if Transporter agrees to do so.
- d) For any Interconnect facilities that Transporter will design and construct without receiving prepayment, Transporter and Interconnecting Party may agree to credit provisions that

21. INTERCONNECTS (Continued)

apply before and after the in-service date of the Interconnect facilities. Transporter may require a credit obligation in an amount up to full Cost Requirement of the Interconnect facilities, but not to exceed the Interconnecting Party's proportionate share of such Cost Requirement. Each year, Transporter will reduce the amount of such credit obligation as necessary to approximate the remaining Cost Requirement payable by the Interconnecting Party under the facilities agreement.

- (e) Subject to Transporter's prior approval, which will not be unreasonably withheld, the Interconnecting Party's payment and credit obligations under the facilities agreement and, if applicable, Exhibit C to the Interconnecting Party's Service Agreement may be assigned to a superseding or replacement Service Agreement(s) ("Successor").
- (f) Notwithstanding anything in this Section 21 to the contrary, Transporter may agree on a not unduly discriminatory basis in the facilities agreement to pay for all or part of the cost to design, construct, own, operate and/or maintain the Interconnect facilities if it determines that it will be economically beneficial for Transporter to do so. Transporter will post notice of any such contribution on its Designated Site for a period of 30 days following any such agreement.
- 21.6 Sharing of Expanded Interconnect Costs. If Transporter agrees, at the request of a later Interconnecting Party ("Second Interconnect Party"), to increase through compression the capacity of the original Interconnect facilities previously constructed by Transporter for an earlier Interconnecting Party ("First Interconnecting Party") pursuant to this Section 21, then the following cost sharing procedures will be applicable:
 - (a) Transporter will add its actual or imputed depreciated cost of the original Interconnect facilities to the incremental expansion cost and apportion the combined costs between the First Interconnecting Party and the Second Interconnecting Party pro rata based on the original and incremental expansion firm design capacity of the Interconnect facilities. (if the First Interconnecting Party reimbursed Transporter by a prepayment method, Transporter will calculate the imputed depreciated cost of the original Interconnect facilities using a depreciation rate equal to the higher of the depreciation rate for the Mainline

21. INTERCONNECTS (Continued)

Transportation System or the depreciation rate based on the term of the First Interconnecting Party's Service Agreement for firm capacity on the original Interconnect facilities, if any).

- (b) If the resulting apportioned cost to the Second Interconnecting Party is greater than the incremental expansion cost, the Second Interconnecting Party's payment responsibility under Section 21.5 will be based on its apportioned cost. Otherwise, the Second Interconnecting Party will only be responsible for the incremental expansion cost.
- (c) If the apportioned cost to the Second Interconnecting Party is greater than the incremental expansion cost, the difference ("Cost Sharing Adjustment") will be credited or refunded by Transporter to the First Interconnecting Party, or its Successor, pursuant to the following procedures:
 - 1) If the First Interconnecting Party and/or its Successor is currently paying Transporter for the original Interconnect facilities through payments over time, such obligation will be amended to reduce the capital cost under lying the facilities construction charge by the Cost Sharing Adjustment, to be effective on the in-service date of the incremental expansion Interconnect facilities.
 - If the First Interconnecting Party has reimbursed Transporter for the original Interconnect facilities through a prepayment, the Cost Sharing Adjustment will be refunded to the First Interconnecting Party within 45 days following the in-service date of the incremental expansion Interconnect facilities. If the Second Interconnecting Party pays its apportioned cost to Transporter using a prepayment method and pays Related Income Taxes, then the portion of the Related Income Taxes payment attributable to the Cost Sharing Adjustment also will be included in Transporter's refund to the First Interconnecting Party. Further, if the First Interconnecting Party is the passive owner of the original Interconnect facilities, its ownership interest will be reduced proportionate to the Cost Sharing Adjustment.

21. INTERCONNECTS (Continued)

- 21.7 Accelerated Payment Provisions.
 - (a) For any receipt or delivery facilities subject to a facilities agreement executed prior to January 16, 2012 and pursuant to which Shipper is paying for such facilities over time, the following provisions will continue to apply:
 - (1) Shipper may elect at any time to cease paying a facilities surcharge by paying Transporter for the then remaining net book value of the facilities, including any Related Income Taxes, at which time the applicable Service Agreement will be amended to reflect the termination of Exhibit C.
 - (2) If Shipper ceases operations at the end-use point or receipt point for which the facilities were installed, or ceases to be a Rate Schedule TF-1, Rate Schedule TF-2, or Rate Schedule TFL-1 Shipper (unless its facility cost reimbursement obligations hereunder are assumed by a Replacement Shipper), Shipper will notify Transporter of such occurrence within five business days of such occurrence. Shipper will pay Transporter for the then remaining net book value of the facilities, including any Related Income Taxes, within 30 days after Transporter submits an invoice to Shipper. Interest calculated in accordance with 18 CFR Section 154.501(d) will accrue on any balance remaining after the due date specified on the invoice.
 - (3) If Transporter has previously paid for all or a portion of receipt or delivery point facilities under a facilities agreement with Shipper, Shipper will, nevertheless, promptly pay Transporter for Transporter's remaining net book value of such facilities, including any Related Income Taxes, when either of the following events occurs: (1) Transporter's ability to fully recover such costs is denied in any Section 4 or Section 5 rate proceeding, or (2) Shipper permanently ceases operations at the end-use point or receipt point adjacent to where the facilities were installed.
 - (b) For any Interconnect facilities subject to a facilities agreement executed after January 16, 2012 and pursuant to which the Interconnecting Party is paying for such facilities over time, any mutually agreed provisions governing the Interconnecting Party's right or obligation to accelerate payment will be set forth in the facilities agreement and, if applicable, Exhibit C to the Interconnecting Party's valid Service Agreement.

22. CAPACITY RELEASE

22.1 CAPACITY RELEASE DEFINITIONS.

For the purposes of this Section 22, the following definitions will apply to Capacity Releases.

- (a) Releasing Shipper: A firm Shipper or Replacement Shipper under an executed Service Agreement subject to Part 284 of the Commission's regulations who intends to have Transporter post to Transporter's Designated Site all or any portion of its firm service to be released to a Replacement Shipper, or to a Prearranged Replacement Shipper through a prearranged capacity release transaction, or which has released its firm service.
- (b) Replacement Shipper: A Shipper which has contracted for service with Transporter subject to Part 284 of the Commission's regulations to utilize a Releasing Shipper's released firm service for a specified period of time. Replacement Shippers must meet all of the terms and conditions of Transporter's tariff as required of any other Shipper.
- (c) Potential Replacement Shipper: A Shipper which desires to acquire released firm capacity.
- (d) Prearranged Replacement Shipper: A party who desires to contract directly for or who has contracted directly for capacity which is subject to Part 284 of the Commission's regulations and who has agreed to the terms of a capacity release transaction directly with a Releasing Shipper, which terms are posted to Transporter's Designated Site. Prearranged Replacement Shippers must meet all of the terms and conditions of Transporter's tariff as required of any other Shipper.
- (e) Bidder: A party submitting a bid(s) to Transporter's Designated Site for released capacity.
- (f) Open Bidding: Full disclosure of the bid(s) submitted to Transporter's Designated Site, excluding the bidder(s) name.

22.1 CAPACITY RELEASE DEFINITIONS (Continued)

- (g) The Economic Value of a bid: The total net present value of the demand and capacity charge bid for the release of storage capacity or of the reservation charge bid for the release of transportation capacity. The net present value shall be computed from the monthly demand and capacity, or reservation revenues to be received over the term of the release contract, using the discounted cash flow rate of return methodology, with the rate of discounting to be equal to the rate set forth in Section 154.501(d) of the Commission's regulations at the time the capacity is posted for bid. NAESB WGQ Capacity Release Related Standard 5.3.3 will apply.
- (h) Open Season: The minimum number of business days allowed for posting releases and accepting bids.

22.2 GENERAL AND INDEX-BASED CAPACITY RELEASE PROVISIONS

NAESB WGQ Capacity Release Related Standards 5.2.4, 5.2.5, 5.3.1, 5.3.3, 5.3.4, 5.3.5, 5.3.7, 5.3.8, 5.3.9, 5.3.18, 5.3.20, 5.3.21, 5.3.24, 5.3.26, 5.3.31 through 5.3.42, 5.3.62, 5.3.62a, and 5.3.63 through 5.3.69 will apply.

22.2 CAPACITY RELEASE GENERAL PROVISIONS (Continued)

- (a) Any firm Shipper (Releasing Shipper) is eligible to release all or any part of its firm transportation capacity, defined, with respect to Rate Schedules TF-1, TF-2 and TFL-1, as capacity between primary receipt and delivery points, for use by another party (Prearranged Replacement Shipper or Replacement Shipper) for a minimum term of one (1) day and a maximum term not to exceed the remaining term of the Releasing Shipper's firm Service Agreement.
 - If a Shipper wishes to temporarily release a part of its capacity and Transporter determines that such a release could create operational problems (such as reducing Releasing Shipper's ability to respond to an OFO pursuant to Section 14.15 of the General Terms and Conditions), then either (i) Releasing Shipper must amend its Service Agreement to incorporate a contract-specific flow requirement on its retained capacity that would enable Transporter to duplicate the displacement capacity available under the Releasing Shipper's Service Agreement as it exists prior to such release, or (ii) Releasing Shipper must condition its release on the Replacement Shipper or Prearranged Replacement Shipper being subject to a contract-specific flow requirement on its newly acquired capacity, or (iii) the Replacement Shipper's Service Agreement must indicate that it is subject to recall in accordance with Section 22.2(a)(2).
 - (2) If Replacement Shipper's Service Agreement indicates that it is subject to recall pursuant to this Section 22.2(a)(2), and if Transporter provides notification that issuance of a general OFO under Section 14.15 is imminent, then Releasing Shipper either (i) must recall sufficient capacity from the Replacement Shipper prior to the Evening Nomination cycle following notification from Transporter to reinstate Releasing Shipper's pre-release ability to respond to a general OFO, or (ii) take other action that is acceptable to Transporter to satisfy such OFO obligation.
 - (3) If a Shipper wishes to temporarily release a part of its capacity that contains a conditional service agreement extension pursuant to Section 11.9 of the General Terms and Conditions of the tariff, then the Replacement Shipper's Service Agreement must indicate that the release is subject to recall.
 - (4) If Replacement Shipper's Service Agreement indicates that a release is subject to recall pursuant to Section 22.2(a)(3) and Transporter receives notice from the base contract holder that it is exercising its conditional service agreement extension, then such base contract holder must recall all of the capacity that goes beyond the later

22.2 CAPACITY RELEASE GENERAL PROVISIONS (Continued)

of the effective date associated with the early termination right or the evergreen rollover period.

- (5) If a Shipper wishes to permanently release only a part of its firm transportation capacity, such Shipper's aggregate MDQs and aggregate MDDOs must be released by the same percentage as the Contract Demand is released. Shipper may specify the MDQs it wishes to release at each Receipt Point and the MDDOs it wishes to release at each Delivery Point.
- (6) A Rate Schedule TF-2 Shipper may only release the Annual Contract Quantity set forth in its executed Service Agreement that is unused. The unused Annual Contract Quantity is such Shipper's Annual Contract Quantity reduced by the amount of transportation service received by such Shipper during the same twelve-month period ending September 30 through the effective date of the release.
- (7) A Releasing Shipper can segment its firm capacity under a Rate Schedule TF-1 or Rate Schedule TFL-1 Service Agreement by posting a release of such firm capacity from any physical Receipt Point to any physical delivery point subject to availability of capacity at and between the designated points.
- (8) Any Replacement Shipper or Prearranged Replacement Shipper which has previously contracted for released capacity may also release the capacity to another party, thereby becoming a Releasing Shipper, subject to the notification and bidding requirements of this Section 22.

22.2 CAPACITY RELEASE GENERAL PROVISIONS (Continued)

- (9) Releasing Shippers that have Service Agreements with aggregate MDDOs in excess of Contract Demand as a result of the grandfathering of pre-existing conjunctive nomination rights under the sales conversion program approved in Docket No. CP92-79 cannot release an amount of MDDOs that would increase the ratio of MDDOs to Contract Demand for the Replacement Shipper above the ratio that exists in the Releasing Shipper's Service Agreement.
- (10) Releasing Shippers with service under Rate Schedules TF-1 (Large Customer), TF-1 25-Year Evergreen, or TF-2, may condition their release upon the Replacement Shipper's assumption of the CRM Surcharge for the released capacity in full, at a fixed rate less than the full CRM Surcharge, or as a percentage of the CRM Surcharge. If no selection is made by the Releasing Shipper and the Replacement Shipper's reservation rate is the Maximum Base Tariff Rate or higher, then the Replacement Shipper will be deemed to have assumed the CRM Surcharge in full. If no selection is made by the Releasing Shipper and the Replacement Shipper's reservation rate is a percentage of the Maximum Base Tariff Rate, then the Replacement Shipper will be deemed to have assumed the CRM Surcharge in the same percentage as it assumed the Maximum Base Tariff Rate. If no selection is made by the Releasing Shipper and the Replacement Shipper's reservation rate is neither the Maximum Base Tariff Rate or higher nor a percentage of the Maximum Base Tariff Rate, then the Replacement Shipper will be deemed to have not assumed the CRM Surcharge.
- (b) Shippers releasing firm storage rights shall be subject to the notification and bidding requirements set forth in Section 22. Unless otherwise specified by the Releasing Shipper, Rate Schedule SGS-2F storage capacity will be released as one package which includes the storage injection and withdrawal rights and the associated storage capacity rights. In addition, the Rate Schedule SGS-2F Releasing Shipper must specify the level of

22.2 CAPACITY RELEASE GENERAL PROVISIONS (Continued)

Working Gas Quantity associated with the release for the purpose of calculating Unavailable Working Gas. Unless otherwise specified by the Releasing Shipper, Rate Schedule LS-2F and LS-3F storage capacity can only be released as one package which includes the storage liquefaction and vaporization rights and associated capacity rights. If Replacement Shipper fails to withdraw or vaporize, as applicable, all gas inventory within three (3) days of termination of the replacement agreement, then Transporter shall take title to any such remaining inventory free and clear of any adverse claims.

(c) Potential Replacement Shippers may post offers to purchase capacity to Transporter's Designated Site, which will remain posted on such site for a period specified by the Potential Replacement Shipper, but not to exceed 90 days. If an offer to purchase results in a capacity release being offered, such release is subject to the bidding requirements of this Section. Offers to purchase capacity must include desired receipt and delivery point(s), contract volume, and contract term.

22.3 CAPACITY RELEASE OFFERS.

All requests to release capacity shall be made through Transporter's Designated Site setting forth the following:

- 1. Releasing Shipper's name, contract number and contract quantity to be released and, for Rate Schedule TF-2 releases, the unused Annual Contract Quantity;
- 2. receipt point(s), delivery point(s) and volumes
 associated with each;
- 3. if a storage release, the volume of storage withdrawal or vaporization and storage capacity rights;
- 4. commencement date for the release, termination date for the release;
- 5. minimum percentage of volume and minimum term for the release which Releasing Shipper is willing to accept;
- 6. any specific conditions of the release, which conditions must be objectively stated, applicable to all bidders, operationally feasible, non-discriminatory to other Shippers and consistent with this Tariff. If Releasing Shipper submits no release conditions, or

22.3 CAPACITY RELEASE OFFERS (Continued)

Transporter finds the requested conditions operationally infeasible or inconsistent with this Tariff, the released capacity shall be subject to the conditions currently effective in this Tariff. Releasing Shipper must notify Transporter if the minimum conditions precedent to the release are not to be posted on Transporter's Designated Site until after the capacity is awarded;

- 7. if Releasing Shipper desires to release capacity on the basis of a volumetric rate, Releasing Shipper must provide the criteria Releasing Shipper desires Transporter to use in the evaluation of volumetric bids and any other terms and conditions of the release;
- 8. if Releasing Shipper has secured a Prearranged Replacement Shipper, the release request must also contain:
 - (a) the name of the Prearranged Replacement Shipper;
 - (b) the rate and term of the bid which such Prearranged Replacement Shipper is willing to pay;
 - (c) whether the transaction is being made pursuant to: i) a standard capacity release, ii) the implementation of an Asset Management Arrangement, as defined in 18 C.F.R. Section 284.8 (h) (3) ("AMA"), or iii) a state-regulated retail access program, as defined in 18 C.F.R. Section 284.8 (h) (4); and
 - (d) whether the Prearranged Replacement Shipper is affiliated with the Releasing Shipper.
- 9. if Releasing Shipper has secured a Prearranged Replacement Shipper pursuant to the implementation of an AMA transaction, the release request must also contain AMA Special Terms and Conditions which: 1) state that the release is being made to implement an AMA; and 2) specifies the quantity and term commitment of the Replacement Shipper's delivery or purchase obligation to the Releasing Shipper;

22.3 CAPACITY RELEASE OFFERS (Continued)

10. whether Releasing Shipper is willing to accept contingent bids that extend beyond the open season and the terms and conditions applicable to such contingency, including the rights and obligations of the next highest Bidder to secure the released capacity in the event that the contingency is not fulfilled or the winning contingent Bidder exercises its option not to take the capacity. The date to which the contingency may be extended must end at least four (4) working days prior to the effective date of the release.

22.4 CAPACITY RELEASE BIDS.

- (a) All bids for released capacity shall be made through Transporter's Designated Site setting forth the following:
 - 1. Bidder's name and capacity release number bidding on;
 - 2. receipt point(s) and delivery point(s) bidding on and the volume and rate associated with each;
 - 3. if bidding on storage, the volume and rates for both the storage withdrawal or vaporization rights and the storage capacity rights;
 - 4. term of bid;
 - 5. minimum percentage of the volume which the Bidder is willing to accept; and

22.4 CAPACITY RELEASE BIDS (Continued)

- 6. any affiliation of the releasing shipper involved in the transaction.
- (b) A Releasing Shipper may specify that bids may be accepted on a volumetric rate basis.
- (c) Releases of Rate Schedule SGS-2F, LS-2F and LS-3F service shall be acquired either through separate bids to pay the applicable Demand and/or Capacity Demand charge(s) or, if released on a volumetric bid basis, through separate bids to pay for usage on a daily basis of Contract Demand withdrawn or vaporized, as applicable, and storage capacity used.
- (d) All bids shall be submitted in amounts which a Replacement Shipper or Prearranged Replacement Shipper agrees to pay to Transporter for the right to demand firm services, including undiscounted reservation surcharges, if applicable. Such bids shall exclude Transporter's volumetric charges and volumetric surcharges, including liquefaction and vaporization volumetric charges, applicable to each service as included in Transporter's rates, which rate components must be paid to Transporter by the successful bidding Replacement Shipper or Prearranged Replacement Shipper in addition to the bid amounts. The Releasing Shipper shall not be responsible for continuing to pay such volumetric charges and volumetric surcharges for the capacity released during the term of each release, nor for penalties incurred by the Replacement Shipper or Prearranged Replacement Shipper.

22.5 CAPACITY RELEASE AWARD PROCESS FOR BIDDABLE TRANSACTIONS.

(a) Transporter shall determine the winning bid. Unless Releasing Shipper specifies otherwise, bids will be evaluated based on the economic value of the demand and capacity or reservation charges bid by each Bidder. After the economic value of each bid for each receipt and delivery point combination has been determined, and if the capacity to be allocated to the highest Bidders would not result in awarding all of the capacity to be released, then Transporter shall award the remaining

22.5 CAPACITY RELEASE AWARD PROCESS FOR BIDDABLE TRANSACTIONS (Continued)

capacity based on the economic value of each bid for each receipt and delivery point combination, from highest economic value to lowest economic value, with the process of awarding capacity to Bidders continuing until all of the capacity has been awarded or until all valid bids have been accepted. Transporter shall break ties and determine the winning bid(s) in those situations by holding a lottery using a computerized random number generator unless the releasing Shipper specifies another non-discriminatory method for determining the winning bid.

A Releasing Shipper may specify and post to Transporter's Designated Site an objectively stated, nondiscriminatory standard for Transporter to use in determining the best bid(s), including the criteria to be used by Transporter in evaluating volumetric bid(s) and, subject to Section 22.5(b), repackaged bids. If Releasing Shipper provides insufficient criteria for Transporter to use in evaluating volumetric rate bid(s), Transporter will evaluate volumetric bids based on the rate which is bid, or, in the case of services for which multiple bids are required, the sum of such bids.

- (b) At Releasing Shipper's request using reasonable, objective criteria for repackaging, Transporter may repackage bids for the same release to maximize the economic value of the released capacity i.e., discrete bids for portions or segments of the released capacity may be repackaged for evaluation; provided, however that Transporter shall be under no obligation to repackage bids if Transporter determines that such repackaging is not administratively or operationally feasible.
- 22.6 CAPACITY RELEASE CREDIT REQUIREMENTS.
 - (a) A Replacement Shipper or Prearranged Replacement Shipper must comply with Section 5.10 of the General Terms and Conditions of this tariff. Such Shipper must submit a request for prequalification for credit as a Shipper to Transporter in order to submit a bid for capacity release service.

22.6 CAPACITY RELEASE CREDIT REQUIREMENTS. (Continued)

(b) Exception. A Prearranged Replacement Shipper will automatically be pre-qualified for credit if a creditworthy Releasing Shipper elects to assume all liability for any unpaid charges (all reservation and commodity based charges) incurred by such prearranged capacity release transaction by posting its assumption of such liability in the special terms and conditions of its Offer.

22.7 CAPACITY RELEASE MISCELLANEOUS PROVISIONS.

- (a) Releasing Shipper is responsible for payment of demand and capacity charges for the release of storage capacity and reservation charges for the release of transportation capacity, as well as any applicable reservation surcharges under the firm Service Agreement as it existed before any release(s), except as provided in subsection 22.4(c) above and in Section 23. Releasing Shipper also is responsible for payment obligations, payment procedures and crediting mechanisms that vary from or are in addition to those set forth herein when those provisions are part of a Negotiated Rate service agreement.
- (b) Transporter shall not be liable to Releasing Shipper for the revenue credits provided for in Section 23 of the General Terms and Conditions in the event any Replacement Shipper fails to pay any applicable demand, capacity, or reservation charges or applicable reservation surcharges to Transporter. However, releasing Shipper shall retain a subrogated right to demand payment from any Replacement Shipper or Prearranged Replacement Shipper in the event any Replacement Shipper or Prearranged Replacement Shipper is delinquent on any such payment owed Transporter.

22.7 CAPACITY RELEASE MISCELLANEOUS PROVISIONS (Continued)

(c) Bids may be submitted for any segment of released Rate Schedule TF-1 or Rate Schedule TFL-1 capacity subject to the availability of metering and pipeline capacity at and between the requested receipt and delivery points. Replacement Shippers or Prearranged Replacement Shippers may change primary receipt or delivery points as specified in this Tariff during the term of a capacity release.

At the expiration of each release such changes shall terminate and the Releasing Shipper shall be entitled to its rights at receipt and delivery points as they existed prior to the release.

- (d) A Releasing Shipper may withdraw its release in accordance with NAESB Standard 5.3.14 and prior to receipt of a valid bid.
- (e) A Bidder may withdraw its bid in accordance with NAESB Standard 5.3.13, provided, however, that the Bidder may not then resubmit a bid with a lower Economic Value for the same capacity. Bidder may not submit multiple bids on the same release package.

22.7 CAPACITY RELEASE MISCELLANEOUS PROVISIONS (Continued)

- Transporter will provide service to a Replacement Shipper or a Prearranged Replacement Shipper pursuant to an electronically executed Service Agreement. The Replacement Shipper's or Prearranged Replacement Shipper's execution of the Service Agreement occurs automatically upon posting of its bid and Transporter's execution occurs upon award. Transporter's accepted bid for release of firm capacity and any conditions of the release contained in the Releasing Shipper's posting of the release will be reflected on the exhibits to the Service Agreement between Transporter and the Replacement Shipper or Prearranged Replacement Shipper. Releasing Shipper's rights under its Service Agreement also will be deemed modified consistent with Transporter's accepted bid for release of firm capacity and any conditions of the release reflected on the exhibits to the Service Agreement between Transporter and the Replacement Shipper or Prearranged Replacement Shipper.
- (g) Transporter may assess a marketing fee to a Releasing Shipper, if agreed upon by both Transporter and Releasing Shipper, pursuant to a separately negotiated marketing contract, for marketing services to be performed by Transporter.
- (h) All terms of Capacity Release transactions between Transporter, the Releasing Shipper and the Replacement Shipper or Prearranged Replacement Shipper, must be agreed to through the public notification and posting procedures contained herein.

22.7 CAPACITY RELEASE MISCELLANEOUS PROVISIONS (Continued)

- (i) In the event a capacity release transaction is not effective prior to the Evening Nomination cycle applicable to the initial day of the capacity release, the Contract Demand released as reflected in the Releasing Shipper's amended Service Agreement and the Contract Demand acquired as stated in the Replacement Shipper's Service Agreement may be modified for the initial day of the release to prevent such Contract Demand from exceeding nineteen/twenty-fourths of the Releasing Shipper's available Contract Demand if the release is effective for the Intraday 1 nomination cycle or fifteen/twenty-fourths of the Releasing Shipper's available Contract Demand if the release is effective for the Intraday 2 nomination cycle or eleven/twenty-fourths of the Releasing Shipper's available Contract Demand if the release is effective for the Intraday 3 nomination cycle.
- (j) For a capacity re-release transaction, both the Releasing Shipper's posting of the offer and the exhibits to the new Replacement Shipper's Service Agreement will reflect the "standard capacity release conditions" included on Exhibit A to the Releasing Shipper's Service Agreement that must also apply to the new Replacement Shipper in order to preserve the previous Releasing Shipper's rights and obligations.
- (k) The "additional capacity release conditions" included on Exhibit A to the Replacement Shipper's Service Agreement will not cause Transporter to modify any existing notification procedures. Transporter does not assume or accept any liability or responsibility to enforce or administer any additional capacity release conditions that a Releasing Shipper may impose. Releasing Shipper and Replacement Shipper will indemnify Transporter for any claims made against Transporter by the other that arise out of any additional capacity release conditions imposed by Releasing Shipper.

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 265-A Through 265-C Superseding Substitute Original Sheet Nos. 265-A Through 265-C

RESERVED FOR FUTURE USE

- 22. CAPACITY RELEASE (Continued)
- 22.8 CAPACITY RELEASE OPEN SEASONS.

NAESB WGQ Capacity Release Related Standards 5.3.10 through 5.3.16, 5.3.19, 5.3.25, 5.3.26, 5.3.28, and 5.3.29 will apply.

22.9 CAPACITY RELEASE TIMELINES.

The following Capacity Release timelines in Central Clock Time (with Mountain Clock Time in parentheses) reflect NAESB WGQ Standard 5.3.2:

Biddable Releases (1 year or less)

Timeline pertains to a Business Day:

Offers Tendered:	9:00 (8:00) A.M.
Open Season Ends:	10:00 (9:00) A.M.
Evaluation Begins:	10:00 (9:00) A.M.
Contingency eliminated.	
Best bid determined.	
Ties broken.	
Evaluation Ends:	11:00 (10:00) A.M.
Award Posted if no Match:	11:00 (10:00) A.M.
Option to Match communicated:	11:00 (10:00) A.M.
Shipper Match Response:	11:30 (10:30) A.M.
Award Posted if Match:	Noon (11:00 A.M.)
Contract Issued:	Within One Hour of
	Award Posting
Nomination Possible:	Next Available
	Nomination Cycle
	for Effective Date
	of Contract

22.9 CAPACITY RELEASE TIMELINES (Continued)

Biddable Releases (more than 1 year)

Offers Tendered:

9:00 (8:00) A.M.

of Contract

Open Season shall include no less than three 9:00 A.M. to 10:00 A.M. time periods on consecutive Business Days:

Timeline pertains to a Business Day:

Open Season Ends: 10	:00 (9:00) A.M.
Evaluation Begins: 10	:00 (9:00) A.M.
Contingency eliminated.	
Best bid determined.	
Ties broken.	
Evaluation Ends: 11	:00 (10:00) A.M.
Award Posted if no Match: 11	:00 (10:00) A.M.
Option to Match communicated: 11	:00 (10:00) A.M.
Shipper Match Response: 11	:30 (10:30) A.M.
Award Posted if Match: No	on (11:00 A.M.)
Contract Issued: Wi	thin One Hour of
	Award Posting
Nomination Possible: Ne	xt Available
	Nomination Cycle
	for Effective Date

22.9 CAPACITY RELEASE TIMELINES (Continued)

Non-Biddable Releases.

The posting of prearranged deals that are not subject to bid are due no later than one hour prior to the nomination deadline for the applicable cycle, pursuant to NAESB WGQ Standard No. 1.3.2.:

$\overline{}$							
-	\sim	C	_	٦.	n	\sim	•
т.	\circ	S	L	_	ΤŢ	q	٠

Nomination Possible:

Timely Cycle:	Noon	(11:00	A.M.)
Evening Cycle:	5:00	(4:00)	P.M.
<pre>Intraday 1 Cycle:</pre>	9:00	(8:00)	A.M.
<pre>Intraday 2 Cycle:</pre>	1:30	(12:30)	P.M.
<pre>Intraday 3 Cycle:</pre>	6:00	(5:00)	P.M.

Contract Issued: Within One Hour of

Award Posting
Next Available
Nomination Cycle
for Effective Date

of Contract

Transporter may extend the open season periods for those releases wherein the Releasing Shipper has requested volumetric rate bids, or where the Releasing Shipper has requested a method of evaluating bids not specified in this Tariff, or where the terms and conditions are too complex to handle through Transporter's automated posting and bidding procedures, or where Releasing Shipper has specified an index-based rate unique formula. Such extended period of time shall end at least two (2) business days prior to the effective date of the release.

22.10 CAPACITY RELEASE TRANSACTIONS EXEMPT FROM BIDDING.

(a) A Releasing Shipper may release capacity to a Prearranged Replacement Shipper for any period of thirty-one (31) days or less without having to comply with the notification and bidding requirements of this Section 22. A Releasing Shipper may not re-release capacity covered by this Section 22.10(a) to the same Prearranged Replacement Shipper without complying with the notification and bidding requirements of this Section 22 until twenty-eight (28) days after the first release period has ended.

22.10 CAPACITY RELEASE TRANSACTIONS EXEMPT FROM BIDDING (Continued)

- (b) For a release involving a Prearranged Replacement Shipper where the rate is equal to the maximum posted tariff rate applicable to capacity releases, and the term is equal to the term described by the Releasing Shipper but is greater than one year in any event, the Prearranged Replacement Shipper shall be awarded the released capacity without the necessity of waiting for competing bids to be submitted. However, Transporter shall be notified of the release by the Releasing Shipper and the notification shall contain the same information as identified in Section 22.3. Transporter shall post a notice of the occurrence of the release on its Designated Site.
- (c) For a release involving a Prearranged Replacement Shipper pursuant to an AMA or pursuant to a state-regulated retail access program, the Prearranged Replacement Shipper shall be awarded the released capacity without the necessity of waiting for competing bids to be submitted. However, Transporter shall be notified of the release by the Releasing Shipper and the notification shall contain the same information as identified in Section 22.3. Transporter shall post a notice of the occurrence of the release on its Designated Site.

22.11 RECALL OF RELEASED CAPACITY.

Releasing Shippers may, to the extent permitted as a condition of the capacity release, recall released capacity. For the recall notification provided to Transporter, the Contract Demand recalled should be expressed in terms of total released capacity entitlements. In the event a capacity recall transaction is not effective prior to the Evening Nomination cycle applicable to the initial day of the capacity recall, the Contract Demand recalled may be modified for the initial day of the recall to prevent such Contract Demand from exceeding nineteen/twenty-fourths of the Contract Demand recalled if the recall is effective prior to the Intraday 1 nomination cycle or fifteen/twenty-fourths of the Contract Demand recalled if the recall is effective prior to the Intraday 2 nomination cycle or eleven/twenty-fourths of the Contract Demand recalled if the recall is effective prior to the Intraday 3 nomination cycle.

22.11 RECALL OF RELEASED CAPACITY (Continued)

The following NAESB WGQ Capacity Release Related Standards, in Central Clock Time, will apply:

Standard 5.3.44: "All Transportation Service Providers (TSPs) should support the following recall notification periods for all released capacity subject to recall rights:

- (i) Timely Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 8:00 a.m. on the day that Timely Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 9:00 a.m. on the day that Timely Nominations are due;
- (ii) Early Evening Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 3:00 p.m. on the day that Evening Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 4:00 p.m. on the day that Evening Nominations are due;
- (iii) Evening Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 5:00 p.m. on the day that Evening Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 6:00 p.m. on the day that Evening Nominations are due;

22.11 RECALL OF RELEASED CAPACITY (Continued)

- (iv) Intraday 1 Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 7:00 a.m. on the day that Intraday 1 Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 8:00 a.m. on the day that Intraday 1 Nominations are due; and
- (v) Intraday 2 Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 12:00 p.m. on the day that Intraday 2 Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 1:00 p.m. on the day that Intraday 2 Nominations are due.
- (vi) Intraday 3 Recall Notification:
 - (a) A Releasing Shipper recalling capacity should provide notice of such recall to the TSP and the first Replacement Shipper no later than 4:00 p.m. on the day that Intraday 3 Nominations are due;
 - (b) The TSP should provide notification of such recall to all affected Replacement Shippers no later than 5:00 p.m. on the day that Intraday 3 Nominations are due."

22.11 RECALL OF RELEASED CAPACITY (Continued)

Standard 5.3.45: "For recall notification provided to the Transportation Service Provider (TSP) prior to the recall notification deadline specified in NAESB WGQ Standard No. 5.3.44 and received between 7:00 a.m. and 5:00 p.m., the TSP should provide notification to all affected Replacement Shippers no later than one hour after receipt of such recall notification.

For recall notification provided to the TSP after 5:00 p.m. and prior to 7:00 a.m., the TSP should provide notification to all affected Replacement Shippers no later than 8:00 a.m. after receipt of such recall notification."

Standard 5.3.54: "The deadline for notifying the Transportation Service Provider of a reput is 8:00 a.m. to allow for timely nominations to flow on the next gas day."

NAESB WGQ Capacity Release Related Standards 5.3.46 through 5.3.53, and 5.3.55 through 5.3.60 will apply.

RESERVED FOR FUTURE USE

22.12 PERMANENT RELEASES OF CAPACITY.

- (a) Parameters, Bidding Procedures and Exit Fee.
 - To the extent that any Shipper desires to release all or any part of its firm contract rights (including the Right-of-First-Refusal specified in Section 25 and its residual rights to firm capacity encumbered with temporary capacity releases, if applicable), on a permanent basis as included in any of Transporter's Part 284 firm Rate Schedules, the bidding procedures specified in Section 22.4, above, will apply. If a Shipper wishes to permanently release only a portion of its firm transportation capacity, the provisions in Section 22.2(a)(5) also will apply. The procedures for establishment of credit must be complied with before Transporter effectuates any permanent releases of capacity under this Section 22.6. In any event, Transporter may refuse to allow a permanent release if it has a reasonable basis to conclude that it will not be financially indifferent to the release. Transporter will notify the Shipper electronically in Northwest Passage if the Shipper's request to permanently release capacity is denied, and will include in the notification the reasons for such denial.
 - (ii) A Replacement Shipper may acquire a Releasing Shipper's grandfathered unilateral evergreen provision under Rate Schedules TF-1, TF-2 and SGS-2F if (i) the primary term of the permanent release from the Releasing Shipper is for a period of at least ten years or (ii) the Replacement Shipper agrees to incorporate a primary term of at least ten years at the time they accept the permanent release. Otherwise the Replacement Shipper has the option to select one of the standard evergreen provisions currently available under the applicable Rate Schedule.
 - (iii) A Shipper may permanently release firm capacity rights acquired under a temporary capacity release in the event such Shipper also holds and is concurrently permanently releasing the corresponding rights under the associated base Service Agreement.

22.12 PERMANENT RELEASES OF CAPACITY (Continued)

- (iv) For any permanent capacity release, the minimum bid acceptable to Transporter will be a bid for the remainder of the Releasing Shipper's contract term at the rate the Releasing Shipper is obligated to pay Transporter for the released capacity.
- (v) As part of its permanent capacity release offer, the Releasing Shipper may offer to make a lump sum exit fee payment, up to a specified amount, directly to the Replacement Shipper as consideration for the Replacement Shipper agreeing to pay Transporter a rate at or above Transporter's minimum acceptable bid rate. If Releasing Shipper's capacity release offer includes an exit fee payment commitment, any bids for the capacity must state the rate the Bidder offers to pay Transporter for the released capacity and the associated exit fee payment, not to exceed the Releasing Shipper's specified amount, that Bidder requires from the Releasing Shipper. The winning bid will be determined based on the net present value of the transportation rate component of the bid less the exit fee component of the bid. The cash flow discount factor for the net present value calculations will be the latest annual interest rate published quarterly by the Commission as posted on the Commission's Internet Website, unless otherwise specified by the Releasing Shipper. The payment of any exit fee will be handled directly between the Releasing Shipper and Replacement Shipper, with both parties agreeing that Transporter will not be responsible for collecting or disbursing such payment.

22.12 PERMANENT RELEASES OF CAPACITY (Continued)

(b) Procedures.

- (i) Replacement Shipper(s) or Prearranged Replacement Shipper(s) and Transporter shall electronically execute a new Service Agreement(s) prior to the commencement of service which confirms the terms of the accepted transaction for the permanent release of capacity. The Replacement Shipper's or Prearranged Replacement Shipper's execution of the Service Agreement occurs automatically upon posting of its bid and Transporter's execution occurs upon award.
- (ii) Upon execution of the new Service Agreement with the Replacement Shipper, the Releasing Shipper's Service Agreement will be deemed amended to reflect the terms contained in the Offer, including either: 1) the reduction in Contract Demand due to a permanent partial release; or 2) the termination of the Service Agreement due to release of all of the Contract Demand. The amendment to the Service Agreement will be reflected in the automatic addition of an Addendum to Service Agreement.
- (iii) If the Releasing Shipper's Service Agreement is encumbered with temporary capacity release transactions at the time of a permanent capacity release, the Service Agreement(s) for such temporary Replacement Shipper(s) will be deemed amended to identify the new Replacement Shipper that replaced the original Releasing Shipper. An Addendum to Service Agreement will be automatically added to such temporary capacity release Service Agreement(s) to document such deemed amendment(s) upon the award of the permanent capacity release. Transporter will notify such Replacement Shipper(s) of the deemed amendment.

- 23. REVENUE CREDITING FOR CAPACITY RELEASES (FIRM)
 - 23.1 Definitions. The definitions in Section 22 of the General Terms and Conditions apply also to this Section.
 - 23.2 Revenue Credits for Released Capacity Eligibility. Transporter shall provide revenue credits to any Shipper, which temporarily releases (Releasing Shipper), under the Capacity Release Provisions of this Tariff, all or any portion of its Contract Demand under a Service Agreement (Released Capacity) governed by the provisions of any Part 284 firm Rate Schedule included in Northwest's Tariff.

- 23. REVENUE CREDITING FOR CAPACITY RELEASES (FIRM) (Continued)
 - 23.3 Monthly Crediting Procedure. Releasing Shipper shall be billed each month for the reservation charge(s) and reservation surcharge(s) (or demand or capacity charges, as applicable, if the release is a release of firm storage service) associated with the released capacity pursuant to its Service Agreement, with a concurrent conditional credit for the payment of the reservation charge(s) and reservation surcharge(s) due from the Replacement Shipper or Prearranged Replacement Shipper. However, a Releasing Shipper that is not subject to facility reservation surcharges pursuant to Section 3.4 of Rate Schedule TF-1 and its Service Agreement will not be credited for facility reservation surcharges paid to Transporter by the Replacement Shipper or Prearranged Replacement Shipper. (The term reservation charge(s) in this paragraph includes volumetric reservation charge payments received from a Replacement Shipper or Prearranged Replacement Shipper where the release was pursuant to volumetric bidding, and shall include all revenues received in excess of the sum of volumetric charges and all surcharges.) This bill shall include an itemization of credits and adjustments associated with each Capacity Release Service Agreement. Releasing Shipper shall also be billed a marketing fee, if applicable.
 - 23.4 Failure to pay. If a Replacement Shipper or Prearranged Replacement Shipper does not make payment to Transporter of the reservation charge(s) and reservation surcharge(s) due as set forth in Section 5 of the General Terms and Conditions, Transporter shall notify the Releasing Shipper within 10 business days of the amount due, including all late charges authorized in Section 5 of the General Terms and Conditions, unless such amount is less than \$50. If the amount due, as described above, remains unpaid in whole or in part for 30 days after the invoice due date, Transporter shall invoice the Releasing Shipper and such amount shall be paid by Releasing Shipper within 10 days of the date of such invoice. In addition, Releasing Shipper may terminate the release of capacity to a Replacement Shipper or Prearranged Replacement Shipper if such Shipper fails to pay all of the amount of any bill for service under the executed capacity release Service Agreement upon at least one week of notice to Transporter and such Shipper. Once terminated, capacity rights and the responsibility to pay all charges specified in the original service agreement applicable to the capacity that was released shall revert to the Releasing Shipper.

23. REVENUE CREDITING FOR CAPACITY RELEASES (FIRM) (Continued)

Transporter shall apply partial payments received from Replacement Shippers or Prearranged Replacement Shippers, in interpreting the provisions of this section, first to reservation, demand and capacity charges, then to volumetric charges, then to volumetric surcharges, then to reservation surcharges, then to penalty charges and finally to late charges.

- 24. CLAY BASIN STORAGE SERVICE REVENUE CREDITING
 - 24.1 Applicability. Clay Basin revenue credits received by Transporter from MountainWest Pipeline, LLC (MWP), or its successor, pursuant to its Clay Basin Interruptible Revenue Sharing tariff provisions will be distributed to Transporter's Rate Schedules TF-1 (Large Customers), TF-1 (Small Customers) for the period they paid a reservation charge due to conversion to Rate Schedule TF-1 (Large Customer) service and TF-2 Shippers (applicable firm Shippers), excluding any Shipper receiving service under a discounted firm transportation service agreement or a capacity release service agreement.
 - 24.2 Timing of Credits. Within 30 days of receipt from MWP, or its successor, of the Clay Basin revenue credits totaling \$5,000 or more, Transporter will determine the distribution, as described below, of such revenue credits due to applicable firm Shippers. Such revenue credits will be reflected as a credit billing adjustment to the next invoices rendered to the applicable firm Shippers. In the event that such credit billing adjustment would result in a credit total invoice to any Shipper, Transporter will refund the excess credit billing adjustment to the Shipper by check within 15 days after determination of the amount of the credit due to the Shipper.

For administrative efficiency no credit billing adjustment or check will be issued until the amount of Clay Basin revenue credits held by Transporter total \$5,000 or more. Credits awaiting distribution until this threshold is reached will accrue interest pursuant to Section 24.4 herein until distributed to the applicable firm Shippers. Interest will be distributed in the same proportion as revenue credits.

- 24. CLAY BASIN STORAGE SERVICE REVENUE CREDITING (Continued)
 - 24.3 Distribution Method. Clay Basin revenue credits will be credited to each applicable firm Shipper on a pro-rata basis in proportion to the Maximum Base Tariff rate reservation charges (exclusive of surcharges) billed to each applicable firm Shipper, during the period coinciding with each twelve-month period utilized by MWP, or its successor, to determine any revenue credits divided by the total reservation charges billed by Transporter to all applicable firm Shippers for the same period. For any applicable firm Shipper paying a negotiated rate above the Maximum Base Tariff rate, reservation revenues will be capped at the Maximum Base Tariff rate when determining such Shipper's pro-rata share of the Clay Basin revenue credit.
 - 24.4 Interest. Transporter will pay interest to Shippers on the Clay Basin revenue credits until the revenue credits are distributed to the applicable firm Shippers. Such interest will be calculated in accordance with the procedures outlined in Section 154.501(d) of the Commission's regulations. Such interest will accrue on the Clay Basin revenue credits commencing on the day Transporter receives such revenue credits from MWP, or its successor.

25. AVAILABLE CAPACITY

25.1 General

This Section 25 sets forth procedures for Transporter to post and offer, and for Shippers to acquire, available capacity for firm service under Rate Schedule TF-1, TF-2, TFL-1, SGS-2F, LS-2F, or LS-3F. New Service Agreement requests for mainline capacity under Rate Schedule TF-1, TF-2 or TFL-1 must include available MDQs at receipt point(s) and MDDOs at delivery point(s). New capacity is requested on Transporter's Designated Site. Modifications to Service Agreement requests utilizing available capacity must be submitted in accordance with Section 11.4 of the General Terms and Conditions.

25.2 Availability of Capacity

Available capacity includes:

- (a) capacity currently available for any period;
- (b) capacity currently available for an interim period, but is reserved for a future period pursuant to Section 25.8 or has been sold for a future period;
- (c) capacity that will become available on a date certain due to the expiration of a firm Service Agreement that does not have an evergreen provision;
- (d) capacity that will become available on a date certain because either Transporter or Shipper has provided termination notice under a firm Service Agreement with an evergreen provision; or
- (e) capacity that will become available as a result of Shipper's request to amend a Service Agreement in such a manner that it creates additional available capacity that otherwise would not be available.

- 25. AVAILABLE CAPACITY (Continued)
- 25.3 Expansion or New Construction Capacity

Expansion or new construction capacity is capacity made available through:

- (a) an expansion of the existing Transportation System, but only pursuant to the non-discriminatory procedures and conditions that will be set forth in Transporter's open season postings for an expansion project; or
- (b) construction of new or upgraded facilities under Section 21 or 29 of the General Terms and Conditions, provided however that prior to the in-service date for such facilities only the party obligated to reimburse Transporter for such facilities may submit a request to acquire firm service rights to any resulting new firm capacity. After the in-service date for the facilities, any associated uncontracted firm capacity will be available to any Shipper pursuant to this Section 25.

If the reimbursing party has an existing transportation Service Agreement that could potentially be amended or segmented in order to use any resulting new or upgraded capacity then the reimbursing party may submit a request to (1) amend Primary Receipt Points or Primary Delivery Points under its existing Service Agreement; or (2) temporarily segment capacity under its existing Service Agreement in accordance with Section 13 of Rate Schedule TF-1 and Section 8 of Rate Schedule TFL-1. If Transporter determines it can accommodate such request, it will execute the requested amendment or segmentation for implementation effective the later of the requested effective date or the in-service date for the associated facilities.

25.4 Informational Postings

Transporter will post information regarding available capacity to its Designated Site utilizing the following reports:

(a) Unsubscribed Capacity Report

Transporter's Unsubscribed Capacity Report will identify all Receipt Points, Delivery Points, mainline capacity, or storage capacity, as applicable, which is available for new firm service, amendment or Capacity Release segmentation requests. Transporter will post capacity as available within two business days of becoming aware of such availability.

25. AVAILABLE CAPACITY (Continued)

(b) TSP Capacity Offering Report

Transporter's TSP Capacity Offering Report will identify Receipt Points, Delivery Points, and/or mainline capacity packages, or storage capacity that Transporter has pre-defined as available for firm service.

(c) Index of Customers Report

Transporter's Index of Customers Report will identify capacity that could become available as a result of Transporter's option to provide termination notice under firm Service Agreements with a bi-lateral evergreen provision.

Transporter has the option to establish a reserve rate for any available capacity before such capacity is posted for competitive bid. Transporter is not required to disclose the reserve rate at the time of the competitive bid posting, but a record of the reserve rate must be maintained for a period of three years for validation purposes.

25.5 Available Capacity Acquisition Methods

Other than acquiring capacity pursuant to Section 25.3, Shippers may acquire capacity utilizing any of the methods identified below:

(a) Prearranged Transactions

On a nondiscriminatory basis, Transporter may enter into a prearranged transaction with any creditworthy Shipper for any available capacity posted pursuant to this Section 25. To request to be the prearranged Shipper in such transactions, a Shipper must submit a prearranged offer request electronically on Transporter's Designated Site, unless otherwise specified by Transporter that the terms and conditions of the prearranged transaction are too complex to be administered on Transporter's Designated Site.

A prearranged offer request must include the bid information specified herein. Such request will be deemed a binding bid which cannot be withdrawn by Shipper once accepted by Transporter. However, Transporter has the discretion to reject any prearranged offer request:

(i) for available capacity requested with contingencies or conditions,

25. AVAILABLE CAPACITY (Continued)

- (ii) for available capacity requested at less than the Maximum Base Tariff Rate, and $\,$
- (iii) for available capacity at the Maximum Base Tariff Rate prior to thirty days of its availability.

For any available capacity, the prearranged offer request that is acceptable to Transporter will be posted as a prearranged transaction and will be subject to competitive bid.

(b) Competitive Bid Transactions

Transporter will hold a competitive bid process for capacity that: (1) has been requested by a prearranged Shipper; or (2) Transporter has elected to post for competitive bid without a prearranged Shipper. Unless otherwise specified, all competitive bid postings will be posted on Transporter's Designated Site.

Each competitive bid posting will include all information outlined below that is pertinent to the offered capacity:

- 1. Contract Demand, Storage Demand and/or Storage Capacity;
- Receipt Point(s), Delivery Point(s), and/or mainline capacity or Storage Facility;
- 3. Date capacity is available, and any availability duration limits, e.g., due to reservation of the capacity for a future period pursuant to Section 25.8;
- 4. Bid open and close dates;
- 5. Minimum acceptable reservation rate, which may be a stated rate or a non-disclosed reserve rate;
- 6. Whether Transporter is willing to entertain Negotiated Rate bids and the specific basis for such bids;

25. AVAILABLE CAPACITY (Continued)

- 7. Reservation rate and term offered by a prearranged Shipper, if any;
- 8. Transporter's bid evaluation criteria;
- 9. Whether capacity is subject to a Right of First Refusal option; and
- 10. Other non-discriminatory conditions that Transporter may elect to specify for the offered capacity.

Competitive Bid Timelines

Transporter's competitive bid postings will be subject to the following bid period timelines:

- (i) Transactions for a term of thirty-one days or less will be posted for a competitive bid period of no less than one hour. Any one hour posted bid period will commence at either 8:45 A.M., 11:45 A.M. or 4:45 P.M. Central Clock Time (7:45 A.M., 10:45 A.M. or 3:45 P.M. Mountain Clock Time) on a Business Day.
- (ii) Transactions for a term greater than thirty-one days but less than a year will be posted for a competitive bid period of no less than twenty-four hours.
- (iii) Transactions for a term of one year or more will be posted for a competitive bid period of no less than seventy-two hours (not including weekends and holidays).

For non-prearranged competitive bid postings, if no acceptable bids are received, Transporter may hold subsequent competitive bid posting(s) for the same capacity.

If Transporter determines that it is willing to accept Negotiated Rate bids for available capacity that is posted for competitive bid pursuant to this Section 25, the evaluation methods set forth in Section 19.3 of the General Terms and Conditions will apply.

(c) First-Come, First-Served Transactions

First-come, first-served requests for point amendments pursuant to Section 11.3 of Rate Schedule TF-1, Section 9.4 of Rate Schedule TF-2 or Section 6.2 of Rate Schedule TFL-1 or for segmentation purposes pursuant to Section 13 of Rate Schedule TF-1 and Section 8 of Rate Schedule TFL-1 may be submitted for capacity that is posted as

25. AVAILABLE CAPACITY (Continued)

available on the Unsubscribed Capacity Report. The Receipt and Delivery Point capacity shown as available on the Unsubscribed Capacity Report for first-come, first-served requests will include: (1) capacity that has been posted on Transporter's TSP Capacity Offering Report without being sold for a minimum period of six months from the date the capacity was posted pursuant to Section 25.4(a); or (2) capacity that is not required to preserve mainline capacity rights to prevent the stranding of mainline capacity and/or the loss of mainline capacity which may create operational problems, unless Transporter, on a non-discriminatory basis, agrees otherwise.

Requests for first-come, first-served point capacity cannot be submitted earlier than thirty days prior to Shipper's proposed effective date unless the request is associated with: (1) a facility modification pursuant to Sections 21 or 29 of the General Terms and Conditions; (2) a Service Agreement term extension pursuant to Section 11.7 of the General Terms and Conditions; or (3) a FERC abandonment of Transporter's facilities. If Transporter determines it can accommodate Shipper's request, then within two Business Days after the date of receiving such request, Transporter will execute the requested amendment and make the requested point change effective on the later of such execution date or Shipper's proposed effective date.

Requests for changes only to receipt and/or delivery points under existing Service Agreements which do not require additional mainline capacity are not subject to the competitive bid procedures described in this Section 25. Amendment requests under existing Service Agreements that do require additional available mainline capacity will be subject to the competitive bid process described herein, unless otherwise agreed to by Transporter.

If Transporter has determined that the requested receipt and/or delivery point change may create operational problems, such Shipper will be subject to a contract-specific flow requirement on its requested segment or must be willing to take other action that is acceptable to Transporter.

25.6 Competitive Bidding Requirements

Prior to submitting a bid, the bidder must first comply with the service request procedures set forth in Section 28 of the General Terms and Conditions and the creditworthiness requirements in Section 5.10 of the General Terms and Conditions.

25. AVAILABLE CAPACITY (Continued)

Unless otherwise specified in Transporter's competitive bid posting, all bids for capacity included in a competitive bid posting must be submitted on Transporter's Designated Site consistent with the bidding periods established under Section 25.5, and must comply with the bid requirements outlined below. A bidder may withdraw its bid at any time prior to close of the bid period; provided, however, that if the bidder is a pre-arranged Shipper, it may not withdraw its bid. A bidder who withdraws its bid may not then submit another bid with a lower economic value for the same capacity. Bidders may not submit multiple bids for the same capacity.

Competitive bids for capacity posted pursuant to Section 25.5 must specify the following, if pertinent to the capacity:

- 1. Bidder's name (which will not be publicly viewable unless and until the capacity is awarded to such bidder);
- 2. Requested Contract Demand, Storage Demand and Storage Capacity, and specific Receipt Point(s), Delivery Point(s) and/or mainline capacity and/ or Storage Facility;
- 3. Quantity of the total posted capacity that bidder is willing to accept in the event of a pro rata allocation in a potential tie-breaker process;
- 4. Reservation rate that bidder offers to pay for the requested capacity;
- 5. Requested primary term of service (including, if applicable, offered term extensions of existing Service Agreements);
- 6. Whether the bidder is an affiliate of Transporter; and
- 7. The terms and conditions of any bid contingencies, if the competitive bid posting indicated that Transporter was willing to accept bids with contingencies.

All bids must be submitted in amounts which a bidder agrees to pay to Transporter for the right to receive firm service, including undiscounted reservation surcharges, if applicable. Such bids will exclude Transporter's volumetric charges and volumetric surcharges applicable to such service included in Transporter's rates, which rate components must be paid to Transporter by the successful bidder in addition to the bid amounts.

25. AVAILABLE CAPACITY (Continued)

Unless the competitive bid posting indicated Transporter was willing to consider Negotiated Rate bids, the rate bid under this Section 25.6 must not exceed the Maximum Base Tariff Rate or be less than the Minimum Base Tariff Rate as set forth in the Statement of Rates.

For bids at the Maximum Base Tariff Rate, bidders may bid any term for which the offered capacity is available. Further, for a prearranged transaction for service commencing at a future date, competing bids at the Maximum Base Tariff Rate will be allowed for service to start either on such future date or on any date between the earliest time the capacity is available and such future date.

Maximum Base Tariff Rate bids may be submitted for any segment of the mainline capacity posted for competitive bid, subject to the availability of capacity at and between the requested physical receipt and delivery points. However, if a Shipper bids on only a segment of the mainline capacity and Transporter has indicated in the special conditions of its competitive bid posting that bids for only a segment of the mainline capacity will create an operational problem, then Shipper will be subject to a contract-specific flow requirement on its requested segment or must be willing to take other action acceptable to Transporter if such Shipper is the successful bidder.

25.7 Competitive Bid Evaluations and Award Process

(a) Best Bid Determination

Within two Business Days following the bid close date, unless otherwise stated in the competitive bid posting, Transporter will evaluate the bids received and determine the winning bid, subject (where applicable) to Section 25.7(b) and (c), Section 25.9 and Section 25.10. Bids will be evaluated based on either the net present value or highest rate of each bid for all bid periods of seventy-two hours or less, as specified in Transporter's competitive bid posting. For bid periods longer than seventy-two hours, bids will be evaluated based on either the net present value, highest rate or the alternative criteria for determining incremental economic value that is specified by Transporter in its competitive bid posting. For prearranged transactions starting a year or more after the underlying capacity becomes available, the incremental economic value of bids will be determined on a net present value basis. For bid evaluation purposes, the value of a Negotiated Rate bid will be capped at the value of such bid at Transporter's applicable Maximum Base Tariff Rate. A bid to pay the Maximum Base Tariff Rate as it may vary from time to time for a given term will be deemed superior to a bid to pay a specified dollar rate which is equal to the Maximum Base Tariff Rate.

25. AVAILABLE CAPACITY (Continued)

For purposes of this Section 25, the incremental economic value of a bid is (1) the total net present value of the reservation charge bid for transportation capacity (including bids to extend the term of existing Service Agreements); (2) the demand and capacity charge bid for storage capacity; or (3) the value determined using either the highest rate or the alternative criteria specified by Transporter in its competitive bid posting. The net present value shall be computed from the monthly demand and capacity, or reservation revenues to be received over the term of the Service Agreement, using the discounted cash flow rate of return methodology, with the rate of discounting, as updated from time to time, to be equal to the rate set forth in Section 154.501(d) of the Commission's regulations at the time the capacity is posted for bid, unless a different discount rate is specified in Transporter's competitive bid posting for available capacity. Bids for capacity under existing Service Agreements will be deemed to have no incremental economic value for award purposes, except to the extent bidder proposes to extend the term of its existing Service Agreement.

(b) Contingent Bids

If the best bid is a contingent bid, Transporter will notify the bidder making the best bid within five Business Days following the bid close date. That bidder will have five Business Days following such notification to satisfy or waive the contingency for bids of six months or longer and one Business Day following such notification to satisfy or waive the contingency for bids for less than six months, or Transporter may disregard such bid.

(c) Tie Break Process

Transporter will break ties and determine the winning bid(s) by providing each of the tying bidders an opportunity to improve their bids by submitting closed bids via electronic means as specified in Transporter's competitive bid posting which may include facsimile, email, or via Transporter's Designated Site, within 24 hours of notification by Transporter. If a tie still exists and the prearranged Shipper has declined to match the tied bids, if applicable, Transporter will allocate the available capacity on a pro rata basis to each tying bidder that has indicated in its bid a willingness to accept a proportionate share of such capacity.

(d) Match Process

For a competitive bid posting with a prearranged transaction, if the highest bid(s) determined by Transporter in the bid evaluation and tie-breaker processes exceeds the economic value of the prearranged transaction, the prearranged Shipper will be provided a one-time

25. AVAILABLE CAPACITY (Continued)

opportunity to match the highest acceptable bid(s) by posting such election on Transporter's Designated Site. For service with a term of less than one year, the posting by the prearranged Shipper must be made within two hours of Transporter's notification to the prearranged Shipper of the best bid(s) and for service with a term of one year or more, the posting by the prearranged Shipper must be made within forty-eight hours, excluding weekends and holidays, of such notification.

If two or more Maximum Base Tariff Rate bids are received for discrete portions of prearranged transaction capacity, the prearranged Shipper will be offered the right to match such bids, one by one, starting with the lowest incremental economic value bid. Accordingly, if the prearranged Shipper declines to match a lower incremental economic value bid for a discrete portion of the prearranged capacity, it will not be offered the opportunity to match any higher incremental economic value bids.

Whether or not the competitive bid posting includes a prearranged transaction, if an existing Shipper holds a Right of First Refusal pursuant to Section 25.9 for such capacity, such existing Shipper will have the final opportunity to match the best bid(s) according to the Section 25.9 procedures.

(e) Award Process

Capacity offered in a competitive bid posting will be awarded to Shippers based on the incremental economic value of the bids, from highest economic value to lowest, until all of the capacity has been awarded or until all valid bids have been accepted, subject to a prearranged Shipper's right to match and/or an existing Shipper's Right of First Refusal under Section 25.9.

For competitive bid postings with a prearranged transaction, if no acceptable higher value bids are submitted, Transporter will award the capacity to the prearranged Shipper within one hour after the close of the bid period, subject to an existing Shipper's Right of First Refusal under Section 25.9.

For any capacity offered under this Section 25, the successful bidder(s) and Transporter will electronically execute a new Service Agreement or amendment prior to the commencement of service which confirms the terms of the accepted bid(s) for the available capacity. The Shipper's execution of the Service Agreement occurs automatically upon posting of its bid and Transporter's execution occurs upon award.

25. AVAILABLE CAPACITY (Continued)

25.8 Reserved Capacity

Transporter may elect to reserve available capacity for future expansion projects. Capacity may be reserved for up to one year prior to Transporter filing for either Commission approval to initiate the National Environmental Policy Act pre-filing review for a proposed expansion or certificate approval for construction of a proposed expansion and thereafter until such expansion project is placed into service. If Transporter elects to reserve capacity for an expansion project, it will notify Shippers of its intent on the earlier of the following dates:

- (a) the date on which expiring or terminating capacity is required to be posted in accordance with Section 25,
- (b) the date on which Transporter executes a letter of intent with a prospective expansion Shipper which requires the reservation of capacity, or
- (c) the date on which Transporter begins an open season for the expansion project.

If Transporter enters into a Service Agreement for available capacity for service starting at a future date or reserves available capacity for a future expansion project, Transporter will post such capacity on its Unsubscribed Capacity Report for use on an interim basis, but will indicate in its posting that such capacity is reserved for future use. Shippers that acquire such capacity on an interim basis will not be eligible for a Right of First Refusal under Section 25.9.

Any capacity reserved for an expansion project that does not go forward for any reason shall be posted on Transporter's Unsubscribed Capacity Report within 30 days of the date when such capacity becomes available, subject to prior interim commitments of such capacity.

25.9 Right of First Refusal

(a) A Shipper under an expiring or terminating Part 284 firm Service Agreement may avoid pre-granted abandonment of service in accordance with the right of first refusal provisions of Section 284.221(d) of the Commission's regulations, provided that: (1) the Service Agreement is expiring under its own terms or Shipper is not the party providing termination notice under an evergreen provision; (2) service is not being provided on an interim basis pursuant to Section 25.8, and (3) service is not being provided on an interim basis using capacity reserved for future service under Section 25.5. To be eligible for a

25. AVAILABLE CAPACITY (Continued)

right of first refusal, a Shipper must be receiving service (1) at the Maximum Base Tariff Rate for a term of twelve consecutive months or longer (or for one year or longer where service is not available for twelve consecutive months); or (2) at any rate for a term of one year or longer when such service is provided under a Service Agreement executed prior to March 27, 2000.

A Shipper eligible for a right of first refusal can continue receiving the contracted service by matching the incremental economic value of competing bid(s) in accordance with the procedures described below; provided, however, that in no event will such Shipper be required to match a bid, including a Negotiated Rate bid, that exceeds the Maximum Base Tariff Rate in order to avoid pre-granted abandonment of service.

(b) Transporter will notify the existing Shipper of the best non-contingent bid(s), or that no bids from creditworthy bidders were received, within one business day after determining the same. The existing Shipper will have a right of first refusal to match the best bid received for such capacity for two business days after receiving the notification of the best bid for bids of six months or longer duration and one business day after receiving notification of the best bid for bids of less than six months.

If the best bid is a Negotiated Rate bid, the existing Shipper can continue receiving the contracted service as set forth in Section 25.9(a) above (1) by matching the highest Negotiated Rate offer submitted by another Shipper that meets or exceeds the lowest rate Transporter is willing to accept for such service, or (2) by paying a Recourse Rate or discounted Recourse Rate per Dth that is equivalent to the highest Negotiated Rate offer submitted by another Shipper that meets or exceeds the lowest rate Transporter is willing to accept for such service. Negotiated Rate offers will be evaluated in accordance with Section 19.3 of the General Terms and Conditions.

(c) If the best bid as determined by Transporter is for less than the total capacity subject to the right of first refusal, the existing Shipper is required to match the incremental economic value for the amount of capacity to which the bid applies in order to retain the right to service for that portion of the capacity. The existing Shipper also may choose to retain only a portion of its capacity rights by matching the incremental economic value of any bid(s) submitted for that portion. If an existing Shipper wishes to retain only a portion of its capacity, such Shipper's aggregate MDQs and aggregate MDDOs must be retained by the same percentage as the Contract Demand is retained. Shipper may specify the MDQs it wishes to retain at each Receipt Point and the MDDOs it wishes to retain at each Delivery Point.

25. AVAILABLE CAPACITY (Continued)

- (d) If an existing Shipper chooses not to match the incremental economic value of the best bid(s) as determined by Transporter, capacity will be awarded to the selected bidder(s), and the existing Shipper's right of first refusal will expire.
- (e) If no creditworthy bids are submitted for any portion of the capacity, an existing Shipper will be entitled to continue its existing service for any portion of the capacity, for any term desired, provided Transporter and Shipper agree to mutually acceptable rates for the service at a level within the posted maximum and minimum tariff rates for the applicable service or at a mutually acceptable Negotiated Rate for the applicable service. If Transporter and Shipper fail to agree on rates within thirty days of the bid close date, Shipper's right of first refusal will be deemed to have terminated. If an existing Shipper wishes to retain only a portion of its capacity and Transporter has indicated in the special conditions of its posting that it has determined that partial use or segmentation of the posted capacity may create operational problems (such as reducing the existing Shipper's ability to respond to an OFO pursuant to Section 14.15 of the General Terms and Conditions), Shipper will be subject to a contract-specific flow requirement on its retained segment or must be willing to take other action that is acceptable to Transporter.

25.10 No Discount Requirement

Notwithstanding anything to the contrary set forth in this Section 25, Transporter will not be required to provide service at any rate less than the Maximum Base Tariff Rate as set forth in the Statement of Rates in the Tariff, as such rate may vary from time to time.

- 26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS
 - 26.1 NAESB WGQ Business Practice Standards.

Transporter has adopted all of the business practices and electronic communications standards, NAESB Wholesale Gas Quadrant (WGQ) Version 3.2, which are required by the Commission in 18 CFR, Section 284.12 (a), as amended from time to time, in accordance with Order No. 587, $\underline{\text{et}}$ al.

NAESB has granted to Transporter a limited self executing waiver to reproduce all or portions of the copyright protected language from the North American Energy Standards Board (NAESB) Standards listed below. With respect to each reproduced standard, Transporter incorporates the following: © 1996 - 2017 NAESB, all rights reserved.

Transporter specifically incorporates by reference the following NAESB WGQ Version 3.2 standards, definitions, and data sets, by reference:

Additional Standards:

```
General
```

```
Definition:
 0.2.5
  Standards:
  0.3.1, 0.3.2, 0.3.16, 0.3.17
Creditworthiness
  Standards:
  0.3.3, 0.3.4, 0.3.5, 0.3.6, 0.3.7, 0.3.8, 0.3.9, 0.3.10
Gas/Electric Operational Communications
  Definitions:
  0.2.1, 0.2.2, 0.2.3, 0.2.4
  Standards:
  0.3.11, 0.3.12, 0.3.13, 0.3.14, 0.3.15
Operating Capacity and Unsubscribed
  Standards:
  0.3.18, 0.3.20, 0.3.21, 0.3.22
  Data Sets:
  0.4.2, 0.4.3
Location Data Download:
  Standards:
```

0.3.23, 0.3.24, 0.3.25, 0.3.26, 0.3.27, 0.3.28, 0.3.29

```
26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)
     Location Data Download (Continued):
       Data Set:
       0.4.4
      Storage Information
        Data Set:
        0.4.1
     Nominations Related Standards
        Definitions:
        1.2.1, 1.2.2, 1.2.3, 1.2.5, 1.2.8, 1.2.9, 1.2.10, 1.2.11, 1.2.13,
        1.2.14, 1.2.15, 1.2.16, 1.2.17, 1.2.18, 1.2.19
        Standards:
        1.3.3, 1.3.4, 1.3.5, 1.3.6, 1.3.7, 1.3.8, 1.3.9, 1.3.11, 1.3.13,
        1.3.15, 1.3.16, 1.3.17, 1.3.18, 1.3.19, 1.3.20, 1.3.21, 1.3.22,
        1.3.23, 1.3.24, 1.3.25, 1.3.26, 1.3.27, 1.3.28, 1.3.29, 1.3.30,
        1.3.31, 1.3.32, 1.3.33, 1.3.34, 1.3.35, 1.3.36, 1.3.37, 1.3.38,
        1.3.39, 1.3.40, 1.3.41, 1.3.42, 1.3.43, 1.3.44, 1.3.45, 1.3.46,
        1.3.48, 1.3.51, 1.3.53, 1.3.55, 1.3.56, 1.3.58, 1.3.62, 1.3.64,
        1.3.65, 1.3.66, 1.3.67, 1.3.68, 1.3.69, 1.3.70, 1.3.71, 1.3.72,
        1.3.73, 1.3.74, 1.3.75, 1.3.76, 1.3.77, 1.3.79, 1.3.80, 1.3.81,
        1.3.82
        Data Sets:
        1.4.1, 1.4.2, 1.4.3, 1.4.4, 1.4.5, 1.4.6, 1.4.7
      Flowing Gas Related Standards
        Definitions:
        2.2.3, 2.2.4, 2.2.5
        Standards:
        2.3.1, 2.3.2, 2.3.3, 2.3.4, 2.3.5, 2.3.6, 2.3.7, 2.3.8, 2.3.10,
        2.3.11, 2.3.12, 2.3.13, 2.3.14, 2.3.15, 2.3.16, 2.3.17, 2.3.18,
        2.3.19, 2.3.20, 2.3.21, 2.3.22, 2.3.23, 2.3.25, 2.3.26, 2.3.27,
        2.3.28, 2.3.29, 2.3.30, 2.3.31, 2.3.32, 2.3.40, 2.3.41, 2.3.42,
        2.3.43, 2.3.44, 2.3.45, 2.3.46, 2.3.47, 2.3.48, 2.3.50, 2.3.51,
        2.3.52, 2.3.53, 2.3.54, 2.3.55, 2.3.56, 2.3.57, 2.3.58, 2.3.59,
        2.3.60, 2.3.61, 2.3.62, 2.3.63, 2.3.64, 2.3.65, 2.3.66
        Data Sets:
        2.4.1, 2.4.2, 2.4.3, 2.4.4, 2.4.5, 2.4.6, 2.4.7, 2.4.8, 2.4.9,
        2.4.10, 2.4.11, 2.4.17, 2.4.18
```

26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)

Invoicing Related Standards

```
Standards:
```

3.3.3, 3.3.4, 3.3.5, 3.3.6, 3.3.7, 3.3.8, 3.3.9, 3.3.10, 3.3.11, 3.3.12, 3.3.13, 3.3.14, 3.3.15, 3.3.16, 3.3.17, 3.3.18, 3.3.19, 3.3.21, 3.3.22, 3.3.23, 3.3.24, 3.3.25, 3.3.26, 3.3.27

Data Sets:

3.4.1, 3.4.2, 3.4.3, 3.4.4

Quadrant Electronic Delivery Mechanism Related Standards Definitions:

4.2.1, 4.2.2, 4.2.3, 4.2.4, 4.2.5, 4.2.6, 4.2.7, 4.2.8, 4.2.9, 4.2.10, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 4.2.15, 4.2.16, 4.2.17, 4.2.18, 4.2.19, 4.2.20

Standards:

4.3.1, 4.3.2, 4.3.3, 4.3.16, 4.3.17, 4.3.18, 4.3.20, 4.3.22, 4.3.23, 4.3.24, 4.3.25, 4.3.26, 4.3.27. 4.3.28, 4.3.30, 4.3.31, 4.3.32, 4.3.33, 4.3.34, 4.3.35, 4.3.36, 4.3.38, 4.3.40, 4.3.41, 4.3.42, 4.3.43, 4.3.44, 4.3.45, 4.3.46, 4.3.47, 4.3.48, 4.3.49, 4.3.50, 4.3.52, 4.3.53, 4.3.54, 4.3.55, 4.3.57, 4.3.58, 4.3.60, , 4.3.62, 4.3.66, 4.3.67, 4.3.68, 4.3.69, 4.3.72, 4.3.75, 4.3.78, 4.3.79, 4.3.80, 4.3.81, 4.3.82, , 4.3.84, 4.3.85, 4.3.86, 4.3.87, 4.3.89, 4.3.90, 4.3.91, 4.3.92, 4.3.93, 4.3.94, 4.3.95, 4.3.96, 4.3.97, 4.3.98, 4.3.99, 4.3.100, 4.3.101, 4.3.102, 4.3.103, 4.3.104, 4.3.105, 4.3.106, 4.3.107, 4.3.108, 4.3.109, 4.3.110

Capacity Release Related Standards

Definitions:

5.2.1, 5.2.2, 5.2.3, 5.2.4, 5.2.5

Standards:

5.3.1, 5.3.3, 5.3.4, 5.3.5, 5.3.7, 5.3.8, 5.3.9, 5.3.10, 5.3.11, 5.3.12, 5.3.13, 5.3.14, 5.3.15, 5.3.16, 5.3.18, 5.3.19, 5.3.20, 5.3.21, 5.3.22, 5.3.23, 5.3.24, 5.3.25, 5.3.26, 5.3.28, 5.3.29, 5.3.31, 5.3.32, 5.3.33, 5.3.34, 5.3.35, 5.3.36, 5.3.37, 5.3.38, 5.3.39, 5.3.40, 5.3.41, 5.3.42, 5.3.46, 5.3.47, 5.3.48, 5.3.49, 5.3.50, 5.3.51, 5.3.52, 5.3.53, 5.3.55, 5.3.56, 5.3.57, 5.3.58, 5.3.59, 5.3.60, 5.3.62, 5.3.62a, 5.3.63, 5.3.64, 5.3.65, 5.3.66, 5.3.67, 5.3.68, 5.3.69, 5.3.70, 5.3.71, 5.3.72, 5.3.73

26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)

Capacity Release Related Standards (Continued) Data Sets:

5.4.14, 5.4.15, 5.4.16, 5.4.17, 5.4.20, 5.4.21, 5.4.22, 5.4.23, 5.4.24, 5.4.25, 5.4.26, 5.4.27

Internet Electronic Transport Related Standards

Definitions:

10.2.1, 10.2.2, 10.2.3, 10.2.4, 10.2.5, 10.2.6, 10.2.7, 10.2.8,

10.2.9, 10.2.10, 10.2.11, 10.2.12, 10.2.13, 10.2.14, 10.2.15,

10.2.16, 10.2.17, 10.2.18, 10.2.19, 10.2.20, 10.2.21, 10.2.22,

10.2.23, 10.2.24, 10.2.25, 10.2.26, 10.2.27, 10.2.28, 10.2.29,

10.2.30, 10.2.31, 10.2.32, 10.2.33, 10.2.34, 10.2.35, 10.2.36,

10.2.37, 10.2.38, 10.2.39

Internet Electronic Transport Related Standards

Standards:

10.3.1, 10.3.3, 10.3.4, 10.3.5, 10.3.6, 10.3.7, 10.3.8, 10.3.9,

10.3.10, 10.3.11, 10.3.12, 10.3.14, 10.3.15, 10.3.16, 10.3.17,

10.3.18, 10.3.19, 10.3.20, 10.3.21, 10.3.22, 10.3.23, 10.3.24,

10.3.25, 10.3.26, 10.3.27, 10.3.28, 10.3.29

Standards for which Waiver or Extension of Time to Comply have been granted:

Waiver: None

Extension of Time:

0.4.1, 0.4.2, 0.4.3

2.4.1, 2.4.2, 2.4.3, 2.4.4, 2.4.5, 2.4.6, 2.4.7, 2.4.8, 2.4.9,

2.4.10, 2.4.11, 2.4.17, 2.4.18

3.4.1, 3.4.2, 3.4.3, 3.4.4

5.4.17, 5.4.20, 5.4.21, 5.4.22, 5.4.23

26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)

Standards not Incorporated by Reference and their Location in the Tariff:

NAESB Standard	GT&C Tariff Section
1.2.4	1
1.2.6	1
1.2.12	1
1.3.1	1
1.3.2	14.1
1.3.14	1
2.2.1	1
2.2.2	1
2.3.9	1
3.2.1	1
5.3.2	22.9
5.3.44	22.11
5.3.45	22.11
5.3.54	22.11

26.2 EDI/EDM Extension of Time. Following written receipt of a bona fide request to implement a NAESB electronic data interchange or electronic delivery mechanism data set, Northwest will within 90 (ninety) days of the request make such data set(s) available for usage. Such extension of time to implement data sets as discussed in this subsection shall be subject to a currently effective extension of time granted by the Commission.

- 26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)
 - 26.3 Electronic Communications. Electronic communication capabilities will be available on a nondiscriminatory basis to any party.

Transporter will provide timely information pertaining to arranging for transportation and storage services, including informing all interested Shippers or other parties about the availability of capacity at Receipt Points, on the mainline, at delivery points and in storage facilities, and whether the capacity is available from the pipeline directly or through capacity release. In addition, Transporter will report design day displacement capacity needed at various constraint points. Transporter will also maintain and provide access to bidding information pertaining to released capacity, and Order No. 497 compliance information. Transporter will update and maintain this information on its Designated Site in accordance with 18 CFR Section 284.12(b)(3). Any party may access this information on Transporter's Designated Site.

There is no fee associated with the use of Transporter's Designated Site.

- 26.4 Information to be Posted. Via Transporter's Designated Site, Transporter will provide to all parties information relevant to the availability of services in accordance with 18 CFR Section 284.13.
- 26.5 Transporter will also support the NAESB approved EDI Standard Business Practice Data Sets referenced below, where applicable, subject to the execution of a Trading Partner Agreement between Transporter and requesting party:

NAESB WGQ Standards:

0.4.1

1.4.1, 1.4.2, 1.4.3, 1.4.4, 1.4.5, 1.4.6, 1.4.7

2.4.1, 2.4.2, 2.4.3, 2.4.4 2.4.5

2.4.6 through 2.4.11, 2.4.17, 2.4.18

3.4.1 through 3.4.4

5.4.14 through 5.4.17

5.4.20, 5.4.21, 5.4.22, 5.4.23, 5.4.24

5.4.25, 5.4.26, 5.4.27

- 26. NAESB WGQ STANDARDS AND ELECTRONIC COMMUNICATIONS (Continued)
 - 26.6 Electronic Communication. Electronic communications between Transporter and Shipper or other parties will be binding and enforceable in the same manner and to the same extent as written communications, and evidence of electronic communications, if introduced on paper, will be admissible to the same extent as other business records originated in written form. Any provisions of this Tariff requiring that certain matters between Transporter and Shipper be written or in writing are satisfied by Transporter and Shipper utilizing electronic communications through Transporter's Designated Site or through other mutually agreed forms of electronic communications.
 - 26.7 Disclaimer. Transporter will not be liable for any damages caused by the incompleteness or inaccuracy of any information posted to Transporter's Designated Site except to the extent such inaccuracy or incompleteness is shown to be the result of negligence or misfeasance of Transporter. Transporter will not be liable for errors in any subsequent re-posting by a third-party vendor.
 - 26.8 Electronic Trading Partner Communications. Transporter will support NAESB WGQ approved EDI communications through Transporter's Designated Site.

NAESB WGQ Nominations Related Standard 1.2.2 will apply.

GENERAL TERMS AND CONDITIONS (Continued)
RESERVED FOR FUTURE USE

28. SERVICE REQUEST PROCEDURES

28.1 Availability. Transporter maintains a Designated Site to provide the information or services required by applicable Commission orders or described in Transporter's FERC Gas Tariff. Transporter's Designated Site is comprised of two internet websites: a non-secure (public) informational postings site and a secure (non-public) customer activities site (Northwest Passage). Information of a general nature is included in the public informational postings site. Confidential customer and interconnect data are accessible only through the non-public customer activities site. Northwest Passage is accessible on Transporter's Designated Site.

Northwest Passage shall be available twenty-four (24) hours per day, subject to maintenance and reasonable downtime. Northwest Passage shall be available on a nondiscriminatory basis to any entity provided that such entity submits a complete request in accordance with Transporter's procedures set forth in 28.2 below. Transporter reserves the right, at its sole discretion, to provide enhancements to Northwest Passage or to discontinue information or services not required by Commission order or otherwise described in Transporter's FERC Gas Tariff.

28.2 Access to Northwest Passage. To gain access to Northwest Passage, a Service Requester must complete and submit a Business Associate Information form to Transporter. The Business Associate Information form is available in the Agreements/Forms area of Transporter's Designated Site.

On the Business Associate Information form, the Service Requester must designate one person as its security coordinator. Upon receipt of the completed Business Associate Information form, Transporter will create a unique Service Requester ID for the Service Requester and assign a userID and temporary password to the Service Requester's security coordinator. The temporary password will be provided electronically to the designated security coordinator and will expire upon the security coordinator's initial login to Northwest Passage, at which time the security coordinator will designate his or her own password. By logging in to Northwest Passage with the temporary password, the Service Requester and its security coordinator agree to the provisions of this Section 28 and any other applicable provisions of Transporter's FERC Gas Tariff, as amended or superseded from time to time.

Service Requester agrees to immediately notify Transporter upon any material change to the information provided on the Business Associate Information form, and to provide Transporter any documentation required to verify such material change (e.g., documentation verifying a legal name change, etc.).

28. SERVICE REQUEST PROCEDURES (Continued)

28.3 Role and Responsibilities of the Service Requester's security coordinator. The security coordinator will perform certain administrative functions in Northwest Passage on behalf of the Service Requester, including (1) identifying user(s) who are duly authorized to access the Service Requester's account and/or transact business on Service Requester's behalf on Northwest Passage ("Authorized Person(s)''), (2) setting up userIDs on Northwest Passage for use by Authorized Person(s), (3) maintaining account information for Authorized Person(s), (4) promptly suspending userID(s) when an Authorized Person is no longer an employee of the Service Requester or is no longer authorized to transact business for that Service Requester, (5) managing or modifying security rights for Authorized Person(s), and (6) ensuring that Transporter's userID and password rules, as detailed in this Section 28, are followed. Transporter shall be entitled to rely upon the security coordinator's performance of these administrative functions in designating Authorized Person(s), including that Authorized Person(s) may (1) transmit information to Transporter via Northwest Passage and/or (2) view the Service Requester's information posted on Northwest Passage in accordance with the security rights granted by the security coordinator.

Either a security coordinator or Authorized Person may instruct Transporter in writing or via email to modify its security coordinator designation. Such instruction shall supersede in its entirety any previously submitted security coordinator designation for that Service Requester. The Service Requester shall be solely responsible for any unauthorized actions due to failure to notify Transporter to modify its security coordinator designation.

28.4 Authorized Person(s). Transporter will provide each Authorized Person with a unique userID and a verification code that will be required to create a password upon first login. Each Authorized Person who wishes to gain access to Northwest Passage must complete and submit a Logon ID Request form. The Logon ID Request form is available in the Agreements/Forms area of Transporter's Designated Site. Alternatively, the security coordinator for Service Requester may submit a unique userID request in Northwest Passage. Each Authorized Person, by logging in to Northwest Passage with his or her userID, agrees to the provisions of Section 28 of Transporter's FERC Gas Tariff and any other applicable provisions of Transporter's FERC Gas Tariff, as amended or superseded from time to time.

28. SERVICE REQUEST PROCEDURES (Continued)

- 28.5 Agent(s). Transporter agrees that it will recognize the appointment of an Agent by Service Requester to access and perform functions in Northwest Passage on Service Requester's behalf. However, Transporter shall only recognize such appointment after Service Requester and the Agent have completed and provided to Transporter an Agency Delegation form as made available on Transporter's Designated Site and Agent, who must also comply with Sections 28.2, 28.3, and 28.4 with respect to its own security coordinator and Authorized Person(s), has completed and provided to Transporter its Business Associate Information form specifying its security coordinator. Thereafter, the Agent will be considered an appointed agent of Service Requester and will be treated as acting on behalf of Service Requester as described herein. Service Requester may cancel the appointment of an Agent by following the procedures specified on the Agency Delegation form and/or name a successor Agent by providing an updated Agency Delegation form. Service Requester represents and acknowledges that any Agent recognized by Transporter has legal authority to act on behalf of Service Requester in performing any functions for which the Agent is authorized, as identified on the Agency Delegation form, and that Transporter is fully entitled to rely upon, and is fully protected in relying upon and acting in accordance with, such representation and acknowledgment.
- 28.6 Requests for Service. All requests for Service Agreements and amendments to existing Service Agreements shall be made to Transporter electronically in Northwest Passage. Each request shall be considered effective on the date the information required in Northwest Passage is entered in Northwest Passage.
- 28.7 Trading Partner Agreement. To conduct transactions using EDI communications, a party must execute a Trading Partner Agreement. A standard NAESB WGQ EDI Trading Partner Agreement will be available in the Agreements/Forms area of Transporter's Designated Site or may be obtained from Transporter's commercial services department.
- 28.8 Requests for Amendments. Amendments to receipt or delivery points or associated volumes under Rate Schedules TF-1 and TFL-1 or amendments to delivery points under Rate Schedule TF-2 shall be considered requests for amended service under Section 28.6 and shall be documented in accordance with the procedures set forth in such rate schedules.

Third Revised Sheet No. 284 Superseding Second Revised Sheet No. 284

GENERAL TERMS AND CONDITIONS (Continued)

28. SERVICE REQUEST PROCEDURES (Continued)

28.9 Compliance with 18 CFR Section 284.102. If service is requested pursuant to Section 284.102 of the Commission's regulations "on behalf of" a local distribution company or an intrastate pipeline, Shipper must provide an original letter signed by an official of such "on behalf of" party on its letterhead. This letter must warrant that such "on behalf of" party is either an intrastate pipeline as defined by Section 2(16) of the Natural Gas Policy Act of 1978 or a local distribution company as defined by Section 2(17) of the Natural Gas Policy Act of 1978. It must also state that such "on behalf of" party has authorized the Shipper to enter into the requested Service Agreement in order to have Section 311(a) transportation service rendered on behalf of such "on behalf of" party by Transporter and that (1) the intrastate pipeline or local distribution company will have physical custody of and transport the Natural Gas at some point; or (2) the intrastate pipeline or local distribution company will hold title to the Natural Gas at some point, which may occur prior to, during, or after the time that the Natural Gas is being transported by the interstate pipeline, for a purpose related to its status and functions as an intrastate pipeline or its status and functions as a local distribution company; or (3) the Natural Gas will be delivered at some point to a customer that either is located in a local distribution company's service area or is physically able to receive direct deliveries of Natural Gas from an intrastate pipeline, and that the local distribution company or intrastate pipeline certifies that it is on its behalf that the interstate pipeline is providing transportation service.

28. SERVICE REQUEST PROCEDURES (continued)

28.10 Establishment of Credit. Before Service Requester may transact business electronically on the Designated Site, Service Requester must establish a credit limit satisfactory to Transporter pursuant to Section 5.10 of the General Terms and Conditions. A credit application form is available in the Agreements/Forms area of Transporter's Designated Site.

28.11 Security. Transporter reserves the right to terminate any userID that has been inactive for more than ninety (90) calendar days. Transporter further reserves the right to deactivate userIDs if the Service Requester or its security coordinator, Authorized Person(s), or Agent(s) breaches any provision in this Section 28. Such deactivation shall only be implemented following ten (10) days' prior notice by Transporter to Service Requester of such intended action and the reason therefore to provide Service Requester a reasonable time to correct its conduct; however, if the conduct results in a serious breach which may immediately jeopardize the security, confidentiality, or viable operation of Northwest Passage, Transporter may immediately deactivate userIDs.

28.12 Confidentiality. Service Requester will keep, and will cause its security coordinator, Authorized Person(s), and Agent(s) to keep, all userIDs and passwords confidential and not disclose the same, either separately or combined. Service Requester agrees that only its security coordinator, Authorized Person(s), and Agent(s) will be given unique userIDs and passwords for that Service Requester, and that only its security coordinator, Authorized Person(s), and Agent(s) will access Northwest Passage on Service Requester's behalf. Likewise, Transporter will keep, and will cause its authorized employees to keep, Service Requester's userIDs and temporary passwords confidential and will not disclose the same, either separately or combined, to any person or entity without authority to access Northwest Passage on Transporter's behalf. Service Requester will immediately notify Transporter if it becomes aware that a security breach has or may have occurred. Any use of Northwest Passage by any person using the userIDs and/or passwords of Service Requester's security coordinator, Authorized Person(s), or Agent(s) shall be deemed to be Service Requester's use and Service Requester accepts full responsibility and liability for such use, whether authorized or unauthorized by Service Requester, except to the extent such use has resulted from the negligence or willful misconduct of Transporter.

28. SERVICE REQUEST PROCEDURES (Continued)

28.13 Indemnification. Except to the extent resulting from the negligence or willful misconduct of Transporter, Service Requester agrees to defend, indemnify and hold harmless Transporter and its officers, directors, employees, agents and representatives from and against all claims, demands, damages, losses, costs and expenses (including court costs and reasonable attorneys' fees) of Transporter, Service Requester, and others arising out of any: (i) failure to keep confidential the userIDs and passwords of Service Requester's security coordinator, Authorized Person(s), or Agent(s); (ii) use of such userIDs or passwords; (iii) use of Service Requester's Northwest Passage account; (iv) failure by Service Requester's security coordinator, Authorized Person(s), or Agent(s) to perform any affirmative duty required by use of the interactive function of Northwest Passage; and/or (v) breach of this Section 28 by Service Requester or its security coordinator, Authorized Person(s), or Agent(s).

28.14 Limitation of Liability. The following limitation of liability is in addition to that in Section 7.3 of the General Terms and Conditions. Except to the extent of Transporter's negligence or willful misconduct, Transporter may rely, without liability upon the actions or inactions of persons utilizing the userIDs or passwords of Service Requester's security coordinator, Authorized Person(s), or Agent(s). Transporter shall not be responsible for any omission or failure of any function accessed through Northwest Passage if such omission or failure is caused by or related to any errors in transmission of data to or from Transporter's computer systems, power failures, failure of any computer systems or backup systems, or any other event beyond the reasonable control of Transporter. If Service Requester requests and receives assistance from Transporter's representatives, such assistance will be at the Service Requester's sole risk and Transporter will not have any responsibility or liability arising therefrom, except to the extent of Transporter's negligence or willful misconduct. Except to the extent of Transporter's negligence or willful misconduct, Transporter will not be responsible for any faulty retrieval of information, inability to retrieve information, or loss of information from Northwest Passage.

28.15 Procedures. Transporter and Service Requester and Service Requester's security coordinator, Authorized Person(s), and Agent(s), if applicable, agree to follow all procedures regarding Northwest Passage as such procedures may be established and announced by Transporter from time to time.

29. MAINLINE PATH ALTERATIONS

- 29.1 General. This Section 29 applies to all facilities required for altering the Mainline Transportation System path of all or part of a Shipper's existing Transportation Contract Demand ("Mainline Path Alteration"). This Section 29 does not apply to facilities required to accommodate the Transportation of new Transportation Contract Demand on Transporter's Mainline Transportation System.
- 29.2 Request for Mainline Path Alteration. A Shipper desiring a Mainline Path Alteration for all or part of its existing Transportation Contract Demand ("Alteration Shipper") will deliver a written request to Transporter that includes the following information: desired receipt and/or delivery point change(s), desired capacity needs for the new path, pressure requirements at desired delivery point(s), and such other information as Transporter may deem relevant. Transporter will respond within 60 days after the foregoing information has been received with either an answer to the request or an estimate of the additional time Transporter will need to provide an answer to the request.
- 29.3 Conditions for Mainline Path Alteration. Transporter is not obligated to perform any Mainline Path Alteration, but may agree to do so if:
 - (a) It will not adversely affect the Mainline Transportation System's operations;
 - (b) Neither it nor the resulting Transportation service will result in diminished services to Transporter's existing customers;
 - (c) It will not cause Transporter to be in violation of any environmental or safety laws or regulations;
 - (d) It will not cause Transporter to be in violation of its right-of-way agreements or any other contractual obligation; and
 - (e) Transporter can acquire all the necessary regulatory approvals, permits and real property rights for Transporter to construct, own, operate and maintain the Mainline Path Alteration.

29. MAINLINE PATH ALTERATIONS (Continued)

29.4 Facilities Agreement.

- (a) If: (i) a Mainline Path Alteration satisfies the conditions set forth in Section 29.3, (ii) Transporter is willing to proceed, and (iii) Transporter and Alteration Shipper can agree upon the terms governing Transporter's design, construction, ownership, operation and maintenance of the Mainline Path Alteration facilities, then such terms will be set forth in a facilities agreement between Transporter and the Alteration Shipper.
- (b) The facilities agreement and Exhibit C or Exhibit D, as applicable, of the Alteration Shipper's Service Agreement will provide that the Alteration Shipper will pay to Transporter over an agreed period of time an incremental facilities charge. Such charge will include, as appropriate, all Transporter's costs associated with the Mainline Path Alteration, including without limitation operating and maintenance expenses, administrative and general expenses, return on equity, return on debt, income taxes, other taxes, depreciation and net negative salvage ("Cost Requirement").
- (c) Notwithstanding Section 29.4(b), Transporter and the Alteration Shipper may agree to a different payment method, as appropriate, for Transporter to design, construct, own, operate and maintain the Mainline Path Alteration facilities if Transporter agrees to do so.
- (d) For any Mainline Path Alteration that Transporter will design and construct without receiving prepayment, Transporter and Alteration Shipper may agree to credit provisions that apply before and after the in-service date of the Mainline Path Alteration. Transporter may require a credit obligation in an amount up to the full Cost Requirement of the Mainline Path Alteration, but not to exceed the Alteration Shipper's proportionate share of such Cost Requirement. Each year, Transporter will reduce the amount of such credit obligation as necessary to approximate the remaining Cost Requirement payable by the Alteration Shipper under the facilities agreement.
- (e) Within 10 days after Transporter and the Alteration Shipper have executed the facilities agreement for the Mainline Path Alteration, Transporter will post notice of the Mainline Path Alteration on its Designated Site for a period of 10 days. If within 30 days after the posting the Alteration Shipper is able

29. MAINLINE PATH ALTERATIONS (Continued)

to acquire capacity that renders all or part of the Mainline Path Alteration unnecessary, then the Alteration Shipper may terminate the facilities agreement by providing written notice of termination to Transporter and paying all costs incurred by Transporter pursuant to the facilities agreement.

- (f) Subject to Transporter's prior written approval, which will not be unreasonably withheld, the Alteration Shipper's payment obligations under the facilities agreement and Exhibit C of the Alteration Shipper's Service Agreement may be assigned to a superseding or replacement Service Agreement(s) ("Successor").
- (g) Notwithstanding anything in this Section 29.4 to the contrary, Transporter may agree on a not unduly discriminatory basis in the facilities agreement to pay for all or part of the cost to design, construct, own, operate and/or maintain the Mainline Path Alteration facilities if it determines that it will be economically beneficial for Transporter to do so. Transporter will post notice of any such contribution on its Designated Site for a period of 30 days following any such agreement.
- 29.5 Mainline Path Alteration Capacity Rights. Prior to the in-service date of the Mainline Path Alteration facilities, the primary receipt point(s) and/or primary delivery point(s) of the Alteration Shipper's Service Agreement may be amended to reflect the new Mainline Transportation System path becoming available as the result of the Mainline Path Alteration facilities. Such amendment, however, will not take effect until the Mainline Path Alteration facilities have been placed in service.

30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM

This Section 30 sets forth the Cost Recovery Mechanism ("CRM") for the Modernization and Emissions Reduction Program ("MER Program") established by Article XIII of the pre-filing Stipulation and Settlement Agreement filed on August 26, 2022 and approved in Docket No. RP22-1155 ("2022 Settlement"). This Section 30 is incorporated by reference into the 2022 Settlement. This Section 30, the MER Program, and the CRM will each terminate when the 2022 Settlement terminates. The provisions of this Section 30, the MER Program, and the CRM may not be amended except as provided in the 2022 Settlement.

The CRM applies to Shippers under Rate Schedules TF-1 (Large Customer), TF-1 (25-year Evergreen Expansion), and TF-2 (such Shippers referred to herein as "CRM Shippers" and such rate schedules referred to herein as "Applicable CRM Rate Schedules") with base contracts having a primary term of one year or more. The CRM establishes a separately tracked surcharge ("CRM Surcharge") to recover Transporter's revenue requirements, as defined in Section 30.4(a) ("Revenue Requirement"), incurred to strengthen the safety, efficiency, reliability, and flexibility of Transporter's system while reducing Transporter's air emissions.

30.1 Scope of CRM

The CRM established by this Section 30 provides for the recovery of the Revenue Requirement for Eligible Facilities, as defined in Section 30.3, placed in service by Transporter from January 1, 2023 through October 31, 2027. Such Revenue Requirement will be recovered through the applicable annual CRM Surcharge every year for the years starting from April 1, 2025 through March 31, 2028, with the determination of each annual Revenue Requirement being subject to both the Cumulative Eligible Capital Investment Limit over entire 5-year period described in 30.3(d), and Annual Eligible Capital Investment Limits described in 30.3(e) below.

30.2 Annual Limited Natural Gas Act Section 4 CRM Surcharge Filings

- (a) Transporter will make its first CRM Surcharge filing with the FERC on or before February 28, 2025 to be effective April 1, 2025, to recover the Revenue Requirement for Eligible Facilities placed into service between January 1, 2023 and October 31, 2024. The first CRM Surcharge established by this filing will be effective from April 1, 2025 through March 31, 2026.
- (b) Transporter will make a second CRM Surcharge filing with the FERC on or before February 28, 2026 to be effective April 1, 2026, to recover the Revenue Requirement for Eligible Facilities placed into service between January 1, 2023 and October 31, 2025. The revised CRM Surcharge established by this second filing will be effective from April 1, 2026 through March 31, 2027. The revised CRM Surcharge filing will

- 30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)
 - 30.2 Annual Limited Natural Gas Act Section 4 CRM Surcharge Filings (Continued)
 - incorporate (1) changes in CRM Revenue Requirement, (2) any over recovery from the preceding year and/or (3) subject to Section 30.4(b), any under recovery from the preceding year.
 - (c) Transporter will make a third CRM Surcharge filing with the FERC on or before February 28, 2027 to be effective April 1, 2027, to recover the Revenue Requirement for Eligible Facilities placed into service between January 1, 2023 and October 31, 2026. The revised CRM Surcharge established by this third filing will be effective from April 1, 2027 through March 31, 2028. The revised CRM Surcharge filing will incorporate (1) changes in the CRM Revenue Requirement, (2) any over recovery from the preceding year and/or (3) subject to Section 30.4(b), any under recovery from the preceding year.
 - (d) The net capital costs for Eligible Facilities placed in service between January 1, 2023 and October 31, 2027 will be rolled into the rate base included in Transporter's next Natural Gas Act ("NGA") Section 4 or 5 general rate case or pre-filing settlement subject to Sections 30.3(d), 30.3(e), and 30.6.
 - 30.3 Eligibility
 - (a) Eligible Facilities. "Eligible Facilities" have been defined in the Eligible Facilities Plan ("EFP") as filed by Transporter in Appendix F-1 of 2022 Settlement.
 - (b) Transporter may unilaterally remove Eligible Facilities from the EFP at any time. Subject to the requisite prior Approval, as defined in Section 30.3 (c), of the CRM Shippers, Northwest may substitute facilities currently included in the EFP and/or add new facilities not currently included in the EFP, provided such substitutions or additions are consistent with the objective of the MER Program and fit within the applicable Annual Eligible Capital Investment Limit in Section 30.3(e) and the total Cumulative Eligible Capital Investment Limit of the EFP in Section 30.3(d).
 - (c) During any year at either the normal annual CRM meeting held pursuant to Section 30.5 or at a special meeting called by Transporter with at least thirty (30) days' prior written notice, Transporter may propose to substitute or add facilities to the EFP consistent with Section 30.3(b). Such meetings may be conducted virtually. If meetings are conducted in person, an option for remote participation via videoconference technology

- 30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)
- 30.3 Eligibility (Continued)

shall be offered to the extent practicable. Each CRM Shipper will have ten (10) days after the meeting to submit to Transporter in writing its vote(s) either supporting, not opposing, or rejecting each proposed substitution and/or addition. Votes received by electronic mail will be considered in writing. Any CRM Shipper who fails to timely submit its vote(s) will be deemed to have abstained from the vote. Transporter will have received the requisite prior "Approval" from the CRM Shippers when CRM Shippers holding at least 75% of the reservation billing determinants assessed the CRM Surcharge at the time of the vote elect either to support or not oppose the proposed substitution or addition. If a CRM Shipper abstains from the vote, then Transporter may calculate the 75% of billing determinants threshold based on the billing determinants of those CRM Shippers that participated in the vote. If the requisite Approval is not received, Transporter may elect to undertake the project outside of the CRM and include these costs as part of its maintenance capital expenditure in its next NGA Section 4 or 5 general rate case filing or pre-filing settlement.

- (d) Cumulative Eligible Capital Investment Limit. The total amount of capital investment for the Eligible Facilities that may be included in the determination of the annual Revenue Requirements, which is estimated at \$389.4 million (\$338.65 million plus 15%).
- Annual Eligible Capital Investment Limit. The Annual Eligible (e) Capital Investment Limit under the CRM shall be equal to the following annual capital dollar limits plus 15%: for 2023, \$9 million limit (\$7.85 million plus 15%); for 2024, \$156.5 million limit (\$136.13 million plus 15%); for 2025, \$91.6 million limit (\$79.63 million plus 15%); for 2026, \$75.5 million limit (\$65.63 million plus 15%); and for 2027, \$56.8 million limit (\$49.41 million plus 15%). If Transporter exceeds the Annual Eligible Capital Investment Limit in any year, Transporter will treat the excess above the Annual Eligible Capital Investment Limit as maintenance capital that will be excluded from the CRM Surcharge calculation. Transporter will treat maintenance capital excluded from the CRM Surcharge calculation as part of Transporter's rate base when determining Transporter's new rates in its next NGA Section 4 or 5 general rate case or pre-filing settlement.
- (f) Minimum Annual Maintenance Capital Expenditures. Transporter will spend on maintenance capital (i.e., expenditures for maintenance and other projects, excluding reimbursements through a customer specific rate as stated on Exhibit C or Exhibit D of the

30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)

30.3 Eligibility (Continued)

customer's transportation service agreements) a minimum annual amount of \$75 million in transmission gas plant. In each annual CRM Surcharge filing, Transporter will include a schedule of transmission gas plant projects with costs that were \$2 million or greater along with a description of each project and the costs per project, which together with transmission gas plant projects under \$2 million shows Transporter's annual maintenance capital expenditures on transmission gas plant in the prior year. For example, for Transporter's first CRM Surcharge filing made on or before February 28, 2025, Transporter will include a schedule of projects with maintenance capital expenditures for year 2024. If Transporter spent less than \$75 million on maintenance capital in the prior year, then the shortfall will be used to reduce the gross plant investment included in the annual CRM Surcharge filing for the upcoming year.

30.4 CRM Surcharge Calculation

- (a) Determination of Revenue Requirement. In each annual CRM Surcharge filing described in Section 30.2, Transporter will calculate the annual Revenue Requirement for Eligible Facilities placed into service through October 31 of the prior year. Transporter will determine the annual Eligible Capital Investment incurred (i.e., gross plant). Transporter will then subtract accumulated depreciation from the gross plant and adjust for accumulated deferred income taxes to determine the cumulative rate base associated with the Eligible Facilities. The Revenue Requirement will then be derived by summing:
 - (1) the cumulative rate base associated with the Eligible Facilities multiplied by the applicable Rate Base Multiplier from Appendix F-3 of the 2022 Settlement (11.50% based on 21% federal corporate income tax rate) that reflects pre-tax return and ad valorem taxes;
 - (2) the cumulative gross plant (subject to the limits in Sections 30.3(d) and 30.3(e) and less retirement of original gross plant of the compressor stations where the scope is for horsepower replacement) associated with the Eligible Facilities multiplied by the applicable depreciation rate from the 2022 Settlement;

- 30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)
- 30.4 CRM Surcharge Calculation (Continued)
 - (3) less operations and maintenance savings in the amounts of \$705,429 for April 1, 2025 CRM Surcharge, \$854,198 for the April 1, 2026 CRM Surcharge, and \$1,279,492 for the April 1, 2027 CRM Surcharge;
 - (4) less revenues derived from greenhouse gas reductions on the Eligible Facilities unless otherwise accounted for in the Carbon/Greenhouse Gas Tax Tracker mechanism included in the 2022 Settlement;
 - (5) any fees or taxes Transporter incurs due to carbon/greenhouse gas emissions on the Eligible Facilities unless otherwise accounted for in the Carbon/Greenhouse Gas Tax Tracker mechanism included in Article IX or Article X of the 2022 Settlement; and
 - (6) for each CRM Surcharge filing except the initial CRM Surcharge filing, any over recovery or under recovery of the prior year's Revenue Requirement.
 - (b) After the annual Revenue Requirement has been calculated, Transporter will calculate the CRM Surcharge as the quotient of the annual Revenue Requirement divided by the current annual actual billing determinants, adjusted for discounts, associated with the Applicable CRM Rate Schedules, excluding any expansion projects that may be excluded pursuant to Section 30.4(c) below; provided, however, for purposes of this section, current billing determinants will be subject to an annual billing determinant floor of 900,000,000 Dth.
 - If the annual billing determinants are lower than 900,000,000 Dth, then Transporter may only roll forward for recovery in its next CRM Surcharge filing any under-recovered Revenue Requirements down to the annual 900,000,000 Dth billing determinant floor. Any under recovered Revenue Requirement below the annual 900,000,000 Dth billing determinant floor may not be rolled forward for recovery in Transporter's next CRM Surcharge filing. For purposes of this section, the annual billing determinants means the actual billing determinants as of October 31 of the prior year under the Applicable CRM Rate Schedules.
 - (c) Lateral Projects. Designated Laterals and projects that are reimbursed through a customer specific rate as stated on Exhibit C or Exhibit D of the customer's transportation service agreement are not subject to the CRM Surcharge.

- 30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)
- 30.5 Annual CRM Shippers Meeting

Upon at least 30 days' prior notice to all CRM Shippers, Transporter shall hold one annual meeting with the CRM Shippers in January of each year. Such meeting may be conducted virtually. If meetings are conducted in person, an option for remote participation via video conference technology shall be offered to the maximum extent practicable. Transporter will review the following at the meeting:

- (a) Upcoming Eligible Facilities and updated cost estimates (including any fees or taxes due to greenhouse gas emissions on Eligible Facilities);
- (b) Prior year Eligible Facilities and actual spend (including any fees or taxes due to greenhouse gas emissions on Eligible Facilities);
- (c) Removals from and proposed additions and/or substitutions of Eligible Facilities;
- (d) Anticipated outages;
- (e) Draft of upcoming CRM Surcharge filing; and
- (f) Anticipated efficiency savings in operations costs, maintenance costs and/or fuel and emissions reductions due to CRM spend for the next year and actual cost savings achieved due to prior year's Eligible Facilities.

Transporter will make available upon request after the annual meeting copies of all updated maintenance and safety plans developed to comply with the regulations or requirements of the Pipeline and Hazardous Materials Safety Administration ("PHMSA"), U.S. Environmental Protection Act ("EPA"), or any state authority regarding implementation of the Pipes Act. Transporter will also make available upon request after the annual meeting copies of any studies, audits, and any information provided to PHMSA, EPA, or any state authority regarding leaks or emissions of greenhouse gases from its facilities as well as any requests for waivers.

- 30.6 Term of CRM and Roll-In of Net Capital Costs
 - (a) The CRM will be in effect from January 1, 2023 through March 31, 2028. However, if a rate case is filed under NGA Section 4 or 5 or a pre-filing settlement is reached with new rates becoming effective prior to March 31, 2028, then the MER Program and CRM will terminate when such new rates take effect.

- 30. MODERNIZATION AND EMISSIONS REDUCTION PROGRAM COST RECOVERY MECHANISM (Continued)
- 30.6 Term of CRM and Roll-In of Net Capital Costs (Continued)
 - (b) Upon termination of the MER Program and CRM:
 - i. Neither Northwest nor any CRM Shipper may seek to recover in new base rates any under recovered or over recovered Revenue Requirement in existence on the date that the 2022 Settlement terminates; and
 - ii. The net capital costs of the Eligible Facilities up to the time of the termination of the MER Program and CRM will be rolled into Transporter's rate base when determining Transporter's new rates in its next NGA Section 4 or 5 general rate case or pre-filing settlement.

- 31. RESERVED FOR FUTURE USE.
- 32. SALES OF EXCESS GAS
 - 32.1 Applicability. Transporter is not providing a sales service under any Rate Schedule of this Tariff, but may buy and sell gas in connection with providing storage and transportation services. Specifically, Transporter may have gas supplies in de minimis quantities or in limited or infrequent situations which it may choose to sell. Such sales are authorized pursuant to Transporter's blanket sales certificate. Circumstances under which Transporter may acquire gas to sell are described below:
 - (a) pursuant to Section 15.11, at the termination of a transportation Service Agreement, the Shipper has 15 days after the invoice date to remove from Transporter's system any imbalance gas resulting from gas received by Transporter in excess of that delivered to Receiving Party for Shipper; otherwise, such gas becomes the property of Transporter;
 - (b) pursuant to Section 15.10(b), any excess Shipper Imbalance gas, resulting from confirmed nominations by Shipper exceeding deliveries by Transporter, that is not reduced to allowed tolerance levels within fifteen days from the date of Northwest's invoice becomes the property of Northwest;
 - (c) pursuant to Section 22.2(b), when a Replacement Shipper fails to withdraw or vaporize, as applicable, all gas inventory within three days of termination of the replacement agreement, such gas becomes the property of Transporter;
 - (d) pursuant to Section 2.2 of Rate Schedule SGS-2I or Section 2.4 of Rate Schedule LS-2I, a Shipper utilizing interruptible storage service generally has seven days in which to withdraw Working Storage Gas Inventory from storage after receiving notice; otherwise, the gas becomes the property of Transporter;
 - (e) as authorized by the Commission, Transporter may reduce the levels of storage gas inventory which it owns and maintains to accommodate transportation balancing;
 - (f) as authorized by the Commission, Transporter may reduce the levels of Jackson Prairie storage facility cushion gas and/or test gas which it owns;

32. SALES OF EXCESS GAS (Continued)

- (g) as system operations dictate, Transporter may buy and sell gas in equivalent Dth amounts for the purpose of moving gas between the Jackson Prairie and Clay Basin storage facilities;
- (h) pursuant to Section 14.12, Transporter may recover gas through its fuel use requirements factors as reimbursement for volumes purchased by Transporter to fund lost and unaccounted-for gas;
- (i) pursuant to Rate Schedule DEX-1, Transporter will take title to any Deferred Exchange Quantity not delivered to Shipper at the end of the Deferred Exchange Period or any extension thereof due to Shipper's failure to arrange for receipt of such gas;
- (j) pursuant to Section 5 of Rate Schedule TPAL and Section 5 of Rate Schedule PAL, any Parked Quantity not removed within the time frame specified by Transporter's notice and any Parked Quantity not removed by the termination date of the Service Agreement will be deemed as a sale to Shipper; and
- (k) pursuant to Section 15.14, any excess LNG Boil-off Balance not brought within tolerance levels within 45 non-entitlement Balancing Approved Days from the date of Transporter's notice becomes the property of Transporter.
- (1) Pursuant to Section 14.14, if, as part of its integrity management program to perform inline inspections, Transporter requires minimum flow levels where scheduled quantities, and balancing flexibility related to line-pack, system storage, and interconnect agreements are insufficient to provide the required flow levels, Transporter may buy and sell gas to create such flows. If, due to market conditions, the purchased gas is sold at a loss, Transporter may recover the loss by selling gas in excess of the purchased gas. If, due to market conditions, the purchased gas is sold at a gain, Transporter will only sell sufficient gas to cover the cost of the gas purchased. Transporter will pass any difference in the quantities purchased verses sold through the fuel use requirements factors as reimbursement.

32. SALES OF EXCESS GAS (Continued)

- 32.2 Revenue Crediting. One hundred percent (100%) of all sales revenues received by Transporter pursuant to Sections 32.1(a) through (d) and Section 32.1(i), and one hundred percent (100%) of all net sales revenues received by Transporter pursuant to Section 32.1(j), Section 5.3 of Rate Schedule TPAL, and Section 5.3 of Rate Schedule PAL, will be credited to the following firm transportation Shippers, except those receiving service under a discounted Service Agreement or a capacity release Service Agreement: Rate Schedule TF-1 (Large Customer) Shippers, Rate Schedule TF-1(Small Customer) Shippers for the months they paid a reservation charge due to conversion to Rate Schedule TF-1 (Large Customer) service, Rate Schedule TF-2 Shippers, and Rate Schedule T-1 Shippers who received service prior to the cancellation of Rate Schedule T-1 (collectively, the specified Shippers). The credits will be allocated to the specified Shippers pro rata in proportion to the Maximum Base Tariff rate reservation charges (exclusive of surcharges) billed to the specified Shippers divided by the total reservation charges billed to all specified Shippers for each applicable month. Such credits shall be reflected as a credit billing adjustment to each March billing for credits accrued during the prior calendar year.
- 32.3 Reporting Requirements. Transporter will post on its Designated Site any quantities of gas available for sale pursuant to Section 32. In the event gas is sold, completed transactions will be posted on the Designated Site within thirty (30) days of the sales transaction detailing the quantity of gas sold, the sales price and whether or not the gas was sold to an affiliate. Transporter will file an annual report with the Commission by March 1 of each year describing for each sales transaction for the preceding calendar year the identities of the parties, the type of service provided, the total volumes sold, and the total revenues received.

FORMS OF SERVICE AGREEMENT

INDEX

Rate Schedule	Description			
TF-1	Firm Transportation	301		
*	Service Agreement Amendment	307		
*	Addendum to Service Agreement	308		
TI-1	Interruptible Transportation	315		
SGS-2F	Storage Gas Service - Firm	322		
SGS-2I	Storage Gas Service - Interruptible	332		
LS-2F	Liquefaction-Storage Gas Service - Firm	340		
LS-2I	Liquefaction-Storage Gas Service - Interruptible	349-A		
LS-3F	Liquefaction-Storage, Vaporization and LNG Delivery Service - Firm	349-I		
LD-4I	Liquefaction-LNG Delivery Service - Interruptibl	e 349-V		
TF-2	Firm Redelivery Transportation	350		
DEX-1	Deferred Exchange of Storage Gas	359		
PAL	Park and Loan Service	363		
TPAL	Term Park and Loan Service	368		
TFL-1	Firm Lateral Transportation	370		
TIL-1	Interruptible Lateral Transportation	385		

^{*} Applicable to all rate schedules.

FORM OF RATE SCHEDULE TF-1 SERVICE AGREEMENT

Rate Schedule TF-1 Service Agreement

			Contra	act No						
THIS SERVICE		MENT	(Agree		•			-		
(Transporter)							e and enter			
	_[.]	(or)	[and r	restates	the	Service	Agreement	made	and	entered
into on		•	.]							
WHEREAS:										

- A. [Insert recital describing how Shipper acquired transportation capacity.]
- B. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule TF-1 and the General Terms and Conditions (GT&C) that apply to Rate Schedule TF-1, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Transportation Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to receive, transport and deliver natural gas for Shipper, on a firm basis. The Transportation Contract Demand, the Maximum Daily Quantity at each Primary Receipt Point, and the Maximum Daily Delivery Obligation at each Primary Delivery Point are set forth on Exhibit A. If contract-specific OFO parameters are set forth on Exhibit A, whenever Transporter requests during the specified time period, Shipper agrees to flow gas as requested by Transporter, up to the specified volume through the specified transportation corridor.
- 3. Transportation Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. Reservation charges apply to the Transportation Contract Demand set forth on Exhibit A. The Maximum Base Tariff Rates (Recourse Rates) set

Rate Schedule TF-1 Service Agreement (Continued)

forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to the Rate Schedule TF-1 customer category identified on Exhibit A, will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D. Additionally, if applicable under Section 21 or 29 of the GT&C, Shipper agrees to pay Transporter a facilities charge as set forth on Exhibit C.

- 4. Transportation Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.

Rate Schedule TF-1 Service Agreement (Continued)

- 8. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	ву:
Name:	Name:
Title:	Title:

a. [Large Customer]

	FORM OF RATE SCHE	EDULE TF-1 SERVIC (Continued)	E AGREEMENT
	[Base	Contract Version]	
	[DatedRate Schedule (Control between No	EXHIBIT A fective (or), Effective to the e TF-1 Service Ag act No) orthwest Pipeline	reement LLC
	SE.	RVICE DETAILS	
1.	Transportation Contract Den	mand (CD):	Dth per day
2.	Primary Receipt Point(s):	Maximum Daily Quantities	
	Point ID Name	(Dth)	
	Total		
3.	Primary Delivery Point(s): Point ID Name	Maximum Daily Delivery Obligation (Dth)	
			pursuant to Section 2.4 of
1	the General Terms and Cond	ICIONS:	[None] or [Description]
4.	Customer Category:		

[Small Customer (subject to temporary conversion to Large Customer

under Section 2.1(d) of Rate Schedule TF-1)]

	FORM OF RATE SCHEDULE TF-1 SERVICE AGREEMENT (Continued)
•	[Base Contract Version]
	EXHIBIT A (Continued)
b.	<pre>Incremental Expansion Customer: [No]</pre>
Rates:	Recourse, Discounted Recourse, or Negotiated Rate Transportation
a.	Reservation Charge (per Dth of CD): [Maximum Base Tariff Rate, plus applicable surcharges]
	(or discounted rate(s) using one or more of the following options, as many times as needed)
	<pre>[[\$, plus applicable surcharges] (or) [% of Maximum Base Tariff Rate, plus applicable surcharges][.] (or) [from [date] through [date]].</pre>
	[Not Applicable] (for use when Customer Category is Small Customer)
b.	Volumetric Charge (per Dth): [Maximum Base Tariff Rate, plus applicable surcharges] (or) [\$, plus applicable surcharges]
c.	Additional Facility Reservation Surcharge Pursuant to Section 3.4 of Rate Schedule TF-1 (per Dth of CD): [None] (or) [Maximum] (or) [\$]
d.	Rate Discount Conditions Consistent with Section 3.5 of Rate Schedule TF-1: [Not Applicable] (or) [(Description)]
e	Negotiated Rate Conditions Consistent with Section 3 7 of Rate

Schedule TF-1: [Not Applicable] (or) [[During the Negotiated Rates Period as outlined on Exhibit D of the agreement, subsections a.

through d. above will not apply] (and/or) [(Description)]]

[Base Contract Version]

EXHIBIT A (Continued)

6. Transportation Term:

[(Specified period before or after a

specified event)]

(and)

Specified conditional service agreement extensions pursuant to Section 11.9 of the General Terms and Conditions of the Tariff:

[None] (or) [(Description)].

c. Evergreen Provision: [No]

(or)

[Yes, standard unilateral evergreen under Section 12.1 of Rate Schedule TF-1]

[Yes, standard bi-lateral evergreen under Section 12.2 [(a)(i) and (b)(i)] (or) [(a)(ii) and (b)(ii)] (or) [(a)(iii) and

(b)(iii)] of Rate Schedule TF-1]

(or)

[Yes, grandfathered unilateral evergreen under Section 12.3 of Rate Schedule TF-1] (or)

[Yes, see Exhibit B]

7. Contract-Specific OFO Parameters and/or Alternative Actions in lieu of a Contract-Specific OFO: [None] (or)

```
Time Period Quantity Transmission Corridor (Date) to Up to Dth per day (Point) to (Point) [(Date) (or) Agreement Termination Date]

[(and)/(or) [(Description)].
```

8. Subordinate rights apply as defined in GT&C Section 1, Secondary Firm Service Rights with a Scheduling and Curtailment Priority per GT&C Section 12.1(b)(ii): [Yes] (or) [No]

[Base Contract Version]

EXHIBIT A (Continued)

9.	Regulatory Authorization	: [18 CFR 284.223]	
		(or)	
		[18 CFR 284.102, on behalf of	

10. Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit C [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Exhibit E [No] (or) [Yes]

		FO:		DULE TF-1 SERVICE Continued)	E AGREEMENT
	[Tem	porar	y Capacity Rele	ease Replacement	Contract Version
		[Date	[Dated and Eff	EXHIBIT A fective(or)	
			(Contra between No: and	to the TF-1 Service Aga act No) rthwest Pipeline	LLC
		(Re	leasing Shipper	/Contract No.: _)
			SI	ERVICE DETAILS	
1.	Transpo	rtati	on Contract Dem	and (CD):	Dth per day
2.	Primary	Rece	ipt Point(s):	Maximum Daily	
	Point	ID	Name	Quantity (Dth)	
		-	Total		
3.	Primary	Deli	very Point(s):	Maximum Daily Delivery	Delivery
	Point	ID	Name	Obligation (Dth)	Pressure (psig)
		_	Total		

Specified conditions for Delivery Pressure, pursuant to Section 2.4 of the General Terms and Conditions: [None] or [Description]

[Temporary Capacity Release Replacement Contract Version]

		EXHIBIT A (Con-	tinued)
4.	Custon	mer Category:	
	a. La	rge Customer	
	b. Ind	cremental Expansion Customer:	<pre>[No] (or) [Yes, Columbia Gorge Expansion] (or) [Yes, 25-Year Evergreen Expansion]</pre>
5.		portation Rates: the version of a.[1], a.[2], a	.[3], or a.[4] that applies.]
	a.[1]	[\$, plus applicable surc	us applicable surcharges] (or)
		Awarded CRM Surcharge (per Dt. [Maximum CRM Surcharge Rate] [\$] (or) [%] of CRM Surcharge Rate [none]	(or)
		(or)	
	a.[2]	[\$, plus applicable surce [% of Maximum Base Tarif (or) [[Maximum Base Tariff Rate, p. [\$, plus applicable surce Base Tariff Rate, plus applicable surce B	us applicable surcharges] (or)
		average load factor volumetri	c commitment of%]
		(or)	
	a.[3]	index reference 1) (+ or -) (In	1 multiplier)*(Index-Based Rate dex-Based Rate reference 2 reference 2)(+ or -)(Index-Based

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

Index-Based Rate Floor: (Insert dollars and cents value not less than Transporter's minimum reservation rate that is the lowest rate acceptable to Releasing Shipper.)

Index-Based Rate Default: (Insert non-biddable rate, not less than the rate floor, to be used for invoicing purposes when the result of the index-based formula is unavailable or cannot be computed.)]

(or)

- a.[4] [See Index-Based Rate Unique Formula specified by Releasing Shipper in Section 12 of this Exhibit A.]
- b. Volumetric Charge (per Dth): [Maximum Base Tariff Rate, plus applicable surcharges] (or) [\$, plus applicable surcharges]
- c. Awarded Additional Facility Reservation Surcharge pursuant to
 Section 3.4 of Rate Schedule TF-1 (per Dth of CD): [None] (or)
 [Maximum] (or) [\$]
- d. Rate Discount Conditions Consistent with Section 3.5 of Rate Schedule TF-1: [Not Applicable] (or) [The following rate discount condition from Releasing Shipper's Contract No. (insert number) shall apply to this Agreement as follows: (insert the same discount condition from the Releasing Shipper's contract, as modified only by the awarded rate that will apply).]

6.	Transportation	Term:	Begin Date:	
			Nomination Cycle:	
			End Date:	

7. Contract-Specific OFO Parameters and/or Alternative Actions in lieu of a Contract-Specific OFO: [None] (or)

Time Period		Quantity		Transmission	Corridor
(Date) to	Up to	Dth p	oer day	(Point) to	(Point)
[(Date) (or)					
Agreement Termination	Date]				
		(and) / (or)			-
	[(Description))].		

8. Subordinate rights apply as defined in GT&C Section 1 Secondary Firm Service Rights with a Scheduling and Curtailment Priority per GT&C Section 12.1(b)(ii): [Yes] (or) [No]

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

- 9. Regulatory Authorization: [18 CFR 284.223] (or) [18 CFR 284.102, on behalf of]
- 10. Additional Exhibits: Exhibit B {Yes] (or) [No] Exhibit C [Yes] (or) [No]
- 11. Standard Capacity Release Conditions:
 - a. Releasing Shipper's recall rights:
 [Use the version of i.[1] or i.[2] that applies]

 - i.[2] [Released capacity may be recalled prior to the [Timely]
 (or) [Evening] (or) [Intraday 1] (or) [Intraday 2] (or)
 [Intraday 3] nomination cycle applicable to the initial
 day of the capacity recall.]
 - ii. Recall notification: [Allowed on any day] (or) [Allowed only on a Business Day] (or) [Not Applicable]
 - b. Reput rights: [Yes] (or) [No]

 - e. Re-releasable: [Yes] (or) [No]

[Temporary Capacity Release Replacement Contract Version]

Exhibit A (Continued)

- 12. Additional Capacity Release Conditions pursuant to Section 22.7(k) of the GT&C: [None] or [Description]
- 13. Index-Based Rate Unique Formula specified by Releasing Shipper: [None] or [Description]

Second Revised Sheet No. 304 Superseding First Revised Sheet No. 304

	FORM OF RATE	(Continued)	CE AGREEMENT	
		EXHIBIT B		
[(Dated and	Effective	, (subject	to Commission	acceptance)]
		(or)		
	[Dated	, Effective _		_,
	(subject	to Commission accep	tance)]	
		to the		
	Rate Sch	edule TF-1 Service A	greement	
	(C	Contract No	_)	
	betwe	en Northwest Pipeline	e LLC	
	and			

NON-CONFORMING PROVISIONS

	(Continued)
	EXHIBIT C
	[Dated and Effective]
	(or)
	, ,
	[Dated, Effective]
	to the
	Rate Schedule TF-1 Service Agreement
	(Contract No.)
	between Northwest Pipeline LLC
	and
	INCREMENTAL FACILITIES PAYMENT OBLIGATION
1.	DESCRIPTION OF FACILITIES:
	The facilities contemplated by Section 21 or 29 of the GT&C to
	provide service under this agreement include the following:
	(description of project)
2.	RESPONSIBILITY FOR FACILITIES COSTS:
۷.	RESTONSIBILITY FOR FACILITIES COSTS.
	Pursuant to Section 21 or 29 of the GT&C, Shipper is responsible
	to pay for the facilities described above and has elected the payment
	option set forth below.
3.	TERMS AND CONDITIONS OF FACILITIES PAYMENT:
J •	Third in book billion of morning.
	a. Type of Charge: .
	[Incremental facilities charge] (or) [Description of other
	payment method agreed to by Transporter and Shipper pursuant
	to Section 21.5 or 29.4]
	b. Charge \$
	b. Charge \(\frac{1}{2} \)
	(Describe the basis of the charges(s) and true-up or
	adjustment provisions, if any, for the stated charge.)
	c. Term of Charge: .
	· Term of charge.
	(Contract term or a term agreed to between the parties,
	whichever is shorter.)
	d Aggelerated Dayment, [Decaription of aggelerated normant accession]
	d. Accelerated Payment: [Description of accelerated payment provision]
	(or) [None]

FORM OF RATE SCHEDULE TF-1 SERVICE AGREEMENT

1.

FORM OF RATE SCHEDULE TF-1 SERVICE AGREEMENT (Continued)
EXHIBIT D
[(Dated and Effective, (subject to Commission acceptance)]
(or)
[Dated, Effective,
(subject to Commission acceptance)]
to the
Rate Schedule TF-1 Service Agreement
(Contract No.)
between Northwest Pipeline LLC
and
NEGOTIATED RATE PROVISIONS
(Pursuant to Section 3.7 of Rate Schedule TF-1)
The Negotiated Rate Provisions will take effect on [the Primary Term
Begin Date] (or) [[(Date)] (and/or) [(Description)]] and will remain in
effect for a period of () [[years], (and/or) [months], (and/or)
[days] (further description as needed)] (Negotiated Rates Period).
Thereafter, the Recourse Rates will apply to the Service Agreement.
(Description of Negotiated Rate Provisions)

Second Revised Sheet No. 306-A Superseding First Revised Sheet No. 306-A

	(Continued)	
	EXHIBIT E [Dated and Effective]	
	(or)	
[Dated	, Effective]
	to the	
	Rate Schedule TF-1 Service Agreement	
	(Contract No)	
	between Northwest Pipeline LLC	
	and	

FORM OF RATE SCHEDULE TF-1 SERVICE AGREEMENT

 ${\tt BUY-OUT\ PROVISION},$ pursuant to Section 11.8 of the General Terms and Conditions:

FORM OF SERVICE AGREEMENT AMENDMENT
Rate Schedule Service Agreement Amendment
Contract No
[Dated, Effective]
THIS AMENDMENT by and between Northwest Pipeline LLC (Transporter) and
WHEREAS:
A. Transporter and Shipper are parties to that certain Rate Schedule Service Agreement dated and assigned Contract No (Agreement).
B. Transporter and Shipper desire to amend the Agreement to [summary of Agreement revisions].
C. [Insert additional recitals as appropriate to memorialize the context of the Amendment.]
THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:
1. As of the effective date set forth thereon, the Exhibit A attached hereto supercedes and replaces the previously effective Exhibit A to the Agreement.
2. The additional exhibits noted on the attached Exhibit A as applicable to the Agreement, if any, also are attached hereto and, as of the effective dates set forth thereon, supercede and replace any previously effective corresponding exhibits to the Agreement.
IN WITNESS WHEREOF, Transporter and Shipper have executed this Amendment as of the date first set forth above.
Northwest Pipeline LLC
By:
Name: Name:
Title: Title:

Second Revised Sheet No. 308 Superseding First Revised Sheet No. 308

ADDENDUM TO SERVICE AGREEMENT
Due to a Merger, Acquisition, Name Change or Assignment
(Dated) (Contract No(Agreement)) (Rate Schedule)
Pursuant to Section 11.5 of the General Terms and Conditions of Transporter's Tariff, this Agreement is modified as follows:
Effective, [New Shipper] replaces [Original Shipper] as the [contract holder] (or) [Releasing Shipper] (or) [contract holder and Releasing Shipper] of this capacity due to a(n) [Merger] (or) [Acquisition] (or) [Name Change] (or) [Assignment].

ADDENDUM TO SERVICE AGREEMENT
Due to a Permanent Release of Capacity
(Dated) (Contract No(Agreement)) (Rate Schedule)
Pursuant to Section 22.12 of the General Terms and Conditions of Transporter's Tariff, this Agreement is modified as follows:
(To Be Added to the Releasing Shipper's Base or Temporary Capacity Release Service Agreement in the Event such Releasing Shipper Permanently Releases All or a Portion of its [Transportation Contract Demand] (or) [Storage Demand and Storage Capacity])
Pursuant to Offer Number, effective, [Original Shipper] has permanently released all or a portion of the [Transportation Contract Demand] (or) [Storage Demand and Storage Capacity] underpinning this Agreement to [New Shipper]. Therefore, this Agreement is [terminated] (or) [modified to reflect the terms contained in such Offer, including the reduction of [Transportation Contract Demand by Dth/d] (or) [Storage Demand by Dth/d and Storage Capacity by Dth]]. [New Shipper's] Service Agreement number is
(To Be Added to the Replacement Shipper's Temporary Capacity Release Service Agreement in the Event that the Related Releasing Shipper Permanently Releases All of its [Transportation Contract Demand] (or) [Storage Demand and Storage Capacity] which is Encumbered by such Replacement Shipper's Temporary Capacity Release Transaction)
Pursuant to Offer Number, effective, [New Shipper] replaces [Original Shipper] as the Releasing Shipper for the [Transportation Contract Demand] (or) [Storage Demand and Storage Capacity] underpinning this Agreement due to a permanent capacity release transaction. Additionally, the original Service Agreement number has been replaced with new Service Agreement number

ADDENDUM TO SERVICE AGREEMENT

Due to a conversion of Rate Schedule LS-2F Capacity

[Dated______]
[Contract No. ____("Agreement")]
[Rate Schedule ____]

(To Be Added to the Shipper's Base LS-2F Service Agreement in the event such Shipper converts all or a portion of its Rate Schedule LS-2F Service Agreement to a Rate Schedule LS-3F Service Agreement)

Pursuant to Section 13.2 of Rate Schedule LS-3F, this Agreement is modified as follows:

Effective _____, Shipper has permanently converted [all] (or) [a portion] of the Storage Demand and Storage Capacity underpinning this Agreement to an LS-3F Service Agreement. Therefore, this Agreement is [terminated] (or) [modified to reflect the reduction of Storage Demand by _____Dth/d and Storage Capacity by ______Dth]. The new LS-3F Service Agreement number is

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet No. 311 Superseding Original Sheet Nos. 311 Through 314

ADDENDUM TO SERVICE AGREEMENT
Due to the Termination of the Agreement
[Dated] [Contract No(Agreement)]
Pursuant to the evergreen provision of the Agreement, as more fully described in the applicable evergreen provisions section of Rate Schedule [] of Transporter's Tariff, this Agreement is modified as follows:
[Shipper] (or) [Transporter] provided notice that this Agreement will terminate at the end of gas day [(Date)].

RESERVED FOR FUTURE USE

FORM OF RATE SCHEDULE TI-1 SERVICE AGREEMENT

Rate Schedule TI-1 Service Agreement Contract No
THIS SERVICE AGREEMENT (Agreement) by and between Northwest Pipeline LLC (Transporter) and (Shipper) is made and entered into on[.] (or) [and restates the Service Agreement made and entered into on]
WHEREAS:
A. [Insert recitals as appropriate to memorialize the context of the Agreement.]
THEREFORE, in consideration of the premises and mutual covenants set forth

1. Tariff Incorporation. Rate Schedule TI-1 and the General Terms and Conditions (GT&C) that apply to Rate Schedule TI-1, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.

herein, Transporter and Shipper agree as follows:

- 2. Transportation Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to receive, transport and deliver natural gas for Shipper, on an interruptible basis. The Maximum Daily Quantity of natural gas and the receipt and delivery points are set forth on Exhibit A.
- 3. Transportation Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule TI-1 will apply to service hereunder unless and to the extent that negotiated rates apply as set forth on Exhibit D. Notwithstanding the foregoing, a temporary discounted Recourse Rate may apply pursuant to the terms set forth on a Rate Schedule TI-1 Temporary Discount form.
- 4. Transportation Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation ${\bf r}$

FORM OF RATE SCHEDULE TI-1 SERVICE AGREEMENT (Continued)

Rate Schedule TI-1 Service Agreement (Continued)

service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

FORM OF RATE SCHEDULE TI-1 SERVICE AGREEMENT (Continued)

Rate Schedule TI-1 Service Agreement (Continued)

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE TI-1 SERVICE AGREEMENT (Continued)
	(concinued)
	EXHIBIT A [Dated and Effective]
	(or) [Dated, Effective]
	to the
	Rate Schedule TI-1 Service Agreement
	(Contract No)
	between Northwest Pipeline LLC
	and
	SERVICE DETAILS
1.	Maximum Daily Quantity: Dth per day
2.	Receipt Point(s): [All Transportation Receipt Points] (or)
	[(List points by point ID and name)]
3.	Delivery Point(s): [All Transportation Delivery Points] (or)
	[(List points by point ID and name)]
4.	Transportation Term:
	a. Primary Term Begin Date: [(Date)] (or)
	[Upon the later of the actual in-service
	date of the required new
	facilities or (Date)]
	b. Primary Term End Date: [(Date)]
	(or)
	[(Specified period before or after a
	specified event)]
	c. Evergreen Provision: [No]
	(or) [Yes, see Section 8 of Rate Schedule TI-1
	(or)
	[Yes, see Exhibit B]

FORM OF RATE SCHEDULE TI-1 SERVICE AGREEMENT (Continued)

EXHIBIT A (Continued)

5.	Regulatory Authorization	n: [18 CFR 284.223]	
		(or)	
		[18 CFR 284.102, on behalf of	

6. Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Second Revised Sheet No. 319 Superseding First Revised Sheet No. 319

FORM OF	RATE SCHEDULE TI-1 SERVICE AGR	EEMENT
	(Continued)	
	EXHIBIT B	
[(Dated and Effective	, (subject to Com	mmission acceptance)]
	(or)	
[Dated	, Effective	<i>'</i>
(sul	oject to Commission acceptance)]
Rate	Schedule TI-1 Service Agreeme:	nt
	(Contract No)	
b	etween Northwest Pipeline LLC	
and	d	_
	NON-CONFORMING PROVISIONS	

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Second Revised Sheet No. 320 Superseding First Revised Sheet No. 320

RESERVED FOR FUTURE USE

FORM OF	RATE SCHEDULE TI-1 SERVICE AGREEMENT	
	(Continued)	
	EXHIBIT D	
[(Dated and Effective	, (subject to Commission	acceptance)]
	(or)	
[Dated	, Effective	,
(sı	ubject to Commission acceptance)]	
	to the	
Rat	e Schedule TI-1 Service Agreement	
	(Contract No.)	
	between Northwest Pipeline LLC	
aı	nd	
	NEGOTIATED RATE PROVISIONS	
(Pursuant	to Section 3.3 of Rate Schedule TI-1)	

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT

Rate Schedule SGS-2F Service Agreement

			Contract No
THIS SERVICE (Transporter)	and		(Agreement) by and between Northwest Pipeline LLC (Shipper) is made and entered into on [and restates the Service Agreement made and entered
into onWHEREAS:		·	j , , , , , , , , , , , , , , , , , , ,

- A. [Insert recital describing how Shipper acquired storage capacity.]
- B. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule SGS-2F and the General Terms and Conditions (GT&C) that apply to Rate Schedule SGS-2F, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Storage Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to inject, store and withdraw natural gas for Shipper, on a firm basis. Shipper may request Transporter to withdraw volumes in excess of Shipper's Storage Demand on a best-efforts basis as provided in Rate Schedule SGS-2F. The Storage Demand and Storage Capacity are set forth on Exhibit A.
- 3. Storage Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to the Rate Schedule SGS-2F customer category identified on Exhibit A will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

Rate Schedule SGS-2F Service Agreement (Continued)

- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights and the amount of the Releasing Shipper's Working Gas Quantity released to Shipper for the initial Storage Cycle, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 8. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.

Second Revised Sheet No. 324 Superseding First Revised Sheet No. 324

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

Rate Schedule SGS-2F Service Agreement (Continued)

9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)
	[Base Contract Version]
	EXHIBIT A
	[Dated and Effective]
	(or)
	[Dated, Effective]
	to the
	Rate Schedule SGS-2F Service Agreement
	(Contract No)
	between Northwest Pipeline LLC
	and
	SERVICE DETAILS
1.	Customer Category: [Pre-Expansion Shipper] or [Expansion Shipper
2.	Storage Demand: Dth per day
3.	Storage Capacity: Dth

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

		[Base Contract	Version]
		EXHIBIT A (Co	ontinued)
4.		rse or Discounted Recourse Sto "Not Applicable" if Exhibit D	<u> </u>
	a.	Demand Charge (per Dth of Sto [Maximum Base Tariff Rate] (or) [\$]	rage Demand):
	b.	Capacity Demand Charge (per D [Maximum Base Tariff Rate] (or) [\$]	
		Rate Discount Conditions Consule SGS-2F: [Not Applicable]	istent with Section 3.2 of Rate (or) [(Description)]
5.	Servic	ce Term:	
	a.	Primary Term Begin Date:	[(Date)] (or)
		[Upon the later of the required new	actual in-service date of the facilities or (Date)

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

[Base Contract Version]

EXHIBIT A (Continued)

b. Primary Term End Date: [(Date)] (or)

[(Specified period before or after a

specified event)]

c. Evergreen Provision: [No]

(or)

[Yes, standard unilateral evergreen

under Section 15.1 of Rate Schedule SGS-2F]

(or)

[Yes, standard bi-lateral evergreen under

Section 15.2 [(a)(i) and (b)(i)] (or) [(a)(ii) and (b)(ii)] of Rate Schedule SGS-

2F] (or)

[Yes, grandfathered unilateral evergreen under Section 15.3 of Rate Schedule SGS-2F]

6. Regulatory Authorization: [18 CFR 284.223]

(or)

[18 CFR 284.102, on behalf of _____]

7. Additional Exhibits: Exhibit B [No] (or) [Yes]

Exhibit D [No] (or) [Yes]

	FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)
	[Temporary Capacity Release Replacement Contract Version]
	EXHIBIT A
	[Dated and Effective]
	(or)
	[Dated, Effective]
	to the
	Rate Schedule SGS-2F Service Agreement
	(Contract No.)
	between Northwest Pipeline LLC
	and
	(Releasing Shipper/Contract No.:)
	SERVICE DETAILS
1.	Customer Category: [Pre-Expansion Shipper] or [Expansion Shipper]
2.	Storage Demand: Dth per day
3.	Storage Capacity: Dth

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

[Temporary Capacity Release Replacement Contract Version]

		EXHIBIT A (Continued)						
4.	Worki Dt	ng Gas Quantity for Replacement Shipper's initial Storage Cycle:						
5.	Storage Rates: (Use the applicable version of (a) and (b))							
	a.	Awarded Demand Charge (per Dth of Storage Demand): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]						
	b.	Awarded Capacity Demand Charge (per Dth of Storage Capacity): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate] (or)						
	a.	Awarded Volumetric Bid Withdrawal Charge (per Dth of withdrawals): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]						
	b.	Awarded Volumetric Bid Storage Charge (per Dth of Shipper's Working Gas Inventory): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]						

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

6.	Servi	ce Ter	n:		
	a.	Begin	Date:	-	Nomination Cycle:
	b.	End Da	ate:		
7.	Regul	atory A	Authorization:	-	CFR 284.223] (or) CFR 284.102, on behalf of]
8.	Addit	ional 1	Exhibits: Exhib	oit B	[No] (or) [Yes]
9.	Stand	ard Ca _l	pacity Release Cor	nditi	ons:
	a.	Releas	sing Shipper's red	call :	rights:
		i.	[Released capacit	y may	y not be recalled.]
		i.	(or) [Evening] (or) [] .natio	y be recalled prior to the [Timely] Intraday 1] (or) [Intraday 2] (or) on cycle applicable to the initial day 1.]
		ii.			[Allowed on any day] (or) [Allowed y] (or) [Not Applicable]
	b.	Reput	rights: [Yes]	(or) [No]
	С.	Re-re	leasable: [Yes]	(or) [No]
	d.	Asset		,	t ("AMA"): [No] (or) ment of the AMA Replacement Shipper]]
			-	Condi	tions pursuant to Section 22.7(k) of

Second Revised Sheet No. 330 Superseding First Revised Sheet No. 330

FORM OF RATE SCHEDULE SGS-2F SERVICE AGREEMENT	
(Continued)	
EXHIBIT B	
[(Dated and Effective, (subject to Commission acce	eptance)]
(or)	
[Dated, Effective,	
(subject to Commission acceptance)]	
to the	
Rate Schedule SGS-2F Service Agreement	
(Contract No.)	
between Northwest Pipeline LLC	
and	
NON-CONFORMING PROVISIONS	

FORM OF RA	ATE SCHEDULE SGS-2F SERVICE AGREEMENT (Continued)	
	EXHIBIT D	
[(Dated and Effective	, (subject to Commission a	acceptance)]
<u> </u>	(or)	_
[Dated	, Effective ,	,
(sub	eject to Commission acceptance)]	
	to the	
Rate	Schedule SGS-2F Service Agreement	
	(Contract No)	
be	etween Northwest Pipeline LLC	
and	L	
	NEGOTIATED RATE PROVISIONS	
(Pursuant t	o Section 3.4 of Rate Schedule SGS-2F)	

FORM OF RATE SCHEDULE SGS-21 SERVICE AGREEMENT

Rate Schedule SGS-2I Service Agreement

			Со	ntract N	·						
THIS SERVICE 2	AGREE	MENT	(Agre	eement) k	y and	d betwee:	n North	west	Pipe	line	e LLC
(Transporter)			[and	(Sh restates		r) is ma Service					
into on WHEREAS:		·	.]				_				
										_	. 1

A. [Insert recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule SGS-2I and the General Terms and Conditions (GT&C) that apply to Rate Schedule SGS-2I, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Storage Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to inject, store and withdraw natural gas for Shipper, on an interruptible basis. The Interruptible Storage Capacity is set forth on Exhibit A.
- 3. Storage Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule SGS-2I will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.

FORM OF RATE SCHEDULE SGS-21 SERVICE AGREEMENT (Continued)

Rate Schedule SGS-2I Service Agreement (Continued)

- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this Agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

Second Revised Sheet No. 333-A Superseding First Revised Sheet No. 333-A

FORM OF RATE SCHEDULE SGS-21 SERVICE AGREEMENT (Continued)

Rate Schedule SGS-2I Service Agreement (Continued)

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
By:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE SG: (Contin				
	EXHIBI [Dated and Effective (or [Dated, Effective to the contract No. Dated, Effective Dated	e] fective] he Service Agreement) t Pipeline LLC			
	SERVICE D	ETAILS			
1.	Interruptible Storage Capacity: _	Dth			
2.	Recourse or Discounted Recourse Storage Rates: (Show "Not Applicable" if Exhibit D is attached.)				
	a. Volumetric Charge (per Dth of [Maximum Base Tage (or) [\$]	f Shipper's Working Gas Inventory): riff Rate]			
	b. Rate Discount Conditions Conscious Schedule SGS-2I: [Not Applied	sistent with Section 3.2 of Rate cable] (or) [(Description)]			
3.	Service Term:	ervice Term:			
	a. Primary Term Begin Date:	<pre>[(Date)] (or) [Upon the later of the actual in-service date of the required new facilities or (Date)]</pre>			
	b. Primary Term End Date:	<pre>[(Date)] (or) [(Specified period before or after a specified event)]</pre>			
	c. Evergreen Provision: [No] (or)				
	[Yes, see Section	n 9.1 of Rate Schedule SGS-2I]			
4.	(0	284.223] r) 284.102, on behalf of]			
5.		[o] (or) [Yes]			

Second Revised Sheet No. 335 Superseding First Revised Sheet No. 335

FORM OF RA	TE SCHEDULE SGS-2I SERVICE	AGREEMENT
	(Continued)	
	EXHIBIT B	
[(Dated and Effective $_$, (subject to	<pre>Commission acceptance)]</pre>
	(or)	
[Dated	, Effective	
(subj	ect to Commission acceptar	nce)]
	to the	
Rate S	chedule SGS-2I Service Agr	eement
	(Contract No.)	
be	tween Northwest Pipeline L	LC
and		
	NON-CONFORMING PROVISIONS	

	FORM OF RA	TE SCHEDULE SGS-21 SERVICE AGREEMENT (Continued)	
		EXHIBIT D	
[(Dated	and Effective	, (subject to Commission	acceptance)]
		(or)	
	[Dated	, Effective	,
	(subj	ect to Commission acceptance)]	_
		to the	
	Rate S	chedule SGS-2I Service Agreement	
		(Contract No.)	
	be	tween Northwest Pipeline LLC	
	and		
		NEGOTIATED RATE PROVISIONS	
	(Pursuant to	Section 3.3 of Rate Schedule SGS-2I)

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Fifth Revised Sheet No. 337 Superseding Fourth Revised Sheet No. 337

RESERVED FOR FUTURE USE

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Fifth Revised Sheet No. 338 Superseding Fourth Revised Sheet No. 338

RESERVED FOR FUTURE USE

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Fourth Revised Sheet No. 339 Superseding Third Revised Sheet No. 339

RESERVED FOR FUTURE USE

FORM OF RATE SCHEDULE LS-2F SERVICE AGREEMENT

Rate Schedule LS-2F Service Agreement

			Contract No
THIS SERVICE	AGREE	MENT	(Agreement) by and between Northwest Pipeline LLC
(Transporter)	and		(Shipper) is made and entered into on
	_[.]	(or)	[and restates the Service Agreement made and entered
into on		·	.]
WHEREAS:			

- A. [Insert recital describing how Shipper acquired storage capacity.]
- B. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule LS-2F and the General Terms and Conditions (GT&C) that apply to Rate Schedule LS-2F, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Storage Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to liquefy, store in liquid phase, and vaporize natural gas for Shipper, on a firm basis. The Storage Demand and Storage Capacity are set forth on Exhibit A.
- 3. Storage Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule LS-2F will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.

Rate Schedule LS-2F Service Agreement (Continued)

- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5, 11.10 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 8. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.

Rate Schedule LS-2F Service Agreement (Continued)

9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

FORM	OF	RATE	SCHEDULE	LS-2F	SERVICE	AGREEMENT
			(Cont	cinued)		

			[Base Contract Version]		
			EXHIBIT A		
		[Dat	ed and Effective]	
			(or)		
		[Dated	, Effective]	
			to the		
		Rate	Schedule LS-2F Service Agreeme	ent	
		,	(Contract No)		
			petween Northwest Pipeline LLC		
		an	d	_	
			SERVICE DETAILS		
1.	Stor	age Demand:	Dth per day		
	2.	Storage Capac	city: Dth		
			Discounted Recourse Storage Ratole" if Exhibit D is attached.)	es:	
	a.	[Ma	<pre>(per Dth of Storage Demand): .ximum Base Tariff Rate] (or)]</pre>		
	b.	[Ma	nd Charge (per Dth of Storage (ximum Base Tariff Rate] (or)	Capacity):	
	c. d. e.	Vaporization (Rate Discount	Charge (per Dth): Maximum Base Charge (per Dth): Maximum Base Conditions Consistent with Sec F: [Not Applicable] (or) [(De	Tariff Rate ction 3.3 of B	Rate

[Base Contract Version]

					EXHI	BIT.	A (Continued)
4.	Ser	vice Te	rm:				
	a.	Primary	y Ter	m Beg	in Date		<pre>[(Date)] (or) [Upon the later of the actual in-service date of the required new facilities or (Date)]</pre>
	b.	Primary	y Ter	m End	Date:		<pre>[(Date)] (or) [(Specified period before or after a specified event)]</pre>
	С.	Evergre	een P	[No] (Yes, Secti (Yes, Secti [Yes, Secti	or) five you not son 11.2 on 11.2 (or) bi-laton 11.2	l of ear: 2 of tera 3 [(notice unilateral evergreen under Rate Schedule LS-2F] notice unilateral evergreen under Rate Schedule LS-2F] l evergreen under a) (i) and (b) (i)] (or) [(a) (ii) and e Schedule LS-2F]
5.	Reg	gulatory	Auth	noriza	tion:		CFR 284.223] (or) CFR 284.102, on behalf of]
6.	Add	litional	Exhi	bits:	Exhil	oit 1	B [No] (or) [Yes]

Exhibit D [No] (or) [Yes]

(or)

FORM OF RATE SCHEDULE LS-2F SERVICE AGREEMENT (Continued) [Temporary Capacity Release Replacement Contract Version] EXHIBIT A [Dated and Effective ____] (or) [Dated _____, Effective _____] to the Rate Schedule LS-2F Service Agreement (Contract No. ____) between Northwest Pipeline LLC (Releasing Shipper/Contract No.:) SERVICE DETAILS Storage Demand: Dth per day 1. Storage Capacity: ____ Dth 2. 3. Storage Rates: (Use the applicable version of (a) and (b)) Awarded Demand Charge (per Dth of Storage Demand): a. [Maximum Base Tariff Rate] (or) [\$______] (or) [$_$ % of Maximum Base Tariff Rate] b. Awarded Capacity Demand Charge (per Dth of Storage Capacity): [Maximum Base Tariff Rate] (or) [\$____] (or) [% of Maximum Base Tariff Rate]

4.

5.

FORM OF RATE SCHEDULE LS-2F SERVICE AGREEMENT (Continued)

[Temporary Capacity]	Release Replacement	Contract	Version]
-----------------------	---------------------	----------	----------

	[Temporary Capacity Release Replacement Contract Version]
	EXHIBIT A (Continued)
a.	Awarded Volumetric Bid Vaporization Demand-Related Charge (per Dth delivered): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]
b.	Awarded Volumetric Bid Storage Capacity Charge (per Dth of Storage Capacity Balance): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]
d.	Liquefaction Charge (per Dth): Maximum Base Tariff Rate Vaporization Charge (per Dth): Maximum Base Tariff Rate ice Term:
a.	Begin Date: Nomination Cycle:
b.	End Date:

(or)

Regulatory Authorization: [18 CFR 284.223]

[18 CFR 284.102, on behalf of _____]

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

- 6. Additional Exhibits: Exhibit B [No] (or) [Yes]
- 7. Standard Capacity Release Conditions:
 - a. Releasing Shipper's recall rights:

 - i. [Released capacity may be recalled prior to the [Timely]
 (or) [Evening] (or) [Intraday 1] (or) [Intraday 2] (or)
 [Intraday 3] nomination cycle applicable to the initial day
 of the capacity recall.]
 - ii. Recall notification: [Allowed on any day] (or) [Allowed only on a Business Day] (or) [Not Applicable]
 - b. Reput rights: [Yes] (or) [No]
 - c. Re-releasable: [Yes] (or) [No]
- 8. Additional Capacity Release Conditions pursuant to Section 22.7(k) of the GT&C: [None] or [(Description)]

Second Revised Sheet No. 348 Superseding First Revised Sheet No. 348

	FORM OF	RATE SCH	EDULE L	S-2F SERVI	CE	AGREEMENT	
			(Conti	nued)			
			EXHIB	IT B			
[(Dated an	d Effective		,	(subject	to	Commission	acceptance)]
			(or	·)			
	[Dated		, E	ffective			,
	(sı	ubject to	Commis	sion accep	tar	nce)]	_
			to t	he			
	Rate	e Schedul	e LS-2F	Service A	gre	eement	
		(Cont	ract No)		
		between N	orthwes	t Pipeline	e L	LC	
	ar	nd		-			
		NON-CO	NFORMIN	G PROVISIO	NS		

Second Revised Sheet No. 349 Superseding First Revised Sheet No. 349

	FORM ((Continued)	
		EXHIBIT D	
[(Dated a	and Effectiv	re, (subject to Commission	acceptance)]
		(or)	
	[Dated _	, Effective	<i>'</i>
	-	(subject to Commission acceptance)]	-
		to the	
	D -	ate Schedule LS-2F Service Agreement	
	No		
		(Contract No)	
		between Northwest Pipeline LLC	
		and	
		NEGOTIATED RATE PROVISIONS	
	(Pursua	nt to Section 3.4 of Rate Schedule LS-2F)	

FORM OF RATE SCHEDULE LS-21 SERVICE AGREEMENT

Rate Schedule LS-2I Service Agreement

		Con	tract No	•	 		
THIS SERVICE				_		-	
(Transporter)		[and r			e and ent Agreement		
into on	 ·	.]					
WHEREAS:							

A. [Insert recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule LS-2I and the General Terms and Conditions (GT&C) that apply to Rate Schedule LS-2I, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Storage Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to liquefy, store in liquid phase, and vaporize natural gas for Shipper, on an interruptible basis. The Storage Capacity is set forth on Exhibit A.
- 3. Storage Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule LS-2I will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.
- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the

Rate Schedule LS-2I Service Agreement (Continued)

established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the evergreen provision.

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
By:	By:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE LS-21 SERVICE AGREEMENT (Continued)
	EXHIBIT A [Dated and Effective] (or) [Dated, Effective] to the Rate Schedule LS-2I Service Agreement (Contract No) between Northwest Pipeline LLC and
	SERVICE DETAILS
1.	Storage Capacity: Dth
2.	Recourse or Discounted Recourse Storage Rates: (Show "Not Applicable" if Exhibit D is attached.)
	<pre>a. Volumetric Charge (per Dth of Shipper's Working Gas Inventory):</pre>
	b. Rate Discount Conditions Consistent with Section 3.2 of Rate Schedule LS-2I: [Not Applicable] (or) [(Description)]
3.	Service Term:
	a. Primary Term Begin Date: [(Date)] (or) [Upon the later of the actual in-service date of the required new facilities or (Date)]
	b. Primary Term End Date: [(Date)] (or) [(Specified period before or after a specified event)]
	<pre>c. Evergreen Provision:</pre>
4.	Regulatory Authorization: [18 CFR 284.223]
5.	Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Second Revised Sheet No. 349-D Superseding First Revised Sheet No. 349-D

FORM OF	RATE SCHEDULE LS-21 SERVICE A	AGREEMENT
	(Continued)	
	EXHIBIT B	
[(Dated and Effective	, (subject to (Commission acceptance)]
[Dated	, Effective	,
(sı	nbject to Commission acceptant	ce)]
	to the	
Rate	Schedule LS-2I Service Agree	ement
	(Contract No.)	
	between Northwest Pipeline LL	C
ar	id	
		
	NON-CONFORMING PROVISIONS	

Second Revised Sheet No. 349-E Superseding First Revised Sheet No. 349-E

		FORM OF	RATE SCI	HEDULE I (Conti	S-2I SERVI nued)	ICE	AGREEMENT	
				EXHIE	IT D			
[(Dated	and	Effective			(subject	to	${\tt Commission}$	acceptance)]
				(0:	r)			
		[Dated		, E	ffective _			_′
		(sı	abject to	o Commis	sion accep	otar	nce)]	
				to ·	the			
		Rate	e Schedul	le LS-2I	Service A	Agre	eement	
	(Contract No.							
			between	Northwe:	st Pipelin	e L	LC	
			nd		_			
		ω						
			NEGOTI	ATED RA	TE PROVISI	ONS		
		(Pursuant	to Sect	ion 3.3	of Rate S	che	dule LS-2I)	

RESERVED FOR FUTURE USE

FORM OF RATE SCHEDULE LS-3F SERVICE AGREEMENT

Rate Schedule LS-3F Service Agreement

			Contract No
THIS SERVICE	AGREE	MENT	(Agreement) by and between Northwest Pipeline LLC
(Transporter)	and		(Shipper) is made and entered into on
	_[.]	(or)	[and restates the Service Agreement made and entered
into on		·	.]
WHEREAS:			

- A. [Insert recital describing how Shipper acquired storage capacity.]
- B. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule LS-3F and the General Terms and Conditions (GT&C) that apply to Rate Schedule LS-3F, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Services. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to liquefy, store in liquid phase, and either vaporize natural gas for on-system delivery or deliver LNG to offsystem delivery points directly connected to the Plymouth LNG Facility storage tanks for Shipper, on a firm basis. The Storage Demand and Storage Capacity are set forth on Exhibit A.
- 3. Storage Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule LS-3F will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.

Rate Schedule LS-3F Service Agreement (Continued)

- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 8. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.

Rate Schedule LS-3F Service Agreement (Continued)

9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	[Base Contract Version]
	EXHIBIT A [Dated and Effective] (or) [Dated, Effective] to the Rate Schedule LS-3F Service Agreement (Contract No) between Northwest Pipeline LLC and
	SERVICE DETAILS
1.	Storage Demand: Dth per day
	4. Storage Capacity: Dth
	5. Recourse or Discounted Recourse Storage Rates: (Show "Not Applicable" if Exhibit D is attached.)
	<pre>a. Demand Charge (per Dth of Storage Demand):</pre>
	<pre>b. Capacity Demand Charge (per Dth of Storage Capacity):</pre>
	c. Liquefaction Charge (per Dth): Maximum Base Tariff Rate, subject to true-up pursuant to Section 14.20 of the GT&C
	d. Vaporization Charge (per Dth): Maximum Base Tariff Rate
	e. Rate Discount Conditions Consistent with Section 3.3 of Rate Schedule LS-3F: [Not Applicable] (or) [(Description)]

[Base Contract Version]

	EXHIBIT A (Continued)
4.	Service Term:
	a. Primary Term Begin Date: [(Date)] (or) [Upon the later of the actual in-service date of the required new facilities or (Date)]
	<pre>b. Primary Term End Date: [(Date)] (or) [(Specified period before</pre>
	<pre>c. Evergreen Provision:</pre>
5.	Regulatory Authorization: [18 CFR 284.223]
6.	Additional Exhibits: Exhibit B [No] (or) [Yes]

Exhibit D [No] (or) [Yes]

	FORM OF RATE SCHEDULE LS-3F SERVICE AGREEMENT (Continued)
	[Temporary Capacity Release Replacement Contract Version]
	EXHIBIT A [Dated and Effective]
	(or) [Dated, Effective] to the
	Rate Schedule LS-3F Service Agreement (Contract No)
	between Northwest Pipeline LLC and
	(Releasing Shipper/Contract No.:)
	SERVICE DETAILS
	3. Storage Demand: Dth per day
	4. Storage Capacity: Dth
3.	Storage Rates: (Use the applicable version of (a) and (b))
	a. Awarded Demand Charge (per Dth of Storage Demand): [Maximum Base Tariff Rate] (or) [\$] (or) [% of Maximum Base Tariff Rate]
	<pre>b. Awarded Capacity Demand Charge (per Dth of Storage Capacity)</pre>
	(or)

ı	Temporary	Capacity	Release	Replacement	Contract	Version
ı	Liemporary	Capacity	METERSE	Mehracement	Contract	ACTOTOLL

EXHIBIT A (Continued)

a.	Awarded Volumetric Bid Vaporization Demand-Related Charge
	(per Dth delivered):
	[Maximum Base Tariff Rate]
	(or)
	[\$]
	(or)
	[% of Maximum Base Tariff Rate]

[Maximum Base Tariff Rate]
 (or)
[\$____]
 (or)
[____% of Maximum Base Tariff Rate]

- c. Liquefaction Charge (per Dth): Maximum Base Tariff Rate, subject to true-up pursuant to Section 14.20 of the GT&C
- d. Vaporization Charge (per Dth): Maximum Base Tariff Rate
- 4. Service Term:

a.	Begin	Date:		
	Non	nination	Cycle:	

b. End Date: ____

5.	Regulatory Authorization:	[18	CFR	284.	223]
			(<	or)	
	[18 CFR 284.102, on	beha	lf o	f	1

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

- 6. Additional Exhibits: Exhibit B [No] (or) [Yes]
- 7. Standard Capacity Release Conditions:
 - a. Releasing Shipper's recall rights:

 - i. [Released capacity may be recalled prior to the [Timely]
 (or) [Evening] (or) [Intraday 1] (or) [Intraday 2] (or
 [Intraday 3] nomination cycle applicable to the initial day
 of the capacity recall.]
 - ii. Recall notification: [Allowed on any day] (or) [Allowed only on a Business Day] (or) [Not Applicable]
 - b. Reput rights: [Yes] (or) [No]
 - c. Re-releasable: [Yes] (or) [No]
- 8. Additional Capacity Release Conditions pursuant to Section 22.7(k) of the GT&C: [None] or [(Description)]

FORM OF	RATE SCHEDULE LS-3F SERVICE AGREEMEN	NT
	(Continued)	
	EXHIBIT B	
[(Dated and Effective	, (subject to Commiss	ion acceptance)]
	(or)	
[Dated	, Effective	,
(su	abject to Commission acceptance)]	
	to the	
Rate	Schedule LS-3F Service Agreement	
	(Contract No.)	
]	between Northwest Pipeline LLC	
an	nd	

NON-CONFORMING PROVISIONS

	FORM OF RATE	E SCHEDULE LS-3F SERVICE AGREEMENT (Continued)	
		EXHIBIT D	
[(Dated	and Effective	, (subject to Commission accept	ance)]
		(or)	
	[Dated	, Effective,	
	(subjec	ct to Commission acceptance)]	
		to the	
	Rate Sch	nedule LS-3F Service Agreement	
		(Contract No)	
	betw	een Northwest Pipeline LLC	
	and		
	NE	GOTIATED RATE PROVISIONS	
	(Durquant to	Spation 2 / of Data Schodula IS-2E)	
	(ruisualit to	Section 3.4 of Rate Schedule LS-3F)	

RESERVED FOR FUTURE USE

First Revised Sheet No. 349-V Superseding Original Sheet No. 349-V

FORM OF RATE SCHEDULE LD-41 SERVICE AGREEMENT

Rate Schedule LD-4I Service Agreement

		Со	ntract No	·					
THIS SERVICE (Transporter)	and	 	•	ipper)	is mad	de and	entered	into	on
into on		 .]	1000000		701 1100	11910011	iorro maa	o arra	circoroa
WHEREAS:									

A. [Insert recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule LD-4I and the General Terms and Conditions (GT&C) that apply to Rate Schedule LD-4I, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Liquefaction Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to liquefy natural gas and deliver the liquefied natural gas for Shipper to off-system delivery points directly connected to the Plymouth LNG liquefiers, on an interruptible basis.
- 3. Service Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule LD-4I will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.
- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the storage service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the

Rate Schedule LD-4I Service Agreement (Continued)

established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the evergreen provision.

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Storage service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE LD-41 SERVICE AGREEMENT (Continued)								
	EXHIBIT A [Dated and Effective]								
	SERVICE DETAILS								
1.	Recourse or Discounted Recourse Rates: (Show "Not Applicable" if Exhibit D is attached.)								
	<pre>a. Volumetric Charge (per Dth):</pre>								
	b. Liquefaction Charge (per Dth): Maximum Base Tariff Rate, subject to true-up pursuant to Section 14.20 of the GT&C								
	c. Rate Discount Conditions Consistent with Section 3.2 of Rate Schedule LD-4I: [Not Applicable] (or) [(Description)]								
2.	Service Term:								
	a. Primary Term Begin Date: [(Date)] (or) [Upon the later of the actual in-service date of the required new facilities or (Date)]								
	b. Primary Term End Date: [(Date)] (or) [(Specified period before or after a specified event)]								
	<pre>c. Evergreen Provision:</pre>								
3.	Regulatory Authorization: [18 CFR 284.223] (or) [18 CRF 284.102, on behalf of]								
4.	Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit D [No] (or) [Yes]								

FORM OF	RATE SCHEDULE LD-41 SERVICE	AGREEMENT						
	(Continued)							
	EXHIBIT B							
[(Dated and Effective	, (subject to	Commission acceptance)]						
	(or)							
[Dated	, Effective	<i>,</i>						
(sı	(subject to Commission acceptance)]							
	to the							
Rate	e Schedule LD-4I Service Agre	eement						
	(Contract No)							
	between Northwest Pipeline L	LC						
aı	nd							

NON-CONFORMING PROVISIONS

	FORM OF	RATE SCHEDULE LD-41 SERVICE AGREEMENT						
		(Continued)						
		EXHIBIT D						
[(Dated a	and Effective	, (subject to Commission acceptar	nce)]					
		(or)						
	[Dated	, Effective ,						
	(su	abject to Commission acceptance)]						
		to the						
	Rate Schedule LD-4I Service Agreement							
		(Contract No)						
]	between Northwest Pipeline LLC						
	an	ıd						
		NEGOTIATED RATE PROVISIONS						
	(Pursuant	to Section 3.3 of Rate Schedule LD-4I)						

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT

Rate Schedule TF-2 Service Agreement

			Contr	act No.					
THIS SERVICE A (Transporter)	and			(S	hippe	r) is ma	n Northwes de and ent Agreement	ered into	o on
into on		·•]				5		
WHEREAS:									

- A. [Insert recital describing how Shipper acquired transportation capacity.]
- B. [Insert recital describing the Shipper's related firm storage service or firm storage ownership.]
- C. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule TF-2 and the General Terms and Conditions (GT&C) that apply to Rate Schedule TF-2, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Transportation Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to receive, transport and deliver natural gas for Shipper, on a firm basis. The Transportation Contract Demand, the Annual Contract Quantity, the Maximum Daily Quantity at the Primary Receipt Point, and the Maximum Daily Delivery Obligation at each Primary Delivery Point are set forth on Exhibit A.
- 3. Transportation Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced

Rate Schedule TF-2 Service Agreement (Continued)

herein. The Monthly Billing Quantity for reservation charges is set forth on Exhibit A. The Maximum Base Tariff Rates (Recourse Rates) for Rate Schedule TF-2 set forth in the Statement of Rates in the Tariff, as revised from time to time, will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D. Additionally, if applicable under Section 21 of the GT&C, Shipper agrees to the incremental facilities charge as set forth on Exhibit C.

- 4. Transportation Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.

Rate Schedule TF-2 Service Agreement (Continued)

- 8. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued)
	[Base Contract Version]
	EXHIBIT A [Dated and Effective] (or) [Dated, Effective] to the
	Rate Schedule TF-2 Service Agreement (Contract No) between Northwest Pipeline LLC and
	SERVICE DETAILS
1.	Transportation Contract Demand: Dth per day
2.	Annual Contract Quantity: Dth
3.	Monthly Billing Quantity: Dth
4.	Primary Receipt Point: Maximum Daily Quantity
	Point ID Name (Dth)

Total

		[Base	Contract Version]		
		EXHI	BIT A (Continued)		
5.	Primary Delivery (Show the followi		mouth LNG is the Receipt	Point.)	
	near the points w	hose loca	y Transporter to Shipper tion are described in Sh t (Contract No) u	ipper's cur	rently
	(or)				
	(Show the followi	ng if Ply	mouth LNG is not the Rec	eipt Point.	.)
	[[[Point ID	Name	Maximum Daily Delivery Obligation (Dth)	Delivery Pressure (psig)	-
	 Tot	 :al			

Specified conditions for Delivery Pressure, pursuant to Section 2.4 of the General Terms and Conditions: [None] or [Description]

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued)

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued) [Base Contract Version] EXHIBIT A (Continued) 6. Recourse or Discounted Recourse Transportation Rates: (Show "Not Applicable" if Exhibit D is attached.) a. Reservation Charge (per Dth of Monthly Billing Quantity):

- c. Rate Discount Conditions Consistent with Section 3.3 of Rate Schedule TF-2: [Not Applicable] (or) [(Description)]
- 7. Transportation Term:

under Section 14.3 of Rate Schedule TF-2]

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued)
[Base Contract Version]
EXHIBIT A (Continued)
8. Contract-Specific OFO Parameters and/or Alternative Actions in lieu of a Contract-Specific OFO: [None] (or)
Time Period Quantity Transmission Corridor (Date) to Up to Dth per day (Point) to (Point) [(Date) (or) Agreement Termination Date] (and)/(or) [(Description)].
9. Regulatory Authorization: [18 CFR 284.223]
10. Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit C [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Exhibit E [No] (or) [Yes]

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued) [Temporary Capacity Release Replacement Contract Version] EXHIBIT A [Dated and Effective ____] (or) [Dated _____, Effective _____] to the Rate Schedule TF-2 Service Agreement (Contract No. ____) between Northwest Pipeline LLC (Releasing Shipper/Contract: _____) SERVICE DETAILS 1. Transportation Contract Demand: Dth per day 2. Annual Contract Quantity: ____ Dth 3. Monthly Billing Quantity: ____ Dth 4. Primary Receipt Point: Maximum Daily Quantity Point ID Name (Dth)

Total

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

5.	Primary Deli (Show the fol	<u> </u>	: ymouth LNG is the R	eceipt Point.)	
	the points wh	nose location fective Servi	by Transporter to S s are described in ce Agreement (Contr	Releasing Shipper	's
	(0	r)			
	(Show the fol	llowing if Pl	ymouth LNG is not t	he Receipt Point.)
	[Maximum Daily Delivery	Delivery]
	[Obligation	Pressure	-
	[Point ID	Name	(Dth)	(psig)]
	_	Total			

Specified conditions for Delivery Pressure, pursuant to Section 2.4 of the General Terms and Conditions: [None] or [Description]

[Temporary Capacity Release Replacement Contract Version]

	EXHIBIT A (Continued)	
6.	Transportation Rates: [Use the version of "Reservation Charge" in "a." which applies.]	
	<pre>a. Awarded Reservation Charge (per Dth of Monthly Billing Quantity): [Maximum Base Tariff Rate, plus applicable surcharges]</pre>	;]
	(or)	
	<pre>a. Awarded Volumetric Bid Reservation Charge (per Dth): [Maximum Base Tariff Rate, plus applicable surcharges]</pre>	3]
	<pre>b. Volumetric Charge (per Dth): [Maximum Base Tariff Rate, plus</pre>	
	c. Volumetric Charge Rate Discount Conditions Consistent with Section 3.3 of Rate Schedule TF-2: [Not Applicable] (or) [(Description)]	1
7.	Transportation Term: Begin Date: Nomination Cycle:	
	End Data:	

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

8. Contract-Specific OFO Parameters and/or Alternative Actions in lieu of a Contract-Specific OFO:

	[None} (or)	
Time Period	Quantity	Transmission Corridor
(Date) to	Up toDth per da	y (Point) to (Point)
[(Date) (or)		
Agreement Termination Date]		
	(and)/(or)[(Descript	cion)].

9. Regulatory Authorization: [18 CFR 284.223] (or)

[18 CFR 284.102, on behalf of]

- 10. Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit C [No] (or) [Yes]
- 11. Standard Capacity Release Conditions:
 - a. Releasing Shipper's recall rights:

 - i. [Released capacity may be recalled prior to the [Timely]
 (or) [Evening] (or) [Intraday 1] (or) [Intraday 2] (or)
 [Intraday 3] nomination cycle applicable to the initial day
 of the capacity recall.]
 - ii. Recall notification: [Allowed on any day] (or) [Allowed only on a Business Day] (or) [Not Applicable]
 - iii. Recall provisions in Section 22.2(a)(2) of the General
 Terms and Conditions of Transporter's Tariff also apply:
 [Yes] (or) [No]
 - b. Reput rights: [Yes] (or) [No]
 - c. Primary Delivery Point may be changed through amendment: $[{\tt Yes}] \ ({\tt or}) \ [{\tt Nol} \ ({\tt or}) \ [{\tt Not} \ {\tt Applicable}]$
 - d. Re-releasable: [Yes] (or) [No]
 - e. Asset Management Arrangement ("AMA"): [No] (or)

[Yes, [Specify commitment of AMA Replacement Shipper]]

12. Additional Capacity Release Conditions pursuant to Section 22.7(k) of the GT&C: [None] or [(Description)]

Third Revised Sheet No. 357 Superseding Second Revised Sheet No. 357

	FORM OF RATE SC.	HEDULE TF-2 SERVICE AGR	REEMENT
		(Continued)	
		EXHIBIT B	
[Dated and Effe	ective	, (subject to Cor	mmission acceptance)]
		(or)	
[Da	ted	, Effective	,
	(subject to	Commission acceptance)]
		to the	
	Rate Schedu	le TF-2 Service Agreeme	ent
	(Cont	cract No.)	
	between :	Northwest Pipeline LLC	
	and		
			_
	NON-CC	NFORMING PROVISIONS	

(or) [None]

	(Continued)
	EXHIBIT C
	[Dated and Effective]
	(or)
	[Dated, Effective]
	to the
	Rate Schedule TF-2 Service Agreement
	(Contract No.)
	between Northwest Pipeline LLC
	and
	INCREMENTAL FACILITIES PAYMENT OBLIGATION
1.	DESCRIPTION OF FACILITIES:
	The facilities contemplated by Section 21 of the GT&C to provide service under this agreement include the following: (description of project)
2.	RESPONSIBILITY FOR FACILITIES COSTS:
	Pursuant to Section 21 of the GT&C, Shipper is responsible to pay for the actual cost of service for the facilities described above and has elected the payment option set forth below.
3.	TERMS AND CONDITIONS OF FACILITIES PAYMENT:
	a. Type of Charge:
	[Incremental facilities charge] (or) [Description of other payment method agreed to by Transporter and Shipper pursuant to Section 21.5]
	b. Charge \$
	(Describe the basis of the charges(s) and true-up or adjustment provisions, if any, for the stated charge.)
	c. Term of Charge:
	(Contract term or a term agreed to between the parties, whichever is shorter.)
	d. Accelerated Payment: [Description of accelerated payment provision

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT

FORM OF	(Continued)	
	EXHIBIT D	
[(Dated and Effective	, (subject to Commission	<pre>acceptance)]</pre>
	(or)	
[Dated	, Effective	_′
(sı	abject to Commission acceptance)]	
	to the	
Rat	e Schedule TF-2 Service Agreement	
	(Contract No)	
	between Northwest Pipeline LLC	
ar	nd	
	NEGOTIATED RATE PROVISIONS	
(Pursuant	t to Section 3.7 of Rate Schedule TF-2)	

Second Revised Sheet No. 358-B Superseding First Revised Sheet No. 358-B

FORM OF RATE SCHEDULE TF-2 SERVICE AGREEMENT (Continued)

	EXHIBIT E	
	[Dated and Effective]	
	(or)	
[Dated	, Effective]
	to the	
	Rate Schedule TF-2 Service Agreement	
	(Contract No)	
	between Northwest Pipeline LLC	
	and	

 ${\tt BUY-OUT\ PROVISION},$ pursuant to Section 11.8 of the General Terms and Conditions:

Fifth Revised Sheet No. 359 Superseding Fourth Revised Sheet No. 359

FORM OF RATE SCHEDULE DEX-1 SERVICE AGREEMENT

	Rate Schedule DEX-1 Service Agreement Contract No
THIS SERVICE AGREE (Transporter) and	EMENT (Agreement) by and between Northwest Pipeline LLC (Shipper) is made and entered into on
[.]	(or) [and restates the Service Agreement made and entered
into on]
WHEREAS:	

A. [Insert recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule DEX-1 and the General Terms and Conditions (GT&C) that apply to Rate Schedule DEX-1, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Deferred Storage Exchange Service. Subject to the terms, conditions, and limitations hereof and of Rate Schedule DEX-1, Shipper agrees to deliver to Transporter, and Transporter agrees to deliver to Shipper, the Deferred Exchange Quantity listed on Exhibit A. Shipper will deliver the Deferred Exchange Quantity to Transporter on a date mutually agreed to by Shipper and Transporter at the Target Storage Facility shown on Exhibit A. Transporter will deliver the Deferred Exchange Quantity to Shipper during the Deferred Exchange Period at the Daily Delivery Quantity and at the Deferred Exchange Storage Facility as shown on Exhibit A.
- 3. Deferred Exchange Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule DEX-1 will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.
- 4. Service Term. This Agreement becomes effective on the effective date set forth on Exhibit A and terminates upon the termination date set forth on Exhibit A. The schedule for the deferred exchange of storage gas hereunder is set forth on Exhibit A.

Rate Schedule DEX-1 Service Agreement (Continued)

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.
- 8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
By:	By:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE DEX-1 SERVICE AGREEMENT (Continued)
	EXHIBIT A
	[Dated and Effective]
	(or)
	[Dated, Effective]
	to the
	Rate Schedule DEX-1 Service Agreement
	(Contract No)
	between Northwest Pipeline LLC
	and
	SERVICE DETAILS
1.	Quantity (Dth):
	a. Deferred Exchange Quantity
	h Daile Daliman Quantity
	b. Daily Delivery Quantity
2.	Exchange Points:
٠.	a. Target Storage Facility:
	b. Deferred Exchange Storage Facility:
3.	Exchange Schedule:
	a. Transfer Date:
	b. Deferred Exchange Period: from to
4.	Agreement Termination Date:
_	
5.	Rate (per Dth of Deferred Exchange Quantity):
	(Show "Not Applicable" if Exhibit D is attached.)
	[Maximum Base Tariff Rate, plus applicable surcharges]
	(or)
	[\$, plus applicable surcharges]
	[Y, plus applicable sulcharges]
6.	Fuel Reimbursement Waiver under Section 4 of Rate Schedule DEX-1:
	[Yes] (or) [No]
7.	Regulatory Authorization: [18 CFR 284.223]
	(or)
	[18 CFR 284.102, on behalf of]
8.	Additional Exhibits: Exhibit B [No] (or) [Yes]
	Exhibit D [No] (or) [Yes]

Second Revised Sheet No. 362 Superseding First Revised Sheet No. 362

FORM OF	RATE SCHEDULE DEX-1 SERVICE (Continued)	AGREEMENT
	EXHIBIT B	
[(Dated and Effective	, (subject to	Commission acceptance)]
	(or)	
[Dated	, Effective	
(su	bject to Commission acceptar	ice)]
	to the	
Rate	e Schedule DEX-1 Service Agre	eement
	(Contract No)	
]	between Northwest Pipeline L	LC
ar	ıd	
	NON-CONFORMING PROVISIONS	

	FORM OF	(Continued)	
		EXHIBIT D	
[(Dated	and Effective	e, (subject to Commission	acceptance)]
		(or)	
	[Dated	, Effective	_′
	(s	ubject to Commission acceptance)]	
		to the	
	Rat	e Schedule DEX-1 Service Agreement	
		(Contract No)	
		between Northwest Pipeline LLC	
	a	nd	
		NEGOTIATED RATE PROVISIONS	
	(Pu	rsuant to Section 19.1 of the GT&C)	

FORM OF RATE SCHEDULE PAL SERVICE AGREEMENT

	Rate Schedule PAL Service Agreement Contract No
THIS SERVICE AGREEMENT	(Agreement) by and between Northwest Pipeline LLC
(Transporter) and	(Shipper) is made and entered into on
[.] (or)	[and restates the Service Agreement made and entered
into on	.1
WHEREAS:	

A. [Insert recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule PAL and the General Terms and Conditions (GT&C) that apply to Rate Schedule PAL, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to provide [Park] (or) [Loan] Service for Shipper up to a Maximum [Park] (or) [Loan] Quantity set forth on Exhibit A at Transporter's [Park] (or) [Loan] Point set forth on Exhibit A.
- 3. Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule PAL will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.
- 4. Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the evergreen provision.

Rate Schedule PAL Service Agreement (Continued)

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE PAL SERVICE AGREEMENT (Continued)
	EXHIBIT A [Dated and Effective]
	(or) [Dated, Effective] to the
	Rate Schedule PAL Service Agreement (Contract No) between Northwest Pipeline LLC and
	SERVICE DETAILS
1.	Maximum [Park] (or) [Loan] Quantity: Dth
2.	<pre>[Park] or [Loan] Point: [Clay Basin]</pre>
3.	Recourse or Discounted Recourse Rate (per Dth): (Show "Not Applicable" if Exhibit D is attached.)
	<pre>a. [Maximum Base Tariff Rate]</pre>
	b. Rate Discount Conditions Consistent with Section 3.2 of Rate Schedule PAL: [Not Applicable] (or) [(Description)]
4.	Term:
	a. Primary Term Begin Date: [(Date)]
	b. Primary Term End Date: [(Date)]
	<pre>c. Evergreen Provision: [No]</pre>
5.	Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Second Revised Sheet No. 366 Superseding First Revised Sheet No. 366

FORM OF	RATE SCHEDULE PAL SERVICE AGREEMENT	
	(Continued)	
	EXHIBIT B	
[(Dated and Effective	, (subject to Commission	acceptance)]
	(or)	_
[Dated	, Effective	,
(sı	ubject to Commission acceptance)]	
	to the	
Rat	te Schedule PAL Service Agreement	
	(Contract No.)	
	between Northwest Pipeline LLC	
ar	nd	

NON-CONFORMING PROVISIONS

FORM OF RA	ATE SCHEDULE PAL SERVICE AGREEMENT
	(Continued)
	EXHIBIT D
[(Dated and Effective	, (subject to Commission acceptance)]
	(or)
[Dated	, Effective ,
(subje	ect to Commission acceptance)]
	to the
Rate S	Schedule PAL Service Agreement
	(Contract No.)
bet	ween Northwest Pipeline LLC
and	
-	
N	EGOTIATED RATE PROVISIONS

(Pursuant to Section 3.1 of Rate Schedule PAL)

Second Revised Sheet No. 368 Superseding First Revised Sheet Nos. 368 Through 369

FORM OF RATE SCHEDULE TPAL SERVICE AGREEMENT

			F	Rate Schedule Contract		.ce Agreer 	nent		
	SERVICE A			(Agreement) b	y and betw hipper) is		_		
(IIa.		-		[and restates					
into	on		·	.]					
WHER	EAS:								
Α.	[Insert r	ecita	als as	s appropriate	to memoria	alize the	context	of the	

Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule TPAL and the General Terms and Conditions (GT&C) that apply to Rate Schedule TPAL, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to provide [Park] (or) [Loan] Service for Shipper up to a Maximum [Parked] (or) [Loaned] Quantity set forth on Exhibit A at Transporter's [Park] (or) [Loan] Point set forth on Exhibit A.
- 3. Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth in the Statement of Rates in the Tariff, as revised from time to time, that apply to Rate Schedule TPAL will apply to service hereunder unless and to the extent that discounted Recourse Rates apply as set forth on Exhibit A or negotiated rates apply as set forth on Exhibit D.
- 4. Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A.

Rate Schedule TPAL Service Agreement (Continued)

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
By:	Ву:
Name:	Name:
Title:	Title:

	FORM OF RAT		E TPAL SERVI tinued)	ICE AGREEME	NT	
	[Dated Rate Sc betw	and Effect , to hedule TP (Contract een North	iBIT A ive (or) Effective the AL Service A No west Pipelin	Agreement _) ne LLC]	
		SERVIC	E DETAILS			
1.	Maximum [Parked] (or) [[Loaned] Q	uantity: _	Dth		
2.	[Park] or [Loan] Point: (or) []	[Clay Ba	asin] (or)	[Jackson Pr	airie],	
3.	[Park] or [Loan] Schedu Dth) (rows inserted as r		course or D	iscounted F	Recourse Rat	ce (per
	(Show "Not Applicable"	if Exhibi	t D is atta	ched.)		
	Service	From Date	To Date	Min Daily Qty	Max Daily Qty	Rate
	Park / Loan			201	21	
	Service	From Date	To Date	Min Qty	Max Qty	Rate
	Park / Loan Balance					
	Service	From Date	To Date	Min Daily Qty	Max Daily Qty	Rate
	Unpark / Loan Payback					
	b. Rate Discount Cond Schedule TPAL: [1					€
4.	Term: a. Primary Term Begin	n Date: [(Date)]			
	b. Primary Term End I	Date: [(Da	ate)]			

Exhibit D [No] (or) [Yes]

5. Additional Exhibits: Exhibit B [No] (or) [Yes]

FORM OF RATE SCHEDULE T	
(Conti	nuea)
EXHIB	IT B
[(Dated and Effective,	(subject to Commission acceptance)]
(or	·)
[Dated, E	ffective,
(subject to Commiss	sion acceptance)]
to t	he
Rate Schedule TPAL	Service Agreement
(Contract No)
between Northwes	t Pipeline LLC
and	

NON-CONFORMING PROVISIONS

	FORM OF RATE	SCHEDULE TPAL SERVICE AGREEME	INT
		(Continued)	
		EXHIBIT D	
[(Dated and E	Effective	, (subject to Commis	sion acceptance)]
		(or)	
[Dated	, Effective	,
	(subject	to Commission acceptance)]	
		to the	
	Rate Sch	edule TPAL Service Agreement	
	((Contract No.)	
	betwe	en Northwest Pipeline LLC	
	and		
	NEG	OTIATED RATE PROVISIONS	
	(Pursuant to S	Section 3.1 of Rate Schedule Ti	PAL)

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT

Rate	Schedule	TFL-1	Service	Agreement
------	----------	-------	---------	-----------

			Contr	act No.						
THIS SERVICE A	GREE	MENT	(Agre	ement) by	y and	d between	n Northwes	t Pipe	eline	LLC
(Transporter)	and			(Sh:	ippe:	r)is made	e and enter	red in	nto o	n
	[.]	(or)	[and	restates	the	Service	Agreement	made	and	entered
into on			.]							
WHEREAS.										

- A. [Insert recital describing how Shipper acquired transportation capacity.]
- B. [Insert additional recitals as appropriate to memorialize the context of the Agreement.]

THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule TFL-1 and the General Terms and Conditions (GT&C) that apply to Rate Schedule TFL-1, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Transportation Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to receive, transport and deliver natural gas for Shipper, on a firm basis, on the Designated Lateral set forth on Exhibit A. The Transportation Contract Demand, the Maximum Daily Quantity at each Primary Receipt Point, and the Maximum Daily Delivery Obligation at each Primary Delivery Point are set forth on Exhibit A.
- 3. Transportation Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. Reservation charges apply to the Transportation Contract Demand set forth on Exhibit A. The Maximum Base Tariff Rates (Recourse Rates) set forth under Rate Schedule TFL-1 in the Statement of Rates in the Tariff, as revised from time to time, that apply to the Designated Lateral set forth on Exhibit A will apply to service hereunder unless and to the extent that discounted Recourse Rates or awarded capacity release rates apply as set

Rate Schedule TFL-1 Service Agreement (Continued)

forth on Exhibit A or negotiated rates apply as set forth on Exhibit D. Additionally, if applicable under Section 21 of the GT&C, Shipper agrees to the incremental facilities charge as set forth on Exhibit C.

- 4. Transportation Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.
- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Capacity Release. If Shipper is a temporary capacity release Replacement Shipper, any capacity release conditions, including recall rights, are set forth on Exhibit A.
- 7. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Sections 11.5 or 22.12 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 8. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.

Rate Schedule TFL-1 Service Agreement (Continued)

9. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
Ву:	Ву:
Name:	Name:
	Title:

	FO	RM OF RATE SCHED	ULE TFL-1 SERVICE Continued)	E AGREEMENT
		[Base Co	ontract Version]	
	[Date	[Dated and EffeedRate Schedule 1	EXHIBIT A ective (or) _, Effective to the TFL-1 Service Agr ct No) thwest Pipeline]
		and		
		SER	VICE DETAILS	
1.	Designated	Lateral:		
2.	Transportat	ion Contract Dema	and (CD):	_ Dth per day
3.	Primary Rece	eipt Point(s):		
	Point ID	Name	Maximum Daily Quantities (Dth)	
		Total		
4.	Primary Del:	<pre>ivery Point(s):</pre>	Maximum Daily	
	Point ID	Name	Delivery Obligation (Dth)	Delivery Pressure (psig)
	-	Total		

Specified conditions for Delivery Pressure, pursuant to Section 2.4 of

the General Terms and Conditions: [None] or [Description]

[Base Contract Version]

EXHIBIT A (Continued)

5. Recourse, Discounted Recourse, or Negotiated Rate Transportation Rates:

(Negotiated Rates are on Exhibit D if attached.)

	D () () () () () () () () () (
a.	Reservation Charge (per Dth of CD):
	[Maximum Base Tariff Rate, plus applicable surcharges]
	(or)
	[\$, plus applicable surcharges]
	(or)
	[\$ x actual load factor, plus applicable surcharges]
	(or)
	[\$ x actual load factor, plus applicable surcharges,
	with the monthly reservation charge determined pursuant to
	Section 3.6(b) of Rate Schedule TFL-1 based on a minimum
	average load factor of %]

[Base Contract Version]

EXHIBIT A (Continued)

- b. Volumetric Charge (per Dth): [Maximum Base Tariff Rate, plus applicable surcharges]
- c. Rate Discount Conditions Consistent with Section 3.6 of Rate Schedule TFL-1: [Not Applicable] (or) [(Description)]
- d. Negotiated Rate Conditions Consistent with Section 3.7 of Rate Schedule TFL-1: [Not Applicable] (or) [[During the Negotiated Rates Period as outlined on Exhibit D of the agreement, subsections a. through c. above will not apply] (and/or) [(Description)]]
- 6. Transportation Term:
 - a. Primary Term Begin Date: [(Date)]

[Upon the later of the actual in-service

date of the required new _____ facilities or (Date)]

b. Primary Term End Date: [(Date)]

(or)

[(Specified period before or after a

specified event)]

(and)

Specified conditional service agreement extensions pursuant to Section 11.9 of the General Terms and Conditions of the Tariff: [None] or [Description].

c. Evergreen Provision: [No]

[Yes, standard unilateral evergreen under

Section 7.1 of Rate Schedule TFL-1]

(or)

[Yes, standard bi-lateral evergreen under Section 7.2 [(a)(i) and (b)(i)] (or) [(a)(ii) and (b)(ii)] (or) [(a)(iii) and (b)(iii)] of

Rate Schedule TFL-1]

7. Regulatory Authorization: [18 CFR 284.223]

(or)

[18 CFR 284.102, on behalf of _____]

8. Additional Exhibits: Exhibit B [No] (or) [Yes]

Exhibit C [No] (or) [Yes]

Exhibit D [No] (or) [Yes]

Exhibit E [No] (or) [Yes]

		FORM	OF RATE SCHE	DULE TFL-1 SERVIC (Continued)	E AGREEMENT
	[Tem	porary	Capacity Rel	ease Replacement	Contract Version]
			[Dated and Ef	EXHIBIT A fective]
	I	Dated		, Effective]
			(Contr between No and	to the TFL-1 Service Ag act No) orthwest Pipeline r/Contract No.: _	LLC
			S	SERVICE DETAILS	
1.	Designat	ed Lat	teral:		
2.	Transpo	rtation	n Contract Der	mand (CD):	Dth per day
3.	Primary	Receip	ot Point(s):	Maximum Daily	
	Point	ID	Name	Quantity (Dth)	
			Total		
4.	Primary	Delive	ery Point(s):	Maximum Daily Delivery	Delivery
	Point	ID	Name	Obligation (Dth)	Pressure (psig)
			Total		

Specified conditions for Delivery Pressure, pursuant to Section 2.4 of the General Terms and Conditions: [None] or [Description]

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

5.		sportation Rates: the version of "Reservation Charge" in "a." which applies.]
	a.	Awarded Reservation Charge (per Dth of CD): [Maximum Base Tariff Rate, plus applicable surcharges] (or)
		[\$, plus applicable surcharges] (or)
		[% of Maximum Base Tariff Rate, plus applicable surcharges] (or)
	a.	Awarded Volumetric Bid Reservation Charge (per Dth):

[Maximum Base Tariff Rate, plus applicable surcharges]

(or)

6.

7.

8.

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT (Continued)

[Temporary Capacity]	Release Replacement	Contract	Version]
-----------------------	---------------------	----------	----------

[\$, plus applicable surcharges]
(or)
<pre>[% of Maximum Base Tariff Rate, plus applicable surcharges]</pre>
[[Maximum Base Tariff Rate,] (or) [\$,] plus applicable
surcharges, and an amount pursuant to Section 3.2(b)(2) of Rate
Schedule TFL-1 using a minimum average load factor volumetric
commitment of%]
<pre>b. Volumetric Charge (per Dth): Maximum Base Tariff Rate, plus</pre>
Transportation Term: Begin Date:
Nomination Cycle:
End Date:

Regulatory Authorization: [18 CFR 284.223]
(or)
[18 CFR 284.102, on behalf of]
Additional Exhibits: Exhibit B [No] (or) [Yes]
Exhibit C [No] (or) [Yes]

- 9. Standard Capacity Release Conditions:
 - a. Releasing Shipper's recall rights:

 - i. [Released capacity may be recalled prior to the [Timely]
 (or) [Evening] (or) [Intraday 1] (or) [Intraday 2] (or)
 [Intraday 3] nomination cycle applicable to the initial day
 of the capacity recall.]
 - ii. Recall notification: [Allowed on any day] (or) [Allowed only on a Business Day] (or) [Not Applicable].
 - iii. Recall provisions in Section 22.2(a)(2) of the General Terms and Conditions of Transporter's Tariff also apply:

 [Yes] (or) [No]

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT (Continued)

[Temporary Capacity Release Replacement Contract Version]

EXHIBIT A (Continued)

- b. Reput rights: [Yes] (or) [No]
- d. Primary Delivery Point may be changed through amendment: $[{\tt Yes}] \ ({\tt or}) \ [{\tt No}]$
- e. Re-releasable: [Yes] (or) [No]
- 10. Additional Capacity Release Conditions pursuant to Section 22.7(k) of the GT&C:[None] or [(Description)]

Second Revised Sheet No. 378 Superseding First Revised Sheet No. 378

FORM OF	RATE SCHEDULE TFL-1 SERVICE	AGREEMENT
	(Continued)	
	EXHIBIT B	
[(Dated and Effective	, (subject to	<pre>Commission acceptance)]</pre>
	(or)	
[Dated	, Effective	,
(su	abject to Commission acceptan	ce)]
	to the	
Rate	e Schedule TFL-1 Service Agree	ement
	(Contract No.)	
	between Northwest Pipeline LI	LC .
ar	ıd	
	NON-CONFORMING PROVISIONS	

	(Continued)
	EXHIBIT C [Dated and Effective] (or)
	[Dated]
	to the Rate Schedule TFL-1 Service Agreement (Contract No) between Northwest Pipeline LLC and
	INCREMENTAL FACILITIES PAYMENT OBLIGATION
1.	DESCRIPTION OF FACILITIES:
	The facilities contemplated by Section 21 of the GT&C to provide service under this agreement include the following: (description of project)
2.	RESPONSIBILITY FOR FACILITIES COSTS:
	Pursuant to Section 21 of the GT&C, Shipper is responsible to pay for the actual cost of service for the facilities described above and has elected the payment option set forth below.
3.	TERMS AND CONDITIONS OF FACILITIES PAYMENT:
	a. Type of Charge:
	[Incremental facilities charge] (or) [Description of other payment method agreed to by Transporter and Shipper pursuant to Section 21.5]
	b. Charge \$
	(Describe the basis of the charges(s) and true-up or adjustment provisions, if any, for the stated charge.)
	c. Term of Charge:
	(Contract term or a term agreed to between the parties, whichever is shorter.)
	<pre>d. Accelerated Payment: [Description of accelerated payment provision</pre>

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT (Continued)
EXHIBIT D
[(Dated and Effective, (subject to Commission acceptance)] (or)
[Dated, Effective,
(subject to Commission acceptance)] to the
Rate Schedule TFL-1 Service Agreement
(Contract No) between Northwest Pipeline LLC
and
NEGOTIATED RATE PROVISIONS
(Pursuant to Section 3.7 of Rate Schedule TFL-1)
1. The Negotiated Rate Provisions will take effect on [the Primary Term Begin Date] (or) [[(Date)] (and/or) [(Description)]] and will remain in effect for a period of () [[years], (and/or) [months], (and/or) [days] (further description as needed)] (Negotiated Rates Period). Thereafter, the Recourse Rates will apply to the Service Agreement.
(Description of Negotiated Rate Provisions)

Second Revised Sheet No. 381 Superseding First Revised Sheet No. 381

FORM OF RATE SCHEDULE TFL-1 SERVICE AGREEMENT (Continued)

	EXHIBIT E	
	[Dated and Effective]	
	(or)	
[Dated	, Effective]
	to the	
	Rate Schedule TFL-1 Service Agreement	
	(Contract No)	
	between Northwest Pipeline LLC	
	and	

 ${\tt BUY-OUT\ PROVISION},$ pursuant to Section 11.8 of the General Terms and Conditions:

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 382 Through 384 Superseding Substitute Original Sheet Nos. 382 Through 384

RESERVED FOR FUTURE USE

FORM OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT

Rate Schedule TIL-1 Service Agreement Contract No
THIS SERVICE AGREEMENT (Agreement) by and between Northwest Pipeline LLC (Transporter) and (Shipper) is made and entered into on [.] (or) [and restates the Service Agreement made and entered
into on]
WHEREAS:
A. [Insert recitals as appropriate to memorialize the context of the Agreement.]
THEREFORE, in consideration of the premises and mutual covenants set forth herein, Transporter and Shipper agree as follows:

- 1. Tariff Incorporation. Rate Schedule TIL-1 and the General Terms and Conditions (GT&C) that apply to Rate Schedule TIL-1, as such may be revised from time to time in Transporter's FERC Gas Tariff (Tariff), are incorporated by reference as part of this Agreement, except to the extent that any provisions thereof may be modified by non-conforming provisions herein.
- 2. Transportation Service. Subject to the terms and conditions that apply to service under this Agreement, Transporter agrees to receive, transport and deliver natural gas for Shipper, on an interruptible basis, on the Designated Lateral set forth on Exhibit A. The Maximum Daily Quantity of natural gas and the receipt and delivery points are set forth on Exhibit A.
- 3. Transportation Rates. Shipper agrees to pay Transporter for all services rendered under this Agreement at the rates set forth or referenced herein. The Maximum Base Tariff Rates (Recourse Rates) set forth under Rate Schedule TIL-1 in the Statement of Rates in the Tariff, as revised from time to time, that apply to the Designated Lateral set forth on Exhibit A will apply to service hereunder unless and to the extent that negotiated rates apply as set forth on Exhibit D. Notwithstanding the foregoing, a temporary discounted Recourse Rate may apply pursuant to the terms set forth on a Rate Schedule TIL-1 Temporary Discount form.
- 4. Transportation Term. This Agreement becomes effective on the effective date set forth on Exhibit A. The primary term begin date for the transportation ${\sf T}$

FORM OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT (Continued)

Rate Schedule TIL-1 Service Agreement (Continued)

service hereunder is set forth on Exhibit A. This Agreement will remain in full force and effect through the primary term end date set forth on Exhibit A and, if Exhibit A indicates that an evergreen provision applies, through the established evergreen rollover periods thereafter until terminated in accordance with the notice requirements under the applicable evergreen provision.

- 5. Non-Conforming Provisions. All aspects in which this Agreement deviates from the Tariff, if any, are set forth as non-conforming provisions on Exhibit B. If Exhibit B includes any material non-conforming provisions, Transporter will file the Agreement with the Federal Energy Regulatory Commission (Commission) and the effectiveness of such non-conforming provisions will be subject to the Commission acceptance of Transporter's filing of the non-conforming Agreement.
- 6. Exhibit / Addendum to Service Agreement Incorporation. Exhibit A is attached hereto and incorporated as part of this Agreement. If any other Exhibits apply, as noted on Exhibit A to this Agreement, then such Exhibits also are attached hereto and incorporated as part of this Agreement. If an Addendum to Service Agreement has been generated pursuant to Section 11.5 of the GT&C of the Tariff, it also is attached hereto and incorporated as part of this Agreement.
- 7. Regulatory Authorization. Transportation service under this Agreement is authorized pursuant to the Commission regulations set forth on Exhibit A.

FORM OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT (Continued)

Rate Schedule TIL-1 Service Agreement (Continued)

8. Superseded Agreements. When this Agreement takes effect, it supersedes, cancels and terminates the following agreement(s): [None] (or) [(List agreement(s))] but the following Amendments and/or Addendum to Service Agreement which have been executed but are not yet effective are not superseded and are added to and become an Amendment and/or Addendum to this agreement: [None] (or) [List Amendments and/or Addendum to Service Agreement]

IN WITNESS WHEREOF, Transporter and Shipper have executed this Agreement as of the date first set forth above.

	Northwest Pipeline LLC
By:	By:
Name:	Name:
Title:	Title:

	FORM OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT
	(Continued)
	EXHIBIT A
	[Dated and Effective]
	(or)
	[Dated, Effective]
	to the
	Rate Schedule TIL-1 Service Agreement
	(Contract No.)
	between Northwest Pipeline LLC
	and
	SERVICE DETAILS
1.	Designated Lateral:
2.	Maximum Daily Quantity: Dth per day
3.	Receipt Point(s): [All Transportation Receipt Points on the Designated Lateral]
4.	Delivery Point(s): [All Transportation Delivery Points on the Designated Lateral]
5.	Transportation Term:
	a. Primary Term Begin Date: [(Date)]
	(or)
	[Upon the later of the actual in-service
	date of the required new
	facilities or (Date)]
	b. Primary Term End Date: [(Date)]
	(or)
	[(Specified period before or after a
	specified event)]
	Specified event)
	c. Evergreen Provision: [No]
	(or)
	[Yes, see Section 6 of Rate Schedule TIL-1]
	(or)
	[Yes, see Exhibit B]
	- ,

FORM OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT (Continued)

EXHIBIT A (Continued)

6.	Regulatory Authorization:	[18 CFR 284.223]	
		(or)	
		[18 CFR 284.102, on behalf of	

7. Additional Exhibits: Exhibit B [No] (or) [Yes] Exhibit D [No] (or) [Yes]

Second Revised Sheet No. 389 Superseding First Revised Sheet No. 389

FORM (OF RATE SCHEDULE TIL-1 SERVICE AGREEMENT (Continued)	1
	(concinaca)	
	EXHIBIT B	
[(Dated and Effective		n acceptance)
[Dated	(or) , Effective	
[Dated _	(subject to Commission acceptance)	
Ra	ate Schedule TIL-1 Service Agreement	
	(Contract No)	
	between Northwest Pipeline LLC	
	and	
	NON-CONFORMING PROVISIONS	

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

Second Revised Sheet No. 390 Superseding First Revised Sheet No. 390

RESERVED FOR FUTURE USE

	FORM OF	RATE SCHEDULE TIL-1 SERVICE AGREEMENT	
		(Continued)	
		EXHIBIT D	
[(Dated	and Effective	, (subject to Commission	<pre>acceptance)]</pre>
		(or)	
	[Dated	, Effective	,
	(su	bject to Commission acceptance)]	_
		to the	
	Rate	Schedule TIL-1 Service Agreement	
		(Contract No.)	
]	petween Northwest Pipeline LLC	
	an	d	
		NEGOTIATED RATE PROVISIONS	
	(Pursuant	to Section 3.6 of Rate Schedule TIL-1)	

Northwest Pipeline LLC FERC Gas Tariff Fifth Revised Volume No. 1

First Revised Sheet Nos. 392 Through 394 Superseding Substitute Original Sheet Nos. 392 Through 394

RESERVED FOR FUTURE USE

NON-CONFORMING SERVICE AGREEMENTS

The following Service Agreements contain one or more currently effective provisions that differ materially from the Forms of Service Agreements contained in this Tariff.

Name of Shipper/Agreement No.	Rate Schedule	Agreement/ Amendment Date	
Avista Corporation (No. 100010)	TF-1	07/31/91 06/25/21(1)	10/4/21
Cascade Natural Gas Corporation (No. 100304)	TF-2	04/01/94 03/31/16(1)	04/21/16
Cascade Natural Gas Corporation (No. 141193)	TF-2	03/31/16	04/21/16
Cascade Natural Gas Corporation (No. 142548)	TF-1	11/1/2018	11/09/18
Cascade Natural Gas Corporation (No. 143078)	TF-1	09/27/2019	10/01/19
Cascade Natural Gas Corporation (No. 143821)	TF-1	11/11/20	12/02/20
Citadel Energy Marketing LLC (No. 144762)	TF-1	11/01/22	09/29/23
City of Enumclaw (No. 100012)	TF-1	07/31/91 07/17/07	07/21/08
Darigold, Inc. (No. 145131)	TF-1	10/01/23	08/01/23

⁽¹⁾ Amendment incorporates the non-conforming provisions.

Name of Shipper/Agreement No.		Agreement/ Amendment Date	Filed
FortisBC Energy Inc. (No. 135562)	TF-1	04/19/07	07/21/08
Idaho Power Company (No. 126442)	TF-1	03/22/01 05/02/07 (2)	06/08/07
Idaho Power Company (No. 126951)	TF-1	08/02/01 03/02/09 (2)	05/01/09
<pre>Idaho Power Company (No. 138369)</pre>	TF-1	04/06/10	04/28/10
<pre>Idaho Power Company (No. 138370)</pre>	TF-1	04/06/10	04/28/10
<pre>Idaho Power Company (No. 138390)</pre>	TF-1	04/15/10	04/28/10
Idaho Power Company (No. 139664)	TF-1	09/25/12	09/28/12
Idaho Power Company (No. 145985)	TF-1	04/01/25	03/20/25

⁽¹⁾ Amendment incorporates the non-conforming provisions.

⁽²⁾ Original Service Agreement, as amended, was restated in the revised Form of Service Agreement format on the listed date.

Name of Shipper/Agreement No.		Agreement/ Amendment Date	Date Filed
IGI Resources, Inc. (No. 137605)	TF-1	10/13/09 02/17/23 (3)	10/28/09
<pre>Intermountain Gas Company (No. 100306)</pre>	TF-2	01/12/94 09/13/23 (2)	9/29/23
<pre>Intermountain Gas Company (No. 136288)</pre>	TF-1	11/20/07 09/29/09 (2)	06/29/12
<pre>Intermountain Gas Company (No. 136289)</pre>	TF-1	11/20/07 09/29/09 (2)	06/29/12
<pre>Intermountain Gas Company (No. 136290)</pre>	TF-1	11/20/07 09/29/09 (2)	06/29/12
<pre>Intermountain Gas Company (No. 142982)</pre>	TF-1	06/28/19	07/23/19
<pre>Intermountain Gas Company (No. 143037)</pre>	TF-1	08/14/19	10/01/19
<pre>Intermountain Gas Company (No. 143072)</pre>	TF-1	09/26/19	10/01/19
Northwest Natural Gas Company (No. 100005)	TF-1	07/31/91 02/14/07 (2)	03/28/07

⁽²⁾ Original Service Agreement, as amended, was restated in the revised Form of Service Agreement format on the listed date.

⁽³⁾ Original Service Agreement, as amended, restates the ROFR end date.

Name of Shipper/Agreement No.	Schedule	Agreement/ Amendment Date	Date Filed
Northwest Natural Gas Company (No. 100308)		01/12/94 05/01/99 (1)	12/03/99
Northwest Natural Gas Company (No. 100310)	TF-2	01/12/94 01/21/08	07/21/08
Northwest Natural Gas Company (No. 138587)	TF-1	09/29/10 06/21/11 (1)	07/15/11
Northwest Natural Gas Company (No. 140964)	TF-1	10/28/15	10/28/15
Occidental Energy Marketing Inc. (No. 121692)	TF-1	03/27/98 03/27/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121696)	TF-1	03/27/98 03/27/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121707)	TF-1	03/27/98 03/27/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121708)	TF-1	03/27/98 03/27/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121714)	TF-1	03/27/98 03/27/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121718)	TF-1	03/28/98 04/06/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121719)	TF-1	03/28/98 04/06/98 (1)	07/13/98
Occidental Energy Marketing Inc. (No. 121720)	TF-1	03/27/98 03/27/98 (1)	07/13/98

⁽¹⁾ Amendment incorporates the non-conforming provisions.

Name of Shipper/Agreement No.	Rate Schedule	Agreement/ Amendment Date	Date Filed
Occidental Energy Marketing Inc. (No. 121798)	TF-1	04/01/98	07/13/98
Occidental Energy Marketing Inc. (No. 121799)	TF-1	04/01/98	07/13/98
Occidental Energy Marketing Inc. (No. 121800)	TF-1	04/01/98	07/13/98
Occidental Energy Marketing Inc. (No. 121801)	TF-1	04/01/98	07/13/98
Occidental Energy Marketing Inc. (No. 122015)	TF-1	05/26/98	07/13/98
Occidental Energy Marketing Inc. (No. 122033)	TF-1	05/27/98	07/13/98
Occidental Energy Marketing Inc. (No. 122034)	TF-1	05/27/98	07/13/98
Occidental Energy Marketing Inc. (No. 122035)	TF-1	05/28/98	07/13/98
Occidental Energy Marketing Inc. (No. 122036)	TF-1	05/27/98	07/13/98
Occidental Energy Marketing Inc. (No. 122037)	TF-1	05/27/98	07/13/98
Occidental Energy Marketing Inc. (No. 122038)	TF-1	03/27/98	07/13/98
Occidental Energy Marketing Inc. (No. 122045)	TF-1	03/27/98	07/13/98

Name of Shipper/Agreement No.	Rate Schedule	Agreement/ Amendment Date	Date Filed
Occidental Energy Marketing Inc. (No. 124046)	TF-1	07/02/99	02/02/00
Occidental Energy Marketing Inc. (No. 124047)	TF-1	07/02/99	02/02/00
Occidental Energy Marketing Inc. (No. 124048)	TF-1	07/02/99	02/02/00
Occidental Energy Marketing Inc. (No. 124049)	TF-1	07/02/99	02/02/00

Name of Shipper/Agreement No.		Agreement/ Amendment Date	
PacifiCorp (No. 129875)	TF-1	08/01/01 11/12/07	07/21/08
Puget Sound Energy, Inc. (No. 135434)	TF-1	03/05/07 03/10/09 (2)	05/01/09
Puget Sound Energy, Inc. (No. 140415)	TF-1	06/26/14	11/19/14
Puget Sound Energy, Inc. (No. 140766)	TF-1	04/22/15	09/21/15
Puget Sound Energy, Inc. (No. 140910)	TF-1	10/13/15	10/28/15
Puget Sound Energy, Inc. (No. 143077)	TF-1	09/27/19	10/01/19
Puget Sound Energy, Inc. (No. 145550)	TF-1	11/02/23	01/24/24
Shell Energy North America (US), LP (No. 134510)	TF-2	05/01/06	05/05/06
Shell Energy North America (US), LP (No. 137104)	TF-1	09/23/09	09/30/09
Southwest Gas Corporation (No. 100048)	TF-1	06/01/91 06/25/19	07/23/19

⁽²⁾ Original Service Agreement, as amended, was restated in the revised Form of Service Agreement format on the listed date.

Name of Shipper/Agreement No.	Rate Schedule	Agreement/ Amendment Date	Date Filed
Southwest Gas Corporation (No. 100049)	TF-1	06/01/91 06/25/19	07/23/19
Spotlight Energy, LLC (No. 145101)	TF-1	04/01/24	03/19/24

NEGOTIATED RATE SERVICE AGREEMENTS

The following Service Agreement contains a Negotiated Rate but does not deviate in any material aspect from the Form of Service Agreement for Rate Schedule TF-1 contained in this Tariff.

Shipper Name/	Negotiated	Rate	Receipt	Delivery	Contract Quantity (Dth/d)
Agreement No.	Rate	Schedule	Points	Points	
Puget Sound Energy, Inc. (No. 127115)	(1,2,3,4)	TF-1	Goldendale In-Line Trans Rec	Goldendale Energy Delivery	50,350

- 1. Shipper will pay Transporter a monthly fee of \$2,366 for the ongoing Operation and Maintenance attributable to the Goldendale meter station and lateral (Delivery Facilities). This charge will be payable on or before the twenty-fifth (25th) day of each month and continue through the term of this Agreement. If Shipper fails to make payment of such invoice(s) within thirty (30) days of the initial invoice date, interest shall accrue on the unpaid portion of the billing(s) at the then applicable FERC Refund Interest Rate, including the collection of any reasonable attorney's fees, which may be incurred in collection of any payments due.
- 2. Annually, Shipper will reimburse Transporter for all otherwise unrecovered capital expenditures incurred for repairs, and/or any modifications that are made to the Delivery Facilities as a result of any laws, rules, regulations and orders of any duly constituted authority having jurisdiction.
- 3. In the event FERC ever requires an allocation of non-Delivery Facilities costs to Transportation service hereunder, such allocated costs will be fully reimbursed by Shipper on a monthly basis, pursuant to the payment procedure set forth in paragraph 1 above.
- 4. For all scheduled nominations under this Agreement involving a receipt and/or delivery point not located on the Delivery Facilities, Shipper will pay a rate equivalent to the then Maximum Currently Effective rate for Rate Schedule TF-1 service under Transporter's Tariff.

NEGOTIATED RATE SERVICE AGREEMENTS

The following Service Agreement contains a Negotiated Rate but does not deviate in any material aspect from the Form of Service Agreement for Rate Schedule TF-1 contained in this Tariff.

Shipper Name/ Agreement No.	Negotiated <u>Rate</u>	Rate Schedule	Receipt <u>Point</u>	Delivery <u>Point</u>	Quantity (Dth/d)
Puget Sound Energy, Inc.	1,2,3,4	TF-1	Olympia Lateral	Olympia	36 , 928
(No. 138077)			In-Line T	rans	

The Negotiated Rate Provisions will take effect on December 04, 2024 and will remain in effect for a period of eight (8) years through March 31, 2033 (Negotiated Rates Period).

- 1. Shipper will pay Transporter a monthly fee of \$1,667 for the ongoing Operation and Maintenance attributable to the Delivery Facilities. This charge will be payable on or before the twenty-fifth (25th) day of each month and continue through the term of this Agreement. If Shipper fails to make payment of such invoice(s) within thirty (30) days of the initial invoice date, interest shall accrue on the unpaid portion of the billing(s) at the then applicable FERC Refund Interest Rate, including the collection of any reasonable attorney's fees, which may be incurred in collection of any payments due.
- 2. Annually, Shipper will reimburse Transporter for all otherwise unrecovered capital expenditures incurred for repairs, and/or any modifications that are made to the Delivery Facilities as a result of any laws, rules, regulations and orders of any duly constituted authority having jurisdiction.
- 3. In the event FERC ever requires an allocation of non-Delivery Facilities costs to Transportation service hereunder, such allocated costs will be fully reimbursed by Shipper on a monthly basis, pursuant to the payment procedure set forth in paragraph 1 above.
- 4. For all scheduled nominations under this Agreement involving a receipt and/or delivery point not located on the Delivery Facilities, Shipper will pay a rate equivalent to the then Maximum Currently Effective rate for Rate Schedule TF-1 service under Transporter's Tariff.